Life Course Research

RESEARCH

Philipp M. Lersch

Residential Relocations and their Consequences

Life course effects in England and Germany



Life Course Research

Herausgegeben von

Prof. Dr. Steffen Hillmert, Universität Tübingen

Philipp M. Lersch

Residential Relocations and their Consequences

Life course effects in England and Germany



Philipp M. Lersch Tilburg, The Netherlands

Dissertation Universität Bremen, 2012

ISBN 978-3-658-04256-1 DOI 10.1007/978-3-658-04257-8 ISBN 978-3-658-04257-8 (eBook)

The Deutsche Nationalbibliothek lists this publication in the Deutsche Nationalbibliografie; detailed bibliographic data are available in the Internet at http://dnb.d-nb.de.

Library of Congress Control Number: 2013953570

Springer VS

© Springer Fachmedien Wiesbaden 2014

This work is subject to copyright. All rights are reserved by the Publisher, whether the whole or part of the material is concerned, specifically the rights of translation, reprinting, reuse of illustrations, recitation, broadcasting, reproduction on microfilms or in any other physical way, and transmission or information storage and retrieval, electronic adaptation, computer software, or by similar or dissimilar methodology now known or hereafter developed. Exempted from this legal reservation are brief excerpts in connection with reviews or scholarly analysis or material supplied specifically for the purpose of being entered and executed on a computer system, for exclusive use by the purchaser of the work. Duplication of this publication or parts thereof is permitted only under the provisions of the Copyright Law of the Publisher's location, in its current version, and permission for use must always be obtained from Springer. Permissions for use may be obtained through RightsLink at the Copyright Clearance Center. Violations are liable to prosecution under the respective Copyright Law. The use of general descriptive names, registered names, trademarks, service marks, etc. in this publication does not imply, even in the absence of a specific statement, that such names are exempt from the relevant protective laws and regulations and therefore free for general use. While the advice and information in this book are believed to be true and accurate at the date of publication, neither the authors nor the editors nor the publisher can accept any legal responsibility for any errors or omissions that may be made. The publisher makes no warranty, express or implied, with respect to the material contained herein.

Printed on acid-free paper

Springer VS is a brand of Springer DE. Springer DE is part of Springer Science+Business Media. www.springer-vs.de To Gertrude Wagner

Foreword

Though central to the sociological concept of the life course, which typically relies on the dimensions of both time and space as basic axes of descriptions, residential relocations have received comparatively little attention by sociological research over the past few years. To date, it has primarily been the social determinants of geographical mobility that have been analysed. While there has been repeated evidence about the substantial effects of individual life situations on residential mobility, we have known relatively little about the other direction of causality, i.e. the individual consequences of residential moves.

In contrast to conventional research, this study by Philipp Lersch now offers a careful and thorough analysis of such consequences in individual life courses. The study is based on micro-level panel data and has special merits in offering a unique combination of regional mobility, social inequality and life course research. Moreover, it is explicitly designed as a comparative study between Germany and the UK - or, in the empirical analyses, England, due to data restrictions - which puts emphasis on the role of institutions when interpreting systematic differences in observed behaviour. Differences in the housing markets of Britain and Germany are well known and obvious sources of specific mobility patterns. However, when looked at in a life course perspective, it becomes obvious that other institutional differences are also highly relevant, concerning, in particular, the labour markets, welfare institutions, and gender relations. It is a special merit of this study to bring these areas, which are too often studied separately, together conceptually and to link them empirically on the level of various life course events. This approach is straightforward as residential moves are typically associated with events like job mobility, union formation and dissolution.

The methodologically advanced analyses yield a multitude of interesting results; some of them may be unexpected, but again the author is able to give them meaning by embedding them in the life course context. Residential moves do not generally improve life situation (e.g., with regard to 'room stress', i.e. available apartment size relative to individual needs), but this depends strongly on the specific kind of the life event that is associated with the move. In this sense, the study is directed at important mechanisms of social inequality. One might go even further and fully include social origin and the parental context in order to study links with intergenerational mechanisms of the transmission of social (dis-)advantage. In any case, the results will provoke the search for additional comparative evidence, and the analytical framework proposed may well enable further research to be built on it.

This book is part of the series Life Course Research (Springer VS Research). In this series we publish empirical studies - in both English and German - which focus on transitions along the life course, and promoting comparative research is a matter of particular concern for us. The work by Philipp Lersch fits therefore perfectly into this series. It is an important book for any researcher interested in the life course in general and links between residential mobility and social inequality in particular. I hope many readers will benefit from it.

Tübingen, September 2013

Steffen Hillmert

Preface

This is a shortened version of my doctoral dissertation at the University of Bremen. I thank my supervisors Michael Windzio, Olaf Groh-Samberg and Mark Taylor for their advice and support. They patiently answered all my questions and made myriads of constructive suggestions. Many discussions with Sergi Vidal, Yvonne Lott, Nate Breznau, Markus Kiesel and Krishantha Kamaladiwala improved my work substantially.

The research was generously funded by the Bremen International Graduate School of Social Sciences (BIGSSS) at the University of Bremen and Jacobs University Bremen. I am grateful to the faculty, staff and all fellows at BIGSSS for creating such an inspiring and cordial atmosphere. Part of the research was conducted at the Institute for Social and Economic Research (ISER), Colchester, and at the German Institute for Economic Research (DIW), Berlin, where Jan Goebel was especially helpful. None of these institutions or persons are in any way responsible for the resulting work and all errors remain my own.

The data from the British Household Panel Survey used in this publication were made available through the ESRC Data Archive. The data were originally collected by the ESRC Research Centre on Micro-social Change at the University of Essex (now incorporated within the ISER). Neither the original collectors of the data nor the Archive bear any responsibility for the analyses or interpretations presented here. The data for Germany used in this publication were made available to me by the German Socio-Economic Panel Study (SOEP) at the DIW, Berlin.

For their unconditional support I thank my whole extended family, but especially Judith, Ruth, Simon and Andreas. Without the encouragement, inspiration and affection of my friends — near and far — this book would have never been written.

I am very grateful to Yvonne for everything.

Bremen and Tilburg, September 2013

Philipp M. Lersch

Contents

1	Intr	oductio	n	1
	1.1	Resear	rch Problem and Research Questions	2
	1.2	Gaps i	n Research and Innovative Contributions	5
	1.3	Organ	isation of Study	7
2	The	orising	Residential Relocations and Their Outcomes	11
	2.1	Theore	etical Approaches Towards Residential Relocations	12
		2.1.1	Early Theories of Residential Relocations	12
		2.1.2	Institutional Approach	14
		2.1.3	Rational Choice Approach	16
		2.1.4	Behavioural Approach	19
		2.1.5	Life Course Perspective	22
		2.1.6	Conclusion	27
	2.2	Explai	ning Residential Relocations	28
	2.3	Explai	ining Variation in Relocation Outcomes	29
		2.3.1	Individual Life Course Conditions	30
		2.3.2	Economic Resources	33
		2.3.3	Situational Conditions of Decision Making	35
		2.3.4	Structural Conditions in the Housing Market	36
		2.3.5	Gendered Norms of Behaviour	38
	2.4	Summ	ary	41
3	Rev	iew of I	Past Literature	43
	3.1	Deterr	ninants of residential relocations	43
	3.2	Chang	es in Room Stress After Residential Relocations	48
		3.2.1	Relevance	49
		3.2.2	Review of Empirical Research	50
	3.3	Neigh	bourhood Effects and Changes in Neighbourhood Quality	
		After 1	Relocations	53

		3.3.1 Relevance	54
		3.3.2 Measures of Neighbourhood Quality	57
		3.3.3 Delimitations of Neighbourhoods	58
		3.3.4 Review of Empirical Research	58
	3.4	Effects of Relocations on Employment in Dual-Earner Couples	63
		3.4.1 Relevance	64
		3.4.2 Review of Empirical Research	66
4	Res	earch Design and Methods	71
	4.1	Country Case Description: England and Germany	72
		4.1.1 Housing Markets	72
		4.1.2 Labour Markets	76
		4.1.3 Welfare Systems	78
		4.1.4 Gender Relations	80
	4.2	Unit of Analysis	82
	4.3	Data	83
		4.3.1 Panel Data at the Individual and Household Level	83
		4.3.2 Neighbourhood Level	86
		4.3.3 Regional Level	89
	4.4	Sample Selection	90
	4.5	Measurement	91
		4.5.1 Dependent Variables	91
		4.5.2 Independent Variables	97
	4.6	Methods 1	101
		4.6.1 Fixed-Effects Regression Models	102
		4.6.2 Multi-Level Regression Models	106
		4.6.3 Actor-Partner Interdependence Models	107
		4.6.4 Shared Frailty Discrete-Time Event History Models	109
5	Sett	ing the Scene: Occurrence and Characteristics of Relocations	111
	5.1	Descriptive Results	112
		5.1.1 Characteristics of Residential Relocations	112
		5.1.2 Characteristics of Movers	114
	5.2	Multivariate Results	115
	5.3	Selective Sample Attrition Due to Residential Relocations	123
	5.4	Summary	126
6	Roo	m Stress and Residential Relocations	129
	6.1	Descriptive Results	130
		6.1.1 Average Room Stress by Groups	130
		6.1.2 Patterns in Changes in Room Stress over the Life Course 1	131
		6.1.3 Average Changes in Room Stress After Relocations 1	133
	6.2	Multivariate Results	135

		6.2.1	Individual Life Course Conditions	136
		6.2.2	Economic Resources and Housing Tenure	145
		6.2.3	Situational Conditions of Decision Making	149
		6.2.4	Immigrant Households and Room Stress	152
		6.2.5	Structural Conditions in Housing Markets	155
	6.3	Summ	ary	158
7	Neig	ghbour	hood Quality and Residential Relocations	161
	7.1	Descr	iptive Results	162
		7.1.1	Objective and Subjective Neighbourhood Quality in	
			Germany	162
		7.1.2	Patterns in Changes in Neighbourhood Quality over the	
			Life Course	164
		7.1.3	Average Changes in Neighbourhood Quality After	
			Relocations by Groups	165
	7.2	Multiv	variate Results	169
		7.2.1	Individual Life Course Conditions	170
		7.2.2	Economic Resources and Housing Tenure	177
		7.2.3	Situational Conditions of Decision Making	183
		7.2.4	Immigrant Households and Neighbourhood Quality	186
		7.2.5	Structural Conditions in Housing Markets	189
	7.3	Summ	ary	192
8	Lon	g-Dista	nce Relocations and Subsequent Employment of	
	Dua	l-Earn	er Couples	195
	8.1	Descri	iptive Results	196
		8.1.1	Leaving Employment for Movers	198
	8.2	Multiv	variate Results	201
		8.2.1	Gross Effect of Long-Distance Relocations	202
		8.2.2	Institutional Context	208
		8.2.3	Robustness Checks	211
		8.2.4	Changes in Wage Rates	214
	8.3	Summ	ary	215
9	Out	comes	of Residential Relocations and Structural Conditions:	
	Eng	land ar	nd Germany Compared	217
	9.1	Institu	tional Features and Their Effects on Residential Relocations	. 219
		9.1.1	Quality of Locations over the Life Course	220
		9.1.2	Role of Housing Tenure	223
		9.1.3	Decommodification of Housing	225
		9.1.4	Gender Relations in England, and West and East Germany	227
	9.2	Life C	Course Regimes and Residential Relocations	229
	9.3	Summ	nary	232

10	Con	clusion
	10.1	Discussion of Hypotheses and Empirical Evidence
	10.2	Limitations of the Analyses
	10.3	Open Questions for Future Research
	10.4	Policy Implications of Findings
	10.5	Final Reflections on Relocations and Social Inequality
A	Add	itional Tables
	A.1	Research Design and Methods
	A.2	Setting the Scene: Occurrence and Characteristics of Relocations 251
	A.3	Roomstress and Residential Relocations
	A.4	Neighbourhood Quality and Residential Relocations
	A.5	Long-Distance Relocations and Subsequent Employment of
		Dual-Earner Couples
Ref	erenc	es

List of Figures

Roadmap of analysis 8
Proportion of downward relocations (room stress) by age group 133
Effects of relocation types on room stress over time
Marginal effect of stable couple relocations on room stress
conditional on household income, women
Predicted changes in room stress by tenure of origin and destination
after relocation, women
Proportion of downward relocations (neighbourhood quality) by
age group
Marginal effect of stable couple relocation on neighbourhood
quality conditional on household income, women
Predicted changes in neighbourhood quality by tenure of origin and
destination after relocation, women
Predicted probabilities for leaving employment by context, gender
and relocation status
Life course regimes and residential trajectories

List of Tables

4.1	Housing tenure structure, 2005 73
4.2	Labour market statistics, 2005
4.3	Principal Component Analysis of IMDs 94
4.4	Principal Component Analysis of Microm variables
5.1	Group averages, movers vs. stayers
5.2	Discrete-time event history models of relocation, England
5.3	Discrete-time event history models of relocation, Germany 120
5.4	Logistic regression models of drop-outs after relocations
6.1	Average room stress by group and country in 2008
6.2	Average changes in room stress after residential relocations 135
6.3	Fixed-effects models of room stress
6.4	Fixed-effects models of room stress with interaction of income and
	presence of children
6.5	Fixed-effects models of room stress with interaction of relocation
	type and income
6.6	Fixed-effects models of room stress for voluntary and involuntary
	relocations
6.7	Fixed-effects models of room stress by immigrant status 154
6.8	Fixed-effects models of room stress with housing market variables 156
7.1	Average neighbourhood quality by subjective neighbourhood
	evaluation in 2004, Germany 163
7.2	Average change in neighbourhood quality by change in subjective
	neighbourhood evaluation, Germany
7.3	Average neighbourhood quality and change in quality by group and
	country
7.4	Fixed-effects model of neighbourhood quality

7.5	Fixed-effects model of neighbourhood quality with interactions of
76	Fixed affasts models of neighbourhood quality with alternative
7.0	income specifications
77	Fixed effects models of neighbourhood quality for voluntary
/./	relocations by distance and for involuntary relocations
78	Fixed affects models of neighbourhood quality by immigrant status 188
7.9	Fixed-effects models of neighbourhood quality by minigrant status . 188
	market variables
8.1	Mobility rates and number of observed long-distance relocations
	by type of couple 197
8.2	Job-motivated long-distance relocations in dual-earner couples 199
8.3	Average proportion leaving employment by mover status
8.4	Average proportion leaving employment until t+2 by mover status201
8.5	Actor-partner interdependence model of leaving employment until
	<i>t</i> +1
8.6	Actor-partner interdependence model of leaving employment until
	<i>t</i> +2
8.7	Actor-partner interdependence models of leaving employment until
	t+1 by context
10.1	Evaluation of hypotheses
A.1	Recoded and original variable names
A.2	Descriptive statistics. England (Chapter 5)
A.3	Descriptive statistics. Germany (Chapter 5)
A.4	Descriptive statistics, England (Chapter 6)
A.5	Descriptive statistics, Germany (Chapter 6)
A.6	Multiple membership multi-level models of room stress with
	housing market variables
A.7	Descriptive statistics, England (Chapter 7)
A.8	Descriptive statistics, Germany (Chapter 7)
A.9	Fixed-effects models of neighbourhood quality with income
	interactions
A.10	Multiple membership multi-level models of neighbourhood quality
	with housing market variables
A.11	Descriptive statistics, England (Chapter 8)
A.12	Descriptive statistics, West Germany (Chapter 8)
A.13	Descriptive statistics, East Germany (Chapter 8)

Chapter 1 Introduction

"Migration is one of the important ways by which people can improve their lives." (DaVanzo 1981: 90)

Individuals' homes are places for numerous vital activities and satisfy basic needs. They offer comfort, protection, privacy and room for self-expression, stimulation and recreation. Individuals eat and sleep, love and hate, work and relax in their homes and homes are status symbols used to distinct oneself from others. Outside their homes, individuals take a stroll, chat with the people next door, see the physician, or go shopping in their neighbourhoods. Workers commute to their work places in the region and if they look for new jobs, they will mostly search the regional labour market first. Thus, individuals' homes, their neighbourhoods and regions are important determinants of their quality of life and constitute a relevant dimension of social stratification. Where individuals live matters (Buck 2000a; Tickamyer 2000; Groh-Samberg 2009: 75). How individuals come to live at particular locations at various times in their lives and under which conditions relocations change the quality of these locations over the life course are therefore important questions in analysing social stratification; and they have not yet received the attention that they deserve.

Individuals relocate throughout their life courses and subsequent residential locations situate life courses in specific places (Elder 2003: 64). At least one relocation from the parental nest is common, but most individuals relocate more often. In Germany 65 per cent of the population aged 15 and over relocated at least twice in their lives and about 18 per cent relocated more than four times. In the United Kingdom (UK), about 71 per cent of the population aged 15 and over moved at least twice and 30 per cent even more than four times in their lives (Bonin et al. 2008: 36, Table 5). Little is known about how these relocations change the quality of individuals' residential locations. Mostly, it is assumed that relocations are positive events that improve the lives of individuals on the whole (Lu 1998: 1474). For example, the prominent model of the 'housing ladder' assumes that household units improve the

P. M. Lersch, *Residential Relocations and their Consequences*, Life Course Research, DOI 10.1007/978-3-658-04257-8_1, © Springer Fachmedien Wiesbaden 2014

quality of their housing with each relocation. In this model, the residential trajectory, i.e. the sequence of residential locations, is a continuous upward trajectory regarding price, size and quality of the dwelling (Kendig 1984; Pickles and Davies 1991). More generally, geographic mobility is believed to be associated with upward social mobility and societal progress in industrial and even more in post-industrial societies. This 'promise of mobility' can be understood as one of modernity's core pillars (Hochstadt 1999: 20): Those leaving their hamlets and travelling to the urban industrial centres, those crossing oceans and making their lives abroad, and those business travellers working in the transit zones of international airports are the key figures of the modern world. They are the *movers* and shakers. Similarly, there is a strong belief in modern societies that residential relocations are positive life events.

Too often overlooked is the variation in outcomes of residential relocations. For some, the promise of mobility may hold and their relocations constitute improvements in their lives. For others, relocations may bring about negative changes in at least some areas of life and some others may be not affected by relocations at all. For example, Clark, Deurloo and Dieleman (2006: 335) show that only one quarter of movers gain in the economic status of their neighbourhoods and housing quality at the same time. Some movers receive better job offers after their relocations, but the labour market outcomes of relocations vary considerably between movers (Jacobsen and Levin 2000). This variation indicates that relocations are *contingent life course events*. Relocations as such may have more or less positive as well as negative outcomes. Which conditions of relocations explain the variation in outcomes is the research puzzle of the present work.

In the following, I use the term 'residential relocation' to denote a change of address for the main place of residence of an individual. Every change regardless of the distance between the old and new address is considered a residential relocation. The terms 'residential relocation' and 'relocation' are used interchangeably. Even though, the term 'residential relocation' is not widespread, I think it is useful to distinguish it from other concepts of movement. For example in the literature, the term 'residential move' is often used to denote changes of address only between places of residence that are in the same local area or region. The term 'migration' is mostly used to describe inter-regional changes of residence, but is also widely used to describe movements across national borders. The term 'mobility' has been used for all types of changes in residence as well as recurrent movements, e.g. between the place of work and the place of residence. None of these alternative terms fully captures the social phenomenon that I analyse in the following chapters.

1.1 Research Problem and Research Questions

In his theory of social production functions, Lindenberg (1996: 169) postulates that individuals are driven by two universal goals: "physical wellbeing and social ap-

proval". These universal goals can be fulfilled by striving for universal instrumental goals: "stimulation and comfort" (ibid.: 171) are instrumental for physical wellbeing, while "status, behavioral confirmation and affection" (ibid.: 171) are instrumental for social approval. These first-order instrumental goals can be achieved by time-variant, lower-order instrumental means. Residential locations, i.e. dwellings, their neighbourhood and regional areas, are such lower-order instrumental means to achieve universal goals. Dwellings provide shelter from the elements and environmental hazards, privacy and a self-controlled space and, thus, help to achieve the instrumental goal of comfort. At the same time, dwellings offer space for selfexpression, social interaction and leisure activities, which supports the goal of stimulation. Dwellings also function as status symbols and, thus, contribute to the goal of social approval. The same can be said about neighbourhoods and regions which may also facilitate the goal attainment of individuals.

The World Bank (1993: 15) defines the following features of a residential location as essential for goal attainment:

"Everyone is housed, with a separate unit for every household. Housing does not take up an undue portion of household income. House prices are not subject to undue variability. Living space is adequate. Structures are safe and provide adequate protection from the elements, fire, and natural disasters. Services and amenities are available and reliable. Location provides good access to employment. Tenure is secure and protected by due process of law. Households may freely choose among different housing options and tenures [...]."

Residential locations meet these criteria to varying degrees. Locations are heterogeneous and some support goal attainment of individuals more than others. Alba and Logan (1993: 1391, emphasis in original) use the term "place stratification [...] [to indicate] that places are ordered hierarchically and consequently are associated with more or less favourable life chances and quality of life for the people who reside in them." For example, some dwellings have leaking roofs and are crowded. Other dwellings are air-conditioned and provide private rooms for all residents. Some neighbourhoods are divided by major roads which increase the number of accidents as well as the level of pollution and noise in the neighbourhood. Other neighbourhoods host spacious green areas and only little traffic moves through the area. Some regions' economies decline and job searchers find no opportunities to take up work. Other regions prosper and regional employers pay competitive salaries to attract labour. It follows that particular locations are less supportive for individual goal attainment, while individuals at other locations can use their location as an instrumental mean to reach their goals of comfort, stimulation and status. Therefore, residential locations are a relevant dimension of social stratification that affects individuals' life chances.

This begs the question of how individuals come to live at particular locations. Individuals are literally born into locations that are consequences of the residential choices of their parents. After their adolescences, individuals normally begin their own residential trajectory by leaving the parental nest and establishing their own (shared) households. After leaving the parental nest, residential relocations determine the locations that individuals live at. Individuals may improve their residential location by relocating to places which offer better opportunities for goal attainment, or by modifying the location itself. Whether individuals improve their situation depends on two issues.

First, individuals need the necessary willingness, resources, and opportunities to be able to relocate and the capability to be residentially mobile is not equally given to all individuals (Kaufmann, Bergman and Joye 2004). Bauman (2000: 86, emphasis in original) argues that globalised societies are to a great extent stratified by the "*degree of mobility*" of their members. Thus, with regard to relocations, some individuals may grasp the opportunities and relocate, while other individuals are stuck and do not relocate.

Second, even if individuals relocate, they will not necessarily improve the quality of their locations to the same degree, since favourable residential locations are scarce and relocations are conditioned by restrictions and constraints. It follows that the classic sociological question about social inequality must be enhanced with spatial aspects of social stratification. The question is not only who gets what, but "who gets what where?" (Lobao and Hooks 2007: 2, emphasis in original). To analyse changes in the quality of locations is thereby closely linked to the explanation of social inequalities over the life course which remains an important endeavour for sociologists (Diewald and Mayer 2009). Relocations must be understood as contingent life course events that may change individuals' qualities of location in various positive and negative ways. Less positive or negative outcomes must be considered not only as irregularities, but as potential outcomes of relocations.

Thus, the main hypothesis of the present analysis is that the *outcomes of relocations vary across movers and that they are systematically stratified due to particular conditions of relocations and characteristics of movers*. By stratified, I mean that some movers may gain more than other movers from relocating, while some movers may reduce the quality of their location. By systematically, I mean that these unequal outcomes are not randomly distributed across all movers, but are associated with certain conditions. These conditions may be expected to be found at three levels: individual conditions, e.g. economic resources; household conditions, e.g. the presence of children; and structural conditions, e.g. the average quality of housing in the local area. To shed light on these conditions is the purpose of the present analysis and motivates the following central research question: *Which individual, household and contextual conditions shape the outcomes of residential relocations over the life course?* In other words: Which factors determine whether households relocate upwardly, horizontally, or downwardly with regard to the quality of their residential location?

The research question is framed within the life course perspective (for a recent overview, cf. Mayer 2009). The life course perspective highlights individual agency over the time of individuals' lives, but also acknowledges that individuals are restricted in their agency due to biographical dependencies, interdependencies between different life domains, interdependencies between individuals, as well as contextual opportunities and constraints resulting from individual embeddedness in particular historical times, places and institutions. By framing the research question in this way, residential relocations can be analysed as events in the flow of time of individuals' lives that are — as much as all biographical events — undetermined and contingent in their outcomes as such, and can only be explained in all their variation by considering individual resources and restrictions and contextual opportunities and constraints.

1.2 Gaps in Research and Innovative Contributions

Various aspects of residential locations have been analysed for some time in the field of sociology. Especially housing has been the subject of extended research including Norbert Elias (e.g. 2003 [1969]: 75ff), who analyses changes in the housing situation in conjunction with general social changes and the civilisation process, and Pierre Bourdieu (e.g. 1984: 372ff), who examines the housing situation as a results of habitus and taste and identifies class differences in the layout, furnishing and style of dwellings. The neighbourhood was a research topic for the Chicago School of Sociology as early as in the 1920s (e.g. Burgess 2007 [1925]). The geographical space of regions was part of sociological research for a long time. Sociological studies of regional effects can be traced, for example, to the work of Emile Durkheim (1951 [1897]: 104ff) who analysed the impact of regional characteristics on suicides. Starting with the spatial turn in the late 1960s which was inspired by the work of Henri Lefebvré (1991 [1974]) amongst other scholars, space played an increasingly prominent role in sociology. However, classic sociological studies of residential locations treated locations as given and static. The first comprehensive study that analysed the dynamics of relocations was the seminal work 'Why families move' of Rossi (1955).

A second strand of social research investigates the determinants of residential relocations in the disciplines of demography, economics, geography, (social) psychology and sociology (Dieleman 2001). The earliest work on relocations was by Ernest Ravenstein (Ravenstein 1876, 1885, 1889) at the end of the 19th century. Until the mid-20th century, research focused on the macro-level determinants of aggregated relocation flows. Beginning with rational choice models of residential relocations, the level of analysis shifted to individuals and their behaviours (e.g. Sjaastad 1962). Most of the research thereafter focussed on the individual motivations for relocations and analysed the impact of individual and household characteristics as well as the life course on the propensity to relocate. Fundamental to most of this research was the assumption that households relocate, because they are better-off at new locations.

A third strand of literature deals with the association between characteristics of individuals and households and the quality of their locations from a mostly cross-

sectional perspective. This research tries to establish links between individual resources such as income and individual heterogeneity such as immigration status on the one hand, and quality of housing and the neighbourhood on the other hand. For example, households with more income are found to live in better housing and neighbourhood quality (Clark and Huang 2003). Most of this research deals with between-individual differences, i.e. scholars try to identify differences between individuals that can explain variation in the quality of locations.

There are a number of gaps in previous research that the present work aims to close. First, this study goes beyond past work by changing the perspective on residential relocations. Instead of conceptualising relocations as inherently positive life events, the *contingency of relocations* is highlighted. In the present research the variation in outcomes is taken into focus, instead of treating downward and horizontal relocations as mere irregularities.

Second, past work on the outcomes of relocations mainly focussed on specific subgroups, e.g. immigrants, specific outcomes, e.g. room stress, and specific characteristics of movers, e.g. economic resources. In the present work, I aim at bringing together these efforts. To this end, I construct a coherent theoretical framework that combines individual, household and structural explanatory factors based on the life course perspective. I test this framework for different outcomes of relocations. The outcomes that are examined in the present analysis are 1) room stress, i.e. the relation between room needs and available rooms, as a measure of housing quality; 2) composite indices to measure neighbourhood quality; and 3) employment status as a measure of labour market outcomes. These outcomes relate to changes in the dwelling, the neighbourhood, and the region respectively. Analysing the employment status also allows examination of within-household differences in relocation outcomes.

Third, past literature mostly focussed on cross-sectional associations between outcomes of relocations and conditions of these relocations. Instead, rich longitudinal data and panel data analysis is used in the present work. Panel data offers in-depth information on the pre- and post-relocation situation of individuals. This allows relating changes in the quality of locations to relocations and to other changes in individuals' life courses, to changes in their households and to changes in structural conditions. Thereby, the dynamic nature of life courses can be captured and more credible claims about determinants of relocation outcomes can be made than is possible with cross-sectional data.

Fourth, little is known about structural conditions that shape the outcomes of residential relocations. At the regional level, it is not well understood, how housing markets affect outcomes of relocations. At the national level, it is unclear how national particularities of housing markets may influence outcomes of relocations. In addition, as long as hypotheses are not tested in different national contexts, it cannot be assessed in which way findings apply generally or are only due to national idiosyncrasies (Diewald and Mayer 2009). Therefore, I examine regional conditions of relocations and compare outcomes of relocations in England and Germany.

1.3 Organisation of Study

The organisation of this study is presented in Figure 1.1. In the present introduction, relocations are conceptualised as contingent life course events. The following research question is raised: which conditions explain the variation in outcomes of relocations? In Chapter 2, I review theoretical approaches that may provide hints to answer this question. After sketching early theoretical approaches to residential relocations, the presentation focuses especially on micro-level approaches that have been developed since the 1960s. The rational choice model, the behavioural model and the life course perspective are described. Macro-level opportunities and constraints for residential relocations are identified using the institutional approach. These theoretical approaches are integrated to derive the theoretical framework. The framework mainly builds on the life course perspective, but incorporates elements from the other described approaches. The framework highlights individual life course conditions, economic resources, situational conditions of decision making, structural conditions in housing markets, and gendered roles in couples as potential explanatory factors for the variance in outcomes of relocations. The framework is used to formulate hypotheses that guide the empirical analysis.

In Chapter 3, past literature is summarised. I consider literature on the determinants of residential relocations, the effects of relocations on changes in room stress, the effects of relocations on changes in neighbourhood quality, and the effects of relocations on employment in dual-earner couples. These findings from past research are placed within the presented theoretical framework and provide first insights about the validity of the hypotheses. The literature review is also used to identify more specific gaps in past research concerning each outcome.

The research design and the empirical strategy are described in Chapter 4. I choose a quantitative empirical approach to answer the research question. After justifying the case selection, I provide a description of England's and Germany's institutional characteristics with regard to their housing markets, labour markets, welfare systems, and gender relations that may effect outcomes of relocations. The two longitudinal datasets, i.e. the British Household Panel Survey (BHPS) for England and the Socio-Economic Panel Study (SOEP) for Germany, used in the empirical analysis are described. Additional data at the level of neighbourhoods and regional housing markets that supplement the individual- and household-level data are introduced. Then the operationalisation of all variables is described and the empirical methods are explicated. Most of the data analysis is conducted using panel regression methods.

The English population is more mobile on average than the German population (Bonin et al. 2008: 36, table 5). Chapter 5 deals with the influence of life course stages and transitions as well as individual resources and structural conditions on the probability to relocate in both countries. The results show which individuals are likely to relocate and whether differences in the determinants of relocations

Chapters:



Fig. 1.1 Roadmap of analysis

exist between England and Germany. In light of these findings, the results from subsequent chapters regarding the outcomes of relocations can be better interpreted.

The next three chapters form the core of my empirical work. In these chapters, the theoretical framework is tested regarding the conditions that shape outcomes of relocations. In Chapter 6, changes in room stress are analysed. First, aggregated changes over time in both countries are described and differences across groups in average room stress are tested. In the multivariate analysis, the hypotheses derived earlier are formally tested using fixed-effects panel regression and multi-level regression models. I especially focus on the effects of residential relocations and various life course conditions in this chapter, but also test for variation in room stress after relocations due to economic resources and housing tenure, involuntary relocations and distance of relocations, immigrant status, and regional conditions in the housing market.

In Chapter 7, changes in neighbourhood quality due to residential relocations are scrutinised. The panel data used in the previous chapters is supplemented with data on the direct neighbourhoods of individuals. Neighbourhood data for England is drawn from the Indices of Multiple Deprivation. For Germany, a composite neighbourhood quality indicator is constructed based on the Microm data set. Again, I test my hypotheses using fixed-effects panel regression and multi-level regression models. The effects of individual life course conditions, economic resources and housing tenure, involuntary relocations and distance of relocations, immigrant status, and conditions in the housing market are examined.

The last empirical chapter deals with the impact of long-distance relocations on employment in dual-earner couples to examine whether both partners are equally likely to leave employment after relocations on average (Chapter 8). In contrast to the two previous chapters, the focus in this chapter is on inter-individual differences between female and male partners, which allows the analysis of intra-household variation in outcomes. I also present findings on the nonindependence of changes of both partners within couples.

Throughout the empirical chapters, conditions of relocations in England and Germany are analysed. In Chapter 9, I summarise cross-national similarities and dissimilarities in my findings for changes in room stress, neighbourhood quality and careers of dual-earner couples. Cross-national similarities are used to describe regularities in relocation outcomes across distinct institutional contexts. Then, I attempt to explain cross-national divergence in the determinants of outcomes after relocations by referring to particular institutional differences in the national housing markets, labour markets and welfare states, and gender relations in both countries. As only two country cases are analysed and institutional differences between these countries are ample, this part of the analysis is exploratory.

In Chapter 10, the main insights from the analysis are highlighted and the hypotheses are assessed in light of the presented empirical evidence. Shortcomings of the present analysis are pointed out and open questions are formulated. Next, policy implications are elaborated by drawing on the findings.

Chapter 2 Theorising Residential Relocations and Their Outcomes

When relocating, individuals shift the centre of their action spaces to a new residential location for a considerable duration of time (Boyle, Halfacree and Robinson 1998: 34). Primarily, this raises two questions. Why do individuals relocate? What are the outcomes of these relocations? Outcomes of relocations are changes in the quality of the present compared to the last residential location. I define the quality of a location as the degree to which the location improves the chances for an individual to achieve physical wellbeing and social approval. Individuals living in low-quality locations will fare worse regarding these goals compared to individuals in high-quality locations on average, because they live in too small and unhealthy dwellings, in unsafe and polluted neighbourhoods and in economically stagnating regions. The quality of a location depends on its features. I use 'features' to refer to objectively observable characteristics of a location, e.g. the density of buildings in a neighbourhood. Features and quality of a location are closely associated and some features will result in better quality than others. I assume that individuals are not always fully aware of the quality of a location, while they may be aware of its features. Individuals are often part of relocating households containing two or more persons with interdependent decisions and outcomes; however households are not constant over time. Therefore, I conceive of and measure residential relocations from an individual perspective.

The first question on the reasons for relocations has received considerable attention for more than a century now. The second question has received less attention explicitly – and is the focus of this analysis. Some expectations can be derived from past theoretical approaches, but there is no coherent theoretical framework explaining variation in outcomes of relocations. The innovative contribution of the present chapter is to draw insights from different theories to develop a framework that treats relocations as genuinely *ambiguous and contingent life course events*. Conditions are derived that determine to which degree relocations improve the chances for an individual to achieve greater physical wellbeing and social approval. The combined theoretical framework is mainly based on the life course perspective, but also integrates elements from other approaches that draw from work in the fields of demography, economics, geography, (social) psychology and sociology (Section 2.1). The combined framework is presented with regard to the occurrence (Section 2.2) and outcomes of relocations (Section 2.3). I also develop hypotheses to test my theoretical framework empirically.

2.1 Theoretical Approaches Towards Residential Relocations

Early theories of residential relocations mainly explain relocations at the macro level (Section 2.1.1). Macro approaches to the explanation of residential relocations link aggregated relocation behaviour to certain aggregated characteristics of populations and environments. The aim is to explain aggregated relocation flows between origin and destination regions. The institutional approach also follows in this tradition, but acknowledges that individuals act within institutional settings and these settings themselves are subject to individual behaviour (Section 2.1.2). In the second half of the 20th century, micro approaches were developed in contrast to the earlier macro approaches. Micro approaches to residential relocations try to explain individual relocation behaviour (Golledge and Stimson 1997: 426). In the rational choice approach, residential relocations are analysed as a function of the returns and costs of relocating (Section 2.1.3). The behavioural approach builds on rational choice theory. The approach explains relocations on the basis of individual decision making following the idea of bounded rationality and emphasising the importance of individual perception of reality (Section 2.1.4). The life course perspective highlights the importance of interdependencies in the life course for residential relocations and is the most influential approach for the theoretical considerations (Section 2.1.5).

2.1.1 Early Theories of Residential Relocations

Common to all historical approaches is the explanation of residential relocations at the macro level, without explicitly considering individual behaviour. Following from this, the approaches reach conclusions that are very limited in their explanatory power for individual behaviour, and are more descriptive than later micro-level theoretical approaches (Kalter 2000: 460). The causes of relocations are superficially covered and expectations about outcomes of relocations can hardly be derived. Therefore, the theories are too limited to be applied in the present analysis. Nonetheless, their presentation helps to understand the development of later theories, and at the same time offers a benchmark to evaluate these approaches.

In three papers in the late 19th century, Ernest Ravenstein developed the first general assumptions about migration and relocation behaviour at an aggregated level (Ravenstein 1876, 1885, 1889). Ravenstein formulated eleven 'laws of migration' that he mainly derived from analysing data on relocation behaviour in the UK, but supplemented with other international data. He found the following regularities (the list follows Grigg (1977: 42ff): 1) Most movers only relocate a short distance. 2) Relocations proceed step by step. 3) Long distance relocations are mainly aimed at industrial and commercial centres. 4) Relocation streams in one direction generate a counter-stream in the opposite direction. 5) Rural dwellers are less mobile than urban dwellers. 6) Women are more mobile than men. 7) Most movers are adults. 8) Large cities grow mainly through inward-relocations. 9) Relocations increase in numbers as industries and commerce develop and transport improves. 10) The main direction of relocations is from rural areas to industrial and commercial centres. 11) The major causes for relocations are economic. Seeing that these are observations about relocations rather than laws, most of the above regularities stood the empirical test in the 19th century and continue to do so today (Lee 1966; Boyle, Halfacree and Robinson 1998: 60). Basic insights formulated by Ravenstein have been reconsidered by other scholars, e.g. in gravity models, the mobility transition and the neoclassical economics approach.

Building on Isaac Newton's law of gravity, *gravity models* of social action were formulated. Zipf (1946) formulated an application to relocations:

$$\widehat{M}_{ij} = \frac{P_i P_j}{D_{ij}},\tag{2.1}$$

where \widehat{M}_{ij} is the estimated number of relocations between regions *i* and *j*, P_i and P_j are the population sizes in the regions *i* and *j*, and D_{ij} is the distance between these regions. The distance may be discounted by an exponent *b*, which can be understood as a "*distance decay* parameter" (Boyle, Halfacree and Robinson 1998: 46, emphasis in original). Theoretically, the measure of distance is used as a proxy for a number of factors, e.g. potential knowledge about the destination and the cost of relocation (ibid.: 46). From this model follows that relocations are rather short than long distance and involve at least one highly populated region, i.e. an urban area. The gravity model mainly rephrases the first, eighth and tenth law of migration formulated by Ravenstein.

The *mobility transition approach* postulates that residential relocations are fundamentally caused by the modernisation of society, which is also stated by the ninth law of migration (ibid.: 60). Similar to the demographic transition that societies go through, a mobility transition can be observed that is deterministic and irreversible. Societies mainly differ in the onset of certain phases in the transition, but all societies are subject to this transition sooner or later. Zelinsky (1971) differentiates five phases where each phase can be characterised by the overall magnitude of mobility, the direction of relocations (urban vs. rural; international vs. internal), the recurrence of mobility (recurrent vs. non-recurrent) and, finally, the mode of mobility (communication vs. transportation).

The *neoclassical economics* perspective is part of the family of 'push-pull' theories, "because they perceive the causes of migration to lie in a combination of 'push factors', impelling people to leave the areas of origin, and 'pull factors', attracting them to certain" (Castles and Miller 2003: 22) locations. In neoclassical economics, residential relocations are explained by factor mobility. The factors of production are land, labour and capital, which are sold by individuals and used by firms to produce goods. Factor markets are regionally bounded. As firms are assumed to be rather immobile, individuals become mobile if they can maximise their utility from selling labour on another regional factor market. Relocations are driven by wage differentials. Relocations stop if an optimal allocation of labour has been reached and the wage level is equal across all regional markets. The higher the wage differential between two regional factor markets, the higher the labour mobility between these markets. Thus, the demand of labour and the expected wages are considered the most important push and pull factors (Boyle, Halfacree and Robinson 1998: 61). The approach builds on the eleventh law of migration.

2.1.2 Institutional Approach

The institutional approach also explains relocations through macro level factors, but at the same time acknowledges that individuals act intentionally within these opportunities and constraints. Thus, the institutional approach examines the structural context influencing individual residential relocations. The structural context refers to "economic and material circumstances of a society including its political and legal framework" (Cadwallader 1992: 19). The institutional approach assumes that individuals can only act within a set of options that is created by certain institutions. Individual preferences may not be actualised due to institutional constraints (Manion and Flowerdew 1982: 11). Furthermore, institutions shape the importance of residential relocations as a behavioural option. That is to say that relocations may be a more viable behavioural option in particular institutional settings, e.g. settings with low transaction costs for changing residence. Institutions do not only constrain options, but also facilitate certain behaviour and are understood as being partly shaped by decision makers. Institutions develop over time, and they can be affected by private and public organisations (Cadwallader 1992: 15f). Not all individuals are affected by institutions in the same way, e.g. high-income individuals may profit more from tax-benefits for home ownership than low-income individuals, and institutions differ between nations, but also at a regional level (Manion and Flowerdew 1982: 24).

A wide range of institutions can be hypothesised as influencing residential relocations. Here, the focus is on the institutions of the housing market, since they are

assumed to be especially relevant (Flowerdew 1982). The housing market affects the location of dwellings, the features of these dwellings and the conditions for relocating between dwellings. The most important dimension of the housing market is the tenure structure. Rental tenure facilitates relocations, but has certain downsides regarding the features of the dwelling, e.g. on average it offers less space. Owner tenure impedes relocations, but has certain other favourable features, e.g. it is more easily adjusted to individual needs (e.g. Megbolugbe and Linneman 1993). In addition, the overall supply of housing affects the behaviour of individuals. A higher and more diverse supply relative to the demand increases the behavioural options of individuals. Housing markets are structured to a very small-scale spatial level (Simmons 1968; Flowerdew 1982: 221). They differ between nation states and are shaped, among others things, by national housing policies. In this regard, social housing and its role in the market is especially affected by policies. However, not only public agents shape the housing market, private agents such as property developers, real estate managers, banks and consumers themselves also affect this institution (Cadwallader 1992: 16).

The following can be theoretically expected regarding relocations according to the institutional approach. At the aggregated level, a high share of rental accommodation in the housing market relative to owner-occupied accommodations increases residential relocations. A high overall supply of dwellings relative to the demand increases the frequency of relocations as well. Concerning the outcomes of residential relocations, the institutional approach does not inform expectations right away. Although, particular institutional characteristics can be assumed to increase the likelihood of positive relocation outcomes. Individuals are more likely to improve their locations if a housing market is characterised by high overall quality, e.g. due to investment in high-quality social housing or a higher share of home ownership. Relocations are likely to improve locations more, if location choices are less restricted by economic resources, e.g. due to overall lower housing prices. To examine these institutional conditions, different institutional contexts at the regional, but also at the national level should be compared.

The institutional approach is less a theory, but more an analytic perspective that emphasises a certain category of determinants of residential relocations. The institutional approach is an extension of other theoretical frameworks to highlight the importance of institutional context effects. To this end, elements of the institutional approach are used in the theoretical synthesis to grasp the institutional differences between different national settings and a cross-national comparative perspective using data from England and Germany is included. However, the institutional framework does not allow for generalisation outside of each institutional context and brings the danger of *ad hoc* explanations (Flowerdew 1982: 222). The approach overstates external constraints for individual behaviour, while underestimating individual decision processes. The framework does not explain individual choices that are made, but only the choice sets available to individuals can be described. Finally, interaction effects of individual characteristics and institutional environments are not considered. All in all, the approach is helpful when extended by other theoretical approaches to explain individual behaviour. These approaches are described in the subsequent sections.

2.1.3 Rational Choice Approach

Moving from the macro to the micro level of analysis, rational choice theory explains residential relocations as a function of returns and costs. Individuals relocate, if they are better off by relocating than by staying (Sjaastad 1962). Rational choice theory assumes that individuals rationally maximise utility and intentionally decide to relocate (Jong and Gardner 1981: 5). In the basic models, individuals have to choose between finite sets of behaviour options including staying or relocating to different locations. Each option generates a certain utility depending on individuals' preferences, which are constant across the population, i.e. a certain behaviour option is preferred by all individuals in the population. Individuals are fully informed about the consequences of their choices and can rank the choice options according to the utility they will generate. The option with the highest utility will be chosen. The choice is restricted by limited resources (Simon 1957: 241). This basic rational choice model has been extended as is shown in the second half of this subsection.

Becker (1995: 53) conceptualises mobility as an investment in human capital, because the resources spent on relocating are an investment in the career and return higher earnings (cf. also Sjaastad 1962).¹ Todaro (1969: 140) emphasises that instead of momentary wage differentials, the expected lifetime utility after relocation is driving a mover's decision. Therefore, an individual relocates, if

$$U_{ri} = R_{ri} - C_{ri} > U_{si}, \tag{2.2}$$

where U_{ri} is some individual lifetime utility from relocation, R_{ri} are individual returns, i.e. income gains, C_{ri} are individual costs of relocating and U_{si} denotes the lifetime utility of staying at the present location. If there are several potential destinations, a location is chosen that maximises U_{ri} . Individuals are assumed to continually assess alternative locations and compare them to their present location (DaVanzo 1981: 124, endnote 12). Relocation decisions are restricted by C_{ri} and by the investment the individual can afford. Costs are not only material, i.e. the actual cost of making the relocation trip and search costs, but also cover psychological and opportunity costs (Sjaastad 1962). More generally, utility from relocations may not be limited to earning gains, but can rather be understood as an overall place utility. The

¹ That is why the approach is also referred to as the human capital approach.

concept of place utility "refers to the net composite of utilities which are derived from the individual's integration at some position in space" (Wolpert 1965: 162).^{2,3}

Place utility can be expected to increase over time spent at a certain location, as the integration at this place grows stronger, more local ties exist and local resources are accumulated. This may also result from active investment of individuals in their residential location. Everything else equal, increasing place utility may result in a decreasing chance of relocation over time. In the literature, this phenomenon has been referred to as the cumulative inertia axiom (Clark 1981: 190). This axiom states that the likelihood of relocating decreases with the time spent in the present location. The axiom has been repeatedly supported by empirical evidence (e.g. Fischer and Malmberg 2001). It is one example of path dependencies over the life course that are further described in Section 2.1.5. The cumulative inertia axiom offers an explanation for why individuals are mostly rather immobile and become more immobile the less mobile they have been in the past.

According to the rational choice approach, individuals relocate if the resulting returns exceed costs and lifetime utility can be expected to be higher at the new location. Therefore, younger individuals are more likely to relocate, because they gain from relocations for a longer time. Returns are also likely to be higher for individuals with more human capital. On the other hand, higher costs will reduce the likelihood of residential relocations. Higher costs can be expected, if somebody has strong local ties. Local ties may be social, e.g. family and friends living close by, material, e.g. home ownership, and psychological, e.g. being emotionally attached to a certain location. Higher costs also arise for longer distances of relocation outcomes. However, the rational choice model can only explain positive relocation outcomes. However, the rational choice approach has been extended to account for collective household decision, incomplete information, and limited search behaviour. These extensions help to explain negative relocation outcomes and are presented now.

Mincer (1978) extends the rational choice explanations of relocations to the household context, especially regarding heterosexual couples (cf. also DaVanzo 1981: 112). Instead of analysing individual gains and costs from residential relocations, the aggregated gains and costs for the household are considered. A household relocates, if

$$U_{rh} = \sum U_{ri} = \sum R_{ri} - \sum C_{ri} > U_{sh}, \qquad (2.3)$$

where the subscript h indicates the household, which is an aggregate of i individuals. Mincer also raises the point that individual household members may be adversely affected by relocations, but relocate with the other members as tied movers. Household members may also benefit from a relocation, but do not relocate, as the household stays put (tied stayer). Tied moving or tied staying occurs, if U_{ri} for household members differ and "the tied partner is one whose absolute loss (gain) is less than

² See Section 2.1.4 for more details on Wolpert (1965) and the behavioural approach.

³ The concept of place utility is similar to what I have defined as the quality of a location.

the absolute value of gain (loss) of the other partner" (Mincer 1978: 751). In general, costs of relocations increase in higher rates than gains with growing household size. With several potential destinations, it is likely that all members of the household fare worse by relocating collectively than by relocating individually. Female partners in heterosexual couples are assumed to be more likely to be negatively affected by household relocations, on the grounds that male partners are assumed to have more human capital on average and, thus, relocations are mainly undertaken for male partners' careers (ibid.).

Mincer does not further elaborate the adverse effects of relocations for tied movers and stayers as long as the collective utility function is maximised (Abraham, Auspurg and Hinz 2010). Game theoretic bargaining models have been applied by family economists and sociologists to analyse these conflicts of individual interests. Fundamental to this approach is the rejection of the ideas of a unitary utility function and pooled income for the whole family (Thomas 1990; Beblo 2001: 12ff). At the same time, the approach assumes that "cooperative behavior enables an increased welfare production" (Ott 1992: 19) in the family. Thus, individuals prefer to cooperate. Bargaining models assume that family members consider their utility in alternative situations to evaluate their present situation within the family. Thus, individuals prefer cooperation in the present family context, but only as long as they do not have a more rewarding alternative. An example for a bargaining model is the divorce-threat model, in which spouses determine their utility outside of their marriages, i.e. their threat points. If their utility within the marriage falls below this threat point, they would divorce. They can use their threat points in bargaining with their partners to assert their interests. It can also be assumed that individuals consider potential changes in their power when making decisions (Lundberg and Pollak 2003).4,5

Further extensions of the classic rational choice approach highlight the importance of information gathering for residential relocations, which formerly was only an implicit part of this approach. The information that is necessary for residential relocation decisions is extensive and includes knowledge of "future preferences [...]; future outcomes at the present and alternative locations [...]; the range of alternative locations available [...]; and the specific characteristics of alternative destinations" (Goodman 1981: 136). The basic rational choice model assumes that individuals have all this information. However, information gathering is costly (Simon 1957: 248). The search costs increase due to the intensity of search, the time spent on searching, the geographic area covered and the number of locations considered. At the same time, these factors determine the completeness of information for the re-

⁴ The most popular solution for these bargaining models has been formulated by Nash (1953). The Nash solution is characterised by "1) Pareto efficiency, 2) symmetry, 3) invariance with regard to linear transformations and 4) independence from irrelevant alternatives" (Beblo 2001: 16, footnote 10).

⁵ Bargaining and gender differences within the household are further addressed in Section 2.3.5.

location decision. Individuals will only continue their search as long as they expect the gain from the search to be higher than the costs. It can be assumed that location search has increasing marginal costs, but only decreasing marginal returns. Thus, it is irrational for individuals to gather complete information. It follows that most potential movers consider only a few destinations before making a decision and most of these will be close to the present location (Goodman 1981: 138ff). Thus, residential relocations are decisions under uncertainty. As the individual makes a decision without being fully informed about the consequences and all alternatives, unintended and adverse consequences may occur (DaVanzo 1981: 95).

Following extended rational choice models, relocations occur since individuals expect to increase their life time utility at an alternative location. Variation in relocation outcomes may follow from limited information of movers and their limited capacities to process this information. Movers may end up in worse locations than other individuals, because they were not completely informed about all features of new locations. Even if they were informed about all features, they may not have correctly inferred the quality of the location from this information. In addition, variation in outcomes may result from restricted resources of individuals. As individuals can only relocate to locations that they can afford, these relocations may increase the quality of location less than for individuals with more resources. The concepts of limited information and restricted resources may be important to explain variance in relocation outcomes and are linked with the life course perspective in my combined theoretical framework.

Many scholars have criticised the assumptions of rational choice theory — especially concerning the homunculus deployed (Lindenberg 1983: 10). The focus on intentionally and goal-oriented acting individuals offers simple explanations for individual behaviour, but with the risk of being too simplistic. Extensions of the rational choice framework already recognise many of these critical points that are raised: satisfying instead of maximising behaviour, limited cognitive capabilities of individuals, limited information, divergence between objective and subjective expected utility, heterogeneity of preferences across the population and regional bounded markets. However, the ever growing extensions of the basic rational choice model result in a theoretical framework that loses much of its original rigour. Two major extensions, bounded rationality and satisfying behaviour, are described in more details in the next section.

2.1.4 Behavioural Approach

The behavioural approach explains relocations on the basis of individual choice and decision making of agents. The approach highlights the idea of bounded rationality and emphasises individual perception of reality grounded in the work of Simon (1957: 256) who introduced the idea of "limited knowledge and ability" of the decision making individual (Manion and Flowerdew 1982: 10). The approach was stimulated by the rational choice framework and shifts the focus from the analysis of optimal to actual behaviour (Simon 1957: 241; Cadwallader 1992: 12). While the approach still assumes that individuals act purposefully and with self-interest "to improve or maintain their quality of life" (Lu 1998: 1474), individuals make decisions under uncertainty and imperfect information. Their decisions may be objectively irrational for the reason that they follow an individually bounded rationality. Rationality is bounded, as the human capability to process information is limited and subjective perceptions of objective environments differ. Individuals are also assumed to be satisfied with less than optimal outcomes. Instead of maximising their utility, individuals satisfy particular situation-specific needs (Wolpert 1965; Golledge and Stimson 1997: 8).

The behavioural approach gives more importance to modelling the information processing and actual decision making before relocation than the rational choice approach. The decision process can be split up into the "migration decision, the location search and the choice process" (Golledge and Stimson 1997: 427). Some incongruence between aspirations and achievements in locations triggers the consideration to relocate. The incongruence is caused by events, e.g. childbirth or loss of job. In migration decisions after such events individuals have three choices. First, aspirations may be adjusted. Second, achievements are modified in the present dwelling, i.e. "in situ adjustment" (Jong and Fawcett 1981: 56). Third, individuals improve their achievements through relocation to re-establish equilibrium between aspirations and achievements (Brown and Moore 1970).⁶ If a relocation is decided for, the search process begins. Information about alternative locations is gathered and processed. The information is added to a cognitive map which is constantly constructed by individuals. In cognitive mapping, cognitive representations of places in geographic realities are created to which value judgements are attached (Downs 1981: 109). These cognitive maps guide spatial decision making. Cognitive maps also determine the awareness space of individuals, i.e. the geographic space known to the individual. The extension of this space is shaped by "the effects of race, family income, education, and occupation" (Wolpert 1965: 165). Relevant considerations of location choices are limited to selections from the cognitive map. Finally, a location is chosen that meets some minimal gain in achievement set by the searcher. The outcome of the decision may be re-evaluated and further spatial behaviour triggered (Golledge and Stimson 1997: 473ff).

Wolpert (1965) introduces the concept of residential stress to describe the "'mismatch' [...] between a household's residential needs and preferences and the characteristics of its current housing and neighborhood" (Lu 1998: 1474).⁷ The concept of stress is important for the behavioural approach and indicates its affinity with

⁶ In situ adjustment may also include buying or selling the present dwelling (Mandić 2001).

⁷ Quigley and Weinberg (1977: 57) suggest to operationalise residential stress as the "the amount of money required at the current residential location and quantity of housing consumed required to

psychology: Residential stress can be understood as a form of cognitive dissonance (Festinger 2001 [1957]; Deane 1990). Mismatches are caused by changes in the needs of individuals, e.g. transitions in the family trajectory of the life course, and external changes in the dwelling and neighbourhood, e.g. decay of buildings (Lu 1998: 1474). Furthermore, the normative standards of individuals regarding quality of locations may change and cause stress (Speare 1974: 175). However, residential stress does not directly cause residential relocations, because the satisficer only reacts to stress if it reaches a certain threshold level (Boyle, Halfacree and Robinson 1998: 64). In addition, residential relocations are not the only way to reduce stress. Stress may also be alleviated in situ (Brown and Moore 1970). The likelihood for individuals to relocate depends on the levels of satisfaction relative to the levels of stress that they experience and the "capacity to find and obtain access to more satisfactory housing" (Lu 1998: 1474).

The behavioural model predicts that individuals will relocate if they face an increase in residential stress above a certain threshold and have a low overall satisfaction with their residential locations. It can be assumed that individuals with larger awareness spaces are more likely to consider a relocation than individuals with smaller awareness spaces. Again, outcomes of relocations will be a matter of the information gathered by individuals. Individuals can make better informed choices regarding relocations, if they gather more information about potential locations and process this information correctly. If individuals make well informed choices, they are more likely to relocate to better locations, as they are aware of differences in the quality of various locations. Variation in relocation outcomes may also occur due to restricted awareness spaces, as locations in the awareness space of individuals may be worse than outside of the awareness space. In addition, particular dimensions of the quality of locations may compete. That is to say, that individuals may decide to improve one dimension of their location and accept negative outcomes in another dimension as a result (Wolpert 1965).

The behavioural approach offers a helpful extension of earlier rational choice models by highlighting the ideas of bounded rationality and satisfying behaviour. The approach is useful for my theoretical synthesis as the ideas of subjective preferences, limited information and bounded rationality provides appealing explanations for variance in relocation outcomes. However, the approach has certain shortcomings. The empirical operationalisation of the behavioural approach tends to be problematic, as subjective perceptions of reality are difficult to measure (Clark 1981: 194). In addition, the model overstates the decision of the mover and underestimates the conditions set by external forces, which may affect individual choices considerably (ibid.: 193). The theoretical framework considers structural constraints of information gathering and processing, but not further limitations of individual behaviour (Golledge and Stimson 1997: 485). The approach assumes that "migration

make the household as well off as it would be if it were currently consuming its preferred quantity of housing services at the optimal location".

is something that is always there and is a 'good thing' because it leads to better living conditions, higher wages and more pleasant residential environments" (Boyle, Halfacree and Robinson 1998: 65). Furthermore, the cornerstone of the behavioural approach is the "*decision-making unit*" (Golledge and Stimson 1997: 33, emphasis in original). For multi-persons households, the behavioural approach ignores the internal power structures of these decision-making units.

Wolpert (1965) linked the behavioural approach to a life cycle model by relating levels of residential stress to certain life cycle stages. The life cycle approach sees the family trajectory as following a set sequence of stages, i.e. marrying, bearing children and an empty nest phase with the same partner, and is rather static in its analysis. At the end of the 1980s and the beginning of the 1990s, the life cycle approach was extended to the life course perspective. This perspective is more dynamic than the life cycle approach and allows for variation in the sequence of events in the family trajectory and places trajectories in their historical times (Elder 1977; Morrow-Jones and Wenning 2005; Geist and McManus 2008). The life course approach is described in more details now.

2.1.5 Life Course Perspective

The life course perspective highlights the links between different life domains of individuals and interdependencies with other individuals. Central to the approach is the time dimension of these interdependencies over the span of individuals' lives (Mayer 2004: 163). One domain of the life course is the residential trajectory. A trajectory is a sequence of states that are linked through transition events (Elder 1985: 31f). Residential locations are states and relocations are transitions in this trajectory. The life course perspective has been joined with aspects of the institutional, rational choice and behavioural approaches in research on residential relocations (Mulder and Hooimeijer 1999: 169). Courgeau (1995: 22) summarises this comprehensive framework: "Individual migratory behaviour [...] develops over time, yet is situated within given historical, geographical, economic and social conditions [...] [and] placed in the time and space of an individual's life". Mayer (2004: 166) stresses four pillars of the life course perspective:

 Embeddedness and interdependence: Life courses are institutionally, historically and spatially embedded. Especially the institutional embeddedness in the context of particular welfare states has been emphasised by European life course researchers (Settersen and Gannon 2009: 456). "It was and still is the modern nation-state which lays the foundation for biographical decision-making and life course policy, with the consequence of national path dependence" (Heinz et al. 2009: 20) and, thus, the need for comparative cross-national research (Mayer and Schoepflin 1989). Furthermore, life courses are embedded in certain local spaces
and periods (Elder and Rockwell 1979). Life courses of individuals are interdependent (Elder 1977). "Individual lives are always linked lives [...]; one person's resources, resource deficits, successes, failures, chronic strains, and (expected and unexpected) transitions can become focal conditions, even turning points, in the lives of others" (Moen and Hernandez 2009: 259) and individuals are mostly situated in multi-person households that move as social convoys through time.

- 2) Multidimensionality: Different life course trajectories are closely linked and interrelated. The residential trajectory is intertwined with other life course trajectories foremost the job and family trajectory (Heinz et al. 2009: 25). The job trajectory has a fundamental influence on the structure of life courses. The job trajectory divides life courses in a pre-work phase of education, a work phase and a post-work phase in late life (Kohli 2009: 67). In addition, wages are major sources of income for most households. The place of work also restricts the potential places of residence depending on how much commuting an individual accepts. The family trajectory links together life courses of different individuals and influences their behaviour, e.g. due to child care responsibilities.
- 3) Self-referential process: Past developments in trajectories influence future transitions (Elder 1985: 31). Individuals act "on the basis of prior experiences and resources" (Mayer 2004: 166). For example, regarding residential relocations, past research shows that the probability of a relocation decreases with the time spent at a location.⁸ This may be caused by the accumulation of resources over time that can not be transferred to other locations, but also by the experiences gained at the present location.
- 4) *Individual agency*: Individuals creatively shape their life courses. That is to say that they *ex ante* act purposefully and intentionally to follow certain goals. At the same time they contribute to the reproduction and change of social structures they are embedded in. Thus, on the one hand, individuals are subject to the social structures that they have internalised and are simultaneously enabled and constrained in their actions by them. On the other hand, social structures are also shaped by actions of individuals (Settersen and Gannon 2009: 458).

As relocating is costly in various ways, individuals will not relocate without significant trigger events, i.e. substantial changes in their lives (Quigley and Weinberg 1977). Triggers are mostly important transitions in other life courses trajectories which induce new needs and preferences regarding residential locations (Myers 1999: 873). Dieleman (2001: 253) lists the following triggers: partnering, birth, union dissolution, death of a partner, entering or finishing stages in one's education and job changes that induce income changes or changes in commuting distance. The residential trajectory is instrumental and subordinated to other trajectories in the life course (Mulder 1993: 23). Different events can trigger distinct forms of residential relocations concerning the destination, e.g. housing adjustments after childbirth are assumed to trigger mainly short distance relocations (Courgeau 1985).

⁸ See cumulative inertia axiom in Section 2.1.3.

A trigger alone is not sufficient to explain residential relocations. Mulder and Hooimeijer (1999: 163) direct attention to the situation in which an individual is confronted with a trigger event. This situation is defined by the resources (enabling conditions at the micro level, e.g. material resources), opportunities (enabling conditions on the macro level, e.g. available housing), restrictions (limiting conditions at the micro level, e.g. discrimination on the housing market). The situational conditions determine whether individuals relocate in response to trigger events. The final decision to relocate is modelled using approaches similar to the rational choice (cf. Section 2.1.3) or behavioural approach (cf. Section 2.1.4).

Regarding restrictions of the relocation decision, the family trajectory is very influential. The lives of individuals in a household are linked through events in the family trajectory. The presence of a partner or child (-ren) in the household as well as their characteristics significantly influence the decisions of individuals to relocate. Especially important in shaping relocation decisions for couples is the employment status of the partner. Household members must negotiate a mutual agreement concerning relocations or split — at least temporarily (Mincer 1978; Nivalainen 2004).⁹ The past residential trajectory can provide resources as well as restrictions for relocating. Resources may result from past experiences with relocations. For example, the mobility socialisation in childhood years affects later life relocations (Courgeau 1985; Myers 1999). The likelihood of relocations increase with the number of past relocations (Clark and Huang 2004). Local social networks and other links to a location that grow over time may restrict relocations. The experience with particular locations in the past shapes the pool of potential future locations (Roseman 1983; McHugh, Hogan and Happel 1995). More generally, the past life course affects the present resources and restrictions of individuals. For example, an individual's economic capital is affected by the past job trajectory. These dynamics of social inequality have been an important theme in life course research as is shown below. Finally, opportunities and constraints of relocations in the life course perspective are similar to what has been described in the institutional approach (cf. Section 2.1.2).

In recent years, the life course perspective has increasingly been linked to sociological theories of social stratification (e.g. Dannefer 2003; Pallas and Jennings 2009; Diewald and Mayer 2009). This is especially promising for the present analysis, in which the dynamics of social inequality due to residential relocations are examined. Inequalities *of* the life course and inequalities *over* the life course can be differentiated.

"The former concerns the manner in which the relationships between the state, the market and the family generate social differentiation and social inequality between different population groups on the basis of various criteria, such as age [...] and gender. The latter refers to processes of differentiation or heterogeneity that unfold during the life course" (Dewilde 2003: 122).¹⁰

⁹ See also description of bargaining models in Section 2.1.3.

Regarding inequalities over the life course, relocations may be turning points in individuals residential trajectories that substantially affect the stratification of the quality of location. Turning points are "events or milestones that substantially alter the direction and/or slope of a trajectory" (George 2009: 169). Thus, turning points delineate distinct and discontinuous phases in trajectories. A turning point indicates an abrupt change between these two phases (Abbott 1997: 92). Turning points "may be crucial opportunities in which an accumulation of resources, for some, and a plateau or a decrease of resources, for others, may eventually create great inequalities between individuals of a single cohort" (Burton-Jeangros and Widmer 2009: 184). The effects of these turning points depend on the resources of individuals, e.g. income, on the social position in various life domains, e.g. the family status, and the general life stage as well as the past experience of individuals (Dewilde 2003; Elder and O'Rand 2009: 440). Some relocations that change relevant characteristics of the residential location are likely to be turning points in residential trajectories, e.g. from rental accommodation into ownership or relocations over long distances. The outcomes of these relocations are undetermined and contingent as such. Only the conditions of these relocations determine the direction that the residential trajectory takes after the event, i.e. the nature of the turning point. The within-changes in quality of locations also implies inequalities between different points in time for the same individual.

Regarding inequalities *of* the life course, residential trajectories of groups within one country may diverge. For example, the residential trajectories of different social classes may follow divergent shapes over the whole life course irrespective of single relocation events. Inequalities of the life course may also occur across nations due to divergent institutional contexts. For example, the role of home ownership in a national housing market will co-determine the age at which individuals seek to buy their first home and may have distinct effects on the overall shape of the residential trajectory. To uncover these cross-national differences, life courses from different countries must be compared.

Following the life course approach, residential relocations will be more frequent in life course stages with many transitions in other life course trajectories (Courgeau 1985; Mulder 1993: 25). For example, the stages between leaving the parental home until establishing a family and leaving school until starting the first job will probably have the highest number of residential relocations. It can also be expected that residential relocations are impeded, if different life courses are intertwined, e.g. in multi-person households. The life course approach does not explain relocation outcomes per se. However, it can be assumed that disruptive life course events, e.g. union dissolutions, trigger urgent relocation responses. These relocations may be less beneficial than relocations with longer prior search periods. Individuals may not compare several potential locations and decide for the best ones, but just relocate as

¹⁰ Inequalities *of* the life course would rather motivate cross-national research, while inequalities *over* the life course motivates longitudinal research (Dewilde 2003).

quickly as possible to the first available location (Dieleman 2001: 256). In addition, the interdependence of life courses in a household makes variance in relocation outcomes more likely, since a new location will only rarely meet the preferences of all household members at the same time (Mincer 1978). Particular life course stages such as the growing family phase may be associated with better relocation outcomes than other stages due to stronger preferences for good locations of parents with children and the willingness to make long-term investments in housing. The research on social inequalities and turning points also shows that outcomes of relocations, e.g. life course stage, and their past biographical experiences. Finally, variance in relocation outcomes may result from the institutional context in which relocations take place. To analyse this variance, cross-national research must be conducted to disentangle institutional context effects from other explanatory variables.

The life course approach is a perspective rather than a fully developed theoretical framework (Mayer 2009), but it is a perspective that is hardly contested in contemporary research on residential relocations. It has - more or less explicitly - become a dominant theoretical instrument to explain residential relocations (Kley and Mulder 2010). The approach is popular, as it offers a comprehensive explanation for the occurrence and non-occurrence of residential relocations. The life course perspective's attractiveness is also grounded in its flexibility and compatibility with other approaches. The life course framework improves on the other presented approaches in a number of ways: 1) Biographical dependencies within individual life courses are taken into account. 2) The timing of relocation events in certain life stages and in relation to particular life events is taken into consideration. 3) The life course perspective has been used to analyse social stratification over time. 4) Also, the life course perspective is based on the concept of individual agency, but at the same time acknowledges embeddedness in particular places, times and societies. Due to these strengths, the life course perspective is better suited to explain variance in relocation outcomes than the other approaches alone.

To recapitulate, the institutional approach is neglecting individual agency. The rational choice approach does not sufficiently account for contextual conditions and is too much focussed on individuals' utility maximising. The behavioural approach already provides a more appropriate explanation for relocation behaviour than the rational choice approach, but still does not sufficiently account for contextual factors, the biographical dependencies in individual life courses or the links between individuals in social convoys. The life course perspective combined with elements from the other approaches overcomes these shortcomings.

However, the shortcomings of the life course perspective itself should be considered as well. The flexibility of the approach is a strength, but may also turn into a problem, since the life course approach offers loosely linked theoretical concepts rather than a coherent and rigorous theoretical framework. Due to its vagueness, the framework is problematic to falsify. The approach's possibility to explain residential relocations may therefore result in a rather shallow account, without describing any one explanatory factor in depth. In addition, the life course approach presents empirical challenges because it calls for very comprehensive and complex life history data to do justice to the concept of biographical dependencies.

2.1.6 Conclusion

The presented theoretical approaches are very elaborate with regard to the causes of residential relocations. A main shortcoming of the presented approaches is that they mostly do not explicitly consider the outcomes of relocations. Most approaches that were presented subscribe to the implicit assumption that voluntary relocations are life course events that yield some kind of positive return for the movers. Instead, I argue that relocations are contingent events in individuals' life courses. Whether relocations become positive or negative turning points with regard to the quality of locations or have no substantive effect at all depends on several conditions. It is important to acknowledge this ambiguous nature of relocation events. Only then is it possible to fully understand the effect that relocations may have on individuals' lives. To this end, I combine elements of the just described theories to develop a combined theoretical framework in the remainder of this chapter. The framework draws mainly from the life course perspective, but incorporates elements from the rational choice, the behavioural and the institutional approaches.

The framework is based on the following basic assumptions about individuals' behaviour. I assume that individuals' behaviour is situated in the flow of time of the life course and always relates to not only the past, but the present situational context and potential futures. Individuals shape their lives by actively and creatively navigating through their life courses to pursue physical wellbeing and social approval (Lindenberg 1996: 169; Emirbayer and Mische 1998; Heinz et al. 2009). However, life courses are only partly "the result of active and free choices, partly created within a fixed set of possibilities and partly imposed from outside" (Settersen and Gannon 2009: 471). As individual behaviour is multi-conditioned, "decisionmaking becomes indeterminate and must be understood from the point of view of situational options, biographical experiences and individual life course expectations" (Heinz 2009: 477, emphasis in original). The situational options are defined by individuals' resources and restrictions as well as constraints and opportunities at the macro level, i.e. regional and national contextual factors (Mulder 1993: 20f). I now describe the determinants of residential relocations (Section 2.2), before turning to the explanation of variation in relocation outcomes (Section 2.3). Whether my theoretical framework is able to contribute to the explanation of variation in the outcomes of relocations is tested in the subsequent empirical chapters.

2.2 Explaining Residential Relocations

Individuals have preferences regarding their locations shaped by biographical experiences and develop in relation to the housing situations of others. Therefore, preferences are conditional on the historical and social context. Housing situations that are considered appropriate may be very different across different societies and historical times (Mulder and Wagner 1998; Rowlands and Gurney 2000). Preferences are dynamic and change over the life course as individuals grow older and their families, job situations and social statuses change. Locations may also change their characteristics over time as dwellings deteriorate, new neighbours move in or regional labour markets change. If individuals live at locations that do not meet their preferences, they experience stress which they want to alleviate if it reaches a certain threshold. To alleviate stress, individuals may relocate to locations that offer better fits with their preferences (Sirgy, Grzeskowiak and Su 2005). These voluntary relocations occur after intentional decision processes. In contrast, involuntary relocations occur, if movers do not have a choice to relocate, e.g. if they receive a notice by their landlord or are evicted. Even in the latter cases, I assume that individuals in European societies can intentionally decide where to relocate to within social, economic, geographical and practical limits.¹¹

Not every mismatch between preferences and actual location will trigger a voluntary relocation. First, stress must reach a certain level before it becomes relevant for the individual. Second, choices regarding relocations are costly and consequential and, thus, individuals will take this into account when considering to relocate. Costs are monetary and non-monetary and arise from one-time costs of relocating, localised capital that cannot be transferred to new residential locations and the risk and uncertainty attached to relocations (Goodman 1981: 138; DaVanzo 1981: 116; Lu 1998: 1476). Thus, individuals only relocate, if they have the necessary resources to meet the costs and they expect that the benefits from relocating will outweigh the associated costs. Individuals' expectations about benefits and costs are confounded by their intellectual capacities, incomplete information and habitual dispositions. Relocations can bring about fundamental changes in the living situations of individuals. Therefore, individuals are likely to weigh the pros and cons of their choices in some ways and make strategic calculations within the limits of their incomplete information and intellectual capacities.¹² The options of individuals are constrained by institutional conditions, e.g. the availability of suitable dwellings.

In their decisions about relocation, individuals also consider the alternative of in situ adjustment (cf. Section 2.1.4). When adjusting in situ individuals stay put, but

¹¹ However, certain subpopulations are deprived of their freedom of movement in European societies, e.g. illegalised immigrants or institutionalised individuals. This highly important issue is not considered in the present analysis.

¹² Jong and Fawcett (1981: 46) argue that residential relocations are more "amenable to a decisionmaking analysis than almost any other demographic behavior".

alter the residential location in some way to reduce discrepancies, e.g. by renovating the dwelling. This alternative will be more likely for individuals that would bear high costs in case of relocation, e.g. owners, or individuals that have strong habitual links to their location. Finally, individuals may adapt their subjective preferences to reduce stress. However, I assume that this alternative is chosen less frequent than the others, because of the habitually anchored preferences. Only if individuals have no other option, they will try to adapt their preferences. This is also done by legitimising their present situation to others and themselves. The habitual dispositions of individuals codetermine to what extent relocations are considered a viable behaviour at all (Myers 1999). For some, relocations is a goal in itself that is associated with new experiences and a desirable 'mobile' life style. For others relocations may be ruptures in their life courses that are avoided as much as possible.

In most cases, individuals are members of multi-person households living together with their partners, children, relatives or other individuals. Then, the relocation decision is made in the 'social convoy' of these households (Moen and Hernandez 2009). Thus, individuals must coordinate their behaviour if they want to stay together. They have to decide whether they want to relocate at all and where they want to relocate to. Finally, whether individuals relocate depends on the conditions on the housing market. Individuals can only relocate if sufficient numbers of locations are available and they only want to relocate if these locations meet their preferences. The characteristics of the housing market also determine the conditions under which individuals can relocate. For example, if transaction costs for buying property are low and only small down payments are necessary, home owners may be more likely to relocate than in housing markets in which transaction costs and downpayments are high. Finally, certain individuals may be discriminated in the housing market, e.g. migrants, and their relocations may be deterred.

2.3 Explaining Variation in Relocation Outcomes

Relocations are ambiguous and contingent life course events, i.e. there is strong variation in the outcomes of relocations.¹³ Following from the discussed theoretical approaches, five sets of conditions are identified that may explain the variation in outcomes after relocations: 1) *individual life course conditions;* 2) *economic resources;* 3) *situational conditions of decision making;* 4) *structural conditions in housing markets;* and 5) *gendered roles in couples.* Each set of conditions is de-

¹³ From a utility maximising perspective it could be argued that an optimal location exists for each individual. Under this assumption, any relocation to somewhere other than the optimal location would have a negative outcome, because the individual gained less than would have been possible. However, due to restrictions and constraints the optimal location is generally out of reach for individuals and the optimal location is merely of theoretical importance. Here the interest is in identifying conditions that cause individuals to relocate to more or less suboptimal locations.

scribed in the following subsections where hypotheses about the effects of these conditions on outcomes of relocations are derived. The set of conditions are not exclusive and overlap partly. This also implies that some of the processes behind these hypotheses are difficult to disentangle in the empirical analysis. Some hypotheses explicitly refer to the effects of relocations. Other hypotheses refer to general divergence in the quality of locations, but make implicit claims about relocations that lead to this divergence in quality. Hypotheses that refer to cross-national divergence anticipate the description of institutional differences given in Section 4.1. Thus, only important institutional differences are mentioned here.

2.3.1 Individual Life Course Conditions

At the micro level, individual behaviours are shaped by their "individual life course expectations" (Heinz 2009: 477, emphasis in original), resources and restrictions. Expectations or preferences mainly follow from habitual dispositions, period- and society-specific social norms and social comparisons with others (Morris, Crull and Winter 1976). Resources are material and non-material objects that are of societal importance and can be used to reach subjective goals. Resources are mainly economic, cultural, social and symbolic forms of capital (Bourdieu and Wacquant 1992: 119).¹⁴ Resources depend on past biographical experiences (Elder 1985). Restrictions are individual conditions that hinder the actualisation of subjective goals. Restrictions may result from a lack of resources and from life course interdependencies. The latter follows from two processes. First, biographical dependencies withinindividual life courses restrict present behaviour. Past choices restrict the potential present choices and past choices make particular present choices more likely. Second, inter-individual relations within and beyond the household restrict behaviour, because individuals have to coordinate their behaviour with others. While individuals are in pursuit of their own goals, they also cooperate with other individuals and their life course are linked to others. This link is especially strong within couples living together and individual behaviour cannot be understood without considering these interdependencies.

Preferences regarding locations differ strongly by life course stage (Lindberg et al. 1992; Jansen 2012). On average, couples have preferences for better locations than singles for several reasons. First, the location is an important component of settling down together as a couple and 'building a nest' (Rossi 1955: 178; Rowlands

¹⁴ Economic capital has the form of material assets and income. Cultural capital can be incorporated, i.e. in the form of dispositions, taste and the knowledge of how to act appropriately in a given situation, objectified, i.e. in the form of books or art works, and institutionalised, i.e. in the form of educational degrees. Social capital refers to the social networks of individuals and the resources they can access through these networks. Symbolic capital "is the form that the various species of capital assume when they are perceived and recognized as legitimate" (Bourdieu 1989: 17).

and Gurney 2000; Kearns et al. 2000). Second, couples put more weight on the quality of their residential locations, as they are less mobile than singles. If they relocate they will try to improve their locations more than singles who may anticipate to relocate again soon. Couples may be more willing to invest into their locations, since they are likely to live there for longer than singles. This may be even more so for married couples, on the grounds that they may expect to stay together for longer than cohabiting couples (Mulder and Wagner 1998; Holland 2012). Third, couples are able to pool their resources and are more likely to afford to relocate to more expensive high quality locations than singles (DeLeire and Kalil 2005). Children may have an additional effect on preferences of parents regarding the quality of locations. Parents will strive to provide good quality locations for their children, e.g. in safe neighbourhoods (Brun and Fagnani 1994). However, children also put financial stress on families. After childbearing, families may not be able to afford good locations or they have to relocate to less expensive areas, e.g. suburban areas. This will be the case foremost for low income families. I formulate the following hypotheses, which should be true on average.

H 1.1. *Relocations of singles to form a couple household and relocations of couples improve the quality of locations more than relocations of singles.*

H 1.2. Partners live in better locations if they are married than before and after being married.

H 1.3. *a) Individuals improve the quality of their location if they have children and b) the positive effect of children on quality of location will increase with income.*

Individuals will live in worse locations after union dissolution, since, first, they do not pool their economic resources any longer and the dissolution may have decreased the individual economic resources due to foregone economies of scale. Second, individuals decide about relocations under stronger uncertainty about their future family trajectory than while being in stable relationships. Third, for childless ex-partners, different preferences about locations may apply that put less weight on particular features of locations such as safe neighbourhoods. Finally, individuals that dissolved their unions have to relocate quickly and do not have time for a thorough strategical calculation of the benefits and costs of various residential locations. It may be expected that union dissolutions have a lasting effect on the quality of locations. Economic resources may be reduced permanently, ex-partners may be less likely to form a new union and they may be less inclined to invest again in building a 'nest' if they already lost one. These processes may differ by gender, because typical life courses of men and women differ. Traditionally, women in couples with children are more engaged in household work and child care than in paid work on the labour market (Bussemaker and Kersbergen 1999: 18). In case of union dissolution, women are often left with less financial resources than the male partner --also in childless couples (Christopher et al. 2002). Due to child care responsibilities that are mainly left with the mother after dissolution, engaging in paid work is

more difficult for women in the post-dissolution phase. Women are less likely to repartner after dissolutions than men with increasing age (Jaschinski 2011). All these factors may cause relocations in association with union dissolutions to have stronger negative effects on women than on men.

H 1.4. *a)* Individuals reduce the quality of their locations, if their relocations are associated with a union dissolution. b) This effect is stronger for women than for men, and c) is persisting over the life course.

Important events in the work trajectory can cause relocations to become turning points in the residential trajectory (Feijten and Mulder 2005). Becoming unemployed may cause a relocation to be a negative turning point, due to the loss of income, income security and status. Upward or downward job mobility may change the income and, thereby, the economic resources that can be used for relocations. Job mobility may also affect the preferences regarding locations. After significant promotions, individuals may want to represent their new social positions by relocating into better residential locations (Rossi 1955: 179). Personal characteristics of individuals may be related to changes in the job trajectory and relocations at the same time. For example, individuals may strive for high incomes and high quality locations simultaneously as a result of their competitive and ambitious character. Income losses in case of unemployment may be cushioned by the welfare state through wage replacement. If the wage replacement is high and close to the former labour income, individuals that become unemployed are more likely to stay in their location. If the wage replacement is low, individuals are more likely to relocate to reduce housing costs. In the latter cases, individuals are likely to reduce the quality of their location.

H 1.5. *Transitions into unemployment and demotions reduce the quality of locations on average, while individuals that become employed or experience a promotion in their job are more likely to improve the quality of their locations than individuals that do not experience the respective job mobility.*

Life courses in England are less stable and homogeneous than in Germany on average, i.e. individuals experience more heterogeneous transitions and changes in their life courses in England than in Germany. Individuals are more likely to experience transitions in their family trajectories, i.e. union forming and dissolution as well as childbearing, in England than in Germany (cf. Section 4.1.3). Thus, changes in their room needs and general residential preferences are more likely for individuals in England than in Germany. English individuals are more at risk to experience downward social mobility than German individuals on average due to weaker job security and social protection in England (cf. Section 4.1.3). English individuals are also more likely to experience disruptive family events on average. Therefore, it may be expected that English individuals are also more likely to experience downward residential relocations that reduce the quality of locations. It follows that the quality

of locations for English individuals may be more volatile over the life course than for Germans. This cross-national divergence will be especially strong in mid- and latelife, when German individuals experience even fewer changes in their preferences and needs compared to English individuals due to their more stable life courses. In the early life course, the divergence will not be as strong, because Germans also experience frequent transitions in their family trajectories.

H 1.6. The quality of locations is more stable over the life course in Germany than in England on average, especially in mid- and late-life.

The quality of locations is closely associated with housing tenure. Ownership is associated with better quality of locations on average compared to social housing and privately rented tenure (cf. below). In England, individuals are more likely to buy property earlier in the life course than in Germany and in England individuals are also more likely to relocate to more than one owned home over the life course than in Germany (Voigtländer 2009: 366ff). In England, home ownership is not as directly linked to marriage and family formation as in Germany (Mulder and Wagner 1998: 707). ¹⁵ Thus, it can be expected that English individuals are likely to increase the quality of locations earlier in their lives and regardless of their family status, while German individuals relocate to better locations later in life and are likely to relocate to better locations especially if they are married or form a family.

H 1.7. Changes in quality of locations over the life course follow distinct patterns in England and Germany: a) on average improvements in the quality of locations through relocations are more likely for young singles and childless individuals in England than in Germany; b) improvements in the quality of location through relocations are more likely in the growing-family phase in Germany than in England; and c) downward relocations are more likely in England than in Germany.

2.3.2 Economic Resources

To reach their goals of physical wellbeing and social approval, individuals constantly compete for scarce material and symbolic objects. Since these objects are rare, only some individuals can use them to meet their goals. As a result, life courses of individuals are unequal in the sense that the degree of goal attainment varies between individuals and for individuals over time. Individuals compete for high quality residential locations in housing markets. The supply of locations is limited and, in general, the scarcity of locations increases with higher quality. Thus, only a limited number of individuals will be able to relocate to high quality locations. In most European countries, housing markets are mainly characterised by free market

¹⁵ Also cf. Section 4.1.1.

exchange. Thus, access to scarce locations is mainly limited through the price of locations. The better a location, the higher its price will be everything else equal. The price of a location is either the rent in rental tenure or the house price respectively land price and construction costs in the owned tenure. Individuals that can afford to pay high prices can relocate to high quality locations. Thus, if individuals have more economic resources, they are more likely to improve the quality of their location on average to increase their degree of goal attainment and relocations will have better outcomes than for individuals with less economic resources.

H 2.1. *a)* Individuals that increase their incomes improve the quality of their location through relocations; b) relocations have an increasingly positive effect on the quality of locations with higher incomes.

The access to different housing tenures depends highly on current and potential income (Kendig 1984; Buck 2000a).¹⁶ On average, quality of location is strongly correlated with housing tenure. Owned dwellings offer the best quality, dwellings rented privately offer less quality and social housing offers the least quality (Easterlow, Smith and Mallinson 2000; Mulder 2006). Owners can better adapt their dwellings to their needs and for many owners home ownership offers a feeling of security (Megbolugbe and Linneman 1993). The reason for these differences are manifold. First, since owners commit for a longer time than tenants to their dwelling, they invest more economic capital. Second, because of lower land prices owned houses are more likely to be situated in less densely populated areas, which offer better neighbourhood quality. Social housing is often associated with the worst locations, as the investment in the dwellings is relatively low as only small returns can be expected.

H 2.2. Individuals that relocate into social housing will improve the quality of their locations the least, individuals relocating into dwellings rented privately will improve the quality moderately and individuals that relocate into ownership with or without mortgage will improve the quality the strongest.

Within each housing tenure there may be high variation in quality — especially across countries. If a housing tenure has a high share in the housing market, it is likely that dwellings in this tenure are more heterogeneous to cater for a wider group of the population. For example, the high share of owner occupied dwellings in England implies that there are more own homes with relatively low quality compared to the owner sector in Germany (cf. Section 4.1.1). To the contrary, the high share of rented dwellings in Germany implies that there are more high-quality rented dwellings in Germany than in England. Thus, Germans that relocate between two rented accommodations may be more likely to improve the quality of their locations than similar movers in England. At the same time, English movers that live

¹⁶ Wealth is an important type of economic resources that may improve outcomes after residential locations (Di Salvo and Ermisch 1997; Mulder and Wagner 1998). Due to problems of measuring wealth over time with the data used in the present analysis, no hypotheses regarding wealth are formulated.

in their own homes before and after a relocation may be more likely to improve the quality of their locations with these relocations. Germans are likely to relocate only once to ownership over their life course, while individuals are more likely to relocate between own homes in England. For Germans, it may be expected that this rare event is thoroughly planned and utilised to increase the quality of location more strongly than for individuals in England.

H 2.3. *a)* Transitions from rented dwellings to home ownership, and b) relocations between privately rented dwellings improve quality of locations stronger in Germany than in England. c) Relocations between own houses improve quality stronger in England than in Germany.

2.3.3 Situational Conditions of Decision Making

Information about potential locations is essential in making a relocation decision, but individuals are not fully informed about all alternatives and consequences in their decision making. This is a matter of the costs of searching and the available time for searching. Costs of information increase with the distance between the present and the potential locations that information is acquired for. The information about distant locations will be limited considerably, but particular forms of information about locations is easier to obtain than others. For example, information on the available rooms of a dwelling is easier to acquire than information on features of the neighbourhood.

H 3.1. *a)* Individuals that relocate farther away improve the quality of their locations less than individuals relocating short distances and b) this effect is stronger for changes in neighbourhood quality than for changes in room stress.

My discussion so far has mainly focussed on voluntary relocations. Involuntary relocations occur, if movers do not have a choice to relocate, e.g. if they receive a notice by their landlord or are evicted.¹⁷ It can be expected that these events have strong negative effects on individuals, since individuals have to relocate without being able to plan the relocation thoroughly. They cannot evaluate different potential locations, but have to relocate somewhere quickly. Thus, the outcome may be more negative than if they would have controlled the situation and would have made a deliberate decision about relocating. In addition, involuntary relocations may be more

¹⁷ At this point, it has to be noted that structurally triggered involuntary relocations, e.g. social and ethnic invasion-succession processes in neighbourhoods, are not explicitly accounted for in the present analysis. In recent years, the gentrification debate has highlighted the consequences of such processes (e.g. Millard-Ball 2002). For example, low-income renters may be driven out from neighbourhoods that are subject to gentrification processes and may have to relocate to lower quality neighbourhoods. I do not explicitly analyse these processes because of problems with measuring neighbourhood-level change in my data.

likely for specific groups of the population, which are more prone to deteriorate the quality of their location through relocating. For example, evictions and resulting involuntary relocations may be more likely for individuals that could not pay their rent and these individuals may also be discriminated against by potential landlords who do not want to rent to individuals associated with the risk of not paying rent.

H 3.2. Individuals improve the quality of their location less on average, if they relocate involuntarily compared to relocating voluntary.

2.3.4 Structural Conditions in the Housing Market

Similar to the institutional approach, I assume that individual behaviour over the life course is also shaped by the institutional context (Mayer and Schoepflin 1989). "Institutionalization of life courses refers to the process by which normative, legal or organizational rules define the social and temporal organization of human lives" (Brückner and Mayer 2005: 32). These rules may hinder the actualisation of subjective goals and constitute constraints. Rules may also facilitate goal attainment and offer opportunities to reach subjective goals. Institutions may affect individuals' life courses differently. For example, the institutional context in the labour market and the welfare system facilitate the genderisation of women's and men's typical life courses (Krüger and Baldus 1999).

In general, the quantity and characteristics of locations available in the housing market will determine the opportunities that individuals have to relocate to better locations. The more high quality locations are available in a housing market, the better are the chances that an individual may relocate to one of these high quality locations, as they are not scarce and thus their price will be modest compared to housing markets in which high quality locations are scarcer. This reflects a fundamental axiom in life course research: "the structural availability of open positions is simply the precondition to participate: without vacant positions there is no career mobility" (Diewald and Mayer 2009: 7). Besides supply, the demand in the housing market is important. Tight housing markets due to high demand of housing will offer less opportunities for individuals, since the competition for high quality locations is strong. If the competition is strong, prices will rise for vacant dwellings and fewer individuals will be able to afford relocating to these locations. Less tight housing markets, on the other hand, will offer more opportunities, because prices for dwellings decrease.

H 4.1. Relocating to housing markets with high demand for housing reduces the chances for individuals to improve the quality of their locations, while relocating to housing markets with high supply of housing increases the chances for individuals to improve the quality of their locations.

H 4.2. Individuals will improve the quality of their location more on average, if they relocate in a housing market with many high quality locations compared to a housing market with few high quality locations.

Housing markets are examples for very imperfect markets.¹⁸ Welfare states heavily intervene in housing markets, due to these limitations of the market and the political goal to provide everybody with adequate housing (Buck 2000a: 135). Housing policies influence the supply, prices and quality of locations and thereby distort the potential direct link between individuals' incomes and their quality of locations.¹⁹ The intervention of the welfare state in housing markets aims at reducing disparities in the quality of locations for individuals. Welfare states differ in the extent to which they intervene on the housing market and, thus, in the extent to which they distort the income-quality link (MacGuire 1981: 115; Balchin 1996a: 9ff; Oxley and Smith 1996: 25ff; Lampert and Althammer 2004: 337). More regulation will result in a weaker effect of income on the quality of the location. In Section 4.1.1, I show that the housing market in Germany is slightly more regulated and the income-quality link is weaker than in England. With regard to other aspects of the welfare state, England can also be characterised as being less decommodified, e.g. with regard to unemployment protection (cf. Section 4.1.3).²⁰

H 4.3. *Changes in the quality of locations through relocations will depend more on changes in income in England than in Germany.*

Ascribed statuses such as ethnicity of immigrants can be associated with discrimination in the housing market and may constitute a strong constraint for individual relocation behaviour (Flowerdew 1976: 56). On average, ethnic minorities live in lower quality locations than the native population (e.g. Özüekren and Kempen 2002; Drever and Clark 2002). There are three main causes for these disparities.

¹⁸ First, housing markets are characterised by price inelasticity of supply. Second, due to the fragmented character of housing markets, the high number of sellers and buyers and the high diversity of dwellings, the housing market only has a low market transparency. Housing markets are also regionally fragmented and conditions for relocations will be very different across regional borders of housing markets. Third, consumers in the housing market are mostly limited in their geographic mobility. Thus, their search is locally bounded and only covers a small fraction of the overall market. Fourth, most locations are not substitutable (Herlyn and Herlyn 1983: 127ff; Mayer 1998: 37).

¹⁹ First, welfare states define legal standards of housing that are supposed to guarantee a minimum quality of dwellings. Second, welfare states provide subsidies and tax benefits for home owners, and pay housing assistance to renters. Thereby, individuals are enabled to improve their locations beyond their individual labour market income and assets. Third, private providers of dwellings are supported financially through subsidies and tax benefits. Fourth, welfare states engage in urban and regional planning. Fifth, states regulate the market exchange between consumers and providers, e.g. by introducing rent regulation and legal protection of renters.

 $^{^{20}}$ Decommodification describes the degree to which life chances of individuals are independent from market forces.

1) Preferences of ethnic minorities regarding features of locations may be different from natives. For example, ethnic minorities may prefer ethnic enclaves. While ethnic concentration in a neighbourhood is not a criterion of neighbourhood quality in itself, it may be a feature associated with low neighbourhood quality on other dimensions (Harris 1999). In the present analysis, I assume that immigrants and natives share similar preferences for residential locations.²¹ This is a strong assumption that should be further validated in future research, but past research supports the credibility of this assumption (e.g. Bolt and Kempen 2002; Wiesemann 2008). 2) Immigrants may be constrained in the actualisation of their preferences, because immigrants on average have less economic capital. 3) Actualisation of preferences may also be impeded through discrimination on the housing market. In this case, gatekeepers such as landlords or real estate agents prefer natives over immigrants. The latter can only relocate to locations that are not preferred by natives (South and Crowder 1998a; Quillian 2003; Schaake, Burgers and Mulder 2010). Discrimination is likely to affect all individuals that live together with an immigrant in the same household, even if they are natives themselves. Even if immigrants relocate to better residential locations, the relative difference to the residential quality of the native population may not change as long as natives improve their locations to a similar or even greater degree. Immigrants may also, on average, fare worse concerning their location outcomes, because of structural inequalities in access to resources. Spatial assimilation, i.e. reaching the same quality of location as natives, is a function of the time that an immigrant has lived in a country and the degree of assimilation into the mainstream society (Bolt, Kempen and Ham 2008). Therefore, it can be expected that first generation immigrants differ stronger from natives than second generation immigrants.

H 4.4. *a)* Individuals in immigrant households improve the quality of their locations less than natives through relocations. *b)* The difference to natives is stronger for individuals in first than for individuals in second generation immigrant households.

2.3.5 Gendered Norms of Behaviour

Individuals in dual-earner couples are constrained in their residential relocations, as their life courses are linked and both careers have to be considered in the decision whether and where to relocate (for linked lives cf. Elder 1998). Since job opportunities are dispersed in geographical space and job offers emerge at relatively random times, it is unlikely that both partners will receive equally good job offers at a new location at the same time (Mincer 1978). Therefore, long-distance relocations can

²¹ I make this assumption, because my data does not allow to measure locational preferences.

be expected to have divergent effects on labour market outcomes of both partners. Long-distance relocations can be assumed to have a stronger effect on labour market outcomes than short-distance relocations, on the grounds that after long-distance relocations the distance to the old place of work will increase substantially in most cases and as individuals usually prefer to limit their commuting they are more likely to change their jobs after long-distance relocations (Mincer 1978; Smits 1999).

Gender is an institution which structures social life (Lorber 1994: 6). Individuals are categorised as women or men and faced with societal norms about genderappropriate behaviour and (re-) produce their gender identities in daily interactions with others. Thus, individuals do their gender in a constant process and gender is not a static feature of individuals (West and Zimmerman 1987; Gildemeister 2008). Traditional gender norms of male breadwinners and female family caretakers are still highly relevant in modern societies. According to these norms, women are more engaged in family work than in paid work on the labour market and even in childless couples men's careers are mostly prioritised (Hochschild 1989; Bussemaker and Kersbergen 1999). In their everyday behaviour, women and men are affected by these norms. Even couples in which partners have non-traditional gender role attitudes live in a gendered social structure and a social environment that mostly follows these gender norms. Thus, even the behaviour of partners with non-traditional gender role attitudes is constrained, e.g. due to better average labour market conditions for men (Risman 2004). Following this perspective, decisions to relocate are not rationally maximising the economic returns for the couple as predicted by the rational choice approach, in which the partner with more human capital is assumed to dominate the relocation decision (cf. Section 2.1.3). Relocation decisions are also not the result of explicit bargaining between partners as predicted by the bargaining models, which assume that partners with more relative power will assert their interests (cf. Section 2.1.3). Instead, the priority of men's careers in relocation decisions is subtle and based on implicit gender norms (Bielby and Bielby 1992; Cooke 2008b; Abraham, Auspurg and Hinz 2010). Women's potential losses after relocations are considered to be less important than men's gains and his career will often be prioritised in the decision to relocate.

Following this argument, I expect couples to prioritise men's careers on average irrespective of both partners' human capital and occupational position as well as the partners' share of the couple's labour income as an indicator for relative resources. Thus, I expect that residential relocations of dual-earner couples benefit men's careers more than women's careers. In addition, I expect women to be tied movers on average. That is to say that their individual careers are adversely affected by the relocation compared to staying put. I derive the following hypotheses.

H 5.1. *Partnered women that relocate are negatively affected in their careers compared to women staying put.*

H 5.2. In couples, men will profit more from long-distance relocations than women with regard to their careers — even controlling for both partners' human capital and past work history.

Gender as a social institution varies between countries and as a result conditions for female labour participation diverge (cf. Section 4.1.4 for a detailed description of these conditions). Thus, one can expect the effect of long-distance relocations on women's careers to vary by country. Tax, welfare and family policies do not hinder women's continuous labour force participation in England. Nonetheless, women are mainly supplementary contributors to the family income and the labour division in the household is more traditional than for example in the Scandinavian countries (Kilkey and Bradshaw 1999: 149; Sainsbury 1999: 250; Crompton and Lyonette 2006; Cooke 2011: 31ff). In Germany, tax policies, extensive family policies and welfare policies that focus on single breadwinners in the family hinder women's continuous employment. Women are pushed into the role of family caretakers and the division of labour in German families follows the male-breadwinner model on average (Bussemaker and Kersbergen 1999: 18; Sainsbury 1999: 247; Blossfeld and Drobnič 2001: 40; Apps and Rees 2005). Societal norms about gender roles differ between England and Germany, but also within Germany. Treas and Widmer (2000: 1421) categorise East Germany as having "work oriented" gender norms, where attitudes are more favourable for working mothers than in other countries. West Germany and England belong to the cluster of countries with "family accommodating" (ibid.: 1422) gender ideology, where mothers of young children are expected to stay home and mothers of school-age children are expected to work only part-time.

The two countries are also highly divergent cases in respect to their labour markets (cf. Section 4.1.2). Germany is a coordinated market economy, while England is a liberal market economy (Hall and Soskice 2001: 8ff). As a result, one can observe higher job mobility in England compared to Germany. Due to the higher overall job mobility and better opportunities for taking up new jobs, relocations may be less disruptive for careers in England than in Germany. As a result of the slightly higher labour market involvement of women in England, more egalitarian gender role norms and the slightly less gendered family policy, I expect long-distance relocations in England to affect women's careers less adversely than in Germany. I also expect West German women to be stronger adversely affected by relocations than East German women, because of the differences in gender norms between East and West Germany. These expectations lead to the following hypothesis:

H 5.3. Gender inequality in effects of long-distance relocations on careers will be weakest in England, modest in East Germany, and strongest in West Germany on average.

2.4 Summary

The presented combined theoretical framework extends previous theoretical approaches by treating relocations as ambiguous and contingent life course events. The occurrence of relocations has been thoroughly theorised in the past, but the variation in relocation outcomes has been mostly neglected. My theoretical framework explicitly aims at explaining this variation. It mainly draws from the life course perspective to explain the occurrence and outcomes of relocations, but also incorporates elements from the institutional approach, the rational choice approach and the behavioural approach. I follow the life course approach in the assumption that individuals actively navigate through their life courses to reach certain goals. I also use the concept of an embedded and self-referential life course that consists of several interdependent trajectories and which is linked to other life courses. Life courses are embedded in institutional contexts at the regional and national level that shape the occurrence and the outcomes of relocations as described in the institutional approach. Following the behavioural approach, it is assumed that individuals have preferences regarding their locations and that they want to alleviate stress that results from incongruity of their preferences and actual location by relocating. It is assumed that individuals make intentional decisions about their relocations by weighing costs and benefits of a relocation, but individuals only have limited information for their choices and restricted resources to relocate as shown in the rational choice approach.

Regarding the occurrence of relocations, it is assume that individuals consider a voluntary residential relocation, as their present locations do not meet their preferences any longer. Mostly, this happens after individuals have experienced transitions in their family and job trajectories. Individuals can decide to relocate to another location, adjust their present location or adjust their preferences. Individuals strategically calculate the pros and cons of this decision in the limits of the available information and their intellectual capacities. The decision is further shaped by individuals' habitus, resources, and restrictions as well as opportunities and constraints at the contextual level. The decision is embedded in the flow of time over the life course. Individuals in multi-person households must coordinate their relocation behaviour in some way, or split.

Relocations may impact individuals' life chances very differently depending on the conditions of these relocations. Five sets of conditions are identified that may affect this variation in outcomes. To recapitulate, these sets of conditions are:

 Individual life course conditions: The residential trajectory is instrumental and depends on transitions in other life course trajectories, e.g. the family trajectory. These instrumental relocations will be shaped by triggering transitions. Especially important are turning points, such as union dissolution, which may have adverse effects on relocation outcomes. Further, life courses of individuals are linked to other individuals and these interdependencies may affect variation in relocation outcomes.

- 2. Economic resources: Housing is a good traded in a market and, therefore, outcomes will be strongly determined by the price that individuals can pay in this market. Individuals may use their current incomes or take on mortgages to relocate to better locations. Depending on which housing tenure they relocate into, the quality of location will differ with home ownership offering the best locations.
- 3. Structural conditions in the housing market: Some individuals are discriminated against in the housing market, e.g. because of their ethnicity. Furthermore, the outcomes of relocations are shaped by the quantity and quality of the supply in the housing market as well as the demand.
- 4. Situational conditions of decision making: The information about potential locations are limited and the processing of this information may be impeded by individuals' capacities.
- 5. Gendered roles in couples: Due to traditional gender norms, men's careers are likely to be prioritised if couples consider long-distance relocations. Therefore, men are likely to benefit more from long-distance relocations than women. Women may also be adversely affected by long distance relocations in their careers compared to stayers.

Chapter 3 Review of Past Literature

In this chapter, past literature on determinants and outcomes of residential relocations in the areas of housing, neighbourhood and labour market are reviewed. By providing an overview of past research, the theoretical argumentation and empirical analysis is contextualised. The literature review also helps to identify specific short-comings in past research that are tackled in the present analysis.

First, I present research on the determinants of residential relocations (Section 3.1). The literature on the causes of residential relocations is abundant and at this point only main findings are reported and recent innovations highlighted. In Section 3.2, literature on residential relocations and changes in room stress is reviewed. Section 3.3 is concerned with the effect of relocations on neighbourhood quality and Section 3.4 deals with literature on labour market outcomes of long-distance relocations for dual-earner couples. In the latter three sections, I briefly discuss the relevance of these outcome areas and then review the literature on the determinants of variation in relocation outcomes in each respective area. Regarding the literature on neighbourhood quality changes, I also briefly review findings on neighbourhood effects and issues in the measurement of neighbourhood quality (Section 3.3.2 and Section 3.3.3).

3.1 Determinants of residential relocations

Derived from my theoretical reflections, individuals are expected to consider a voluntary residential relocation, if their present locations do not meet their preferences any longer (cf. Section 2.2). Mostly, preferences and present locations are not congruent, if major transitions in the family and job trajectories occurred. Past research shows that transitions in the family and job trajectories increase the probability of residential relocations (e.g. Mulder 1993: 95ff; Myers 1999; Li 2004).^{1,2} Regarding the family trajectory, the probability of relocation increases strongly after the first conception, but falls again before the birth and is back to normal shortly after the birth. Thus, individuals relocate anticipating events in their family trajectory. Subsequent fertility events do not affect the probability to relocate to the same degree (Kulu 2008; Michielin and Mulder 2008). Changes in partner status are important triggers for relocations (e.g. Clark and Huang 2004; Aarland and Nordvik 2009). Union formation and especially marriage is often followed by a relocation (Wagner 1989: 143ff; Mulder 1993: 183), or a relocation is undertaken in anticipation of such an event (Feijten and Mulder 2002). Union dissolutions trigger relocations of at least one of the partners in almost all cases.³

It is shown that changes in job status may cause relocations (Böheim and Taylor 2002; Clark and Huang 2004). Triggers from the job trajectory are more likely to cause relocations in early stages of the career (Wagner 1989: 188). Vlist et al. (2002) find job changes to be less important than family changes to explain residential relocations. While the unemployed tend to be more mobile than the employed (Parkes and Kearns 2003; Kulu and Billari 2006), longer unemployment spells reduce chances of relocations (Böheim and Taylor 2002; Windzio 2004). Recent research highlights the endogeneity of the residential trajectory and other life course domains such as the job trajectory, i.e. both may be subject to linked decisions of individuals (Mok 2005). Similarly, Böheim and Taylor (2002) show that unobserved heterogeneity that affects residential relocations is correlated with unobserved heterogeneity that affects job changes.

Age is found to be an important explanatory factor in almost every empirical analysis of residential relocations and across diverse national contexts (e.g. Kearns and Parkes 2003; Clark and Huang 2004; Kulu and Billari 2006). However, it can be argued that age itself does not have a causal effect on the likelihood of relocations, but instead age is a proxy for life course stages and life course transitions (Morrow-Jones and Wenning 2005). Young adults from the beginning of their twenties to the middle of their thirties are found to be the most mobile age group, because of frequent transitions in their family and job trajectories. With growing age after this phase, the probability for relocations decreases due to cumulative inertia (cf. Section 2.1.3). When reaching retirement age, the probability increases again (Gobillon and

¹ The education trajectory is understood as a part of the job trajectory here.

² It has been argued that transitions in the family trajectory mainly trigger short-distance relocations, while transitions in the job trajectory causes long-distance relocations. However, this differentiation seems to be too rigid, because, e.g., a considerable share of long-distance relocations are motivated by family reasons (Boyle, Feng and Gayle 2009: 111).

³ See Section 5 regarding the endogeneity of residential relocations and union formation and dissolution.

Wolff 2011). Furthermore, age functions as a proxy for health and local capital as well as cohort effects.⁴

Life course research highlights the importance of biographical dependencies and it can be expected that present transitions in the residential trajectory are shaped by past developments. The time in residence is found to reduce the probability of residential relocation significantly (Wagner 1989: 158; Frick 1996: 183). This supports the cumulative inertia axiom (see section 2.1.3). Similar to age however, the variable time in residence is a proxy for local ties that cannot be transferred to new locations and may not have a causal effect on relocations. The probability for further residential relocations increases with the number of past relocations (Kulu 2002: e.g. Clark and Huang 2004).

In general, being in a partnership reduces the risk of relocations (Kearns and Parkes 2003; Clark and Huang 2003; Michielin and Mulder 2008). Böheim and Taylor (2002) show that especially being in a partnership with an employed partner reduces the probability of relocation in the UK. The negative effect of being in dual-earner couples is also found for Germany (Jürges 1998). Findings on the number of children are ambiguous. In Germany, the overall number of children aged 16 or younger does not have a significant influence on residential relocations (Clark, Deurloo and Dieleman 2000), but the presence of children younger than three years has a positive effect (Frick 1996: 178). It is also found that the presence of older children that can be anticipated to move out of the parental nest soon reduces the likelihood of parents to relocate (Deurloo, Dieleman and Clark 1997). In the UK, the risk of residential relocations is increased if the youngest child in the family is younger than 16 years old (Böheim and Taylor 2002). More generally, the probability of residential relocation decreases with household size, because this increases the costs of relocations (Kearns and Parkes 2003).

The effects of individual resources on the probability of residential relocations are described now. Significant increases or decreases in income raise the probability of relocations (Frick 1996: 184). The overall level of income does not have a consistent effect on the probability of residential relocation (Wagner 1989: 120). Vlist et al. (2002) find a negative effect for tenants and a positive effect for owners, while other studies do not find a significant effect of income at all (Kearns and Parkes 2003; Clark and Huang 2003). For Germany, Frick (1996: 184) finds a positive and significant effect of income on the probability of residential relocation over short distances (see also Clark and Drever 2000). This is also supported by Böheim and Taylor (2002) for the UK.

Concerning education, higher qualification is found to influence the probability of residential relocations positively (Li 2004; Kulu and Billari 2006). This is

⁴ Morrow-Jones and Wenning (2005) raise the issue that age can be a proxy for very different background variables, which may have contrary effects on relocations. However, comparing different models with age as a proxy or certain life course variables, only small differences in model fit were found.

probably due to disperse job locations for highly-qualified jobs, the concentration of qualified jobs in metropolitan agglomerations and past experience with relocations that were motivated by attending higher education. Manual workers are found to be less mobile than other occupational groups in the UK (Böheim and Taylor 2002). However, for Germany, Wagner (1989: 194) shows that workers without any qualification are more likely to relocate over short distances than other occupational groups.

I now turn to the effect of locational characteristics on residential relocations. Locational characteristics may be indicators of the potential (mis-) match of preferences and actual locations, e.g. with regard to room stress. Certain locational characteristics such as home ownership may also reflect localised forms of capital that impede mobility. Indeed, owners are found to be much less mobile than tenants across various countries (e.g. Frick 1996: 182; Li 2004; Clark and Huang 2004). Owners with mortgages are especially immobile (Böheim and Taylor 2002). Social housing has been found to reduce the probability of relocations for the unemployed in the UK (Battu, Ma and Phimister 2008). Kearns and Parkes (2003) find social tenants in general to be more mobile than owners, but less mobile than private tenants. Also, social tenants are less mobile than private tenants and owners in Germany (Frick 1996: 24).

The literature shows that the size of the dwelling is the single most important reason for relocating stated by movers (Michelson 1979: 114; Parkes and Kearns 2003). Individuals that experience room stress are much more likely to relocate than individuals without room stress (Clark and Huang 2004; Aarland and Nordvik 2009).⁵ Room stress does not only depend on the sheer floor space of a dwelling, but more importantly on the layout and the number of separate rooms that are available to the household. Not only objective room stress, but also subjective room stress, i.e. a subjectively perceived lack of housing space, increases the probability for residential relocations (Frick 1996: 180f). Clark and Huang (2003) show that not only negative room stress, but also positive room stress triggers relocations.

Rabe and Taylor (2010) show that objectively deprived neighbourhoods increase the probability to relocate for couples, but not for singles in the UK. However, these neighbourhood characteristics explain only a small share of actual relocations. Clark and Huang (2003) find no significant effect of the perceived quality of the neighbourhood on relocations in general, but Clark and Huang (2004) find a negative effect for long-distance relocations. Kearns and Parkes (2003) find subjective satisfaction with the neighbourhood to affect the intention to relocate, but not the actualisation. Some households may prefer living in objectively worse neighbourhoods for subjective reasons (Bolt and Kempen 2010). Thus, neighbourhood quality seem to be a weak predictor of relocations.

⁵ The ratio of the actual housing space relative to the need in housing space is termed room stress. See Section 4.5.1.2 for a more detailed definition.

In general, the share of ethnic minorities in the neighbourhood is found to increase the probability of residential relocations from the neighbourhood for natives (Ham and Feijten 2008; Schaake, Burgers and Mulder 2010). This has been labelled 'white flight' in the US literature as the white population relocates away from ethnically concentrated areas. While research finds evidence for 'white flight', the effect is modest and depends on the ethnic heterogeneity and changes in ethnic composition in the neighbourhood as well as individual characteristics (Crowder 2000). These findings support the theoretical expectations that the neighbourhood quality affects relocations through the 'fit' between locations and individuals' preferences. The inconsistencies in findings may result from the difficulties to measure households' preferences regarding neighbourhood qualities and difficulties to measure neighbourhood quality itself.

Life courses are embedded in particular spaces and historic times and this may affect residential relocations. Past research shows that period effects shape residential relocations, i.e. historic events such as war, natural catastrophes and economic recessions affect relocation behaviour distinctively (Wagner 1989: 190). Furthermore, the time-specific conditions in the housing and labour markets influence the occurrence of residential relocations. Considering housing market indicators in Germany, Frick (1996: 188) finds a significant and negative effect for construction costs and a negative but non-significant effect for the tightness in the local housing market on the probability to relocate. For the UK, Böheim and Taylor (2002) find that higher regional housing prices and more vacant dwellings increase the probability for relocations. At the same time, high average housing prices at potential destinations deter mobility of owners (Rabe and Taylor 2012). The overall situation in the labour market also affects residential relocations. For the unemployed in Germany, the regional unemployment rate is negatively related to the probability of residential relocation (Windzio 2004). In the UK, the regional unemployment rate only has little impact on the probability to relocate (Böheim and Taylor 2002). Rabe and Taylor (2012) find that the employed are more likely to relocate to regional labour markets that offer more job stability, while the unemployed are more likely to relocate to regional labour markets that offer higher wages.

The effect of institutional contexts on residential relocations is also shown by findings on cross-national differences in relocation behaviour (Dieleman 2001). The UK has a more mobile population than Germany. According to the special Eurobarometer 64.1 survey in 2005, 65 per cent of Germans aged 15 and older moved at least twice in their lives until the interview. About 18 per cent relocated even more than four times. In the UK, about 71 per cent of those aged 15 and older moved at least twice and about 30 per cent more than four times in their lives. In Sweden 81.7 per cent of the population aged 15 or over moved at least twice, while 53.5 per cent moved five times or more, which is the highest share for the EU-25 countries. On the other extreme one can find Italy. Here, 32.3 per cent of the population aged 15 or over moved at least four times (Bonin et al. 2008: 36, Table 5). The characteristics of relocations are also found to differ.

In Germany, a higher share of the population experiences local relocations than in the UK, i.e. relocations within the same town or region, while a lower share of the population experienced long distance relocations in Germany than in the UK, i.e. relocations across regional borders (Bonin et al. 2008: 36, Table 5).

3.2 Changes in Room Stress After Residential Relocations

The dwelling provides a shelter for its inhabitants and arguably "of all the necessities of life, shelter is among the most basic of our needs" (Cairney and Boyle 2004: 162). But housing is much more than just a roof. One's dwelling provides vital resources for safety, recreation, reproduction and self-expression. Often, the home is subjectively perceived as a safe haven within an otherwise hostile world. This is not the least reflected in the proverbial 'My home is my castle'. The dwelling also caters for the individual need of privacy. Individuals use their dwellings to create an environment not under observation by others. To create this environment, individuals spend significant amounts of time in furnishing and designing their dwellings to adapt them to their needs and preferences.

A wide range of indicators can be used to characterise dwellings: number of rooms and room stress, layout of rooms, overall floorspace, equipment of dwelling, condition of built structure, degree of privacy, tenure and type of building (Townsend 1979: 476; Herlyn and Herlyn 1983: 31; Howden-Chapman 2004). Research shows that some characteristics are more favourable than others. In this section, the importance of room stress is highlighted (Section 3.2.1) and the empirical findings on changes in these characteristics after residential relocations are reported (Section 3.2.2).

What is assumed to be 'good' housing is subject to social and historical changes (Townsend 1979: 476). Recent decades have seen an rapid improvement in the average housing quality in most developed countries. For example, the average floor-space per person increased in Germany from about 36 m^2 in 1995 to 41 m^2 in 2008.⁶ As a result of increasing affluence, "housing consumption has improved and crowding has been alleviated over time because of shrinking household size, overall expansion in the housing stock, and the 'trading up' process' (Huang 2003: 592). Thus, understanding individuals' housing situations necessitates comparison to period-specific standards.

⁶ Own calculation based on Statistisches Bundesamt (2011). Average floor space per person is calculated as housing space of all residential buildings divided by population. 1995 is the earliest date for which data on floor space is available.

3.2.1 Relevance

The high importance of a dwelling for the life of its residents can hardly be disputed. The effects of substandard housing conditions — especially high room stress — and their impact on health have already been described by Friedrich Engels at the end of the 19th century (Engels 1970 [1872]). The importance of adequate housing becomes especially apparent, if the effects of being homeless are analysed. Rough sleeping can cause devastating health outcomes, because "shelter, warmth and privacy [...] do not exist; good food may be hard to find [...]; it is almost impossible to keep clean; minor illnesses are hard to cure" (Ineichen 1993: 82). Apart from these extreme situations, other housing quality disparities have been linked to a wide range of outcomes. In Engel's time, "acute, infectious disease, which could be transmitted by casual contact [were the main concern and] [...] it was clear that housing and sanitary conditions were instrumental in the differential distribution of health status by social status" (Dunn 2000: 360). Today the link between housing and health is more complex, because chronic illnesses are a main concern in current debates and their relationship to housing is harder to establish, as various intervening factors may play a role (Pevalin, Taylor and Todd 2008).

Since Engel's early comment, high room stress or crowding has been a recurrent topic in the literature on housing quality. Room stress occurs if a household's need of space is not met by the available space in the dwelling. Important is the number of separated rooms, rather than the overall floorspace of dwellings (Gove and Hughes 1983: 222). In the past, high room stress has been mainly linked to physical health problems, e.g. caused by infections. Today, problems with mental wellbeing following from high room stress are highlighted (Ineichen 1993: 45). Several activities like sleeping, cooking, eating, and relaxing need space and privacy. If this space or privacy is not available, the activities are impeded (Herlyn and Herlyn 1983: 45). The effect of room stress on wellbeing probably works through the degree of personal control that individuals have and their ability for restoration in the dwelling (Gove and Hughes 1983: 222). Room stress affects these processes negatively and causes psychological stress (Dunn 2000; Evans 2003). In an extensive study, Gove and Hughes (1983: 223) reports a number of problems resulting from high room stress for wellbeing through excess demands and lack of privacy:⁷

"The effects of crowding were very strongly related to a number of direct responses to crowding, ranging from physical withdrawal from the home to psychological withdrawal in the home [...], to a lack of planning, and to feeling physically and psychologically drained. Crowding was also found to be strongly related to poor mental health, to having poor social relationships in the home, to a number of aspects of poor child care, to an inability to adopt the sick role, and to being dissatisfied with one's home. Crowding had some effects on reducing the frequency of sexual intercourse, was related to some extent to the ineffective

⁷ These findings remained after controlling for sex, ethnicity, marital status, education, family income and age.

use of birth control, and was probably related to having a miscarriage or abortion. Crowding was also related to poor social relationships outside the household, but here the relationship was modest."

Stress resulting from crowding or high room stress can also result in physical health problems: "Stress can undermine the immune system and increase susceptibility to a very wide range of health problems" (Easterlow, Smith and Mallinson 2000: 369). The consequences of room stress are especially severe for children (Coggon et al. 1993; Marsh et al. 2000; Conley 2001). Reported problems for children include delays in the locomotory development, lower school attainment, increased intensity of child-parent conflicts and increase in passive behaviour. High room stress in the past has a negative effect on future health outcomes (Marsh et al. 2000). However, it is difficult to disentangle the effect of room stress from other social background variables, e.g. social status, and room stress may work as a facilitator of other problems children have (Herlyn and Herlyn 1983: 65).

3.2.2 Review of Empirical Research

For the majority of relocations in modern societies, the housing quality after relocations is at least as good as before. Households are likely to relocate to dwellings that are better suited for their needs (Clark, Deurloo and Dieleman 2000). However, the degree of improvement may vary. Some face deteriorating relocations. For example, in the US, about 10 per cent of the population did not participate in the general increase of housing consumption over the last decades and about one in four housing careers show a downward trajectory with regard to housing quality (Clark, Deurloo and Dieleman 2003). Only about one in four movers reports a reduction in room stress in the UK, while the other movers do not change or increase their room stress on average (Pevalin, Taylor and Todd 2008).

Home ownership is associated with better housing quality on average (cf. Section 2.3.2), and research shows that home ownership and room stress are especially closely linked (e.g. Clark and Drever 2000). The following reviews literature on room stress, albeit scant, in order to bring relationships into focus. Findings regarding relocation to ownership are also reported due to the close association between both qualities.

Life course stages and family events are important determinants of relocations into home ownership (Clark, Deurloo and Dieleman 1997; Mulder and Wagner 1998). In general, cohabitators are less likely to live in ownership than married couples, but more likely than singles. Smaller, unmarried households with less children are more likely to experience deteriorating relocations within the owner tenure (Morrow-Jones and Wenning 2005). Married couples and families with children are more likely to relocate into home ownership, because they are more likely to

be committed to stay together and make a long-time investment (Michelson 1979: 138ff; Deurloo, Clark and Dieleman 1994; Dieleman and Everaers 1994). The postponement of childbirth increases the resources that can be accumulated for the housing career and couples that have children later in life are more likely to end up in home ownership (Mulder 2003). Widows and divorcees are more likely to relocate to worse housing (Dieleman, Clark and Deurloo 1995; Morrow-Jones and Wenning 2005). Union dissolutions have a long lasting negative effect on ownership (Feijten and Mulder 2005; Helderman and Mulder 2007). Older home owners are more likely to relocate to worse tenures, if they relocate at all (Morrow-Jones and Wenning 2005).

Other researchers have operationalised life course stages using age groups. While on average relocations improve the housing quality in the UK, this is not the case for those above 49 years old and singles (Clark and Huang 2003). This may indicate that families and especially growing families improve their housing qualities on average, while singles and households in the post-family phases are less likely to improve their locations. One of the reasons for deteriorating housing quality in late-life may be that households reduce their use of abundant housing space after children have left the household. There is evidence for this reduction in the US and France (Clark, Deurloo and Dieleman 2006; Gobillon and Wolff 2011).

Regarding life course events in the job trajectory, current or recent unemployment affects ownership negatively. In the UK, unemployment reduces the chances to enter or to be in ownership (Di Salvo and Ermisch 1997; Ermisch and Halpin 2004). For the US, no effect of unemployment on crowding is found in Latino households (Memken and Canabal 1994). Receiving income assistance is associated with higher room stress in the US (Cook and Bruin 1994). Thus, it seems that while employment status has a substantial effect on ownership, its impact on crowding controlled for income changes is small. The number of earners in the family is found to be important in explaining home ownership. Dual-earner couples are more likely to enter home ownership and are less likely to drop out of this tenure (Deurloo, Clark and Dieleman 1994; Dieleman, Clark and Deurloo 1995).

The transition to more spacious dwellings is highly dependent on economic resources (Michelson 1979: 135; Clark, Deurloo and Dieleman 1997, 2000; Buck 2000a). In the US, low income households are more likely to relocate to worse dwellings in terms of housing prices (Morrow-Jones and Wenning 2005). Also in the US, low-income households have not benefited to the same extent from the overall increase in housing consumption like higher-income households. While the share of low-income households in room stress is not high, it remained constant over the period 1968-1992 whereas it decreased for the other income groups (Clark, Deurloo and Dieleman 2000: 61). Measuring housing quality as a combination of tenure and number of rooms in the Netherlands, Feijten and Mulder (2005) find an increase in quality with higher incomes. Changes in income are also found to be significant predictors of relocations to home ownership (Deurloo, Clark and Dieleman 1994). Decreases in income make deteriorating relocations more likely (Clark, Deurloo and Dieleman 2003). People coming from families with high social status are more likely to end up in home ownership (Di Salvo and Ermisch 1997). Intergenerational wealth accumulation is very important for this transition (Mulder and Wagner 1998).

In the US, households with lower educated members are more likely to move to worse dwellings in terms of housing prices (Morrow-Jones and Wenning 2005). The effect of education is not straightforward, however. People in higher education are more likely to live in worse housing at the beginning of their residential career when they are still at university. Then over time their investments in education lead to higher incomes and improvements in their housing quality, and the effect is stronger than for people that did not attend higher education (Mulder 2003; Feijten and Mulder 2005). The findings are in accordance with theoretical expectations. As shown in the previous paragraphs, economic resources and education as a proxy for resources in addition to certain preferences are associated with better housing.

If information about potential locations is scarce, relocations may have less positive outcomes. This may be due to high information costs or due to limited search times. Information costs increase with the distance of relocations and it may be expected that long-distance relocations have less positive outcomes than short-distance relocations with regard to housing quality. For the UK, there is no evidence for shortdistance adjustment relocations to improve housing quality after long-distance relocations (Clark and Huang 2004). This may be due to the fact that information costs do not increase prohibitively high with relocation distance. Limited search times may result from involuntary relocations after which a new place of residence must be found quickly. For the Netherlands, involuntary movers that had to relocate because of renovations of their old dwellings reported that time stress caused them to relocate to worse than their desired locations (Kleinhans 2003).

The theoretical framework assumes that immigrants are constrained in their relocation behaviour due to discrimination in the housing market. This is found to be true for Germany on average. In a qualitative study in Germany, all participating Turkish households experienced discrimination in their location searchs and were restricted in their choices (Horr 2008). While members of ethnic minorities tend to improve their housing situation through residential relocations, they do not achieve the same quality level as the native population (Clark and Drever 2001). This relocational disadvantage exists when economic statuses are otherwise equal between natives and immigrants (Drever and Clark 2002). For the UK, Daley (1998) finds immigrants to be less likely to relocate to ownership. Discrimination in the UK, both expected and experienced, leads Pakistani away from certain areas with better housing qualities (Bowes, Dar and Sim 2002).

Housing market conditions, national institutional contexts and period effects shape transitions to home ownership (Dieleman and Everaers 1994; Deurloo, Clark and Dieleman 1994; Mulder and Wagner 1998). In the Netherlands, relocations increase housing quality more in local housing markets with greater availability of owned dwellings and in rural areas when compared with urban areas (Feijten and Mulder 2005). For the UK, higher regional unemployment and housing prices affect

the first time entry into home ownership and social housing negatively, "primarily by slowing departure from the parental home" (Di Salvo and Ermisch 1997: 16). In housing markets in the US with high population inflows, e.g. urban agglomerations, income is more important as a means to exchange crowded dwellings for more spacious dwellings than in less tight housing markets (Clark, Deurloo and Dieleman 2000). Also in the US, relocations to suburban neighbourhoods are more likely to result in housing improvement (Morrow-Jones and Wenning 2005). In Germany, the particularities of the housing market, i.e. a large rental sector, high prices for homes and high down-payments in purchasing property, causes the transition to home ownership to take place later in life and be less directly linked to trigger events such as childbirth (Deurloo, Dieleman and Clark 1997).⁸

A general lack of longitudinal and cross-national research with regard to changes in room stress is apparent from the literature review. Ultimately, we know little about changes in individuals' life courses that determine changes in room stress. Instead, previous literature mostly uses cross-sectional methods that do not allow differentiation between intra- and inter-individual variation in room stress both over time and across socio-economic strata. This may be especially problematic, if individuals have unobserved, time-constant preferences regarding room stress that are associated with observed variables. We also know little about potential cross-national differences in the average level of room stress and changes in room stress over time. It may be expected that conditions in national housing markets affect these variables. More importantly, past research provides little insights in the effect of relocations on room stress. General determinants of room stress are analysed in past research, but it is unclear how relocations directly affect changes in room stress. This is a striking gap in the literature, because relocations are the most important life course events affecting room stress besides family events.

3.3 Neighbourhood Effects and Changes in Neighbourhood Quality After Relocations

The neighbourhood provides the direct spatial environment for a household. It is a cluster of residents (social environment), buildings (built dimension) and natural conditions (environmental condition). Local neighbourhoods are complex social entities that can provide important resources and opportunities for their residents. Ideally, neighbourhoods are places of social interaction with neighbours, provide essential goods and services, as well as safe and healthy environments. Neighbourhoods are stratified by the amount of opportunities and resources that they offer and this may affect their residents' life chances (Section 3.3.1). These neighbourhood effects are defined as "the net change in the contribution to life-chances made by

⁸ See also Section 4.1.1.

living in one area rather than another" (Atkinson and Kintrea 2001: 2278). Still the evidence for causal effects of neighbourhoods on residents is weak. Findings regarding neighbourhood effects also depend on the characteristic of the neighbourhood that is analysed (Section 3.3.2) and the chosen spatial delimitation of neighbourhoods (Section 3.3.3). However, if one assumes that place matters and being able to live in better neighbourhoods can improve individuals' life chances, then it is also of high sociological interest to show whether people can relocate to better neighbourhoods. Past research indicates that this path may not be open to all (Section 3.3.4).

3.3.1 Relevance

In this section, I review the literature on neighbourhood effects to show the relevance of changes in neighbourhood quality on individuals' lives. In Section 3.3.1.1, I describe the mechanisms through which neighbourhoods may affect individuals. In Section 3.3.1.2, I review the evidence for neighbourhood effects.

3.3.1.1 Mechanisms of Neighbourhood Effects

Atkinson and Kintrea (ibid.) distinguish compositional and contextual neighbourhood effects. Compositional effects result from a concentration of specific subpopulations in a neighbourhood which induces a lack of local cultural and social resources. This is what has been described by Wilson (1987) and social disorganisation theory (Shaw and McKay 1969; Sampson and Groves 1989) where the emphasis is on "norms and collective efficacy" (Sampson, Morenoff and Gannon-Rowley 2002: 457) that characterise 'good' neighbourhoods. The concentration of deprived individuals in neighbourhoods may lead to stigmatisation of areas, adaptation of deviant behaviours by inhabitants, limited social networks and stresses on local services. Especially the concentration of economically deprived individuals is reported in the literature as causing these problems (e.g. Friedrichs and Blasius 2003; Galster et al. 2008; Hastings 2009). Some research also points to ethnic segregation as a potential cause for negative neighbourhood effects (e.g. Clark and Drinkwater 2002; Drever 2004; and see the US literature on ethnic assimilation, e.g. Alba and Logan 1993), but it can be argued that not the concentration of ethnic minorities itself is an issue, but the presence of ethnic minorities may rather be an indicator for other problems in the neighbourhood that feed into negative neighbourhood effects (Harris 1999).

Contextual effects result from the built environment of the neighbourhood itself and its environmental conditions (Atkinson and Kintrea 2001). The built environment may affect the inhabitants of the neighbourhood by not providing resources that promote safety and health, e.g. due to air pollution (Voigtländer, Berger and Razum 2010). The influence of lacking services and goods can be understood as a compositional as well as contextual effect. Goods and services may be of lower quality in a neighbourhood, because inhabitants are not able to voice their needs or local businesses do not make enough profit to provide certain products, thus being the result of the composition of the population in the neighbourhood (Hastings 2009). Other aspects of services and goods, e.g. public transport, are also a matter of the built environment.

The effect of neighbourhoods on individuals is not uniform across the whole population and these mechanisms may affect individuals differently. First, they depend on the life course stage. For example, pre-school children may not be affected by stress on local schools, while this will have a stronger effect on adolescents. Adults' behaviour will be less influenced by role models in the neighbourhood than children and adolescents, but the lack of services or information will be more influential for adults than for younger residents. Some individuals will spend more time in the neighbourhood and may be more exposed to potential negative neighbourhood effects, e.g. if they do not work. Some individuals may also be more mobile, e.g. if they own a car, and do not depend on the services in the neighbourhood, because they can use services in other areas (Ellen and Turner 1997: 836).

3.3.1.2 Evidence for Neighbourhood Effects

Identifying genuine neighbourhood effects is a complex endeavour. Most empirical research falls short of using appropriate methodological tools to capture the complexity of individuals nested in neighbourhoods. Therefore, most research on neighbourhood effects is flawed by fundamental methodological problems, e.g. the selection into neighbourhoods, and the evidence has to be treated with extreme caution (Dietz 2002: 549ff).⁹

Quasi-experimental studies such as government-sponsored relocation programmes offer potential ways to tackle the methodological problems of neighbourhood effect research (ibid.: 568). One of the most famous programmes of this kind is Move To Opportunities (MTO). MTO was conducted as a housing voucher lottery in five cities in the US from 1994 to 1997 to examine neighbourhood effects. The treatment group could only use their vouchers to relocate to non-poor neighbourhoods. Research finds ambivalent results of the programme. For example, labour market outcomes were not affected by relocations to other neighbourhoods, but mental health improved significantly in the treatment group (Leventhal and Brooks-Gunn

⁹ Other methodological problems are, e.g., that neighbourhoods and residents influence each other at the same time and it is difficult to establish the direction of causality (problem of simultaneity). In addition, it is complicated to measure the quality of the neighbourhood and analyses may be flawed because important neighbourhood variables are not observed (omitted variable problem). See Buck (2001) for a discussion.

2003; Turney et al. 2006; Kling, Liebman and Katz 2007). Another intervention placing families in diverse neighbourhood settings is the Gautreaux programme conducted in the late 1970s in Chicago. For example, Mendenhall, Deluca and Duncan (2006) find significant effects of racial segregation and resources in neighbourhoods on participating African American women's employment status and welfare receipt using the Gautreaux data. The authors consider residential relocation to better neighbourhoods as an important approach in reducing social inequality for the African American population. However, even these quasi-experiments have methodological problems which may obscure the identification of neighbourhood effects. For example, participation in the MTO programme is voluntary and non-random selection into the treatment group may be a result (Clampet-Lundquist and Massey 2008).

Neighbourhood effects may be smaller in the European context compared to the US for several reasons. First, disparities in living conditions are less in Europe than in the US. Second, public transport in Europe is well developed in most cities facilitating mobility across neighbourhoods. Third, more developed welfare states may soften spatial inequalities (Schönwälder and Söhn 2009). Hence, research on neighbourhood effects in the European context should be considered as well. To my knowledge, there are no quasi-experimental studies in England or Germany testing for neighbourhood effects. In Germany, Helbig (2010) finds good neighbourhoods to improve educational attainment for children in Berlin, but no negative neighbourhood effects. There does not seem to be a negative neighbourhood effect of ethnic concentration on social ties of ethnic minorities in Germany (Drever 2004). In England, McCulloch (2001) finds some effect of neighbourhood deprivation on individual outcomes - satisfaction with neighbourhood, employment status, and men's health status - using multi-level models. But Bolster et al. (2007) do not find a clear effect of neighbourhood deprivation at a very small neighbourhood scale (enumeration districts) on individual income trajectories. Propper et al. (2007) find social tenants that lived in more deprived neighbourhoods in 1991 to experience lower income and worse mental health in 2001 than those social tenants that lived in less deprived neighbourhoods. However, neighbourhood deprivation does not affect change in income or mental health and the authors conclude that the identified association is most likely not causal. Buck (2001) finds some evidence for a positive association of neighbourhood deprivation with non-monetary poverty of residents, the subjective expectation of not starting a job in the next year, not starting a job, entering poverty and not leaving poverty. The study questions if these associations are causal, because of the methodological problems of identifying pure neighbourhood effects.

In summary, the findings on neighbourhood effects are inconclusive. In general, evidence for neighbourhood effects on health and subjective wellbeing seem to be stronger than for material outcomes or employment status. If neighbourhood effects are found at all, they are mostly very small and have substantially less effects on individuals than individual- or household-level characteristics. From this one cannot conclude that there are no important neighbourhood effects on individuals. It may well be that analyses up to now could not tap the causal effects despite all efforts due to the methodological problems involved with identifying neighbourhood effects. In addition, even relatively small neighbourhood effects may add up over the life course and result in significantly different outcomes for individuals over time (ibid.). So far, there is little research on the cumulative effect of various neighbourhoods over time. Thus, neighbourhood quality seems to be modestly relevant for individuals to the best of current knowledge.

3.3.2 Measures of Neighbourhood Quality

Neighbourhoods are complex social entities that can be characterised along numerous dimensions. In past research various variables have been used to describe these dimensions (Andersson et al. 2007). Often the choice of variables is strongly limited by the availability of data at the neighbourhood level, but recent years have seen an increase in data that can be used to analyse neighbourhoods. Measures of neighbourhood quality can be differentiated in those aiming to tap compositional or contextual neighbourhood effects. With regard to compositional effects, most research uses measures of the economic composition of the population, e.g. the share of poor residents (e.g. Brännström 2004; Crowder and South 2005). Some authors argue that instead of poverty measures, indicators of affluence are better suited to describe neighbourhood quality (e.g. Browning and Cagney 2003). The second most widely used set of indicators measures the ethnic composition of the neighbourhood's residents. Either the overall share of ethnic minorities or shares of various ethnic subgroups are used (e.g. Morenoff 2003; Crowder, Hall and Tolnay 2011). Following social disorganisation theory, residential stability is used as a measure to describe neighbourhoods by measuring the average population turnover in a given year or the average time that residents lived in a given neighbourhood (e.g. Sampson and Groves 1989; Browning and Cagney 2003). Several other indicators are regularly used to analyse compositional neighbourhood effects: average education of residents, prevalent housing tenure, average employment status and average family status (Clark, Deurloo and Dieleman 2006; Andersson et al. 2007; Baum, Arthurson and Rickson 2010).

Fewer studies explicitly consider the contextual dimension of neighbourhoods by including measures of the built environment or environmental conditions. Research on environmental inequality has used measures of environmental pollution, e.g. noise (Brainard et al. 2004) or emission of toxic chemicals (Crowder and Downey 2010). More popular is the consideration of measures of building density and degree of urbanisation (e.g. Sampson and Groves 1989; Clark 2007). Services and goods in the neighbourhood can be understood as part of the contextual dimension of the neighbourhood. For example, Pinderhughes et al. (2001: 944) use the self-reported satisfaction with several services in measuring quality in the contextual dimension.

3.3.3 Delimitations of Neighbourhoods

No universal criteria for the delimitation of neighbourhoods exist. (Quasi-) official demarcations - often used for administrative purposes - coexist with subjective boundaries of inhabitants and other users of neighbourhoods. When analysing neighbourhoods, the investigator must consider the relevance of the chosen delimitation of the neighbourhood for residents' behaviour (Barrett, Oropesa and Kanan 1994). Depending on which neighbourhood dimension should be investigated, different demarcations can be utilized that vary in geographic and population size and it has been found that neighbourhood effects may vary by spatial delimitation (Hipp 2007). At the one extreme, neighbourhood delimitations that are too large may include conditions that are very heterogeneous and residents are only exposed to particular subareas or subconditions (Blasius and Friedrichs 2007). At the other extreme, neighbourhood delimitations that are too small may be meaningless for residents, because they can easily leave the neighbourhood. Also, if the neighbourhood is small, a higher variance of its characteristics over time is likely. In general, such small geographical areas will also result in more data 'noise' due to small sample size (Meen 2009).

Empirical evidence shows that smaller neighbourhood delimitations are superior to larger delimitations to identify neighbourhood effects. Hipp (2007) shows that economic status and residential stability in very small delimitations of neighbourhoods (so-called "microneighbourhoods" (ibid.: 663)) in the US, which are constituted by the 10 closest neighbours, have a stronger explanatory power for satisfaction with the neighbourhood than larger areas such as the census tract (about 4,000 residents on average, and normally used for analysis of neighbourhood effects in the US). This is consistent with findings from Farwick (2009: 231) showing that very small areas are more relevant for the analysis of individual behaviour of ethnic minorities than larger, more heterogeneous areas in Germany. Farwick analyses neighbourhoods of about 5 households.

3.3.4 Review of Empirical Research

Surprisingly little research has been conducted with regard to changes in neighbourhood quality over the life course: "A clear weakness of research on neighbourhoods is that we do not know the degree to which people move between different types of neighbourhood across the lifecourse" (McCulloch 2012: 98). Not the least, this is due to the complex data that is needed to compare pre- and post-relocation situations. Nevertheless, over the last decades research on changes in neighbour-
hood quality due to relocations has developed along four lines:¹⁰ 1) neighbourhood choice literature; 2) leaving deprived neighbourhoods; 3) spatial integration of ethnic minorities; and 4) urban-rural relocations.

First, there is general research on neighbourhood choice. This strand of research aims to answer the question of what determines the choice of certain neighbourhoods for individuals. As individuals choose neighbourhood with particular qualities, insights from this strand of literature may be important to understand changes in neighbourhood quality. In addition, findings from this strand of literature put more specific findings from the other strands of literature into perspective. It has been shown that characteristics of neighbourhoods are influential in the location choice of individuals (e.g. Barrett, Oropesa and Kanan 1994; Ham and Feijten 2008). In a survey about residential choice, respondents were asked to chose one of two potential, equally priced residential locations: a very good dwelling in a less favourable neighbourhood or a less favourable dwelling in a very good neighbourhood. More than two thirds of the respondents chose the location with the better neighbourhood (Butler et al. 1969: 22; cited in Lee, Oropesa and Kanan 1994: 251).

The main finding of the neighbourhood choice literature can be summarised as 'birds of a feather flock together'. Individuals relocate to neighbourhoods in which the residents resemble individuals' own characteristics with regard to income, family status, ethnic status, education, and employment status (e.g. Ham and Feijten 2008; Hedman, Ham and Manley 2011). This helps to explain the sorting of neighbourhoods in terms of quality. However, the causal mechanisms behind this sorting may be twofold. First, the sorting may be driven by the preferences of individuals that want to live close to other individuals of their kind. Second, the sorting may be driven by constraints that limit the neighbourhood choice of individuals and cause individuals with similar constraints to end up in the same neighbourhoods. Both lines of argumentation have been followed in empirical research. As it is hard to disentangle these mechanisms there is little convincing evidence so far that shows the relative importance of both mechanisms. Contrary to the just described findings, there is also the theoretical approach of place stratification that assumes that all individuals have preferences to relocate to similar better neighbourhoods. This approach is further described below.

Second, research is concerned with relocations in and out of poor or deprived neighbourhoods, defined as having a share of poor or deprived individuals above a certain threshold. In general, "residents in poor areas respond to negative residential conditions in the same way as the rest of the population" (Kearns and Parkes 2003: 849). However, past research shows that not all residents in poor neighbourhoods can improve their situations to the same degree in cases of relocation. It is found that income positively affects relocations out of poor or deprived neighbourhoods (South and Crowder 1997a; Crowder and South 2005; Clark, Deurloo and Dieleman 2006).

¹⁰ I only report literature that is concerned with changes in neighbourhood quality, i.e. longitudinal research.

Individuals with more income are more likely to leave deprived neighbourhoods than individuals with less income. However, changes in individual income are not as clearly related to changes in deprivation of the neighbourhood. Sampson and Sharkey (2008) is one of the few studies that use panel methods to relate individual change to change in neighbourhood quality. The paper finds a very small positive effect of changes in income on changes in neighbourhood deprivation in Chicago, US. South and Crowder (1997a) do not find changes in income to have a significant effect on leaving poor neighbourhoods in the US.

Research finds uniformly that members of ethnic minorities are less likely than natives to leave poor neighbourhoods and improve the economic dimension of their neighbourhoods if they relocate (Crowder and South 2005; South, Crowder and Chavez 2005; Sampson and Sharkey 2008). Ethnic minorities are also more likely to relocate to poor neighbourhoods (Quillian 2003; Briggs and Keys 2009). Most of this research has been conducted for the US, where the race gap between African Americans and whites in the likelihood to leave poor neighbourhoods decreased significantly between 1970 and 1997, but remains significant even in 1997 (Crowder and South 2005). In other countries, the negative association between ethnic minorities and change in neighbourhood deprivation is also found, e.g. in France (Pan Ké Shon 2010). The findings have also been replicated with regard to environmental deprivation of a neighbourhood, i.e. high levels of environmental pollution. In the US, ethnic minorities are less likely than native majority members and individuals with higher income are more likely than individuals with lower income to leave polluted areas (Crowder and Downey 2010; research also shows environmental inequality in Germany: Raddatz and Mennis 2013).

Being in a partnership increases chances to leave poor neighbourhoods (South and Crowder 1998b). A higher number of children makes it more likely to stay in poor neighbourhoods — especially for ethnic minorities (Gramlich, Laren and Sealand 1992; Crowder and South 2005). Sampson and Sharkey (2008) find an increase in family size to have a positive effect on being in a high-income neighbourhood. Partnership status changes to cohabitation or marriage do not have significant effects. On the contrary, Crowder and South (2005) find divorce to affect the probability of relocation to affluent neighbourhoods negatively and marriage to affect chances to relocate to affluent neighbourhoods positively. Regarding other life course events, the event of birth is associated with an improvement of neighbourhood quality in England, while becoming unemployment makes it less likely to leave deprived neighbourhoods (Rabe and Taylor 2010). For the US, Sampson and Sharkey (2008) do not find changes in job status or prestige to affect average neighbourhood quality.

Regarding the housing market context, it is found that in tighter housing markets in the Southeast of England, even high-educated individuals have a harder time relocating away from deprived neighbourhoods than in looser markets in the rest of England (Bailey and Livingston 2008). In the US, individuals are less likely to leave poor neighbourhoods with higher poverty rates. This may be explained through the geographical distance that most likely exists between areas with high poverty and areas with low poverty (Crowder and South 2005: 1739). Housing tenure and neighbourhood poverty are closely related. At the contextual level, neighbourhood quality is strongly correlated with the tenure mix in the neighbourhood and, on average, more home ownership in a neighbourhood is associated with better neighbourhood quality (Kearns and Mason 2007). Individual transitions into home ownership are often associated with an improvement in neighbourhood quality, while relocating to rental tenure increases neighbourhood deprivation on average (Sampson and Sharkey 2008; Rabe and Taylor 2010).

The third line of research deals with the neighbourhood situation of ethnic minorities. This strand is probably the most elaborate one but mainly focusses on the effect of ethnic minority status on neighbourhood outcomes. The main goals of this research is to analyse the spatial integration of minorities, i.e. whether minorities relocate to the same neighbourhoods as the majority. Researchers in this area measure neighbourhood quality based on the ethnic composition in the neighbourhood. Often, ethnic concentration in a neighbourhood is a proxy for low neighbourhood quality rather than a direct cause of low neighbourhood quality (Harris 1999). Research in this strand of literature is mainly based on four theoretical approaches:

- 1. The spatial assimilation model assumes that individuals prefer to live in neighbourhoods with residents similar to their own social statuses and life course stages. Thus, changes in preferences are caused by social mobility and family events. The model builds on early work of Rossi (1955) on the impact of the family cycle on relocations as well as on the ecological sociology of the Chicago school (Burgess 2007 [1925]). Preferences are assumed to be homogeneous for ethnic minorities and the majority. The actualisation of preferences is assumed to be primarily a matter of financial resources (Charles 2003). As differences in neighbourhood outcomes are only due to different financial resources, it is assumed that no discrimination constrains the relocations of ethnic minorities.
- 2. The *ethnic enclave model* was first introduced by Wilson and Portes (1980) and assumes that new immigrants settling in a city will cluster in certain neighbourhoods. After being a necessity in the beginning, this concentration soon provides amenities and support for the ethnic community such as the ethnic economy, close social ties and cultural affinity (Zhou and Logan 1991). These ethnic enclaves become attractive to other minority members who relocate into the neighbourhoods later on. To benefit from living in an ethnic enclave, ethnic minorities may accept a lower neighbourhood quality on other dimensions, e.g. the quality of the built environment (Schaake, Burgers and Mulder 2010).
- 3. Contrary to the assumption that individuals want to live close to others that are similar to themselves, the *place stratification model* assumes that neighbourhoods are stratified by the life chances they offer to their residents (Alba and Logan 1993). To improve their life chances, individuals prefer to relocate

to better-off areas independent of their similarities or differences to their potential neighbours. However, specific groups of the population, e.g. whites in the US, manage to constrain access to these areas for members of outgroups, e.g. African Americans, and secure the best areas for themselves. In stark contrast to the space assimilation model, the place stratification model assumes that while ethnic minorities and the majority have the same preferences, they are not equally able to meet their preferences even with the same financial resources, because of structural constraints (South and Crowder 1997a).

4. The *housing availability model* suggests that the structure of the housing market, e.g. the number of newly constructed dwellings, affects changes in neighbourhood outcomes (South and Crowder 1997a; Kearns and Parkes 2003). For example, a high supply of new high-quality dwellings might accelerate filterdown processes of dwellings (Millard-Ball 2002: 836). That is to say that households with high socio-economic status relocate to the newly constructed dwellings. Their old, now vacant dwellings can then be occupied by other households that are likely to improve their neighbourhood situations by relocating into these places.

Most of the research testing these theoretical approaches has been conducted in the US. No approach is fully supported, but rather a mix of different approaches is suggested by the evidence. Uniformly, the literature finds that members of ethnic minorities are more likely than the majority to relocate to neighbourhoods with high concentrations of minorities and are less likely to relocate to 'white' neighbourhoods (e.g. Alba and Logan 1993; South and Crowder 1998b; Magnusson and Özüekren 2002). Higher income and education makes relocations to segregated areas less likely, but ethnic minorities seem to be more likely to relocate to segregated areas even after controlling for economic status (Clark and Ledwith 2007; Bolt and Kempen 2010). The unemployed are more likely to relocate to segregated neighbourhoods (Bolt and Kempen 2010). Singles are more likely to relocate to ethnically segregated areas than couple households (Ham and Manley 2009). Older individuals are less likely to relocate to segregated areas.

Fourth, research on urban-rural relocations and suburbanisation is relevant for the present analysis as average neighbourhood quality differs strongly by degree of urbanisation or population density (Adams 1992; Ham and Feijten 2008; McCulloch 2012). Large increases in neighbourhood quality are observed for those relocating out of urban centres into rural areas (Sampson and Sharkey 2008). Leaving urban areas for less densely populated areas is more likely with older age, higher income and more education (South and Crowder 1997b; Feijten, Hooimeijer and Mulder 2008: 156). Relocating to urban areas is more likely for younger, single and well educated individuals (Courgeau 1989; Nilsson 2003; Kulu and Billari 2006). Life stages are important predictors of urban-rural relocations, because families prefer to live in less densely populated areas where they can buy cheaper, higher-quality prop-

erty with sufficient space (McCulloch 2012). Anticipation of first birth increases probability for relocations to rural areas. Especially couples with children in large cities relocate to less densely populated areas (Kulu 2008). In the US, African Americans have been found to be less likely to relocate to suburban areas than whites (South and Crowder 1997b; Sampson and Sharkey 2008). Individuals that have lived in rural areas before are more likely to relocate to rural areas again compared with those that never lived in rural areas (Feijten, Hooimeijer and Mulder 2008). Housing tenure choice affects relocations to suburban and rural settings as ownership is more prevalent in these areas (Deurloo, Clark and Dieleman 1990). But those that own a home in the city are less likely to relocate to rural areas than those not owning a home (Sampson and Sharkey 2008). Many of those who relocate from rural to urban areas will relocate back later in life. In Sweden, a study finds more than half of all individuals that relocated from rural to urban areas return within a 10-year period (Nilsson 2003).

As I stated at the beginning of this section, surprisingly little research analyses changes in neighbourhood quality for individuals. Thus, more longitudinal research is needed to replicate and extend past findings. This is especially so for general population, because a lot of research has focussed on ethnic minorities and low-income households. Neighbourhood quality changes for these two groups are well analysed — at least for the US. Patterns of neighbourhood quality changes for other groups in the population are less clear. This gap in research is especially wide in Germany, where almost no research has been conducted regarding neighbourhood quality changes. This may be due to data limitations of the past. By aiming to narrow these gaps, the present analysis makes a substantial contribution to the literature.

3.4 Effects of Relocations on Employment in Dual-Earner Couples

In neoclassical economics, relocations are conceptualised as investments in human capital (cf. Section 2.1.3). Following from this human capital approach, individuals' long-distance relocations are mainly assumed to be motivated by potential increases in lifetime earnings. Mincer (1978) was among the first to point out the potential adverse effect that relocations of couples may have on one partner and that both partners may not enjoy equal returns from relocations.¹¹ Because it is unlikely that both partners find equally good jobs after a relocation, one partner may lose earnings or leave employment by relocating compared to staying put and, thus, she or he becomes a tied mover.¹² Some partners may profit from relocations, but the couple

¹¹ Mincer's (1978) idea is more general and extends to all household members.

¹² Alternatively, the terms 'leading spouse' and 'trailing spouse' (for tied movers) are used to characterise the roles of both partners in relocation decisions in the literature.

stays put because they cannot assert their interests and these partners become tied stayers. In the literature, it is assumed that gender constitutes one of the most important determinants of the relative difference in economic returns from relocations in couples with women disproportionately losing in the event of a relocation.

Starting with the work of Mincer (1978), a comprehensive literature on the gender differences in economic returns and employment effects of relocations for couples has emerged. Two, broad strands of literature can be differentiated. The first deals with the analysis of relocation decisions in couples. A central research interest is whether potential gains or losses from relocations of men affect couples' decisions to relocate more than potential gains or losses of women. Potential gains or losses are mainly operationalised using human capital and employment status proxies, e.g. analyses test whether couples with highly-educated men are more likely to relocate than couples with highly-educated women. Recent work predicts returns from relocations to model relocation decisions instead of using proxies for returns (e.g. Rabe 2011). The advantage of this strand of literature is that potentially tied stayers can be identified, i.e. individuals that may profit from a relocation but do not relocate. A clear disadvantage of this literature is that the actual returns of relocations are not tested. Even if couples are equally likely to relocate in response to women's and men's characteristics, this does not mean that the actual returns after relocation are equal as well.

The second strand of literature deals with ex-post outcomes of relocations, i.e. the actual economic returns and employment effects of relocations are analysed. To do so, scholars compare pre- and post-relocation employment statuses of individuals. The advantage of this approach is that the actual return of relocations are analysed. The disadvantage of this approach is that tied stayers cannot be identified. For example, those individuals that cannot realise potential earnings by not relocating are not considered in the analysis. I present the empirical findings from these two strands of research in Section 3.4.2. Before, I briefly describe the relevance of analysing economic returns and employment effects of residential relocations for dual-earner couples (Section 3.4.1).

3.4.1 Relevance

It is highly relevant to analyse divergence in relocation outcomes of dual-earner couples for a number of reasons. First, it is commonly accepted in the study of couple behaviour that the idea of a unitary utility function and unconditionally pooled income is false (Beblo 2001: 12ff). One partner's gains and another partner's losses cannot simply be summed up to analyse the welfare of couples and one partner's income is not unconditionally shared for the benefit of both partners. Individuals within couples represent their own particular interests which may be opposed to interests of the other partner. Partners negotiate with each other and

use power to assert their interests (Blood and Wolfe 1960: 29ff). At the same time, behaviours are influenced by gender (e.g. Bielby and Bielby 1992). In contrast to changes in neighbourhood quality and room stress, employment effects of relocations may differ strongly for partners in couples sharing a residence.¹³ Therefore, relocations' effects on careers should not be averaged over the whole household, but instead individual outcomes should be analysed.

Second, the effect of relocations on individuals' careers in dual-earner couples may affect the relative power of both partners, because relocations may affect the relation between both partners' incomes. Thus, the 'size of the pie' that both partners produce has to be differentiated from the change in the 'distribution of the pie' for each partner after relocations (Lundberg and Pollak 2003). Because even if women recover from initial losses after relocations, the relative difference between men and women may increase permanently. "It may be that over the long course, family migration drives a growing wedge between husbands and wives occupational and earnings attainment" (LeClere and McLaughlin 1997: 330). Similarly, Abraham, Auspurg and Hinz (2010: 889) state that small differences in initial relocation outcomes may be sufficient to prioritise the more successful career in future relocation decisions and widen the gap between women and men in dual-earner couples.

Third, the analysis of dual-earner couples is especially relevant. When Mincer (1978) published his article, women's labour market participation was much lower than it is today in 2013 and dual-earner couples have become much more frequent. Hence, the coordination of two careers is necessary for a growing number of individuals. Still, gender differences can be expected, because while women "now have a deeper commitment to the labour market than in the past they are still left performing two roles, while they excel outside the home, they must cope within it" (Hardill and Wheatley 2010: 257). In addition, women may be more adversely affected by relocations today than they have been in the 1970s on average, because "their greater attachment to the labor force may imply a greater loss in earnings as they leave jobs in the initial location" (Blackburn 2010: 88). Therefore, it is highly relevant to analyse the economic returns of relocations for women and men in dual-earner couples.

Fourth, and more generally, there is the widely shared idea in the literature that long-distance relocations — especially of men — increase earnings and are an important way to propel one's career (e.g. Blau, Duncan and Tyree 1967: 243ff). Indeed there is evidence that geographical mobility is, for example, an integral part of professional and managerial careers (e.g. Markham et al. 1983; Hardill 2002: 11ff). However, evidence also shows that there is high variance in economic returns and employment effects after relocations over the whole population (Jacobsen and

¹³ One may also expect intra-household differences in *subjective* neighbourhood quality and room stress. But these differences can be assumed to be smaller than with regard to career outcomes. Also, the data in the present analysis provide only limited information on subjective housing and neighbourhood measures.

Levin 1997, 2000; Clark and Withers 2002). For many individuals, long-distance relocations seem to have no effect on earnings or labour market participation (Clark and Withers 2002). Further research is needed to qualify the determinants of this variation in outcomes.

3.4.2 Review of Empirical Research

To understand the differences between women and men in couples, it is instructive to look at single households first. By analysing singles, gender differences that are independent of being constrained by a partner can be examined. Jacobsen and Levin (1997) show that single women that relocate have a higher individual wage growth than stayers. Also, sinlge women and men do not differ in job-related determinants of their relocations in Germany (Jürges 2006: 316). Thus, single women and men seem to profit similarly from relocations at the beginning of the 21st century. Does this also hold for women in couples?

Women in couples profit less from relocations than single women. Evidence is strong for earning losses of partnered women after residential relocations for the US and the UK (Sandell 1977; McKinnish 2008; Blackburn 2010). Other research reports no significant changes for women, but even then women profit less than men from relocations (LeClere and McLaughlin 1997; Cooke 2003; Clark and Huang 2006; for East Germany: Nisic 2010). These losses are in contrast to earning gains for partnered men (Clark and Huang 2006; McKinnish 2008; Blackburn 2010; for East Germany: Nisic 2010). The effect of relocations on earnings varies in magnitude across the studies. For example, Cooke et al. (2009) find the immediate effect of long-distance relocations on women's earnings to be about half as large as the effect of a childbirth. Because of the simultaneous gains in earnings of men, "migration contributes as much to the intrafamily earnings gap as does childbirth" (ibid.: 165). The differences between women and men remain significant in most cases when controlled for both partners' human capital, occupational position and family status.

How can these gender differences be explained? Findings suggest that norms about gender-appropriate behaviour affect relocation decision. Traditionally, it is gender-appropriate for men to be active on the labour market, while women are supposed to be homemakers. Non-traditional norms, on the other hand, state that "family and employment roles are the joint responsibility as well as a source of reward for both spouses" (Bird and Bird 1985: 754). If a couple has traditional gender role attitudes, women's losses after relocations are considered to be less important than men's gains (Bielby and Bielby 1992). This is in contrast to the human capital model that assumes that differences between women and men are only due to their specialisation in paid labour and unpaid house work (McKinnish 2008). The findings regarding the importance of gender roles for relocation outcomes are re-

peatedly supported (Bird and Bird 1985; Shihadeh 1991; Boyle et al. 1999; Cooke 2008a).

The literature shows that the negative effect of relocations on female careers is mainly due to a reduction in labour market participation. Employed women are likely to leave work after a relocation or to reduce their work hours (Long 1974; Taylor 2007; Boyle, Feng and Gayle 2009). LeClere and McLaughlin (1997) show that the average wage penalty for women after relocation in the US is about US-\$ 650 in the first year and almost 97 per cent of this penalty is due to lower labour market participation rates. After long-distance relocations in the UK, the number of annual work weeks drops for partnered women and annual income is lower than before the relocation mainly for this reason (Blackburn 2006). Thus, "the loss in income suffered by married women is caused more by their relative inability to find a full-time job" (Jacobsen and Levin 1997: 699). Those women that stay in employment after a relocation do not experience earning losses in the US (LeClere and McLaughlin 1997). In France, women that stay in employment can even increase their earnings slightly (Pailhé and Solaz 2008: 356).

Relocations also lead to unemployment and reduced work hours of partnered women in the UK and this explains a big share of the loss in earnings (Boyle, Feng and Gayle 2009; Rabe 2011). The disruptive effect of relocations on employment is also apparent for men. In the UK, male movers are twice as likely (7.6 per cent in contrast to 3.5 per cent) to experience an interruption in their career than stayers (Böheim and Taylor 2007: 105). But men seem to find a new job more quickly than women. Conditional on staying in employment, wage rates are higher for both, women and men, after relocation. In the UK, Rabe (2011) finds women in dual-earner couples to increase their wage rate by 17.3 per cent on average (8.6 per cent for stayers). Men in dual-earner couples increase their wage rate by 21.1 per cent (7.7 per cent for stayers). For a sample also including single men, Böheim and Taylor (2007) find a wage premium of 3.4 per cent at the first observation after a relocation for movers compared to stayers. Men that changed their jobs in the same time interval had a wage growth premium of 3.8 per cent if they relocated compared to staying.

Two characteristics of couples that may moderate the effect of relocations on earnings and employment status are mainly tested in the literature. First, the relative differences in education and occupational status of both partners are tested. Higher education and higher occupational positions reduces chances of job loss after relocations for women in the US, UK and Netherlands (Boyle et al. 1999; Smits 1999). Women with high-education are not significantly affected in their earnings, if their male partners work in a mobile occupation in the US (McKinnish 2008). Women are more likely to leave the labour market after relocations if they worked in non-professional occupations before in the US (Lichter 1980). Women are strongly negatively affected by relocations in the UK and US, if their male partner is the main breadwinner (Boyle et al. 1999).Women with higher education profit more from long-distance relocations with regard to earnings than women with lower education

in West Germany, but not in East Germany (Nisic 2010). Relocations of highlyeducated women have an additional positive effect if they end in metropolitan areas in Germany (Nisic 2009). For men in the UK, returns from relocations seem to be more substantial for highly-educated men than for low-educated men (Böheim and Taylor 2007).

Second, the effect of the family status is tested. Because women are still often considered the main caretaker for children, women's labour market participation may be lower and less stable, if children are around. Hence, mothers may be more likely to leave employment in cases of relocations than childless women. In addition, relocations of families may be mostly directed to suburban and rural areas. Because of the local labour market opportunities in these areas (cf. below), women may be less likely to find adequate jobs. Childbearing and the presence of children is not found to moderate the effect of relocations on women's returns to relocation in the UK and US (Boyle et al. 2003; Cooke 2003). Also, children leaving the household do not moderate the effect of relocations in the US (Cooke 2003). However, McKinnish (2008) finds a stronger negative effect of having a male partner in a mobile occupation in earnings of women in couples with children under the age of 18 than in couples without children. In the Netherlands, the negative effects of relocations for women's labour force participation are stronger if couples have children (Smits 1999).

In the literature, there is evidence that long-distance relocations affect careers more than short-distance relocations (e.g. Boyle, Feng and Gayle 2009). For example, Smits (1999) shows that the negative effect on women's careers is considerably stronger if a relocation crosses at least two provincial boundaries in the Netherlands. It is intuitive that long-distance relocations have a stronger effect on careers. Only if individuals relocate for a considerable distance, then the old places of work are too far away to commute there regularly. Therefore, in most of the reported empirical analyses, only long-distance relocations are considered. The operationalisation of long-distance relocations differs. In the UK, relocations across local authority districts' borders are used as a proxy for long-distance relocations (e.g. Böheim and Taylor 2007). In the US, mostly inter-state relocations are used (e.g. Jacobsen and Levin 2000). In Germany, relocations across county borders whose capitals are at least 50 km apart are used (Jürges 2006). Instead of analysing long-distance relocations are also considered in the literature (e.g. Taylor 2007).

There is some evidence that destinations of relocations influence career outcomes for dual-earner couples. For example, metropolitan areas may provide better opportunities for both partners because of the higher density of jobs. In Germany, higher educated women can profit especially from relocating into metropolitan areas. East German, partnered women that relocate into metropolitan areas improve their wage significantly, women in West Germany do not (Nisic 2010). In France, high regional unemployment deters labour participation after relocations for women and men (Pailhé and Solaz 2008). In the Netherlands, the negative effect of relocations on female labour participation is weaker in Randstad, which is an economically prosperous area, than in other areas (Smits 1999: 144). In Sweden, relocations into larger urban areas are associated with gains in earnings, while relocations out of urban areas are associated with smaller gains in earnings (Korpi, Clark and Malmberg 2011).

As time passes after a relocation, the effect of the event may vanish. Most studies suggest that negative as well as positive effects of relocations on earnings and employment status level out a few years after a relocation. Thus, in contrast to expectations from the human capital approach, relocation do not seem to increase lifetime earnings considerably. For example, for men in the UK, relocations initially increase earnings for movers compared to stayers. But the difference is no longer significant three years after a relocation took place (Böheim and Taylor 2007). Negative effects also seem to be only temporary. Clark and Withers (2002) show that participation rates of women fall immediately after a relocation, but within 10 months after the event, participation rates are back to pre-relocation levels in the US. Similarly, women in the US face a strong decline in work hours in the year right after a relocation. This drop causes income losses of about 20 per cent for partnered women. In the subsequent years, women adjust their situation and work hours as well as earnings return to pre-relocation levels. Thus, relocations seem to cause only short-term interruptions in female careers in the US (LeClere and McLaughlin 1997; Blackburn 2010). If the observation window is long enough, long-run returns may even turn out to be positive for women in the US. Lichter (1983) shows that while in the mediumrun (1966-71) married women that relocate lose income compared to stayers, their incomes are slightly higher in the long-run (1966-76). In contrast to findings from the US, evidence for France suggests a persisting negative effect of relocations on women's earnings. Two years after a relocation, women's participation rates are still below pre-relocation levels. Women's earnings also remain lower in the second year after a relocation compared to before the relocation, while men's earnings increase compared to the pre-relocation levels (Pailhé and Solaz 2008: 356).

I have only discussed the returns from relocations, but did not consider the issue of forgone returns due to tied staying so far. For example, Nisic (2010) argues that many partners in German dual-earner couples are tied stayers. Especially in West Germany partners restrict each other's mobility in dual-earner couples. The research claims that especially men's mobility is increasingly deterred. This also implies that those dual-earner couples that relocate form a highly selective group of couples and movers. For example, one partner in the mobile dual-earner couple may be less determined to succeed in her or his career and may agree to sacrifice individual career prospects than in dual-earner couples that stay put (Boyle et al. 2001). In Germany, dual-earner couples are in general less mobile than single-earner couples, and men's job characteristics are more important to explain relocations than women's characteristics (Jürges 1998: 371ff). Human capital and employment status for women influence relocation behaviour stronger in non-traditional couples than in traditional ones, where gender role attitudes of couples are measured with partners' share of

the couple's time spent on housework on a Sunday (Jürges 2006). In the UK, Rabe (2011) finds dual-earner couples' relocations to be equally motivated by both partners' potential wage gains. However, about 12 per cent of women and only 1 per cent of men may be tied movers, i.e. movers that experience income losses while their partners gain. At the same time, there is also evidence for tied stayers among male partners in the UK (ibid.).

Past research on labour market outcomes of dual-earner couples after longdistance relocations has four short-comings: First, only few studies have analysed pooled samples of female and male partners and directly tested the relative differences in outcomes by gender. Instead, often researchers assume gender differences or postulate them without presenting statistical tests. Second, if pooled samples are analysed, research seldom accounts for the nonindependence between partners in the analysis. This may bias estimates and analysing nonindependence may provide further insights into the phenomenon under study. Third, cross-national research is rare. Comparative research has only been conducted for the United States (US) and the United Kingdom (UK) so far, but both countries share similarities in their welfare systems and labour markets (Esping-Andersen 1990: 27; Hall and Soskice 2001: 27–31). Comparisons with other countries such as Germany with divergent institutional settings are lacking. In the next chapter, I describe the institutional differences between England and Germany in more detail (Section 4.1). I also thoroughly describe my research design and empirical methods.

Chapter 4 Research Design and Methods

I apply a quantitative research design to be able to test the hypotheses developed in Chapter 2 and to generalise findings to a wider population than would be possible with qualitative methods. To this end, I use longitudinal micro-data that allow tracking of individuals over time. Thereby, the quality of locations before and after residential relocations of the same individuals can be examined. Individuals' residential trajectories are further contextualised by considering characteristics of their households and of the regions they live in. To establish claims about social regularities, it is important to test hypotheses in more than one national context and, thus, individuals from two national contexts with heterogeneous conditions for residential relocations are observed. This also provides the opportunity to analyse the effect of institutional contexts on relocation outcomes.

In the next section, I describe the two institutional cases that I selected for my analysis, England and Germany (cf. Section 4.1). I describe the housing markets, labour markets, welfare systems, and gender relations in both. The focus of my elaboration is on differences between housing markets, since I assume that these differences are most important to understand differences in outcomes of residential relocations. In Section 4.2, I briefly justify why I choose individuals instead of households as the central unit of analysis. Next, I present the data sources that I use at the level of individuals, households, neighbourhoods and regions in Section 4.3. In Section 4.4, I describe the analytic sample and in Section 4.5 I describe the construction of my dependent and independent variables. Finally, in Section 4.6, I discuss my choice of methods and describe the specific multivariate regression models that I use in the empirical analysis.

4.1 Country Case Description: England and Germany

Cross-national research is needed to establish regularities across societal contexts, but is also needed to bring national particularities to light (Kohn 1987; Mayer 2005; Blossfeld 2009: 287). To achieve these goals, I compare the cases of England and Germany in the present analysis. I concentrate on England rather than the whole UK in my analysis, due to of data limitations.¹ In the description of the institutional context below, I refer to the UK in general, as for the institutional areas that I consider in my analysis there are no systematic differences between the countries in the UK. In the following, I give more details about the institutional differences between the UK and Germany that I consider relevant for the outcomes of residential relocations. Due to the manifold institutional differences, direct causal links between institutional differences and outcomes of relocations are impossible to establish. Instead, possible links are described and alternative explanations evaluated in later chapters.

4.1.1 Housing Markets

The housing markets in the UK and Germany constitute very different contexts for residential relocations. The characteristics of these markets will have substantial influence on the outcomes of relocations as they determine who can when relocate where. First of all, the tenure structure differs strongly between the UK and Germany. Table 4.1 shows the share of different housing tenures in the UK and Germany and provides the average for the EU27 countries as a reference for the year 2005. The share of dwellings rented privately is only 16 per cent in the UK, but 37 per cent in Germany. In contrast, the share of dwellings owned with a mortgage is 44 per cent in the UK and only 29 per cent in Germany. In both countries, the share of dwellings owned outright is about 25 per cent. Social housing is slightly more popular in the UK (14 per cent) than in Germany (10 per cent). The term social housing is used to refer to dwellings owned by local authorities or housing associations in the UK. In Germany, social housing refers to dwellings that are subject to rent control due to public subsidization of the construction of the dwelling. Both types of social housing are particular and not directly comparable, but they have in common that access to them is regulated not only by price, but also by administrative allocation decisions (Stephens, Burns and MacKay 2003; Fitzpatrick and Pawson 2007).

Kemeny (1995: 16ff) argues that the UK has a dual housing market with the private and social housing sectors being strictly separated (cf. also Kemeny 2006). The social sector is strongly regulated. Supply of dwellings in the private rented

¹ The data on neighbourhoods is not available for the other countries in the UK except for Scotland. The data for Scotland is not directly comparable with the data for England, however.

4.1 Country Case Description: England and Germany

	UK	Germany in per cent	EU27
Social housing ^a	14	10	12
Rented privately	16	37	20
Owned with mortgage	44	29	27
Owned outright	26	24	41

Table 4.1 Housing tenure structure, 2005

Data: Eurostat (2012)

Note: ^a: Dwellings owned by local authorities or housing associations in the UK, dwellings that are subject to rent control due to public subsidization in Germany.

sector has been reduced by strong regulation in the 1960s and 1970s, which caused rents in 1981 to be clearly below market price level (Voigtländer 2009). These regulations were introduced in response to the acute housing shortage after World War II. There is only weak legal protection for private renters in the UK (Balchin 1996b: 225). Short-term rent contracts are very common and "there is a strong emphasis in government housing policy in England on the role of private renting in providing a flexible source of mainly short-term accommodation" (Kemp and Keoghan 2001: 22). It is assumed that newly established households may spend some time in dwellings rented privately, before moving on to buy a home or being allocated a dwelling in social housing.

In general, social housing is more popular and offers better quality than the privately rented tenure in the UK (ibid.), but social housing is increasingly used by "economically inactive, less skilled, unemployed and low-income people" (Mullins and Murie 2006: 258) that cannot afford to buy homes. The increasing marginalisation of social housing is associated with a symbolic devaluation. Better-off house-holds try to avoid social housing (Lee and Roseman 1999). The concentration of poor households in social housing is also due to allocation decisions and facilitates spatial segregation (Fitzpatrick and Pawson 2007).

Home ownership is very popular in the UK. The housing market is characterized by strong competition in the mortgage market, small transaction fees and high housing price to income ratios for home owners, i.e. loans can be comparably high relative to the income of borrowers — at least before the economic crisis in 2008 (Diamond and Lea 1992: 118ff). Mortgages can be re-paid early without high penalties and mortgage interest rates are mostly fixed only for short terms. Therefore, households in the UK are able to "buy several properties during their life time" (Voigtländer 2009: 368). Owner occupation is cheaper than renting in the UK on average (Freeman 1997). Home ownership is strongly supported by public policies in the UK, e.g. through tax benefits and the Right to Buy for tenants in the social housing sector (Balchin 1996b: 227).² The high share of owners and the strong political support of ownership have the effect that "owner-occupation is 'normalised" (Smith, Easterlow and Munro 2004: 597) in the UK. Therefore, tenants may make a stronger effort to relocate into home ownership to comply with this norm than in other countries in which ownership is not the norm (Mulder and Wagner 1998; Ronald 2008: 119ff). In the UK, relatively young people already relocate into ownership and, in 1994, 74 per cent of childless couples with a household head aged 25 were already home owners in the UK (Freeman 1997).

In contrast to the UK, Kemney describes the German case as an "unitary rental market" (Kemeny 1995: 49, emphasis in original), i.e. the social housing sector is not regulated strongly and competes with the privately rented sector. In Germany, rents are relatively low due to the competition between the private and publicly subsidized housing sectors and general rent control (ibid.: 49). The legal protection of tenants is very strong in Germany and landlords cannot easily terminate rental contracts. Rents are capped according to local average rents and increases in rent for sitting tenants are strongly regulated (Kirchner 2007). As a consequence, sitting renters enjoy lower rents on average than tenants that recently signed a new rent contract (Voigtländer 2009). Privately rented dwellings are on average in good condition, as landlords can realise sufficient returns of their investment in property (Mulder and Wagner 1998; Kirchner 2007). However, regulation of the private rented sector is low in Germany compared with other countries, and the market was liberalised much earlier after World War II than in the UK. "This allowed the private rental housing market to develop further and ensured the availability of a steady supply" (Voigtländer 2009: 362).

Social housing in Germany is constituted by dwellings which have been publicly subsidised in their construction and regulated in terms of access and rent (Kirchner 2007). Rents in social housing are lower than in the private market on average, but may not be the lowest rents in the market.³ Access to social housing is subject to administrative regulation, e.g. access is mostly denied for high-income households (Dorn 1997). Every developer can claim subsidies and only few dwellings in social housing are temporarily fixed and after a certain period, dwellings enter the private market. Experts expect that only 4 per cent of social housing dwellings in 2002 will still be rent-regulated in 2020 (Kirchner 2007). Due to the shrinking share of social housing in Germany and their unattractive locations mainly in the periphery of cities, social housing is mostly a safety net for households that do not have access to dwellings in the private market due to low income or discrimination (Häußermann and Siebel 2000: 165).

 $^{^2}$ The Right to Buy from 1980 allowed sitting social tenants to buy their dwellings for reduced prices (Lowe 2007: 366).

³ For example, old, privately rented buildings in urban centres that have not been modernised often have lower rents.

Buying property is relatively expensive in Germany compared to the UK, due to high construction and land prices and the need for high down payments to purchase property (Freeman 1997; Mulder and Wagner 1998; Kurz 2004). Further characteristics of the German housing market include long term contract saving schemes (*Bausparen*), only medium tax benefits for home owners and high transaction costs in the housing market.⁴ Home owners, however, can gain extra tax credits if they have children (Deurloo, Dieleman and Clark 1997). Yet, the credit line for home buyers depends much more on the current income situation than in the UK (Diamond and Lea 1992: 79ff). There is almost no 'sub-prime' mortgage market in Germany. Particular to the German mortgage market are high penalties for early repayment of mortgages and long-time fixed interest rates, which make repeated home purchases less attractive than in the UK (Voigtländer 2009).

"Germans need more time to save before buying a home" (Mulder and Wagner 1998: 702) and once they bought a home, buying a second time becomes difficult. In addition, public subsidies for owners are mostly granted only once for individuals (Voigtländer 2009). Thus, buying property is a once-in-a-lifetime investment for most households. Due to the longer time needed for saving, individuals in Germany relocate into ownership much later in life than in the UK. In 1993, only 17 per cent of childless couples with a household head aged 25 were in ownership in Germany. The differences between the UK and Germany in home owner rates are much weaker, if home owner rates in mid- and late-life are compared (Freeman 1997). The rate of outright ownership is strongly connected with events in the family life course" (Mulder and Wagner 1998: 707), e.g. marriage. This connection is weaker in the UK. Germans are more likely to make relocations within the rental tenure before they buy homes than in the UK (Deurloo, Dieleman and Clark 1997; Mulder and Wagner 1998).

Housing markets in the UK and Germany differ in their degree of decommodification. The degree of decommodification may indicate how dependent outcomes of relocations are on current market incomes of individuals in the UK and Germany. Doling (1999) argues that housing is decommodified, if access to housing and maintenance of one's home is relatively independent from current household income. Regarding access, I showed above that access to ownership is more decommodified in the UK than in Germany. Access to property is mainly a function of the household income in Germany, while home buyers in the UK can more easily obtain loans even with low incomes. Access to private rented accommodation is more decommodified in Germany than in the UK. Access to social housing is relatively decommodified in both countries. Regarding maintenance of the current housing status, housing allowance in the UK covers the total rental payment of tenants with

⁴ The term *Bausparen* denotes publicly subsidised saving schemes that are coupled with attractive loans that can only be obtained after some time of saving. On average, almost every German household owns a *Bauspar* contract (Mulder and Wagner 1998).

low or no income as long as they do not exceed certain limits, but is not paid to home owners (Doling 1999: 162). In Germany, low-income households are eligible to receive housing support for rent and costs of homes. Recipients of unemployment benefits and social assistance receive housing benefits that only cover 'appropriate' rents, which are determined by local authorities. Households with higher rents must relocate or pay the excessive part themselves (Busch-Geertsema 2004).

There is regional heterogeneity in housing markets in the UK and Germany. Regarding the spatial structure for England, scholars describe a strong regional North-South divide that can be traced in various indicators, e.g. unemployment rate and affluence, where the South is more successful in most of these indicators (O'Leary et al. 2005). London and the South-East has been described as an escalator region which propels young peoples' careers (Fielding 1992). At the level of cities, the UK is more segregated than Germany, i.e. in the UK the concentration of low-income individuals and migrants in certain neighbourhoods is higher (Musterd 2005).

Regional patterns of inequality are more complex in Germany. A North-South is overlaid by an East-West divide after re-unification, but also strong disparities on lower regional levels persist (Kilper 2004). The North and the East are less affluent than the rest of Germany on average. The housing markets in East and West Germany differ strongly. The "East German housing stock still bears the mark of state socialism" (Tomann 1996: 52). Home ownership rates are very low in East Germany. The conditions of buildings are worse on average in East Germany at least in the 1990s. Yet, convergence of standards in East and West Germany is apparent over time (Frick and Grimm 2009). Population dynamics are different between East and West Germany as well. The population declines in many East German regions due to of low fertility and outbound relocations. After re-unification in the 1990s, high mobility from East to West Germany occurred (Rees and Kupiszewski 1999).

4.1.2 Labour Markets

Conditions in labour markets can be expected to affect the occurrence and outcomes of relocations, as relocations may be occupationally motivated and because labour income is the most important source of income for most individuals. Especially with regard to labour market outcomes of dual-earner couples, institutional differences between labour markets in the UK and Germany are highly relevant. The labour market in the UK can be described as a liberal market economy. It is characterised by "firms [that] coordinate their activities primarily via hierarchies and competitive market arrangements" (Hall and Soskice 2001: 8). Firms make decisions mainly based on current revenues and shareholder values and lack close-knit information networks linking them to other firms. Unions are weak on average, worker representation is not institutionalised and the relations between employees and firms are individualised. Managers can make top-down decisions and hiring and firing is frequent. Firms provide inhouse training, but no intense apprenticeships. Highly mobile and fluid labour markets result (ibid.: 27–31). Outsiders can relatively easily enter the labour market and unemployment spells are shorter than in Germany on average (Mills and Blossfeld 2006: 78).

The German labour market is part of a coordinated market economy. "Firms [in these economies] depend more heavily on non-market relationships to coordinate their endeavors with other actors and to construct their own core competencies" (Hall and Soskice 2001: 8). Coordinated market economies are characterised by close-knit information networks between firms. In firms, coordinated instead of unilateral decision making is common and employee representatives participate to some degree in decision making, e.g. through work councils. Employees share their skills and knowledge with the firm and to avoid head-hunting or exploitation of employees, industry-wide wages are set in bargains between unions and industry associations. Since employees need high industry-specific skills, public training systems are in place. Job tenures are long on average and firms organise inter-firm networks to share knowledge without sharing employees through job mobility. Job mobility is lower in Germany compared to the UK and entry for outsiders is more difficult.

	UI Women	K 1 Men	Germ Women in per	any Men cent	EU2 Women	7 Men
Labour force participation	66	78	60	71	56	71
Part-time employed	42	9	43	7	30	7
Gender pay gap ^a	21	-	23	_	17	-
Job tenure ^b						
0 < 5 years	51	46	40	36	43	38
6 < 10 years	21	21	21	19	20	20
\geq 10 years	28	33	39	45	37	42

Table 4.2 Labour market statistics, 2005

Data: Eurostat (2012); Organisation for Economic Co-operation and Development (2012) Note: Population aged 15-64. ^a: 2008; Gender pay gap is defined as difference between average gross hourly wage of men and of women as a percentage of average gross hourly wage of men. ^b: EU15 only.

Table 4.2 provides an overview how these institutional similarities and dissimilarities play out and also hints at characteristics of the welfare systems and gender relations in both countries. First, labour force participation is slightly higher in the UK than in Germany and Europe on average. This hints at the importance of gainful employment in the UK that is further discussed in Section 4.1.3 regarding differences in the welfare systems. Second, in both countries, labour participation of women is lower than for men, but higher than the average in all EU27-countries. Gender differences also exist with regard to part-time employment. While only a very small share of men are working part-time, almost one in two women in the UK and Germany work part-time. In both countries, this is higher than the average of the EU27-countries. By 2005, the share of women working part-time was similar in the UK and Germany after a strong increase in female part-time work in Germany. In the 1990s, female part-time work was more frequent in the UK than in Germany. Working part-time is associated with a wage penalty in the UK which is not the case in Germany (McGinnity and McManus 2007; Cooke and Gash 2010; Cooke 2011: 111). The high share of part-time work in the UK is partly due to the low statutory regulation of employment (Crompton and Lyonette 2006). The Organisation for Economic Co-operation and Development (OECD) has developed an index of employment protection ranging from 0 to 6, where 0 indicates the least protection and 6 indicates the most protection. In the UK, labour protection varies between 0.60 in 1990 and 0.75 in 2008. Labour protection in Germany varies between 3.17 in 1990 and 2.12 in 2008 (Venn 2009: 8). Thus, employment protection is much stronger in Germany, but has eroded between 1990 and 2008.

The gender pay gap, i.e. the differences between women's and men's hourly gross wage rates as a percentage of men's wage rates, is 21 per cent in the UK and 23 per cent in Germany, which indicates that women earn about one fifth less than men on average in both countries. This is higher than the average for all European countries. The last three findings show that labour markets in the UK and Germany are clearly gendered. I describe these gender differences in more detail in Section 4.1.4. Third, Table 4.2 clearly shows higher job mobility in the UK than in Germany. Only one third of men in the UK had their current job for at least 10 years, but 45 per cent of German men had their job for less than 5 years, but only 36 per cent of German men worked in their job less than 5 years. For women, the job mobility is higher than for men in both countries, but shows similar differences between countries.

4.1.3 Welfare Systems

Welfare systems in the UK and Germany are substantially different. The welfare system will affect the way negative life course events such as unemployment are moderated and will thereby alter the effect of these events on relocations and their outcomes. For example, Esping-Andersen (1990), who uses the dimensions of decommodification and stratification to classify welfare states, describes the UK as a liberal welfare state.⁵ Liberal welfare states are characterised by "individualism and the primacy of the market" (Arts and Gelissen 2002: 141). Decommodification is low and stratification between individuals is high, which results in a residual

⁵ Regarding decommodification see Footnote 20 above. Stratification refers to the social inequalities and to the kind of solidarities that are generated by the welfare state (Arts and Gelissen 2002).

welfare provision. Due to low decommodification, economic security outside of the labour market is low (Dingeldey 2007). Women are encouraged to participate in the labour market. Benefits are mostly means-tested and private insurance is encouraged (McGinnity 2004: 8). In case of unemployment, the system offers minimal levels of income protection, but does not cover all unemployed. The system is mainly aimed at preventing poverty (ibid.: 12).

Germany is described as a conservative welfare state by Esping-Andersen (1990: 27). Decommodification is moderate. Stratification is characterised in three ways: 1) Income maintenance support is linked to labour market participation. Those that leave work due to unemployment, illness or pre-retirement are protected against high income losses. Individuals in need without prior labour market participation receive less support (Arts and Gelissen 2002; Mills and Blossfeld 2006: 13). 2) The conservative welfare state is committed to the male-breadwinner model, i.e. a traditional division of labour in which men work for pay to support their family and women mainly work unpaid in the household and only do part-time paid work (Mills and Blossfeld 2006: 13, cf. next Section for more details on gender relations). 3) Following the principle of subsidiarity, the conservative welfare state only steps in, if the resources of the family are exhausted (Arts and Gelissen 2002). Benefits are mostly insurance-based in Germany rather than means-tested as in the UK. Welfare for the unemployment provides a higher level of protection in Germany than in the UK, but only for those with continuous employment histories (McGinnity 2004: 12). Wage replacement rates for unemployed are higher for singles in Germany in the 1990s, but for families with children they are slightly lower than in the UK. This is for the reason that benefits are based on earned contributions in Germany, while they are based on need in the UK (ibid.: 80f).

The German conservative welfare state has seen fundamental changes in the early 2000s which marked an important step away from the conservative welfare model towards a more liberal model. These changes were introduced by the so called Hartz laws in 2002 and 2003 The primary goal of these laws was "getting people back to work" (Ludwig-Mayerhofer and Wroblewski 2004: 500) by using enabling or activating instruments. The duration of unemployment benefits were shortened and eligibility restricted. Similar measures had been introduced in the UK in 1996/1997 under the 'New Deal' (Dingeldey 2007).

The welfare systems mediate life course risks differently in the UK and Germany. DiPrete (2002: 300) shows that in Germany "institutions suppress the rate of classaltering events but do not uniformly suppress the consequences of negative events". For example, while the likelihood of union dissolution in Germany is relatively low, the negative financial consequences of dissolution for women are not as strongly cushioned in Germany as in other countries. Similarly, employment protection is high for those in work in Germany. The long-term financial consequences of unemployment are high, though, because re-entry into the labour market is difficult. Those on favourable tracks in the labour market benefit from this security, while labour market outsiders face difficulties to improve their situations (Mayer 2005). The mediation of life course risks has consequences for inequalities over the life course of individuals. The German institutional context "places people on a track early in their life course (stratified by class background) and keeps them their for the duration of the journey" (Brückner 2004: 20). While inequalities between individuals are relatively moderate in Germany, these inequalities are relatively stable over the whole life course (Leisering 2003: 218; Mayer 2005).

Regarding life course risks, "Germany, although a different welfare regime, is close to Britain as a 'welfare state with integrated safety nets', which promotes a continuous life course" (Leisering 2003: 216). Nevertheless, the UK is characterised by less standardisation and security over the life course. Transitions in the job and family trajectory are more common in the UK than in Germany (Allmendinger and Hinz 1998: 76; Mayer 2005: 30). For example, the average duration of marriage until divorce in 2008 in England is 11.5 years (Office for National Statistics 2010). In Germany the average duration is 14.1 years (Statistisches Bundesamt 2012b). The total fertility rate was 1.96 in the UK and 1.38 in Germany in 2008 (Eurostat 2012).⁶ The UK and Germany are characterised by stratified educational systems which determine much of later life inequalities, but in the UK downward class mobility is higher than in Germany due to "the unstandardised vocational training available in Britain which provides little protection against downward class mobility" (Allmendinger and Hinz 1998: 76). On average only 4 per cent of German men, but 7 per cent of British men experience changes in their class position in a given year (ibid.: 76). Jarvis and Jenkins (1998: 151) show that downward income mobility is slightly more likely than upward income mobility in the UK. Thus, in the UK life course risks are more prevalent and individuals are more likely to experience changes in their life courses (Leisering 2003: 216).

4.1.4 Gender Relations

Gender relations can be defined as "the set of mutually constitutive structures and practices which produce gender differentiation, gender inequalities, and gender hierarchy in a given society" (Orloff 1996: 52). The UK and Germany differ in these structures and practices and, thus, gender inequalities over the life course diverge. The gender relations in both countries are likely to affect differences in labour market outcomes of long-distance relocations between women and men. In the UK, the welfare policies "emphasize the principle of individual freedom" (Blossfeld and Drobnič 2001: 43) and this is also reflected in family policies and gender differences. Female labour participation is neither actively supported nor hindered by welfare state incentives. The tax burden for singles relative to married couples has

⁶ The total fertility rate indicates the number of childbirth for a women in her life under the assumption of the fertility rates per age in a given year.

been reduced and family members are taxed individually since 1991. Tax regulations facilitate dual-earner couples (Sainsbury 1999: 247; McCulloch and Dex 2001: 197; Apps and Rees 2005). The British welfare state — contrary to the German case — is not committed to the support of the family as such and only rudimentary family policies are in place, e.g. parental leave is weaker (Ruspini 1998: 299; Kilkey and Bradshaw 1999: 149; Ferrarini 2006: 66). Public childcare is also weakly developed in the UK compared to Germany and most parents must privately organise additional daycare, especially for children under 3 years old (Sainsbury 1999: 247; Meyers, Gornich and Ross 1999: 127). Ferrarini (2006: 13) characterises this combination of little direct support for dual-earner couples and rudimentary public family support as a "market-oriented family policy model".

The German welfare policies "emphasize the value of the family based on marriage" (Ostner and Lewis 1995: 184). For couples, the welfare state supports malebreadwinner families in which women prioritise unpaid housework and carework (ibid.: 187). Partnered women's work is often part-time and supplements family income, which mainly relies on the male partners' incomes (Blossfeld and Drobnič 2001: 39). Social benefits are allocated to the family and not individuals. Nonworking female partners have no economic independence (Kilkey and Bradshaw 1999: 254; Bussemaker and Kersbergen 1999: 18). Maternal leave is extensive, but has been reformed in recent years to become slightly less gendered (Lewis et al. 2008). Public day-care for children is more extensive in Germany than in the UK and provision is better in East than West Germany, but still day-care is limited (Sainsbury 1999: 247). The couple is taxed jointly, which undermines women's labour participation (Apps and Rees 2005). Ferrarini (2006: 13) describes these policies as the "general family policy model" and Strauß (2008: 67) argues that the institutional setting in Germany makes it more likely for German women to completely withdraw from the labour market than for British women.

Leaving aside important issues of intra-household welfare distribution, division of labour in the household may not be a problem as long as couples stay together. Low or no labour market participation of women is a much bigger problem in case of union dissolution, which affects 20 to 30 per cent of marriages in Europe within 15 years after marriage (Andersson 2003). Differences in women's outcomes after dissolution will depend on their role in the labour market and their dependence on their partners' incomes (Ruspini 1998). Germany as well as the UK fail to provide independence of partners' incomes for a vast share of women (Kilkey and Bradshaw 1999: 254). While separations have adverse material effects on both ex-partners by reducing income and increasing the risk of poverty, women lose more due to their lower labour income on average — also in childless couples — in both countries (DiPrete and McManus 2000; Vandecasteele 2011). The strong division of labour between the genders increases these effects in Germany especially for women not in the labour market, but also for men due to joint taxation during marriage. German women need a long time to recover materially from union dissolutions. In Britain, women rely on the labour market more strongly to secure their living and repartnering quickly reduces financial losses after dissolution (Andreß et al. 2006). The event of dissolution increases poverty risk stronger in Germany than in the UK for women (Ruspini 1998).

Gender relations are also shaped by norms, values and attitudes that are prevalent in a society and affect gender-appropriate behaviour (Pfau-Effinger 1997: 517; Lück 2008: 428). Treas and Widmer (2000) show that attitudes towards labour participation of women differ between the UK and Germany, but more strongly within Germany between East and West. About 90 per cent of East Germans agree that married women should work full-time before the first child, while 88 per cent of Britons agree and only 78 per cent of West Germans. With a child in pre-school age, only 21 per cent of East Germans agree that mothers should stay home and quite work, but 63 per cent of Britons and even 69 per cent of West Germans agree. Treas and Widmer (ibid.) categorise East Germany as having a "work oriented" gender ideology, where attitudes are more favourable for working mothers than in other countries. West Germany and the UK belong to the cluster of "family accommodating" gender ideology, where mothers of young children are expected to stay home and mothers of school-age children are expected to work only part-time (ibid.: 1427).

4.2 Unit of Analysis

After the description of my country cases, I now elaborate on further details of my research design. In the literature on residential relocations, the unit of analysis varies between individuals and households. Some researchers treat the household as the unit of analysis arguing that "moving is ordinarily a group rather than an individual action" (Rossi 1955: 15; similarly DaVanzo 1981: 112). Other researchers treat the individual as the central unit, since households are not stable over time (Mulder 1993: 24). Dissolving and forming households are important processes that are often associated with residential relocations. To be able to consider such relocations that are associated with household changes, the individual instead of the household must be treated as the unit of analysis. I therefore apply a combination of both approaches and treat individuals as the central unit of analysis in the present analysis, and, at the same time, I acknowledge that individuals mostly relocate together with other individuals in the social convoys of their households. To this end, I include information about the household and other household members in the analysis of individual relocation behaviour (Mulder 1993: 24; Winstanley, Thorns and Perkins 2002).

4.3 Data

Different data sources are combined in the present analysis. At the individual and household level, data is drawn from two nationally representative panel surveys (Section 4.3.1). These data sets are supplemented with additional data sets at the neighbourhood level (Section 4.3.2) and the regional level (Section 4.3.3).

4.3.1 Panel Data at the Individual and Household Level

For panel data sets, the same individuals are interviewed in at least two — often more frequent — discrete points in time with broadly the same interview instruments to facilitate comparison across time. Panel data sets enjoy increasing popularity in social science, as they can be used to analyse dynamic relationships, i.e. changes in independent variables can be related to changes in dependent variables (Giesselmann and Windzio 2012: 10ff). This makes causal claims at least more credible than in cross-sectional analysis, since the time sequence of changes can be considered as well. Panel data can also be used to control for unobserved heterogeneity in surveyed individuals. Finally, estimators from panel data may be more efficient than estimators from cross-sectional data, due to repeated observations of respondents (Frees 2004: 5ff).

However, panel data come with a number of down-sides. If respondents are asked the same questions repeatedly, their responses may be affected by having answered the question before. In addition to potential non-random sample selection that may affect all survey data, panels are also subject to attrition, i.e. some respondents drop out of the panel over time. If drop-out is selective, the remaining sample population is non-random. While these problems can be tackled with appropriate data collection and weighting procedures, there are more fundamental problems to panel data that cannot be solved that easily. Panel data provides only limited information about the ordering of events and especially on the time passed since the onset of a given process, due to the discrete time structure of data collection. The shape of processes evolving over time can hardly be captured using panel data. This is especially true regarding long-time processes, since panels may not survey retrospective life history data. In these cases, only limited sections of different life course trajectories can be analysed (Blossfeld and Rohwer 2002: 13ff). In addition, panel data offer only limited opportunities to analyse rare events such as residential relocations, because of the relative small sample sizes compared to topic-specific and census-type surveys. For the present analysis, the advantages of panel data outweigh the limitations, seeing that panel data allow comparison of the same individuals before and after relocations. This is necessary to examine the effect of relocations on outcomes. For these reasons, panel data is the best possible format for empirically researching

the life course event of residential relocation across a population and panel data are used to conduct the empirical analyses. The limitations of the data must be kept in mind in interpreting the results, though. I now introduce the two panel data sets for England (Section 4.3.1.1) and Germany (Section 4.3.1.2).

4.3.1.1 British Household Panel Survey

The British Household Panel Survey (BHPS) is run by the Economic and Social Research Council UK Longitudinal Studies Centre with the Institute for Social and Economic Research (ISER) at the University of Essex and started in 1991.⁷ The original sample for the BHPS was drawn in 1990 and is representative of the British population at that time in England, Wales and Scotland south of the Caledonian Canal. For the sample 8,167 issued addresses were drawn from the small users Postal Address File. A two-stage clustered probability design and systematic sampling were used.

Additional samples for specific subpopulations, e.g. low-income households, were drawn and added to the BHPS later on, but these sample members are not used in the current analysis. The BHPS has no refresher or booster samples. The data sets provides longitudinal and cross-sectional weights to adjust the sample.^{8,9} In 2008, 5,391 households of the original sample were still in the panel of which 4,509 were interviewed. In these households, 9,249 eligible individuals lived of which 7,500 completed a full interview (response rate: 81 per cent). The field period for data collection runs from early September to early December each year. In 1999, data collection was changed from Pen and Paper Interviews (PAPI) to Computer Assisted Personal Interviews (CAPI). Few households are also interviewed via telephone. If a person is too infirm or absent at the time of the interview, a proxy interview is conducted, which covers only a reduced set of topics.

All household members that were sampled at the start of the survey are referred to as Original Sample Members (OSMs). Each OSM over the age of 16 has been interviewed annually since 1991 as long as she or he did not refuse to take part in the panel or her or his address is unknown. Household members that moved into a household with an OSMs are interviewed as well, but not followed if they leave the household again. They are referred to as Temporary Sample Members (TSMs). Children born to OSMs become OSMs as well. Parents of OSMs that are TSM become

⁷ The BHPS data is freely available from the UK Data Archive.

⁸ See Taylor et al. (2010: A5-1ff) for details on the construction of weights.

⁹ In the descriptive analyses of BHPS and SOEP data, I use cross-sectional weights to adjust for non-random sample selection in both data sets. I refrain from using weights in the multivariate, longitudinal analyses, because including design weights in these models is very complex, e.g. period-specific weights cannot be included in ordinary fixed-effects regression models (Bjerk 2009: 9, footnote 13). In these models, non-randomness of the sample is controlled for, if selection processes are due to observables or time-constant unobserved heterogeneity (cf. Section 4.6.1).

Permanent Sample Members (PSMs) and are also followed if they leave the household. At each interview, all eligible household members are interviewed individually and supplementary information on the household is collected. The BHPS covers a wide range of topics including neighbourhood characteristics, intentions for residential relocations and actualised relocations, and characteristics of the dwelling. Additional topic modules are asked at certain waves, e.g. regarding the cohabitation history, and additional neighbourhood characteristics.

OSMs and PSMs are followed when they leave the household or relocate. Several measures are taken to track movers: Change-of-address and address-confirmation cards are issued to respondents before field work starts. If the field team did not receive information about a new address beforehand, a forwarding address or phone number is searched from other respondents, new residents at the old address, neighbours, local phone directories, shops or post offices. As a last resort, the panel maintenance team tries to identify new addresses through contact persons that were provided by the respondents (Taylor et al. 2010). Buck (2000b) and Rabe and Taylor (2010) found no attrition bias due to residential relocations in the panel.

4.3.1.2 Socio-Economic Panel Study

The Socio-Economic Panel Study (SOEP) was established in 1984 and is run by the German Institute for Economic Research (Deutsches Institut für Wirtschaftsforschung, DIW) in Berlin.¹⁰ The SOEP is a representative survey of the German population. The original samples of 1984 were drawn separately for the autochthonous (sample A) and foreign population (sample B); the latter was over-sampled to achieve a sufficient number of cases.¹¹ The sample selection for sample A was based on a random selection of 584 sample points from a market research address register through a multi-stage stratified sampling procedure. For each sample point, interviewers selected households by means of a random-walk procedure. Sample B was selected using a random draw of metropolitan areas and counties for each nationality as PSUs. Households were then selected from the respective foreigner register in each PSU. Before reunification, an East German sample was added, which was drawn from the East German central population register.

In addition, five supplementary and boosting samples were drawn until 2006, which followed slightly divergent sampling strategies.¹² I do not use the high-income sample G and the innovation sample I in my analysis, as they are very selective samples and use specific instruments respectively. The data set provides

¹⁰ The SOEP data can be obtained from the DIW.

¹¹ Sample B includes households with a Turkish, Greek, Yugoslavian, Spanish or Italian household head.

¹² See Wagner, Frick and Schupp (2007: 149ff) for a description of the sample strategies.

longitudinal and cross-sectional weights to adjust the sample.¹³ In 2008, 10,959 of 12,651 households were successfully interviewed. In these households, 22,594 individuals lived of which 19,684 completed a full interview (response rate: 87 per cent). The field period for data collection starts in February (January before 2005) each year. Interviews run until October, but by March normally about two thirds of all interviews are collected. Data is collected in a mixed mode approach. The survey is mainly conducted through CAPI, but also a considerable number of self-administered PAPI are carried out.¹⁴ No proxy interviews are conducted.

Data for each participating household is collected annually and each household member above the age of 17 is interviewed. In contrast to the BHPS, there is no difference between OSMs and TSMs in the SOEP.¹⁵ Every member of a household in which a SOEP sample member lives is followed if this household is left again.¹⁶ This also applies to children born to household members. At each interview, all eligible members of the household are interviewed individually. One household member provides supplementary information on the household in general. The range of topics in the SOEP is similar to the BHPS. The intention to relocate is not recorded in the SOEP (Wagner, Frick and Schupp 2007).

Relocated households and members leaving the household are traced through various means. A database of the respondents addresses is updated regularly based on information from the respondents themselves by keeping in contact with the households throughout the year, or from the interviewers' fieldwork, e.g. by talking to neighbours, and from searches in other (official) address databases. For the year 2008, 96.3 per cent of the addresses of relocated households could be identified. Due to the high rate of identified movers, panel attrition due to residential relocations is not considered a problem (TNS Infratest 2008: 28).

4.3.2 Neighbourhood Level

In Section 3.3, I have described, how past research measured neighbourhood quality. In the following, I describe the data that I use to measure neighbourhood quality in England (Section 4.3.2.1) and Germany (Section 4.3.2.2). There are no equivalent data sets for both countries. The selected data sets provide the best available information about neighbourhoods for each country. Although there are some impediments

¹³ See Kroh (2009) for a short overview.

¹⁴ CAPI was introduced in 2000.

¹⁵ Schonlau, Watson and Kroh (2010) examine the effect of such differences in following rules on the sample population. The paper finds following rules to have relatively little effect on sample size.

¹⁶ This following rule was established in 1989.

to direct cross-national comparison. In Section 4.5.1.3, I describe the construction of neighbourhood quality measures from these data.

4.3.2.1 Indices of Multiple Deprivation

The Indices of Multiple Deprivation (IMDs) provide the most comprehensive data on neighbourhood quality in England.¹⁷ In 2000 the IMDs were introduced and renewed in 2004 (IMD2004) and 2007 (IMD2007).¹⁸ The IMDs have been created by Noble et al. (2003) and Noble et al. (2008) and were intended to inform policy and planning. As the construction of the indices fundamentally changed between 2000 and 2004, this elaboration focuses on the two recent editions. The IMDs consist of seven domain-specific indices (Income deprivation: Employment deprivation; Health deprivation and disability; Education, skills and training deprivation; Barriers to housing and services; Living environment deprivation; Crime) which cover the social and built dimension as well as the environmental conditions of the neighbourhood and a general, weighted index combining these different domains.¹⁹ The domain-specific indices are built on a set of about 40 diverse indicators. The indicators are mainly drawn from administrative sources and were selected based on their appropriateness to measure the respective dimension of deprivation, their availability at a small-scale geographical level, recent measurement points, expected recurrent measurement in the future and their statistical robustness (Noble et al. 2006). For each domain, a score is computed using factor analysis. The higher the score, the more deprived a neighbourhood is. Then, the neighbourhoods are ranked according to the score for each domain. The lower the rank, the more deprived a neighbourhood is. The ranks range from 1 (the most deprived neighbourhood) to 32,482 (least deprived neighbourhood).

The IMDs provide only cross-sectional information. Most of the underlying data refers to the years 2001 (IMD2004) and 2005 (IMD2007) respectively. While the computing of the general index and the derived indices stayed the same between 2004 and 2007, some variables were changed in the Income deprivation domain and the Education, skills and training deprivation domain, but neighbourhood deprivation measured by IMD2004 and IMD2007 is rather stable over time. I decided to

¹⁷ This presentation of the IMDs is based on Noble et al. (2003) and Noble et al. (2008) if not otherwise stated. Only the English IMDs are described.

¹⁸ The IMDs can be retrieved freely from the Internet (Office for National Statistics 2012b).

¹⁹ The overall score of deprivation is constructed using the ranks of the domain-specific indices, applying a logarithmic transformation and a weighting procedure. The transformation is X = -23 * log(1 - R * (1 - exp(-100/23))), where *R* denotes the domain-specific rank which is scaled to the range [0,1] by R = (-1 * domain-specific rank + number of LSOAs)/number of LSOAs. The income and employment indices are more heavily weighted than the other indices and account for 45 per cent of the overall score. The resulting score ranges by definition from [0,100].

use both indices combined for the period 2000 to 2008, because for these years neighbourhood data for Germany is available.

The IMDs can be linked to the BHPS using the Lower Super Output Areas (LSOAs) identification number.²⁰ LSOA are the smallest geographical unit that IMDs are available for. LSOAs contain 1,500 inhabitants on average. In 2007, the population size in LSOAs ranged from 400 to 7,000 inhabitants. Using the LSOA identification number assigned to all households in the BHPS, the IMDs can be easily matched to the individual and household level information.²¹

4.3.2.2 Microm Data Set

The SOEP data can be merged with information on neighbourhoods from the Microm data set. The Microm data set has been generated by a market research company (Microm Micromarketing-Systeme und Consult GmbH) for commercial purposes. The data can be merged to the SOEP data set at the household level from the year 2000 onwards using household identification numbers.²² Data before 2004 is taken from the year 2004, i.e. if the household does not relocate in that time it has the same data values assigned. Most of the data is available at the building (on average 8 households) or street level (on average 25 households). If a building hosts less than 5 households, the building is pooled with similarly structured buildings in the direct proximity to protect privacy (Goebel et al. 2008).²³ Thus, neighbourhood units at the street and building level are considerably smaller than LSOAs in England. The advantages and disadvantages of such "microneighbourhoods" (Hipp 2007: 663) have been described in Section 3.3.3. There it was argued that small neighbourhood delimitations are appropriate to measure neighbourhood quality.

About 60 variables are available in the Microm data set that describe the neighbourhood (Goebel et al. 2008). The data are mainly aimed at identifying potential consumers. Nevertheless, there are numerous variables that are highly relevant for measuring neighbourhood quality. The variables in the data set cover the following topics: purchasing power, consumer behaviour, built structure, demo-

²⁰ LSOAs have been designed to provide constant geographical units over time for statistical purposes and delimitations have been chosen to account for a socially homogeneous population and obvious boundaries (Office for National Statistics 2007). It can be argued that LSOA are a relatively good representation of the neighbourhood concept outline in Section 3.3.3 (Noble et al. 2006).

²¹ The LSOA identification number for respondents in the BHPS is not part of the public distribution of the data set, but can be obtained free of charge under special licence from the UK Data Archive.

²² The data can only be analysed on site under special license from the DIW.

 $^{^{23}}$ Additional data is available on the "*Marktzellen*" level. This unit contains 450 households on average. However, there are only six variables available at this level which cover mobility and the purchasing power in the neighbourhood.

graphic characteristics and residential mobility. The data set also provides a number of comprehensive variables describing dominant life styles and social milieus in neighbourhoods. Information about the generation of all these variables by Microm Micromarketing-Systeme und Consult GmbH is quite limited. In the following, I describe the generation of the variables central for my analysis in detail.

The *purchasing power* variable is based on official tax data which is enhanced through estimations by the data provider. Official register data on income and taxes at the municipality level is broken down to lower levels of data aggregation using regression models (ibid.). An internet search brought to light a Microm documentation from 2000, in which the following variables were listed as entering the regression for modelling purchasing power on lower levels of data aggregation: age structure, credit risk, share of foreigners, Microm-generated variables on social milieus, family structure, building types, street type, cars in area (Microm Micromarketing-Systeme und Consult GmbH 2000). The variable *average population turnover* is based on a data base of movers maintained by the Deutsche Post AG (primary German postal service) and extented with further, not specified data sources (Microm Micromarketing-Systeme und Consult GmbH 2012). For *newspaper readers* and *donations*, there is no information on the generation of the variables.

4.3.3 Regional Level

I also include regional level data in my analysis. The regional data is mainly used to describe housing markets. Data are drawn from official sources.

4.3.3.1 English Regional Data

For England, data at the level of 88 counties is used. I use the variable *share of owned dwellings* which is based on information from the 2001 census (Office for National Statistics 2012b). Unfortunately, there is no information on home owner rates in counties over time from official sources. Thus, the data from 2001 is used for all years.²⁴ The variable *number of new constructions* is based on information from the Department for Communities and Local Government's housing statistics (Department for Communities and Local Government 2012b). Missing information for singles years is linearly interpolated for each county. The variable *population balance* is based on official data from the Office for National Statistics' annual population estimates (Office for National Statistics 2012a). The variable *regional unem*-

²⁴ At the regional level, the share of home owners is available for the years 2000 to 2007. In most regions, home owner rates do not change substantially over time and it can be assumed that this is also true at the county level.

ployment rate is based on data from the official labour market statistics of the Office for National Statistics (Office for National Statistics 2012c). The regional data has been merged with the BHPS data using the Local Authority District code available under special license from the UK Data Archive.

4.3.3.2 German Regional Data

For Germany, data at the level of counties (*Kreise und kreisfreie Städte*) is used. Germany is divided into 413 counties. The information can be matched to the SOEP data using an official identification number (*Gemeindekennziffer*), which is available for all SOEP households under special license. The data is drawn from the INKAR data set from 2010. The INKAR data set is a comprehensive collection of official data provided by the Federal Institute for Research on Building, Urban Affairs and Spatial Development. More specificly, I use the variable *share of buildings with only one or two dwellings* which is based on official statistics from the projection of the dwelling stock in Germany of the federal states and the federal government (*Fortschreibung des Wohngebäude- und Wohnungsbestandes des Bundes und der Länder*). The variable *number of new constructions* is drawn from data on constructions of the federal states and the federal government (*Statistik der Baufertigstellungen des Bundes und der Länder*). The variables *population balance* and *regional unemployment rate* are based on official data from the Eurostat Regio database.

4.4 Sample Selection

My analytic sample consists of data for the years 1991-2008 and is further restricted in a number of ways: First, I exclude all individuals that do not live in private households. For example, students that live in dorms and older people in retirement homes are excluded from the analysis by this rule. I exclude these observations, because the relocation behaviour of individuals in institutional homes is different from other individuals. In addition, measures of housing quality that are used for other households cannot readily be applied for institutional homes, e.g. room stress indicators. I also exclude members of armed forces, as their choice of location is not voluntary. Second, I only include information on the household head and her or his partner. This rule excludes young children living in a shared household with their parents from the analysis. I only consider household heads and their partners, since I concentrate on individuals that have left the parental nest and started their own residential trajectories. Furthermore, the household head and her or his partner are most likely to make decisions regarding relocation behaviour in the household. To identify household heads, I use the indicators provided in the BHPS and SOEP. In the BHPS, the household head is the principal owner or renter of the property. In case of shared ownership, the eldest owner takes precedence (Taylor et al. 2010: App2-3). In the SOEP, the household head is the person that considers her- or himself as being the person that can best provide information about the household (Haisken-DeNew and Frick 2005: 21).

Third, I exclude all individuals younger than 18 years and older than 79 years. Individuals below the age of 18 that relocate independently from their parents are too few. Individuals above the age of 79 are likely to have a particular relocation behaviour. For example, older individuals are more likely to relocate because of their health. Fourth, I do not consider households of unrelated adults on the grounds that in these households it is not clear who makes decisions about relocations. In addition, in households of unrelated sharers, the clustering of observations in households cannot be avoided by separating the analysis by gender. Due to this problem, I also exclude homosexual couples from the analysis. Finally, I only consider individuals that have been observed for at least two observations. Individuals that have only been observed in one panel wave cannot contribute information about changes in the quality of their location and, thus, are excluded. Further, more specific restrictions of the sample are described in the introductions to each empirical chapter.

4.5 Measurement

I now describe the construction of the dependent (Section 4.5.1.1) and independent variables (Section 4.5.2) used in the empirical analysis. Descriptive statistics of these variables are presented in each empirical chapter, as the analytic samples differ between the chapters. The original names of the variables in the BHPS and SOEP that are used in the present analysis are presented in Table A.1 in the appendix.

4.5.1 Dependent Variables

4.5.1.1 Residential Relocation

In Chapter 5, I analyse the occurrence of *residential relocations*. Beside being the dependent variable in Chapter 5, residential relocations is also the most important independent variable in Chapter 6 and Chapter 7. I identify residential relocations through the address log files that the field work teams maintain for each respondent. Through these files, respondents that live at a new address compared to the last interview can be identified. I consider all address changes as residential relocations irrespective of the distance that the respondents relocated. Using the address log files instead of subjective measures of mobility is especially helpful for also gathering in-

formation on respondents that have not been interviewed, but for which a address log file entry is available. This is relevant to analyse the selectivity of attrition after residential relocations. However, for these cases, there is no information available from the interview after the residential relocation. For the analysis of the occurrence of residential relocations, the event is measured at the survey year, before the relocation takes place, i.e. relocations are measured at *t* if it occurs until t + 1.²⁵ For the analysis of the effects of relocations on other variables, the relocation variable is coded 1 at *t*, if a relocation occurred between t - 1 and *t*. Otherwise the variable is coded 0.

For the analyses in Chapter 6 and Chapter 7, I differentiate types of relocations depending on whether households relocate together, or whether households change their compositions with the relocation: 1) *Stable single* (\leq 35 years): A single under the age of 36 relocates and stays single after the relocation. 2) *Stable single* (\geq 36 years): A single above the age of 35 relocates and stays single after the relocation. 3) *Stable couple*: A couple relocates collectively and still lives together after the relocation. 4) *Single to couple*: A single relocates and lives together with a partner after the relocation. 5) *Dissolved couple*: A couples relocates and the partners live in separate households after the relocation. 6) *Other relocations*: If the household of origin or destination is neither a single nor a couple household, I categorise the relocation could not be identified clearly. All variables are binary and coded 1 if the particular event took place and 0 otherwise.

In Chapter 8, I analyse the effect of *long-distance relocations* on dual-earner couples. I operationalise long-distance relocations using a proxy. I treat relocations across borders of Local Authority Districts (LADs) in England and relocations across county borders in Germany as long-distance relocations. This approach has also been used in past research in England (e.g. Rabe 2011) and Germany (e.g. Jürges 2006: 306, but author also takes into account the distance between counties' capitals). Arguably, relocations across administrative borders is a relatively imprecise proxy for long-distance relocations. Relocations across county and LAD borders may be very short, if the former and new location is close to a mutual border. At the same time, relocations within administrative borders may be actually long, if the county or LAD covers a large area. Therefore, I re-estimated all models in Chapter 8 using the actual relocation distance with a cut-off value of 50 km to indicate long-distance relocations. 50 km is a common threshold for long-distance relocations in the literature (Boyle et al. 2003). Unfortunately, the information on relocation distances is not available in the SOEP before 2001. Therefore, the main

²⁵ Due to the fact that data collection for both panels is conducted on a yearly basis, not more than one relocation between two consecutive interviews is observable. However, certain individuals may move more often. To circumvent this problem I considered monthly observations of the dependent variable, since the month of relocating into the current dwelling is recorded in both panels. However, the vast majority of relevant covariates are only collected yearly.

results are presented using the cross-border proxy for long-distance relocations. I refer to the results using actual relocation distances, if they differ substantially.

4.5.1.2 Room Stress

In Chapter 6, I analyse the effect of residential relocations on *room stress*. In Section 3.2.1, I identified room stress as one of the most important measures of housing quality. Room stress measures the relation of room needs to available rooms for a household and is a standard measure of housing quality (e.g. Rossi 1955: 179; Townsend 1979: 476). In this analysis, the room need of a household is defined as follows:

- Two rooms are added for the household head.²⁶ If the head is living with a partner, they also have a combined room need of two rooms.
- For every other household member above the age of 9, one more room is added.
- For every child below the age of 10 in the household, half a room is added. The room need is rounded upwards.

The construction of the room need measure follows Clark (1992) and Huang (2003: 599), but the room need of adolescents is weighted more heavily. I call this room need the 'objective room need'. The available rooms are then subtracted from the objective room need for the household to compute the room stress variable.²⁷ Thus, a positive room stress value indicates that a household has less available rooms than its objective room need, while a negative room stress value indicates that a household has excess rooms relative to the objective room needs. I understand room stress as a relative measure. While the scale has a zero point at which objective room needs and number of actual rooms match, I do not claim that this is the optimal room stress situation for each household. As becomes apparent in the empirical analysis, most individuals live in 'negative' room stress, i.e. their dwellings have more rooms than they would need according to their objective room need. Thus, the room stress indicator may be rather interpreted as a measure of wealth than deprivation in England and Germany.

²⁶ A single person is in need of two rooms to separate basic functions of the dwelling, e.g. eating and sleeping, in different rooms and to be able to sustain their privacy even when hosting guests.

²⁷ The variable for the number of rooms in the dwelling is measured slightly different in the BHPS and SOEP. In the BHPS, respondents are asked for the number of all rooms excluding kitchens, bathrooms and sublet rooms. In the SOEP, respondents are asked for rooms larger than 6 square metres, excluding kitchens and bathrooms. Thus, a similar dwelling may be reported slightly different in the two surveys and English households are likely to report more rooms than German households. This has to be kept in mind in comparing descriptive measures of room stress across the two countries.

4.5.1.3 Neighbourhood Quality

In Chapter 7, I analyse the effect of residential relocations on *neighbourhood quality*. To measure neighbourhood quality in England, I use data from the IMD2004 and IMD2007 (cf. Section 4.3.2.1). However, Niggebrugge et al. (2005) and Rabe and Taylor (2010) report problems with the indices of Barriers to housing and services and Living environment deprivation, since the indices do not behave in the expected way. In Table 4.3 a principal component analysis of the domain specific indices is presented. All indices but Barriers to housing and services and Living environment deprivation load highly and uniquely on the first component, indicating that these indices of Barriers to housing and services and Living environment deprivation load highly and services and Living environment deprivation load on a separate component. Therefore, I exclude both indices, Barriers to housing and services and Living environment deprivation, from the analysis of neighbourhood quality.

IMDs	Factor 1	Factor 2	
Income	0.917	0.147	
Employment	0.935	-0.011	
Health deprivation and disability	0.921	-0.006	
Education, skills and training	0.850	-0.191	
Barriers to housing and services	-0.107	0.902	
Crime	0.748	0.082	
Living Environment	0.555	0.555	

Table 4.3 Principal Component Analysis of IMDs

Data: IMD 2007

Note: Only data from 2007; Method: Principal Component Analysis, Rotation: Varimax; Extraction criteria: Eigenvalue > 1; overall KMO= 0.83

I use the remaining domain-specific indices to compute a general neighbourhood quality measure. To this end, I rescale the score of each index to vary between 0 and 100. Then, I sum the scores. I weigh the income deprivation score twice as much as the other scores together to account for the high importance of the economic situation in a neighbourhood. I use the logged scores for the indices of income deprivation, employment deprivation and education deprivation, because they have a very skewed distribution. Finally, I standardise the variable with a mean of 0 and a standard deviation of 1. I invert the variable, so that a higher value indicates a better neighbourhood quality. I assign the data from IMD2004 to the years 2000 to 2003 and the data from IMD2007 to the years 2004 to 2008.

The neighbourhood quality measure for Germany is based on four variables from the Microm data set (cf. Section 4.3.2.2): average purchasing power (used as a measure of affluence), residential turnover rate (used as measure of residential stability),
prevalence of national newspaper readers (used as proxy for cultural capital), prevalence of donations (used as proxy for social capital and affluence). Purchasing power is a continuous variable. I standardise the variable separately for West and East Germany and each observation year. I standardise separately for East and West Germany, because of the high average difference in purchasing power between the two parts of Germany. The other variables are categorical and have 9 categories each. The categorical variables are based on underlying continuous variables, which are themselves not part of the Microm data set. For each of the categorical variables, 1 indicates a very low value, 5 indicates an average value and 9 indicates a very high value on the underlying continuous variable. Purchasing power is measured at the street sector level. No lower level information is available for this variable. The other variables are measured at the building level. Besides the residential turnover rate, no higher level information for these variables are available. I rescale all variables so that they vary between 0 and 100. To test the assumption that these variables measure the latent concept neighbourhood quality, I perform a principal component analysis. The results in Table 4.4 show that for the year 2008, two principal components are extracted. Purchasing power, newspapers and donations load on the first component, while only residential stability loads on the second component. Due to the theoretical relevance of residential stability and the lack of better measures I still decide to combine these variables in the neighbourhood quality measure. To construct the neighbourhood quality measure, all variables are summed and the average purchasing power is weighed twice as much as the other variables. The resulting index is standardised in the same way the English neighbourhood quality measure is standardised. A higher value indicates a better neighbourhood quality.

In the analysis, the focus is on changes in neighbourhood quality due to residential relocations. The issue of changes in the neighbourhood itself is left aside. These changes may affect the neighbourhood quality for the residents in the area, but may also trigger relocations. For example, gentrification processes may affect the average neighbourhood quality for the residents in a neighbourhood. At the same time, individuals may leave the neighbourhood because of these changes. These changes are very interesting and important to analyse. Unfortunately, the available data does not adequately capture these dynamic processes. The IMDs are only measured at two points in time. The Microm data only varies over time after 2003. Therefore, I average the neighbourhood quality across residential spells, i.e. the time that an individual lives in a particular neighbourhood. Thus, by definition, neighbourhood quality can only change, if the individual relocates. As long as individuals live in the same neighbourhood, the have the same neighbourhood quality assigned.

The neighbourhood quality measures for England and Germany use the available data to its maximum. They describe neighbourhoods in England and Germany in the best way possible given data limitations. However, this makes a direct comparison of changes in neighbourhood quality between England and Germany impossible, because very different measures for neighbourhood quality are applied. Only the

Variable	Factor 1	Factor 2	
Purchasing power	0.744	0.298	
Residential stability	-0.028	0.975	
Newspapers	0.887	-0.286	
Donations	0.825	0.101	

Table 4.4 Principal Component Analysis of Microm variables

Data: Microm 2010

Note: Only year 2008; Method: Principal Component Analysis, Rotation: Varimax; Extraction criteria: Eigenvalue > 1; overall KMO= 0.48

direction of changes can be interpreted meaningfully, if it is assumed that the variables measure the latent variable neighbourhood quality.

4.5.1.4 Career Characteristics

In Chapter 8, I analyse the effect of long-distance relocations on subsequent employment of dual-earner couples. I analyse whether partners *leave employment* after relocations. The leaving employment variable is binary and coded 1 if the respondent is unemployed, inactive or in education at t + 1 and coded 0 if the respondent is still in employment, or entered other activities at t + 1. Individuals that are in other activities at t + 1 are treated as not having left employment, since they may still have a legal work contract, e.g. when in maternal leave. The measure of employment status depends on the status at the time of the interview irrespective of changes between the interviews.

Additionally, to describe the quality of pre- and post-relocation jobs for those that maintain employment, the hourly gross *wage rate* is analysed in accordance with recent literature (e.g. Böheim and Taylor 2007). The hourly wage rate is a more accurate measure of potential lifetime earnings than the gross labour income, for the reason that the hourly wage rate is independent from the work hours of individuals. Past literature shows that especially women temporarily decrease work hours after relocations (Boyle, Feng and Gayle 2009). By using the hourly wage rate, this drop in work hours does not affect my results. Imputed, current, monthly labour wages before taxes at the time of interview deflated to prices of 2006 are used. The wages are made comparable across countries by using purchasing power parity. Reported normal working hours and overtime are used to compute hourly wage rates. I assume that for paid overtime respondents receive a 50 per cent wage premium (see similarly Rabe 2011).²⁸

²⁸ The amount of paid overtime is not reported in the SOEP before 2001. Instead a categorical variable measures whether respondents are compensated with free time, extra pay, or both for overtime, or whether they are not compensated at all. I use this variable as a proxy for paid overtime

4.5.2 Independent Variables

In my analyses I include the following independent variables.

- Women is coded 1 if the respondent is female and 0 if the respondent is male.
- The *age* of respondents in years is considered. To model non-linear relationships with age, I also include $age^2/100$ in some analyses. I divide the squared term by 100 to make the scale of the variable more similar to the other variables in the analyses. The variables are continuous.
- I include variables for the *presence of children aged 0 to 4*, *presence of children aged 0 to 5*, *presence of children aged 0 to 9*, *presence of children aged 5 to 10* and *presence of children aged 10 to 17*. All variables are binary and indicate whether at least one household member is in the given age range (coded 1), or whether no household member is in the given age range (coded 0). In some analysis, I also include a variable indicating the event of childbearing. To avoid multicollinearity between the childbearing variable and the variable *presence of children aged 0 to 9*, I include the variable *presence of children aged 1 to 9* instead of latter variable. The variables are measured at the household level.
- I differentiate several life course stages with regard to age, partnership status and presence of children: *single* (≤ 35 years old), *single* (≥ 36 years old), *single parent, couple parent, childless couple* and *multigenerational household*. In multivariate analysis, these life course stages are included as dummy variables coded 1 if the respondent is in this stage and 0 otherwise. The variable is measured at the household level.
- I include various family events in my analysis. The variable *child born* is coded 1 if respondents experienced a childbirth between t 1 and t and 0 otherwise. If a child left the household between t 1 and t, the variable *child moved out* is coded 1 and 0 otherwise. If respondents live in a household in which a partner moved in after the last interview, the variable *partners moved together* is coded 1 and 0 otherwise. If respondents lived in a shared dwelling with a partner at t 1, but not anymore at t the variable *partners moved apart* is coded 1 and 0 otherwise. If respondents lived in a shared dwelling with a partner at t 1, but not anymore at t the variable *partners moved apart* is coded 1 and 0 otherwise. If respondents' partners died between t 1 and t, the variable *partner died* is coded 1 and 0 otherwise.
- I differentiate three categories in the partnership status. The variable *single* is coded 1, if respondents live without a partner in the same household.²⁹ If respondents cohabit with a partner, but are not married, the variable *cohabiting* is coded 1 and otherwise 0. If respondents live together with their spouses, the

before 2001 (no paid overtime if not compensated or compensated with free time; all overtime is paid if compensated with extra pay; 50 per cent of overtime is paid if compensated with extra pay and free time).

²⁹ Living-apart-together (LAT) couples can only be identified in a few waves in the BHPS. While respondents in LAT couples could be identified in the SOEP, for reasons of comparability, LAT couples are treated as singles in the present analysis.

variable *married* is coded 1 and 0 otherwise. Because I do only include one single or one couple per household in my analytic sample, the partnership status variables are constant at the household level.

- I measure the presence of immigrants in the household. Immigrants have been born outside of England and Germany respectively (first generation immigrant) or have been born in England and Germany respectively by parents of which at least one party immigrated (second generation immigrant). *Immigrant in HH* is a binary variable that takes the value 1 if at least one household member is a first or second generation immigrant and 0 if no household member is an immigrant.
- For England, the binary variable *egalitarian couple* is coded 1, if both partners strongly agree or agree to the statement "Both the husband and wife should contribute to the household income" and is coded 0 otherwise. For Germany, the variable *egalitarian couple* is coded 1, if the male partner contributes at least one third of the couples' time use on chores, shopping and childcare on a regular Sunday and is coded 0 otherwise (cf. Jürges 2006).
- The variable *she earns* ≥ 50 % is coded 1, if the female partner earns at least 50 per cent of the couple's annual labour income and is coded 0 otherwise. *She earns* ≤ 25 % is coded 1, if the female partner earns not more than 25 per cent of the couple's annual labour income and is coded 0 otherwise. Both variables are included at the same time in the analysis and the reference category is women that earn more than 25 per cent and less than 50 per cent of the couples' annual labour incomes (even though the operationalisation of the variable is different, the basic idea follows from Sørensen and McLanahan 1987).
- *Household income (log)* is measured as annual household income after taxes and including transfers and benefits.³⁰ To adjust the income for economies of scale in the household and age-specific needs, I equivalise the income using the modified OECD-scale. This scale assigns a value of 1 to the household head, a value of 0.5 to each additional household member above the age of 13 and a value of 0.3 to each child under the age of 14 (Eurostat 2011).³¹ I deflate the income to prices of 2006. I adjust the income using purchasing power parity to make it comparable across countries. Thus, the unit of income is PPP-\$. Finally, I transform the variable using the logarithm, because in general income has decreasing marginal returns. The variable is measured at the household level and is continuous.

³⁰ For the SOEP I use post-government income information from the Cross-National Equivalent Files. For the BHPS, I use post-government income data from the 'Derived Current and Annual Net Household Income Variables, BHPS waves 1-18' (Levy and Jenkins 2012). For the BHPS, post-government income information is only available for households in which all household members responded.

³¹ I use an equivalised income measure in my analysis, as I do not include counts of the number of household members in the relevant age groups in my multivariate models.

- I use a continuous variable *relocation distance* that measures the distance between the present and the previous locations of respondents in kilometres. The distance measure is constructed by the survey teams using geo-coded address data of the respondents. The distance variable is available with the public BHPS file and for the SOEP it is available under special license. Geo-coded addresses in the SOEP are only available starting in 2000.
- The variable *involuntary relocation* is coded 1, if respondents report at least one of the following reasons for their relocations: 'notice given by landlord', 'housing costs too high' and 'partnership dissolution'. The reason 'housing costs too high' cannot clearly be identified in the BHPS data, because the reason is not coded alone, but rather as 'Wanted smaller/cheaper accommodation'.
- The variable *years in residence* measures the time that respondents lived in a dwelling. In the SOEP, only the household head reports the time she or he has lived in the dwelling. I assign this value to all household members, but I restrict the maximum time in residence to respondents' age. This is a continuous variable.
- I measure the *housing costs* of households. The housing costs cover monthly gross rent and mortgage payments. I adjust the costs using purchasing power parity to make them comparable between England and Germany and divide the costs by 100. Thus, the variable indicates the monthly housing costs in PPP-\$ 100. The variable is continuous and measured at the household level.
- The tenure status of respondents is measured at the household level. In England, the variable *social housing* indicates whether respondents live in accommodation rented from local authorities or housing associations (coded 1, coded 0 otherwise). In Germany, the variable *social housing* indicates whether respondents live in rental accommodation with rent control due to public subsidization of the dwelling (coded 1) or not (coded 0). The variable *rented privately* is coded 1 if respondents live in rented accommodation that does not fall in the category of social housing just described and coded 0 otherwise. If respondents live in own homes for which they still pay mortgages, the variable *owned with mortgage* is coded 1 and 0 otherwise. If respondents live in own homes without paying mortgages, the variable *owned outright* is coded 1 and 0 otherwise.
- The employment status is measured with three binary variables. The variable *employed* is coded 1 if the respondent is in paid employment or self-employed and coded 0 if the respondent is not in paid employment or self-employed. The variable *unemployed* is coded 1 if the respondent reports being unemployed and coded 0 otherwise. The variable *inactive* is coded 1 if the respondent is not in paid employment, self-employed or unemployed and coded 0 otherwise. For example, respondents are inactive, if they are retired, in education, or in parental leave.
- I include variables describing events in the job trajectory in the analyses. The variable *promotion* indicates whether respondents' real hourly wage rate increased by more than 10 per cent since the last interview (coded 1, coded 0

otherwise). The variable *demotion* is coded 1 if respondents' real hourly wage rate decreased by more than 10 per cent since the last interview and is coded 0 otherwise. The variable *became unemployed* is coded 1 if respondents are unemployed at the time of the interview, but were not unemployed at the last interview (coded 0 otherwise).

- I use the Comparative Analysis of Social Mobility in Industrial Nations (CAS-MIN) classification scheme (König, Lüttinger and Müller 1988). The CAS-MIN scheme differentiates nine levels of qualification: inadequately completed education (1a), general elementary school (1b), basic vocational qualification (1c), intermediate general qualification (2a), intermediate vocational qualification (2b), higher general qualification (2c_gen), higher vocational qualification (2c_voc), lower tertiary qualification (3a) and higher tertiary qualification (3b). I combine these into three categories: *basic* (1a, 1b, 1c), *intermediate* (2a, 2b, 2c), and *higher education* (3a, 3b). In the analysis, I include basic and higher education as dummy variables with intermediate education as the reference category.
- To measure the attachment to the labour market I use the variable *continuous work history*. The variable measures the time the respondent spent being employed since she or he first left education as a share of the time since she or he left education. Thus, a respondent that worked continuously since she or he left education has a value of 1, while a respondent that was out of employment for some time since she or he left education will have a value smaller than 1. A respondent that never worked will have the value 0.
- I measure the occupational status of respondents using four binary variables. I use the one-digit International Standard Classification of Occupations (ISCO) code to generate the variables. The variable *professional/manager/technician* is coded 1, if respondents work as legislators, senior officials and managers (ISCO 1), professionals (2) or technicians and associate professionals (3). Otherwise the variable is coded 0. If respondents work as clerks (4), service workers and shop and market sales workers (5), the variable *skilled non-manual* is coded 1 and 0 otherwise. If respondents work as skilled agricultural and fishery workers (6), craft and related trades workers (7), or plant and machine operators and assemblers (8), the variable *skilled manual* is coded 1 and 0 otherwise. For respondents in elementary occupations, the variable *partly and unskilled occupations* is coded 1 and for all other respondents the variable is coded 0. Members of armed forces (0) are dropped from the analysis (cf. Section 4.4).
- *Time with employer* measures the time respondents worked in their current company in years. The variable is continuous.
- The binary variable *permanent position* is coded 1, if respondents have a permanent work contract and is coded 0 otherwise.
- The continuous variable *work hours* measures the number of hours worked including overtime in an average week in the current job.

- To control for regions, I include binary variables for *East* Germany (coded 1 for East Germany, coded 0 for West Germany) and *Southeast* England (coded 1 for Southeast England including London and coded 0 for the rest of England). See Section 4.1.1 for a justification.
- I measure the *degree of urbanisation*. The variable is measured at the level of counties in Germany and at the level of LADs in England. In Germany, I derive the degree of urbanisation from the municipality classification system (*Siedlungsstrukturelle Gemeindetypen*) of the Federal Institute for Research on Building, Urban Affairs and Spatial Development (Bundesinstitut für Bau-Stadt- und Raumforschung 2009). I join the categories of core cities (*Kernstädte*) to form the metropolitan category. Rural counties are joined to form the rural category. The rest of the municipalities make up the urban category. In England, the variable is based on the 'Defra Classification of Local Authority Districts and Unitary Authorities in England' (Rural Evidence Research Centre 2005). 'Major Urban' and 'Large Urban' cities make up the metropolitan category. 'Other urban' and 'Significant Rural' LADs are joined in the urban category. 'Rural 50' and 'Rural 80' form the rural category.
- At the level of counties in Germany and LADs in England I measure three characteristics of housing markets. In England, I use the share of *owned dwellings* in a housing market as a measure for the general quality of locations in the market. In Germany, I use the share of buildings with only one or two dwellings as a proxy for house ownership, because no information on house ownership itself is available. The variable *new constructions* measures the number of newly constructed dwellings per 1,000 existing dwellings in each year. The *population balance* is measured as the change in total population of a given year from the year before per 1,000 inhabitants averaged over the last five years. This is a proxy for the demand side on the housing market. All three variables are continuous.
- At the level of counties and LADs, I also measure the *regional unemployment rate* for the civilian labour force.

4.6 Methods

The theoretical model developed in Chapter 2 motivates the choice of quantitative, empirical methods in the present analysis in the following ways: First, the theoretical model makes assumptions about the effect of a relocation event on individuals' quality of location. To estimate the effect of the event as accurately as possible with the available observational data, appropriate methods of regression analysis should be used that account for heterogeneity and self-selection of individuals. Furthermore, the theoretical model relates changes in individuals' life courses to changes in the quality of locations. The empirical analysis should be able to capture these dynamics. Fixed-effects regression models meet these requirements. Fixed-effects regression models are the working horse of the present analysis. The models are described in detail in Section 4.6.1. They are applied throughout Chapter 6 and 7. Second, the theoretical model assumes that individuals are embedded in housing markets that shape individuals' opportunities and constraints. Therefore, methods must be applied that allow modelling the contextual effect of housing market characteristics on individuals. Multi-level regression models are appropriate for this kind of analysis. I describe these models in Section 4.6.2. They are applied in Chapter 6 and 7.

Third, the theoretical model assumes that partners in couples are nonindependent in their behaviour. Kenny, Kashy and Cook (2006: 5) highlight four reasons for this nonindependence: 1) "compositional effect": partners are not randomly selected but are similar in a variety of characteristics; 2) "partner effect": behaviour or characteristics of one partner affects the other partner; 3) "mutual effect": both partners' behaviour or characteristics influence each other; and 4) "common fate": both partners are exposed to the same external conditions. Therefore, methods must be applied that estimate the effect of relocations taking this nonindependence and unobserved, shared characteristics of partners into account. The actor-partner interdependence model, which can be set up as a multi-level model, deals appropriately with these problems and I describe the method in Section 4.6.3. The actor-partner interdependence model is applied in Chapter 8. For the analysis of the occurrence of residential relocations in Chapter 5, I use event history analysis which is further motivated in Section 4.6.4. ³²

4.6.1 Fixed-Effects Regression Models

Sociological research uses mainly non-experimental, observational data, in which systematic differences in the observed individuals may obscure the effect of explanatory variables, i.e. some kind of treatment, on the dependent variable. First, individuals may be systematically different because of heterogeneity, i.e. selection bias. Second, individuals may be different due to endogeneity, i.e. self-selection into the treatment group (Gangl 2010). It follows that "because [in] observational data analysis [...] the researcher does not have the capacity to assign individuals to treatments [...], researchers must instead investigate how individuals end up in alternative treatment states" (Morgan and Winship 2010: 41). Simple regression analysis may be used, to control for the non-randomness of treatment. If relevant variables are controlled in regression models so that treatment "is as good as randomly assigned" (Gangl 2010), causal effects can be estimated. However, unobserved heterogeneity

 $^{^{32}}$ The software Stata version 10.1 (StataCorp 2007) has been used for all empirical analyses, if not otherwise stated.

cannot be accounted for in such regression models (Halaby 2004). In addition, individuals intentional choice of the treatment is mostly not considered (Gangl 2010).

As an example for the present analysis, individuals living in low-quality neighbourhoods may select themselves into the group of movers. If changes in neighbourhood quality after relocations are analysed without taking account of this selection process, the coefficients for relocations may be biased. To alleviate this problem, panel data sets such as the BHPS and the SOEP have the great advantage to offer several observations for each individual. These subsequent observations can be used to identify the effects of certain changes controlling for self-selection and timeconstant, unobserved heterogeneity by using special panel regression methods such as fixed-effects regression models. The fixed-effects model is a within-estimation procedure. That is to say that only within-individual variance in the independent variables is used to account for within-individual variance in the dependent variable. Following from this, the model provides a stronger support for causal claims not biased by self-selection processes and time-constant, unobserved heterogeneity, as only changes in the dependent and independent variables are considered.³³ Fixedeffects regression models have repeatedly been praised and considered superior to conventional cross-sectional regression models in the literature (Halaby 2004; Allison 2009: 1ff; Gangl 2010).

The fixed-effects model for individual i and time period t can be derived starting from the ordinary least squares (OLS) regression model for pooled observations, i.e. several t are recorded for each individual:

$$y_{it} = \alpha + \beta x_{it} + \gamma z_i + u_{it}, \qquad (4.1)$$

where α is a constant, β and γ are vectors of coefficients, x_{it} is a vector of timevariant covariates and z_i a vector of time-constant covariates (Allison 2009: 12ff; Giesselmann and Windzio 2012: 40ff). However, in OLS estimation for linear regression, it is required that u_{it} is not correlated across time and is independent from x_{it} as well as z_i to estimate unbiased coefficients among other assumptions (Menard 2008: 4ff). The assumption that u_{it} is not correlated with z_i and x_{it} is very strong, as certain unobserved characteristics of individuals are probably correlated with other variables in the equation. For example, this may be a problem if time-constant preferences for certain types of dwellings are not measured in a model, but are correlated with income and both affect outcomes of relocations. For time-constant unobserved heterogeneity, this problem can be solved by decomposing the error term u_{it} into v_i , which is fixed for each individuals and periods. By eliminating v_i from the equation, the assumption of no correlation between the error term and the independent

³³ The causal claim may be strengthened even more by explicitly considering the time order of events and processes. Thus, a prior change in the independent variables can be related to a later change in the dependent variable.

variables is weakened to the degree that the idiosyncratic error term should not be correlated with the predictors and with itself over time.

To eliminate v_i from the equation, several solutions are possible. The easiest way would be to include individual-specific dummy variables that control for v_i and a pooled OLS regression could be estimated. However, with large samples the computational burden of estimating all individual-specific dummy variables would be high. An alternative and equivalent approach is to de-mean all time-variant variables by using the individual-specific means for all covariates over time such that

$$\overline{y}_{it} = \alpha + \beta \overline{x}_{it} + \gamma z_i + v_i \tag{4.2}$$

and then subtract Equation 4.1 and 4.2. This results in

$$y_{it}^* = \beta x_{it}^* + \varepsilon_{it}, \qquad (4.3)$$

where the time-constant individual error term v_i has been eliminated from the equation and the time-constant unobserved heterogeneity no longer poses as problem. The time-constant α has been eliminated. Furthermore, time-constant covariates cannot be modelled anymore, because they are differenced out in Equation 4.3. However, if the research question motivates the analysis of a time-constant covariate, the covariate may be interacted with a time-varying variable. Equation 4.3 can be estimated using OLS.

It should be noted that the model still relies on the assumption that ε_{is} is not correlated with x_{it}^* . The subscript s indicates that no ε in the past, present or future is allowed to correlate with x_{it}^* in the present. This is the assumption of strict exogeneity. However, due to unobserved time-varying heterogeneity this assumption may be violated. In this case, the causal effect of x_{ii}^* cannot be identified. Further, the assumption of exogeneity can also be violated due to reversed causality and measurement error. The effect of measurement error may be stronger in fixed-effects regression than in other types of regression models, since measurement error tends to vary from year to year (Angrist and Pischke 2009: 225). Fixed-effects models also have the same trend assumption. This means that individuals just differ in the level of their individual time-constant variables, but not in their slopes, e.g. fixedeffects models control for selection in the level of time-constant variables but not for selection in trends in time-constant variables (ibid.: 221). The further assumptions that ε_{it} should not be auto-correlated within individuals and the assumption of homoscedasticity can be dropped by computing standard errors for cluster-correlated data (Williams 2000).

One of the potential problems of the fixed-effects model is the lower efficiency of the model compared to other specifications, because, at least implicitly, an additional parameter for the number of individuals minus 1 has to be estimated in the model. In addition, fixed-effects models only make use of a relatively small portion of the available sample, on the grounds that only cases with within-variation in the dependent and independent variable are used to draw inference from. That is to say that the model estimates the average treatment effect on the treated (ATT) in contrast to the population average treatment effect (ATE) (Gangl 2010). Individuals not experiencing a treatment are not contributing to the estimation of the treatment effect. Therefore, generalisations can only be made for those that experienced a treatment. This is especially relevant for the present analysis. Relocations are rare and only those individuals that experience a relocation contribute to the estimated coefficient for relocations. However, the research question of the present analysis is only concerned with outcomes of relocations and, therefore, it is sensible to draw inference only from movers.

An alternative way to model panel data would be random-effects models, which are more efficient than fixed-effects models, because they also consider individuals without within-variation. However, in most cases, random-effects estimations will be biased, as random-effects models rely on the assumption that v_i and all independent variables are not correlated. Only in the case that this assumption is met, random-effects models provide unbiased and more efficient estimations than fixed-effects models. To test this assumption, a Hausman test can be conducted based on the null hypothesis that the coefficients of a random-effects model are not significantly different from a fixed-effects specification.³⁴ Only if this null hypothesis cannot be rejected, the random-effects specification should be preferred. However, if the research question motivates a between-individual analysis, e.g. if effects of time-constant characteristics of individuals should be examined, the random-effects model should be preferred over the fixed-effects model. In cases of unbalanced panel data, random-effects models are more efficient than pooled OLS regression to analyse between-individual effects (Giesselmann and Windzio 2012: 91).

It was mentioned above that panel attrition may pose a problem for the analyses of longitudinal data. Results from fixed-effects models are unbiased from attrition, if attrition is random, i.e. unbalanced panel designs are not a problem for fixed-effects regression. Estimation is also unbiased, if non-response is correlated with time-constant characteristics, but problems arise if attrition is correlated with the idiosyncratic error (Wooldridge 2002: 579ff). To test the effect of attrition on the estimation results, Wooldridge (ibid.: 581) recommends to include a lagged indicator for non-response or a lead indicator of panel attrition in the estimation model. If these indicators are not significantly affecting the outcome, this is evidence against correlation of time-varying unobserved characteristics that are correlated with the dependent variable and the selection process controlled for all independent variables.

³⁴ Hausman (1978) suggests a test to compare two sets of coefficients of which one may be inconsistent but more efficient than the other.

4.6.2 Multi-Level Regression Models

In recent decades, special regression methods have been developed to capture the multilevelled nature of human society, i.e. individuals and other units of analysis are nested within hierarchical groups. A classic example for a three-level data structure are students who are nested within classes which are nested within schools. In multi-level regression, the different levels of data are explicitly taken into account. For longitudinal research questions, time can be modelled as an additional level of data. For particular longitudinal research questions in which units of analysis are also nested in groups, multi-level regression may be a more appropriate analysis tool than fixed-effects regression. Panel regression and multi-level regression models share many similarities as becomes apparent in the following.

Multi-level regression can be understood as a two-step regression analysis (Kenny, Kashy and Cook 2006: 82ff). Here, a two-level model with random intercepts that I use in my analysis is described. In the first step of multi-level regression, the following linear regression equation is estimated:

$$y_{ji} = \beta_{0j} + \beta_1 x_{ji} + \varepsilon_{ji}, \qquad (4.4)$$

where y_{ji} is the outcome for individual *i* in group *j*, β_{1j} is a vector of regression coefficients, x_{ji} are variables at the level of individuals and ε_{ji} is random error. What distinguishes Equation 4.4 from a simple linear regression is the index *j* for the constant β_{0j} . The index indicates that the constant is allowed to vary by group. This means that the intercept may shift the regression line for each group up or down, while the regression lines of all groups remain parallel. In the second step of multi-level regression, β_{0i} is treated as the outcome of another regression equation:

$$\beta_{0j} = a_0 + a_1 z_j + v_j. \tag{4.5}$$

In this equation, the intercepts β_{0j} are a function of the constant a_0 , group-level predictor variables z_j and random error v_j . Equations 4.4 and 4.5 can be joined so that

$$y_{ji} = a_0 + a_1 z_j + \beta_1 x_{ji} + \nu_j + \varepsilon_{ji}.$$

$$(4.6)$$

Equation 4.6 has two random error terms: v_j is a normally distributed random variable that represents variation in the group-specific intercepts that is not due to z_j . ε_{ji} is a normally distributed random variable that represents variation across individuals controlled for x_{ji} and z_j . Thus, the intercept consists of two parts in this equation. The fixed part is a_0 and is constant across all individuals, while the random part is v_j and varies between groups (Twisk 2006: 28).

The levels of analysis may not always be as clearly distinguishable as in the student-class-school example. For example, individuals can be members in more than one housing market over time if they relocate. This kind of data structure can be

accommodated by multiple membership multi-level models (MMMLMs) (Browne, Goldstein and Rasbash 2001; Windzio 2008: 136).³⁵ In MMMLMs the membership in different groups is weighted. One common way is to use the time which respondents are exposed to the respective groups as a weight. In the estimation procedure, each respondent receives a weighted combination of residuals of all groups the respondent belonged to over time (Rasbash et al. 2009a: 283). To estimate MMMLMs, Markov Chain Monte Carlo (MCMC) methods are used in MLwiN (Browne 2012: 231).³⁶

4.6.3 Actor-Partner Interdependence Models

Nonindependence in dyads, e.g. couples, prohibit the naive use of OLS regression to analyse dyadic data, because one basic assumption of OLS is independence of observations in the data. Instead, Kenny, Kashy and Cook (2006: 144) suggest actorpartner interdependence models (APIMs) for analysing mixed independent variables, i.e. variables that vary between couples and within couples. In APIMs each partner is at the same time "*agent* of influence" and "*target* of influence" (Oriña, Wood and Simpson 2002: 460, emphasis in original). The model allows estimation of actor effects (i.e. intrapersonal effects) and partner effects (i.e. interpersonal effects) at the same time by including variables describing the respondent and her or his partner (Kenny, Kashy and Cook 2006: 146). Thus, the model allows estimation of the effect of variables on one partner controlled for characteristics of the other partner and the other way around. Furthermore, by including couple-level random effects" (Cook and Kenny 2005: 103).

APIMs can be estimated using different strategies. I describe a multi-level regression approach for distinguishable dyads, i.e. dyads in which the two members can be differentiated based on a variable such as gender. Dyadic data can be characterised as multi-level data with individuals at the lower level and couples at the higher level (Kenny, Kashy and Cook 2006: 87, see previous section on more details about multi-level regression). Multiple observations for each individual over time can be included as an additional level, so that individual-year observations are at the first level, individuals at the second level and couples at the third level. The use of multi-level APIM for such longitudinal data is more appropriate than fixed-effects regression, since 1) individuals are nested within couples, 2) random

³⁵ MMMLMs models cannot be estimated in Stata. I used the software MLwiN version 2.24 (Rasbash et al. 2009b) to estimate the models.

³⁶ MCMC is a simulation procedure that uses repeated random draws from distributions of unknown parameters to derive estimates for these parameters. Each repeated draw is informed by prior draws (Browne 2012: 3ff).

intercepts can be correlated, and 3) between-individual differences of women and men can be estimated.

Following from my theoretical model, I expect differences between women and men in the effects of relocations on leaving employment. To test this expectation in Chapter 8, I follow the strategy of Raudenbush, Brennan and Barnett (1995) and include separate intercepts for women and men in my model (cf. also Kenny, Kashy and Cook 2006: 176).³⁷ I also interact these intercepts with all dependent variables in the model. These interactions result in having two different coefficients for each variable — one for women and one for men. Gender-differences can be tested by performing Wald tests under the null hypothesis that the difference between the coefficients for women and men is 0. The dependent variable leaving employment is binary (cf. Section 4.5.1.4).

A logistic regression framework is used to model the probability that the dependent variable is 1 (Pr(y = 1)) conditioned on the independent variables (Pampel 2008: 3ff). The logistic multi-level APIM with two intercepts for individual *i* in couple *c* at time *t* can be written in the following form:

$$log\left(\frac{Pr(y_{cit+1}=1)}{1-Pr(y_{cit+1}=1)}\right) = \alpha_w + \alpha_m + (\alpha_w + \alpha_m)(\beta_1 z_{ct} + \beta_2 x_{cit} + \beta_3 p_{cit}) + v_{wc} + v_{mc} + \varepsilon_{cit},$$

$$(4.7)$$

where the first line on the right-hand side represents the fixed part of the equation and the second line represents the random part of the equation.³⁸ α_w and α_m are intercepts for women and men respectively. The intercepts represent the baseline probability for women and men that y = 1, if all variables in the model equal 0. z_{ct} is a vector of couple-level variables, e.g. marital status, x_{cit} is a vector of individuallevel variables, e.g. education, and p_{cit} is a vector of characteristics of the partner, e.g. education of partner. Because, the terms for all these variables are multiplied by the two gender-specific intercepts, vectors of coefficients β_1 , β_2 and β_3 are estimated separately for women and men. v_{wc} and v_{mc} are random variables that represent the couple-specific variation from the intercepts for women and men respectively. The two terms are allowed to correlate to allow for selection into couples that is due to unobserved characteristics of both partners. Finally, ε_{cit} is random error.

In contrast to linear regression coefficients, logit coefficients are affected by residual variation. If residual variation varies between subpopulations, estimates for these subpopulations cannot be compared as differences in the estimates are confounded by differences in the residual variation (Allison 1999). To compare effects

 $^{^{37}}$ The intercepts are binary variables coded 1 if the respondent is a women respectively men and 0 otherwise. I include these variables in the model and suppress the overall intercept.

³⁸ The maximum likelihood (ML) function has no closed form for logistic random-effects models. Therefore, the ML must be evaluated with approximate procedures. Stata's xtmelogit uses Gauss-Hermite quadrature approximation. The accuracy of the approximation depends on the number of quadrature points used (Rabe-Hesketh and Skrondal 2005: 124ff).

across subpopulations or between models, average marginal effects (AMEs) instead of the logit coefficients can be used (Mood 2010). AMEs indicate the population average changes in P(y=1) given a change in the respective variable while holding all other variables at their sample values (Cameron and Trivedi 2010: 343f).³⁹

4.6.4 Shared Frailty Discrete-Time Event History Models

In Chapter 5, I analyse the occurrence of relocations. Event history analysis (EHA) is an especially handy tool for this, since relocations should be understood in relation to other life course events and state dependencies, as relocations and the life course are closely related (Buck 2000b: 251). Furthermore, the method allows for time-dependent covariates and censored data (Heaton and Call 1995).⁴⁰ In EHA, data is used that records the transition from one state to another at certain points of time (Hougaard 2000: 2). The hazard of a transition in interval t is modelled and it is interpreted as the probability of transition occurrence, given that no transition has occurred before the start of interval t. In regard to relocations, the states are different residential locations, while events or transitions are relocations. As the BHPS and SOEP data is collected in annual interviews, i.e. the data is grouped annually, and for most of the variables only annual information is available, I use a discrete time model to analyse the data. The event history model is estimated using logistic regression. Furthermore, repeated events have to be taken into account, as persons may relocate more than once (Allison 1982; Singer and Willett 2003: 357ff; Box-Steffensmeier and Jones 2007: 69ff). I extend the basic discrete-time model by including individual-specific, time-constant random-effects to account for unobserved heterogeneity. These models are called shared frailty models (Box-Steffensmeier and Jones 2007: 146; Steele 2011).⁴¹ I use random-effects instead of fixed-effects models to model relocations, since I aim to describe between-individual differences between movers and stayers as well as intra-individual determinants for relocations.

The model can be specified as

 $^{^{39}}$ For binary variables, the change in the respective variable is from 0 to 1 and for continuous variables the AME is computed for an infinitesimal change in the respective variable.

⁴⁰ An observation is right-censored, if the observation ends before the completion of a spell. A spell is the time a respondent stays in the same state which is ended by an event. In the present analysis, the relocation history is analysed which is only completed with the death of a respondent. Thus, nearly all observed respondents are right-censored at the end of their observation periods. A high share of the observed spells are also left-censored, i.e. observations start after respondents' spells began. Guo (1993) recommends inclusion of these cases starting with their first observation and their event time as is.

⁴¹ Frailty refers to the assumption that some individuals are more likely than others to experience an event due to unobserved characteristics. In the present case the frailty is shared by all observations that an individual contributes to the sample.

4 Research Design and Methods

$$log\left(\frac{Pr(y=1)}{1-Pr(y=1)}\right) = \alpha + \tau d_{it} + \beta x_{it} + \gamma z_i + \nu_i + \varepsilon_{it}, \qquad (4.8)$$

where y_{it} is coded 1 if the event occurs between t and t + 1 and coded 0 if the event does not occur or the observation is censored. α is the constant baseline probability to experience the event. The variable d measures the duration that a respondent has spent in the current state and τ captures the duration dependency of an event, e.g. to account for the cumulative inertia described in Section 2.1.3. Here the duration is only included as a linear term, but it is straight-forward to extend the model by including duration dummies, polynomials or splines (Box-Steffensmeier and Jones 2007: 75f). x_{it} is a vector of time-varying covariates, z_i is a vector of time-constant covariates, v_i is the individual-specific frailty to experience the event — similar to random-effects described before — which is drawn from a normally distributed random variable and ε_{it} is random error.

Chapter 5 Setting the Scene: Occurrence and Characteristics of Relocations

I showed that England and Germany differ in their average residential mobility rates (cf. Section 3.1). To gain a better picture of these differences, I analyse the characteristics of relocations in both countries in this chapter in an explorative way. This is important for understanding who is relocating and to contextualise the outcomes of relocations for subsequent chapters. To analyse the occurrence of relocations, I proceed as follows. First, I present descriptive findings regarding the frequency and characteristics of residential relocations in Section 5.1. Next, I analyse the determinants of residential relocations using an event history analysis framework in Section 5.2. In Section 5.3, I examine potentially non-random panel attrition after relocations. Finally, I summarise my findings in Section 5.4. The observation period for the analysis in the present chapter is 1991 to 2008 (cf. Section 4.4 for a description of further sample rules).¹

While transitions in the family trajectory are important trigger events for residential relocations (cf. e.g. Booth and Amato 1993; Flowerdew and Al-Hamad 2004; Feijten and Ham 2007), I restrict the analysis to individuals in stable households, i.e. households that remain single or couple households between t and t + 1. I exclude major household changes from the analysis in the present chapter, because it can be expected that the determinants of relocations in these cases are fundamentally different from those for other relocations (Pickles and Davies 1991). In addition, relocations and household changes are endogenous processes, i.e. the causal link between the processes can be imagined to go in both directions, and it is hard to disentangle, which decision precedes the other: ending the partnership or moving out of the shared dwelling. Outcomes of relocations accompanied by major household changes are analysed in the subsequent chapters.

¹ Descriptive statistics for all variables are reported in Table A.2 for England and Table A.3 for Germany.

P. M. Lersch, *Residential Relocations and their Consequences*, Life Course Research, DOI 10.1007/978-3-658-04257-8 5, © Springer Fachmedien Wiesbaden 2014

5.1 Descriptive Results

Evidence for a higher residential mobility of the English population that has been found in past literature cannot be reproduced with the current analysis for stable households. For the years 1991 to 2008, the pooled and weighted average mobility rate for individuals is about 6 per cent for England and Germany. In interpreting these rates, it has to be kept in mind that I analyse only stable households. As shown in Section 4.1.3, household changes are more frequent in England than in Germany. Thus, the overall mobility in England is higher than in Germany when also considering these relocations associated with household changes. I now present descriptive findings about who relocates and the average characteristics of relocations in England and Germany. In Section 5.1.1, I analyse characteristics of relocations as well as the re-occurrence of relocations over the observation period. Section 5.1.2 shows differences in the characteristics of movers and stayers in both countries.

5.1.1 Characteristics of Residential Relocations

Residential relocations differ in their characteristics. Here, I describe relocations of stable households along three dimensions: 1) the distance of relocations; 2) the motivation for relocating; and 3) the re-occurrence of relocations. These characteristics are selected, because it can be expected that they relate to different outcomes after relocations. For example, long distance relocations may cause deteriorating relocations more often than short-distance relocations. Concerning distance, relocations in England are shorter than in Germany on average.² In England, the average distance is about 28 kilometres (SD = 69.34, N = 6,895).³ In England, about 48 per cent of all relocations are less than 3 kilometres. About 13 per cent are at least 50 kilometres. In Germany, the average distance of relocation is about 37 kilometres (SD = 86.42, N = 8,232). In Germany, about 52 per cent of all relocations are less than 3 kilometres. Thus, Germans relocate over a short distance more often than movers in England. However, the average distance of relocations is higher in Germany because of a small share of very long relocations. This is in accordance with earlier findings (Bonin et al. 2008: 33).

There are different motivations for relocating. In the literature, it is common to differentiate housing, family, job and area reasons (e.g. Clark and Huang 2003;

² For Germany, the relocation distance is only available from 2000 to 2008.

³ It has to be kept in mind that the potential maximum distance of relocations in England is smaller than in Germany. For example, the distance between Newcastle upon Tyne in the North-East to Portsmouth in the South of England is about 466 kilometres. From Flensburg in the North to Constance in the South of Germany it is 792 kilometres.

Bonin et al. 2008: 71). In the BHPS and the SOEP, respondents that relocated are asked for their subjective reasons for the relocation.⁴ In both countries, a high share of movers are motivated by housing related issues.⁵ In Germany, 70 per cent of all respondents stated that they relocated for housing reasons. In England, 58 per cent stated housing as one of the two main reasons for relocating. The single most stated reason for relocations is insufficient space in the old dwelling. This reason has been stated by 17 per cent of movers in England and 27 per cent in Germany. Across the two countries, singles and couples are, on average, motivated by different reasons to relocate. Singles are more likely to be motivated by family and job reasons, while individuals in couples are more likely to be motivated by housing and area reasons. These differences are aligned with my theoretical expectations. Stable couples will mainly relocate if their housing situations deteriorate because of insufficient housing space after childbirth or if they live in unattractive areas, while singles experience more trigger events in the job and family trajectory which motivate their relocations. The triggers of residential relocations are further scrutinised in the multivariate analysis in Section 5.2.

In the literature on residential relocations, it is very common to differentiate between job motivated relocations that are believed to be of longer distances and housing motivated relocations that are more likely to be shorter distances (e.g. Böheim and Taylor 2002). At first sight, this differentiation is supported by the present analysis. Households relocating for job reasons relocate longer distances than those relocating not for job reasons. In Germany, the average distance of relocation for those relocating for job reasons is 143 kilometres (SD = 156.67, N = 604), while the distance for those relocating not for job reasons is only 15 kilometres (SD = 50.88, N = 4,934). In England, the average distance of relocation for those relocating for job reasons is 79 kilometres (SD = 92.59, N = 627), while the distance for those relocating not for job reasons is only 16 kilometres (SD = 45.59, N = 3,548). However, only looking at those that relocated for at least 50 kilometres — which is a common cut-off value for long-distance relocations (e.g. Boyle et al. 2003) — and analysing their motivations for relocations reveals that a significant share of long distance relocations are at least partly motivated by housing and family reasons. For example, in Germany about 25 per cent of those that relocate more than 50 kilometres are motivated by housing reasons. Thus, treating all long distance relocations as motivated by job mobility is problematic as these relocations are also motivated by other reasons and I do not follow the differentiation of relocations by distance in the subsequent analyses of housing and neighbourhood quality. Instead, I consider all changes in the addresses of individuals as residential relocations.

⁴ In the BHPS and SOEP, housing reasons include, e.g.: dwelling too big or too small, dwelling too expansive, notice given, dwelling inadequate furnished. Family reasons include, e.g.: separation, marriage, child left and other family reasons. Area reasons include, e.g.: quality of location and surroundings.

⁵ In comparing the shares across countries, note that in the SOEP several reasons could be stated, while in the BHPS only the two most important reasons after job reasons are recorded.

Unfortunately, retrospective information about the residential history is very limited in the BHPS and SOEP and I have no information about the relocations before the start of the observation period. Thus, I most likely underestimate the number of relocations that individuals experienced in their life until the end of the observation period. For example, almost all individuals will have experienced at least one relocation out of the parental nest. Within the observation period, 41 per cent of individuals in England experienced at least one relocation. About 2 per cent relocated at least four times. In Germany, I observe more individuals that do not relocate within the observation period (67 per cent). Only about 1 per cent relocated at least 4 times. Thus, English individuals are slightly more likely to relocate repeatedly within the observation period, while Germans are more likely to not relocate at all. Other retrospective data shows that 18 per cent of the German population and 30 per cent of the English population have relocated more than four times over their whole life course since childhood until the interview (Bonin et al. 2008: 36).

5.1.2 Characteristics of Movers

Table 5.1 presents group averages for individuals in England and Germany by their relocation status. The characteristics are measured at the interview before the relocation took place. There are no significant differences between stayers and movers regarding their gender. Movers are more likely to live in Southeast England than in the rest of England, but there are no significant regional differences between stayers and movers in Germany. English movers are more likely to live in metropolitan and less likely to live in rural areas than stayers. In Germany, movers are more likely than stayers to live in urban but less likely to live in metropolitan areas on average.

In accordance with the cumulative inertia axiom, individuals that live for a shorter time in their residences are more likely to relocate on average. In England and Germany, individuals that relocate lived for about 7 years in their residence on average, while it is at least twice as long for stayers. In both countries, movers are more likely to be young singles or single parents than stayers. Movers are less likely to be older singles and childless couples than stayers. Movers are non-significantly less likely to be couples with children than stayers in England, but they are more likely to be couples with children in Germany. Seven per cent of movers in England experience a childbirth, while only 3 per cent of stayers do. In Germany 6 per cent of movers and 2 per cent of stayers experience a childbirth. Movers are as likely as stayers to experience a child moving out in both countries. Individuals that relocate are more likely to live with young children than those staying, but movers are less likely to live with older children. Movers live more often in households in which at least one immigrant lives than stayers in both countries.

About 61 per cent of stayers and 66 per cent of movers are employed in England. In Germany, only 56 per cent of stayers are employed, while 66 per cent of movers are employed. Movers are more likely to be unemployed in both countries, but less likely to be inactive. Movers and stayers are equally likely to be in dual-earner couples in both countries. The average household income for stayers is about PPP-\$ 1,821 in England and does not differ significantly between stayers and movers. In Germany, stayers have a significantly higher income (PPP-\$ 1,761) than movers (PPP-\$ 1,608).

The distribution of housing tenure is very different in England and Germany. This is reflected in the strong differences in the shares of renters and owners. Owner occupancy is much more prevalent in England than in Germany. In Germany, the majority lives in dwellings rented privately or in social housing. In both countries, movers live as often in social housing as stayers on average. Living in privately rented accommodation is much more common for movers than for stayers in both countries. In Germany, 85 per cent of all movers rent privately, while only 47 per cent of stayers rent privately. In England, 27 per cent of movers and only 5 per cent of stayers rent privately. Stayers in both countries live more often in their own homes than movers whether or not they own with a mortgage or outright. Movers in both countries live in significantly higher room stress than stayers. The average neighbourhood quality is significantly lower for movers in both countries. The difference is especially strong in Germany, where movers live in neighbourhoods whose quality levels tend to be 0.31 standard deviations lower than for stayers. I now test, if the differences identified in the descriptive analyses also hold in a multivariate framework

5.2 Multivariate Results

I analyse the event of residential relocation using shared frailty discrete-time event history analysis (EHA; cf. Section 4.6.4). The dependent variable measures whether an individual relocated (coded 1) or not (coded 0) between t, i.e. the current wave, and t + 1, i.e. the next wave, for each pair of subsequently observed interview waves. All independent variables are measured in the wave before a potential relocation and events are measured synchronously with relocations between t and t + 1. Table 5.2 reports the estimation results for individuals in England and Table 5.3 for individuals in Germany. The unstandardised coefficients b and the average marginal effects (AME) are listed (cf. Section 4.6.3). In the estimation table, ρ indicates the proportion of the total variance that is contributed by the individual-specific, time-constant random term. A higher ρ indicates that a higher proportion of the overall variance in the model can be explained by the random term. The random term is identified by repeated relocations.

For English women and men, the chances of relocations to occur decrease significantly with the time they spent in their residences in accordance with the cumulative inertia axiom and past literature (cf. Section 2.1.3). Age also has a strong

Variables	England			G	Germany	
	Stayer		Mover	Stayer		Mover
	I	Avera	ge	1	Avera	ge
Women	0.54		0.53	0.53		0.52
Southeast England/East Germany	0.35	**	0.37	0.19		0.20
Urbanisation						
Metropolitan	0.41	***	0.44	0.21	***	0.18
Urban	0.34		0.33	0.48	*	0.49
Rural	0.25	*	0.23	0.32		0.33
Years in residence	13.11	***	6.64	15.66	***	6.99
Life stage						
Single (\leq 35 years old)	0.02	***	0.09	0.04	***	0.12
Single (> 36 years old)	0.13	***	0.09	0.17	***	0.13
Single parent	0.05	*	0.06	0.04	***	0.06
Couple parent	0.42		0.40	0.38	**	0.40
Childless couple	0.38	**	0.35	0.37	***	0.29
Family events						
Child born	0.03	***	0.07	0.02	***	0.06
Child moves out	0.04		0.04	0.02		0.02
At least one child						
aged 0 to 9	0.21	***	0.30	0.18	***	0.30
aged 10 to 17	0.20	***	0.15	0.18	***	0.16
Immigrant	0.18	***	0.21	0.17	***	0.24
Employment status						
Employed	0.61	***	0.66	0.56	***	0.66
Unemployed	0.03	***	0.04	0.05	***	0.08
Inactive	0.36	***	0.30	0.39	***	0.26
Dual-earner couple	0.39		0.40	0.29		0.29
Household income	1,821.22		1,799.37	1,761.00	***	1,608.43
Housing costs	331.37	***	421.86	375.56	***	437.10
Tenure						
Social housing	0.16		0.17	0.05		0.06
Rented privately	0.05	***	0.27	0.47	***	0.85
Owned with mortgage	0.50	***	0.43	0.25	***	0.05
Owned outright	0.29	***	0.13	0.23	***	0.04
Room stress	-2.08	***	-1.57	-1.29	***	-0.52
Neighbourhood quality ¹	0.02	**	-0.05	0.04	***	-0.27

Table	5.1	Group	averages,	movers	vs.	stayers	
-------	-----	-------	-----------	--------	-----	---------	--

Data: BHPS waves 1-18, SOEP v26 waves 8-25 (cross-sectionally weighted)

Note: Within country differences: *** significant at 0.1% two-tailed, ** significant at 1%,* significant at 5%. ¹: Only 2000-2008.

negative effect on relocations for both genders in England in accordance with previous findings (e.g. Böheim and Taylor 2002). The effect of an additional year of age becomes slightly weaker with growing age according to the coefficient of age squared, but the hazard rate to relocate decreases for the observed age range of 18 to 79 years. Cohabiting men are slightly more likely to relocate than single or married men. For the interpretation of the effect of partnership status on relocation behaviour, it is important to keep in mind that only stable households are considered in the present analysis. The presence of children aged 1 to 9 and 10 to 17 have negative effects on the hazards for relocations for English women which is aligned with research showing the increasing costs of relocations for larger households (Kearns and Parkes 2003). English men are less likely to relocate, if they live together with older children, but not if they live with young children. At the same time, childbearing has a positive effect on the chances to relocate (cf. also Kulu 2008). English women and men that experience a childbirth are 2 percentage points more likely to relocate in the same year than those not experiencing a childbirth. The same applies for those experiencing a child moving out.

English women and men are less mobile if they are employed compared to those inactive. Job mobility has almost no effect on geographical mobility. Only women that become unemployed have a hazard rate that is 2 percentage points higher than for those not becoming unemployed, but the status of being unemployed has no significant effect on the hazard rate of relocations compared to inactivity for women. Together with the significant effect of becoming unemployed, this may indicate that the hazards to relocate decrease with the duration of unemployment (Windzio 2004). English men are less likely to relocate if they are unemployed compared to being inactive. Unexpectedly, women and men in dual-earner couples are not less likely to relocate compared to single-earner couples and singles. Income has a positive effect on the chance to relocate, but the effect is very small and not significant in accordance with past findings (Kearns and Parkes 2003; Clark and Huang 2003).

As expected from past research, housing tenure has a strong effect on the hazards to relocate in England (Kearns and Parkes 2003). Compared to owning with mortgage, individuals in both rental tenures have a higher hazard to relocate. For renters in social housing, the hazard rate is 1-2 percentage points higher on average. Private renters have a 9 percentage points higher hazard rate to relocate compared to owners with mortgages. The effect of living in privately rented accommodation on the chances to relocate is the highest for women and men in England compared to all other effects in the model. Room stress also has a significant positive effect on the hazard of relocation for both genders. The fewer rooms available relative to the number of residents, the more likely individuals relocate in England (cf. also Böheim and Taylor 2002). Other variables in the model do not significantly affect the hazard for relocation in England.

In Germany, individuals are less likely to relocate the longer they live in their present dwelling and the older they are (cf. Table 5.3). For German women, I find a significant coefficient for age squared. This again indicates that the effect of an

	Model 5.1 England			
	Women		Mei	n
	b	AME	b	AME
Time in residence/100	-2.937**	-0.155**	-1.895*	-0.102*
	(-2.92)	(-2.84)	(-2.09)	(-2.08)
Time in residence/100 sq.	2.164	0.114	-0.915	-0.049
-	(0.86)	(0.85)	(-0.43)	(-0.43)
Urbanisation (ref: Rural)	~ /	× /	· /	· /
Metropolitan	0.038	0.002	0.024	0.001
1	(0.61)	(0.62)	(0.36)	(0.36)
Urban	-0.057	-0.003	-0.004	-0.000
	(-1.05)	(-1.04)	(-0.06)	(-0.06)
Age/100	-5.135***	-0.271***	-6.023***	-0.325***
	(-3.41)	(-3.49)	(-4.04)	(-4.05)
Age/100 sa	1 706	0.090	2.896	0.156
1190,100 54.	(1.09)	(1, 10)	(1.73)	(1.73)
Partnership status (ref: Single)	(110))	(1110)	(1172)	(11/2)
Cohabiting	0.061	0.003	0 199*	0.011*
Condoning	(0.63)	(0.63)	(2.07)	(2.08)
Married	0.019	0.001	0.118	0.006
Warned	(0.26)	(0.26)	(1.38)	(1.38)
Child aged (ref: No child)	(0.20)	(0.20)	(1.50)	(1.50)
1 to 9	_0.133*	-0.007*	0.004	0.000
1.00 /	(-2.40)	(-2.37)	(0.07)	(0.07)
10 to 17	_0.491***	-0.026***	-0.513***	-0.028***
10 10 17	(-7.30)	(-7.13)	(-5.56)	(-5.67)
Family events (ref: No event)	(-7.39)	(-7.15)	(-5.50)	(-3.07)
Child horn	0.405***	0.021***	0.440***	0.024***
Child bolli	(4.46)	(4, 40)	(4.80)	(4.84)
Childlaft	(4.40)	(4.49)	(4.60)	(4.64)
Clinia lett	(2, 28)	(2, 20)	(2.01)	(2.01)
Ich mahilita (naf: No mahilita)	(3.28)	(3.20)	(2.91)	(2.91)
Dramation	0.117	0.006	0.000	0.000
Promotion	-0.11/	-0.006	(0.000)	(0.000)
Demetion	(-1.41)	(-1.41)	(0.00)	(0.00)
Demotion	-0.019	-0.001	-0.123	-0.007
	(-0.23)	(-0.23)	(-1.60)	(-1.59)
Became unemployed	0.434***	0.023***	0.177	0.010
	(3.63)	(3.65)	(1.43)	(1.42)
Job status (ref: Inactive)	0.400*	0.010*		0.04 = ***
Employed	-0.183*	-0.010*	-0.324***	-0.017***
	(-2.08)	(-2.08)	(-3.31)	(-3.32)
Unemployed	-0.186	-0.010	-0.307^{*}	-0.017^{*}
	(-1.24)	(-1.24)	(-2.22)	(-2.23)
	continued or	verleaf		

Table 5.2 Discrete-time event history models of relocation, England

	continue	ed		
	Model 5.1 England			
	Wome	en	Mei	ı
	b	AME	b	AME
Dual-earner couple	0.034	0.002	-0.017	-0.001
	(0.38)	(0.38)	(-0.26)	(-0.26)
Income (log)	0.008	0.000	0.053	0.003
	(0.22)	(0.22)	(1.27)	(1.27)
Housing tenure (ref: Owned with n	nortgage)		. ,	
Social housing	0.287***	0.015***	0.268***	0.014***
C C	(3.61)	(3.59)	(3.36)	(3.36)
Rented privately	1.620***	0.086***	1.741***	0.094***
· ·	(23.79)	(24.80)	(24.54)	(24.09)
Owned outright	0.158	0.008	0.141	0.008
-	(1.65)	(1.65)	(1.41)	(1.40)
Room stress	0.102***	0.005***	0.111***	0.006***
	(5.74)	(5.84)	(6.38)	(6.53)
Constant	-0.787^{*}	· · · ·	-0.935**	~ /
	(-2.01)		(-2.94)	
ρ	0.02*		0.02	2
Individual-year observations	38,906		33,199	
Individuals	4,498		3,94	3
Relocations	2,48	5	2,177	
LL Null-model	-9,081	.84	-7,921	.05
LL	-8,036.62 -6,958.04			.04

Data: BHPS 1-18 (individual level, unweighed)

Note: Logistic random-effects model, dependent binary variable: relocation (coded 1 if relocates until t + 1, coded 0 otherwise); unstandardised coefficients, z statistics in parentheses based on bootstrapped standard errors (50 replications); *** significant at 0.1% two-tailed, ** significant at 1%,* significant at 5%; models also include covariates for period and region not shown here.

additional year of age decreases with higher age. The effect of time in residence is weaker in Germany compared to England, which may indicate that cumulative inertia has a stronger effect in England. German men are less likely to relocate if they live in metropolitan areas compared with rural areas. German women and men are less likely to relocate if they live in urban areas compared to rural areas, but the effect is very small with less than 1 percentage point difference in the hazards to relocate between rural and urban areas. Cohabiting and married men are more likely to relocate than single men in Germany. Again, it is important to note that only stable households are considered. Married women are less likely to relocate than single women. The presence of young children and childbearing clearly increases

	Model 5.2 Germany			
	Women		Me	n
	b	AME	b	AME
Time in residence/100	-1.721***	-0.089***	-1.589**	-0.082^{**}
	(-3.58)	(-3.51)	(-2.75)	(-2.73)
Time in residence/100 sq.	0.646	0.033	0.454	0.023
	(0.58)	(0.58)	(0.35)	(0.35)
Urbanisation (ref: Rural)		· · · ·	· · /	× /
Metropolitan	-0.058	-0.003	-0.077^{*}	-0.004^{*}
1	(-1.93)	(-1.94)	(-2.08)	(-2.08)
Urban	-0.089**	-0.005**	-0.098**	-0.005**
	(-2.86)	(-2.88)	(-2.88)	(-2.89)
Age/100	-5.392***	-0.278***	-4.894***	-0.253***
8	(-7.49)	(-7.55)	(-5.24)	(-5.22)
Age/100 sq.	2.273**	0.117**	1.638	0.085
1.80,100 04.	(2.78)	(2.78)	(1.60)	(1.59)
Partnership status (ref: Single)	()	()	()	()
Cohabiting	-0.018	-0.001	0.230***	0.012***
connorming	(-0.38)	(-0.38)	(3.76)	(3.77)
Married	-0.118**	-0.006**	0.139*	0.007*
munica	(-2.66)	(-2.65)	(2, 25)	(2.26)
Child aged (ref: No child)	(2.00)	(2.05)	(2.23)	(2.20)
1 to 9	0.156***	0.008***	0.176***	0 009***
1 10 7	(4.23)	(4.20)	(5.14)	(5.16)
10 to 17	_0.198***	_0.010***	_0.234***	_0.012***
10 10 17	(-6.25)	(-6.31)	(-5.48)	(-5.45)
Family events (ref: No event)	(-0.23)	(-0.51)	(-5.48)	(-5.45)
Child born	0.508***	0.026***	0.463***	0.024***
Child bolh	(8.67)	(8.62)	(7, 75)	(7.65)
Child loft	0.550***	(0.02)	0.350***	0.018***
Child left	(6.52)	(6.40)	(4.15)	(4.07)
Joh mahility (rafe No mahility)	(0.32)	(0.49)	(4.13)	(4.07)
Dramation	0.125**	0.006**	0.002	0.000
FIOIIIOUOII	(2.71)	(2, 72)	(0.003)	(0.08)
Domotion	(2.71)	(2.72)	(0.08)	(0.08)
Demotion	(0.42)	(0, 42)	-0.041	-0.002
Decome un energlosse d	(0.43)	(0.43)	(-0.88)	(-0.88)
Became unemployed	(7.41)	(7.25)	(4.08)	0.013
Tale states (self: Leasting)	(7.41)	(7.35)	(4.08)	(4.12)
Job status (ref: Inactive)	0 100***	0.010***	0.022	0.000
Employed	-0.189	-0.010^{+++}	-0.032	-0.002
TT 1 7	(-5.66)	(-5.70)	(-0.53)	(-0.53)
Unemployed	-0.002	-0.000	0.079	0.004
	(-0.05)	(-0.05)	(1.30)	(1.30)
	continued or	verleaf		

 Table 5.3 Discrete-time event history models of relocation, Germany

	continue	ed			
	Model 5.2 Germany				
	Wome	en	Mer	1	
	b	AME	b	AME	
Dual-earner couple	0.079	0.004	0.009	0.000	
	(1.67)	(1.67)	(0.27)	(0.27)	
Income (log)	0.051**	0.003**	0.063**	0.003**	
	(2.70)	(2.69)	(2.75)	(2.78)	
Housing tenure (ref: Owned with	mortgage)				
Social housing	1.501***	0.077***	1.553***	0.080***	
C C	(20.13)	(19.49)	(21.80)	(20.29)	
Rented privately	1.895***	0.098***	1.903***	0.098***	
- · ·	(33.39)	(32.16)	(34.47)	(30.42)	
Owned outright	0.373***	0.019***	0.352***	0.018***	
-	(4.76)	(4.75)	(3.44)	(3.41)	
Room stress	0.164***	0.008***	0.171***	0.009***	
	(12.63)	(13.09)	(11.73)	(12.25)	
Constant	-2.261***	. ,	-2.617***	· · · ·	
	(-12.62)		(-13.45)		
ρ	0.03***		0.02*	**	
Individual-year observations	116,464 102.848			48	
Individuals	14,43	31	13,04	4	
Relocations	7,124	4	6,21	6	
LL Null-model	-26,337	7.75	-23,157	7.46	
LL	-22,909.57 -20,041.26			1.26	

Data: SOEP v26 waves 8-25 (individual level, unweighed)

Note: Logistic random-effects model, dependent binary variable: relocation (coded 1 if relocates until t + 1, coded 0 otherwise); unstandardised coefficients, z statistics in parentheses based on bootstrapped standard errors (50 replications); *** significant at 0.1% two-tailed, ** significant at 1%,* significant at 5%; models also include covariates for period and region not shown here.

the hazard of relocation. German women and men that experience a childbirth are 2-3 percentage points more likely to relocate than those not experiencing birth. The presence of older children reduces the hazards to relocate, but the effect is only half as strong as in England. If children move out of the household, German women are 3 percentage points more likely to relocate themselves, while men are only 2 percentage point more likely.

Promotions positively affect the hazards to relocate for women in Germany, but the difference in the hazard to relocate between women with promotion and those without is less than 1 percentage point. Women and men are more likely to relocate if they became unemployed. Women have a 2 percentage point higher hazard rate to relocate on average if they become unemployed, while men have a 1 percentage point higher hazard rate. Being employed reduces the chances to relocate significantly for women but not for men compared to the inactive. Being unemployed has no significant effect on the hazard rate. Dual-earner couples do not significantly differ in their hazards to relocate from other couples and singles. Women and men with higher incomes are more likely to relocate, but the effect is very small.

Housing tenure has a strong effect on the hazard rate to relocate in Germany as well. Compared to owning with mortgages, social renters' hazard to relocate is 8 percentage points higher on average. The difference between owners and social tenants in their hazards to relocate is much stronger in Germany than in England. Those renting privately, are about 10 percentage points more likely to relocate than those owning with mortgage. Owners that do not pay a mortgage are 2 percentage points more likely to relocate than owners with a mortgage. The strong difference between private tenants and owners can be observed in both countries, but in contrast to England, the average mobility of home owners is much lower in Germany (cf. Table 5.1). Finally, women and men in higher room stress are significantly more likely to relocate than those in lower room stress in Germany.

The hazard rate to relocate may be affected by unobserved, time-constant characteristics, e.g. openness to new experiences and past relocation history. Indeed, I find ρ to be significantly different from 0 at least at the 95 per cent confidence level for all but English men. However, ρ is not larger than 0.03 in any of the models.⁶ Thus, controlling for time-constant individual unobserved heterogeneity is important in modelling relocation behaviour. The estimation of the random error term should be treated with caution, though, because the estimation may be inconsistent as only few individuals in my sample experience repeated relocations ⁷

The models also include year dummies, for which the coefficients are not reported here. The models do not control for past relocations, because the retrospective residential history before the first interview is not surveyed in the BHPS and SOEP.⁸ The random terms may capture some of the inter-individual differences in unobserved residential histories. Neighbourhood quality is not included as an independent variable in the model, because the information is only available after 1999. I also estimate an alternative specification including neighbourhood quality

⁶ This is substantially smaller than ρ reported in Rabe and Taylor (2010) for a relocation model in the UK, but this may be due to the fact that I control for more individual characteristics in my model that may be absorbed in the random term than in the model reported in Rabe and Taylor (ibid.).

⁷ I estimate a pooled cross-sectional logistic regression with clustered standard errors and without random error terms as an alternative specification to check the robustness of my estimation results. As the alternative estimation results without random error terms do not differ in their interpretation, I do not elaborate on the alternative model here.

⁸ The estimation of a variable measuring observed number of relocations may be obscured, because for each individual the number of observed relocations may diverge more or less dramatically from the actual number of past relocations (Taylor 2007).

for the years after 1999. The models show that the chances of residential relocations decrease with higher neighbourhood quality in Germany, but in England neighbourhood quality has no significant effect on the chances to relocate.⁹ Rabe and Taylor (2010) only find a significant effect of neighbourhood quality on chances to relocate for couples in England.

Leaving major household changes aside, the results of the multivariate analysis support the theoretical framework developed in Section 2.2. Relocations occur, because the preferences of individuals no longer match their residential locations. The results regarding the presence of children and family events support this assumption. Transitions in the job trajectory may also affect preferences regarding residential locations and I find job status and job mobility to have some effect on relocations. I find support for the assumption that relocations depend on resources of individuals only for Germany. The more economic resources Germans have, the more likely they are to relocate according to my model. The residential biography is also found to be important. Individuals become less mobile the longer they live at a location. This supports the assumption that the costs of relocations increase with the time spent at a certain location. The costs of relocations also increase if individuals live in ownership compared to rental tenure as shown in the model. I find some support for the assumption that relocations depend on the contextual opportunities and constraints in Germany, where individuals in urban areas are less likely to relocate than individuals in rural areas.

5.3 Selective Sample Attrition Due to Residential Relocations

When analysing panel data, selective attrition poses a problem. Selective attrition occurs, if respondents that leave the panel are systematically different from those respondents that stay in the panel (Frees 2004: 11). Individuals that relocate and drop out of the panel in the next wave may form a selective group of all movers. If this is the case, the results for my analysis in subsequent chapters would be biased as I only consider outcomes of relocations for those households that have been interviewed at wave *t* and t + 1. Both datasets, the BHPS and the SOEP, record if a respondent has not been found at the same address from the last interview even if the respondent did not take part in an interview in that year. I examine whether those taking part in an interview significantly differ from those that did not take part in an interview after a residential relocation. I analyse differences between the two groups regarding age, income, education, immigrant status, household type and tenure at time *t*, i.e. the interview before the potential drop out. The individual response behaviour is analysed. It should be kept in mind that for the BHPS only OSMs and

⁹ English women: b = 0.057, z=1.23 ; English men: b = 0.045, z=1.13; German women: b = -0.092, z=-3.82; German men: b = -0.084, z=-3.36.

PSMs are traced to the new address (see Section 4.3.1.1). Only these sample members are included in the present analysis. The response behaviour is examined by running a pooled binary logistic regression model with clustered standard errors where the dependent variable takes the value 0, if the respondent relocated between t and t + 1 and took part in an interview in t + 1. The dependent variable takes the value 1, if the respondent relocated between t and t + 1 and did not take part in an interview in t + 1. The estimation results are shown in Table 5.4. Cases that are not clearly identified as movers or stayers are not considered in the analysis.

Across the years 1991-2008, about 9 per cent of respondents in the BHPS selected under the sample rules for the present chapter that are identified as movers are not interviewed for at least one wave. In the SOEP, about 11 per cent of respondents in the current sample that are identified as movers are not interviewed in the next wave. The corresponding numbers for non-movers are much lower with 4 per cent in the BHPS and 6 per cent in the SOEP. Respondents with unclear mover status are not reflected in these numbers. Non-response is only in small part due to untraced addresses of movers. Only 2 per cent of individuals which are identified as movers could not be traced to their new address in England and 4 per cent in Germany.

According to the models in Table 5.4, in England, women are more likely to drop out, if they have children aged 10 to 17, live in an immigrant household or are in a dual-earner couple. They are less likely to drop out if they are married or are employed. For men in England, the probability to relocate and leave the panel is higher if respondents live in an immigrant household and the probability is lower with increasing household income. None of the modelled variables increases chances for German women to drop out. Having children aged 0 to 9 and a higher room stress significantly reduces the chances to drop out for this group. For German men, chances for drop out are increased if they cohabit, live with children aged 10 to 17, or live in an immigrant household. Chances for drop out are reduced for private tenants and with higher room stress. No other variable in England or Germany has a significant effect on the probability to leave the panel. I also test whether neighbourhood quality is associated with a higher probability for panel attrition for the years 2000-2008. I do not find any significant effects of neighbourhood quality on dropping out of the panel.

Thus, I find some evidence for selective attrition after relocations for particular groups. First, respondents in immigrant households seem to be more likely to leave the panel at least temporarily after a residential relocation than other respondents. This may bias my results upwardly, because immigrant households can be expected to live in lower quality housing and neighbourhoods and may not improve their locations as much as other respondents after relocations. Second, for Germany, I find a selection of respondents with low room stress into dropping out of the panel. It is not clear how this selection may affect my results concerning room stress. For example, these respondents are more likely to increase their room stress after relocation. For these cases, my estimation of changes in room stress would be biased upwardly.

Other selection processes seem to apply only to particular samples, but may nonetheless affect results for these samples. For example, the higher attrition for women in dual-earner couples in England may affect my results on labour market outcomes of dual-earner couples presented in Chapter 8. Thus, to be able to infer results regarding relocation outcomes to the whole population of movers, it is important to test for the sensitivity of my results regarding selectivity of respondents at wave t + 1 after a relocation. I do so for the analyses in Chapter 6, 7 and 8.

	Model 5.3 England		Mode Germ	l 5.4 any
	Women	Men	Women	Men
Age/100	-4.246	7.006	2.227	0.262
	(-0.73)	(1.24)	(1.18)	(0.12)
Age/100 sq.	4.800	-10.084	-2.153	0.668
	(0.72)	(-1.47)	(-1.00)	(0.28)
Partnership status (ref: Single)				
Cohabiting	-0.701	-0.084	0.120	0.310*
	(-1.86)	(-0.22)	(1.04)	(2.53)
Married	-0.821^{*}	-0.645	0.010	-0.072
	(-2.51)	(-1.71)	(0.10)	(-0.62)
Child aged (ref: No child)	· /	· · · ·	· · ·	× /
0 to 9	-0.269	0.247	-0.299^{**}	-0.020
	(-1.11)	(0.97)	(-3.10)	(-0.22)
10 to 17	0.821**	0.259	0.023	0.336***
	(3.09)	(0.97)	(0.24)	(3.38)
Education (ref: intermediate)		· · /	· · ·	× /
Basic	0.191	0.284	-0.006	-0.020
	(0.68)	(1.22)	(-0.08)	(-0.24)
Higher	-0.353	-0.234	-0.165	-0.142
	(-1.30)	(-0.91)	(-1.54)	(-1.37)
Job status (ref: Inactive)		. ,	· · ·	. ,
Employed	-0.933^{*}	0.472	0.041	-0.074
	(-2.46)	(1.04)	(0.35)	(-0.56)
Unemployed	-0.552	0.676	0.169	-0.021
	(-0.76)	(1.37)	(1.22)	(-0.13)
Household income (log)	-0.017	-0.197^{*}	0.035	-0.014
	(-0.18)	(-1.97)	(0.78)	(-0.33)
Immigrant	1.422***	0.518*	0.110	0.287***
	(6.96)	(2.33)	(1.32)	(3.54)
	continued or	verleaf		

Table 5.4 Logistic regression models of drop-outs after relocations

continued				
	Model 5.3 England		Model Germ	l 5.4 any
	Women	Men	Women	Men
Housing tenure (ref: Owned with m	ortgage)			
Social housing	-0.043 (-0.14)	0.235 (0.85)	-0.138 (-0.71)	-0.356 (-1.86)
Rented privately	(-0.220)	-0.586	-0.178 (-1.32)	(-2.59)
Owned outright	-0.128	0.253	(0.014)	-0.368
Dual-earner couple	(-0.29) 1.355*** (3.41)	(0.30) 0.083 (0.29)	-0.206	(-1.09) -0.181 (-1.91)
Room stress	0.048	(0.23) 0.088 (1.18)	(-0.113^{***}) (-3.85)	(-0.101^{***}) (-3.37)
Constant	(0.03) -2.075^{*} (-2.06)	(-2.805^{*}) (-2.49)	(-5.75) (-5.75)	(-3.79) (-3.79)
Individual-year observations Individuals LL Null-model LL Pseudo R ²	2,435 1,538 -447.35 -409.32 0.09	2,348 1,388 -489.99 -462.73 0.06	9,284 5,716 -3,019.39 -2,989.73 0.01	8,161 5,213 -2,882.45 -2,847.27 0.01

Data: BHPS waves 1-18, SOEP v26 waves 8-25 (unweighted)

Note: Binary logit models, dependent variable: relocated and still in panel (0), relocated and dropped out (1); unstandardised coefficients, clustered standard errors, t statistics in parentheses. *** significant at 0.1% two-tailed, ** significant at 1%,* significant at 5%.

5.4 Summary

In this chapter, I analyse the occurrence and characteristics of residential relocations in England and Germany. I focus on stable households, because major household changes such as union dissolutions and relocations are endogenous events. Average relocation rates in England and Germany for stable households of singles or couples are very similar. The theoretical expectations outlined in Section 2.2 are supported by the empirical analysis of relocation behaviour in the present chapter. The determinants of residential relocations of individuals in stable households differ only slightly between England and Germany. Detailed analyses in the next chapters show, whether relocations also have similar outcomes for individuals in both countries. In these analyses, I have to control for potential selectivity of respondents that stay in the panel after a relocation, because I show that drop-outs after residential relocation are a selective group in some characteristics.

In England and Germany, most relocations are motivated by housing reasons and limited space is the single most relevant motivation for relocations. Thus, one can expect relocations to have a strong effect on room stress in both countries. On average couples are more often motivated by housing and neighbourhood reasons than singles. The latter are more often motivated by family and job reasons than couples. This may have the effect that couples are more likely to improve their locations with regard to room stress and neighbourhood quality than singles. Individuals that relocate because of job reasons relocate farther away than movers motivated by other reasons on average, but a high share of long distance movers are also motivated by family and housing reasons. The high share of movers that report occupational motives for their decision to relocate underline the link between residential and job mobility that is further analysed in Chapter 8. A major difference between both countries is the higher average mobility of owners in England compared to Germany which will be further illuminated in Chapter 9.

Chapter 6 Room Stress and Residential Relocations

In this chapter, I analyse the effect of residential relocations on individuals' room stress and test the theoretical framework laid out in Section 2.3 with regard to room stress. The space a dwelling offers is one of its most fundamental characteristics. The number of separate rooms is relevant for a dwelling's function to offer privacy and the possibility for retreat rather than the sheer floor space. The number of available rooms in relation to the room need of the household is referred to as room stress in this Chapter (cf. Section 4.5.1.2 for the construction of the variable). High room stress reduces psychological and physical wellbeing — especially for children (cf. Section 3.2.1 for more details).

I begin with a descriptive analysis of room stress in Section 6.1. I examine crosssectional differences in room stress, before turning to a longitudinal analysis of changes at the national and individual level. Then, I analyse room stress in a multivariate framework (Section 6.2). In Section 6.2.1, I test for individual life course conditions and I analyse how singles and couples differ in their changes in room stress and in what way children affect these changes. I also scrutinise the effect of certain life course events in the family and job trajectory on changes in room stress. Tests of individual life course conditions is a special focus in the present chapter, since room stress can be expected to change after family events or relocations. In Section 6.2.2, I test for the stratification of room stress changes due to income and housing tenure. Section 6.2.3 is devoted to the analysis of involuntary relocations and the effect of relocation distance on changes in room stress. Next, I examine differences in changes of room stress between migrants and natives (cf. Section 6.2.4). I analyse structural conditions in regional housing markets and their effect on changes in room stress in Section 6.2.5. In Section 6.3, I summarise my findings.

6.1 Descriptive Results

According to offical statistics for 2001, every household in England had 5.3 rooms on average and every person had access to about 2.3 rooms (Office for National Statistics 2012b). In Germany, the average number of rooms per dwelling was 4.4 and each person had about 2.1 rooms (Statistisches Bundesamt 2004).¹ Thus, dwellings in England offer more rooms and individuals have more rooms per capita than in Germany on average. However, this gross average is of little help in understanding the room stress situation of individuals in both countries. It can be expected that certain subgroups of the population will deviate strongly from the national average and I provide a more detailed picture of the room stress situation in both countries now.

6.1.1 Average Room Stress by Groups

In Table 6.1, the average room stress levels for households with certain characteristics are reported for England and Germany for the year 2008.² The figures underline that the average room stress is higher in Germany than in England, but in both countries households live in dwellings that are more spacious than their objective room need suggests on average.³ In 2008, an average English household had about 2.1 excess rooms, i.e. 2.1 more rooms than its objective room need, while in Germany it was about 1.3 excess rooms. Average room stress does not differ between Southeast England and the rest of England. In Germany, room stress is much higher in East Germany than in West Germany. This reflects the initially low housing standard in East Germany after reunification (cf. also Frick and Grimm 2009).

Tenants experience a higher room stress than owners on average in both countries and social tenants live in especially crowded dwellings. Social tenants have only slightly more rooms than they need, with 0.7 rooms extra on average for England and 0.2 in Germany. In both countries, outright owners have the lowest room stress with 2.8 and 2.4 excess rooms. The relative differences between tenures within both countries are stronger in Germany than in England. In Germany, outright owners have on average more than twelve times as many excess rooms than social tenants. In England, outright owners have four times as many excess rooms.

¹ In the official statistics in both countries, kitchens are counted as separate rooms. In the BHPS and SOEP, though, kitchens are not considered as separate rooms.

 $^{^2}$ For the first sections of the descriptive results, I consider data at the household level. From Section 6.1.2 onwards, I turn to analyses of individual level data, because I also account for changes in the household composition from a longitudinal perspective.

³ Part of the differences between England and Germany are due to the particularities of measurement, which are described in Section 4.5.1.2.

I find strong differences in room stress by life course stages. Young singles below the age of 36 years and single parents experience the highest room stress in both countries. In England, both groups have about one excess room on average. In Germany, young singles have about 0.3 rooms more than their objective room need. Single parents have about 0.4 rooms more than their objective room need. Older singles above the age of 35 years have an average room stress of -1.8 rooms in England and -0.9 rooms in Germany. Couples with children live in slightly higher room stress than couples without children. The latter group experiences the lowest room stress in both countries. These findings show that couples are in lower room stress than singles on average. Families with children experience relatively high room stress, even though children are also especially vulnerable to the effects of room stress.

I expect that access to high quality housing is closely associated with household income (cf. Section 2.3.2). Indeed, I observe strong differences in the average room stress by income groups. In England, the lowest income quintile lives in dwellings that offer about 1.4 more rooms than the objective room need on average. In Germany, households in the lowest income quintile only have about 0.7 more rooms than their objective room needs. In England, the highest income quintile has about twice as many excess rooms than the lowest income quintile. In Germany, the ratio is about 1:3. Individuals in households with at least one immigrant have a higher room stress than the native German population on average. I do not find significant differences between households with an immigrant in the household and other households in England regarding room stress.

6.1.2 Patterns in Changes in Room Stress over the Life Course

I now present the country-specific within-individual and between-individual standard deviations of room stress. The within-standard deviation describes the variation in the de-meaned data used for fixed-effects regression (cf. Section 4.6.1). The between-standard deviation describes the variation in the person-specific means. These statistics provide information about the magnitude of inequalities *over* the life course (within) and about inequalities *across* life courses (between). The within-individual standard deviation in England is 0.883 (95 per cent confidence interval (CI): [0.876,0.884]). That is to say that roughly 68 per cent of English individuals live with room stress that varies about \pm 0.9 rooms around the average over their observed life course. Only about one third of the population differs more from their individual standard deviation is slightly smaller (within-SD= 0.823, CI: [0.821,0.826]). About 68 per cent of the German individuals have a room stress within about \pm 0.8 rooms of their individual average over their observed life course. The difference between the variation in England and Germany is statistically significant as the confidence intervals do not overlap. The
Group	Average room stress England Germany	
Total	-2.14 -1.28	
Region		
Southeast England/East Germany	-2.15^{\dagger} -0.90	
Rest*	-2.13 -1.36	
Tenure		
Social housing	-0.72 -0.21	
Rented privately	-1.39 -0.63	
Owned with mortgage*	-2.30 -2.08	
Owned outright	-2.79 -2.42	
Life stage		
Single (\leq 35 years old)	-0.94 -0.25	
Single (\geq 36 years old)	-1.77 -0.86	
Single parent	-0.94 -0.44	
Couple parent*	-2.13 -1.53	
Childless couple	-2.84 -1.91	
Other	$-1.46^{\$\dagger}$ $-1.58^{\$\dagger}$	
Household income		
1st quintile (lowest)	-1.37 -0.72	
2nd	-1.49 -1.12	
3rd*	-1.97 -1.28	
4th	-2.29 -1.68	
5th quintile (highest)	-3.00 -2.12	
Immigrant in HH		
yes	-2.20^{\dagger} -0.81	
no*	-2.12 -1.38	

Table 6.1 Average room stress by group and country in 2008

Data: BHPS wave 18, SOEP v26 wave 25 (household level, cross-sectional weights) Note: * indicates reference category, † indicates mean not statistically different from reference category. § : N <100.

between-standard deviation is quite similar in England and Germany. About 68 per cent of the individual-specific means of room stress are not more than 1.4 rooms different from the average of all individual-specific averages in both countries. Thus, the inequality between life courses is higher in both countries than the inequality over individuals' life courses on average with regard to room stress.

Figure 6.1 shows the proportion of relocations in each age group that increase room stress. Between 23 to 57 per cent of relocations in each age group are downward, i.e. they increase room stress. The risk of relocating into more crowded dwellings is lowest in the age range 26 to 45 years. For individuals aged 25 and below the risk is slightly higher. Relocations for those aged 56 and above are especially likely to increase room stress. In all age groups the share of downward relocations is higher in England than in Germany and the difference is especially strong in late-



Data: BHPS waves 1-18, SOEP v26 waves 8-25 (household level, cross-sectional weights) Note: Whiskers indicate 95 %-confidence interval of proportion.

Fig. 6.1 Proportion of downward relocations (room stress) by age group

life. In the next sections, I turn to the analysis of residential relocations that shape these patterns. Changes at isolated times in the life courses of individuals are also the focus of the remainder of this chapter.

6.1.3 Average Changes in Room Stress After Relocations

Table 6.2 shows the effects of residential relocations on room stress in the year immediately after a relocation. In this table, I report the average change in room stress for all relocations and for different types of relocations separately that have been introduced in Section 4.5.1.1. It turns out that this differentiation is important. While relocations in general reduce room stress for women and men in England and Germany, a closer look at different types of relocations reveals that some types of relocations actually increase room stress on average and that some of these effects are different for men and women. I will therefore consider the types of relocations separately in the subsequent analysis.

Young stable single households significantly reduce their room stress if they relocate compared to stayers (Table 6.2). For example, English women reduce their room stress by about 0.3 rooms. Only young, male singles in England do not significantly change their room stress if they relocate compared to staying put. Female and male singles aged 36 and above increase their room stress on average if they relocate in England. After a relocation, singles' average room stress is about 0.5 rooms higher than before the relocation. In Germany, female singles aged 36 and above slightly reduce their room stress in case of relocation, while men in Germany do not change their room stress. Relocations of stable single households are rarely motivated by adaptations to changes in objective room need, which only changes if a child joins or leaves the household. The former will rarely happen without a partner also joining the household. The latter may not trigger a relocation as long as the housing costs are not too much of a burden.

Couples that relocate reduce their room stress significantly. In England, couples reduce their room stress by about 0.5 rooms on average. In Germany, couples reduce their room stress by about 0.6 rooms on average. In general, couples are much less likely to relocate than singles (cf. Chapter 5). If couples relocate, they will often respond to increased room stress due to childbearing. Couples may also be more interested in investing into excess rooms as they anticipate future room needs after childbearing. Another reason for couples to relocate is the transition into ownership. Owned houses offer more rooms than apartments on average. In this regard, the difference in the changes in room stress between England and Germany are interesting. Couples in Germany reduce their room stress slightly more than couples in England through relocations. This may be due to the fact that couples in England are more likely to live in home ownership from the start of their partnerships, while couples in Germany make the transition into ownership during their partnerships (Mulder and Wagner 1998).

Relocations from single into couple households reduce room stress the most except for English men. For women in England room stress is reduced by about 0.6 rooms. In Germany, the change in room stress is even stronger and room stress is reduced by about 0.9 rooms on average for women and men. English men do not significantly change their room stress if they move together with a partner compared with not experiencing a relocation. These findings seems to be counter-intuitive at first sight, but are aligned with my expectations formulated in H 1.1. Forming a couple household seems to go along with an additional investment in excess rooms compared to singles with English men as an exception. First, this may be an anticipatory investment to meet future room needs after childbearing. Second, this may also indicate that couples have a higher preference for spacious dwellings than singles. Thus, the room need of two singles forming a couples may not just be additive, but may increase even more.

Significant gender differences for both countries exist in the effect of relocations that are accompanied by union dissolutions. Ex-partners that relocate into a single or new couple household experience an increase in room stress across all groups. However, in both countries, women face a greater increase in room stress than men. In England, women increase their room stress by about 0.9 rooms if they relocate

6.2 Multivariate Results

Relocation type	Average change in room stress				
•••	Eng	land	Ger	many	
	Women	Men	Women	Men	
No relocation	-0.02	-0.02	-0.01	-0.01	
	(40,212)	(32,359)	(114,278)	(103,363)	
Relocation	-0.17	-0.19	-0.27	-0.34	
	(3,435)	(3,035)	(9,997)	(9,027)	
Stable single (\leq 35 years old)	-0.30	-0.09^{\dagger}	-0.19	-0.26	
	(387)	(277)	(1,221)	(945)	
Stable single (\geq 36 years old)	0.52	0.51	-0.09	-0.05^{\dagger}	
	(323)	(189)	(1,355)	(945)	
Stable couple	-0.53	-0.52	-0.65	-0.62	
	(1,745)	(1,697)	(4,636)	(4,770)	
Single to couple	-0.56	-0.29^{\dagger}	-0.88	-0.91	
	(188)	(155)	(739)	(624)	
Union dissolved	0.85 *	0.38	0.78	** 0.46	
	(251)	(172)	(945)	(716)	
Other	0.36	0.43	0.28	0.36	
	(541)	(545)	(1,101)	(1,027)	

Table 6.2 Average changes in room stress after residential relocations

Data: BHPS waves 1-18, SOEP v26 waves 8-25 (individual level, cross-sectional weights) Note: Within country differences: *** significant at 0.1% two-tailed, ** significant at 1%,* significant at 5%. [†] indicates mean not statistically different from event not occured. N in parantheses.

and dissolve their union. In Germany, they increase room stress by 0.8 rooms. English men only increase their room stress by about 0.4 rooms after relocations with union dissolution and German men increase their room stress by about 0.5 rooms.⁴

6.2 Multivariate Results

In this section, the effect of residential relocations is analysed in a multivariate framework to test the hypotheses outlined in Section 2.3. I mainly use fixed-effects panel regression models to analyse the average treatment effect on the treated of different types of residential relocations on room stress (cf. Section 4.6.1 for a description of fixed-effects regression models). These models allow relation of changes in independent variables to changes in the dependent variable. The focus on within-individual change provides the opportunity to tap into determinants of changes in room stress that would not be possible with cross-sectional regression tools, but inference only relies on those individuals that experience changes in the dependent

⁴ The category of other relocations is too heterogeneous to interpret it meaningfully and instead functions as a control category.

and independent variables. Due to the limitations of fixed-effects regression outlined in Section 4.6.1, I refrain from making causal claims, but merely talk about associations between variables.

Table A.4 and Table A.5 in the appendix show descriptive statistics for all covariates used in the multivariate analysis. In the following sections, I split the sample by gender to allow the effects of covariates to vary between women and men. Furthermore, separating models by gender allows for inclusion of information at the household level while still treating individuals as the research unit. Thereby, the problem is tackled that households, in most cases, are the decision unit for relocation choices, while at the same time households are not constant over time.

6.2.1 Individual Life Course Conditions

In Table 6.3, I present estimation results for fixed-effects regressions on room stress.⁵ I report models separately by country and gender. The independent variables of most interest measure the occurrence of different types of residential relocations. These variables measure the immediate effect of relocations on room stress. All independent variables are measured at the same time as the dependent variable. Relocations, family and job events are measured between t - 1, i.e. the interview before the current interview t, and t. All models have a satisfactory explained within-individual variance of 10 to 12 per cent. In all models, the period dummies increase the explained within-variance the most and between 2 and 4 percentage points of the within-change are attributable to period effects. In Germany, the housing tenure variables are also important in explaining within-change and contribute about

⁵ I performed the following diagnostic analyses for the main fixed-effects models presented in Table 6.3. 1) I tested whether the more efficient random-effect model would be preferable over the fixed-effects model, but the Hausman test rejected the assumption that the random-effect model is consistent with the fixed-effects estimations for all models, 2) I further tested for autocorrelation and heteroscedasticity in the data. Both tests indicated that the use of clustered standard errors is in order. 3) I analysed the variance inflation factor (VIF) for a pooled OLS regression model. Only the age variables showed a VIF exceeding the common cut-off value of 10. 4) I also tested for a non-response bias by including a lead indicator respectively a lagged indicator of non-response in my analysis. Including the non-response variables does not have substantial influence on the estimated effects of other covariates. 5) Finally, I examined influential individual-year observations with Cook's distance value exceeding 4/N and absolute standardised Pearson residuals exceeding 3 (cf. Kohler and Kreuter 2005: 211). In England and Germany, influential cases had very low room stress on average and were more likely to own their dwelling outright. This may be caused by the strong heterogeneity of this group. After checking for data inconsistencies, I tested whether excluding the cases changes estimation results. The estimates were not substantially changed by excluding the cases in England. For German women, the negative effects of being employed or unemployed turned significant after excluding influential cases. For German men, relocations with union dissolution do not longer significantly affect room stress. I decided to keep all cases in the estimation sample.

3 percentage points to the explained variance. If included in a stepwise fashion, all variables increase model fit according to log-likelihood ratio (LR) tests with the exception of the job mobility variables that do not increase model fit in any specification.⁶ The variables are still included in the final model to provide evidence against H 1.5 concerning job mobility. Inclusion of the relocation variables increases model fit significantly, but has only a small impact on the size of the explained withinvariance of about 0.04 percentage points.⁷ All models include dummy periods not shown in the table. The effects of periods are consistent with the descriptive findings and show a constant decrease in average room stress over time. The models also include natural cubic spline specifications of age that are not shown in the table.⁸

The relocation type variables are interactions of partnership status and residential relocation, which are grouped by age for singles. The reference category for all relocation types are single stayers that do not change their partnership statuses. The effect of relocations is estimated controlling for changes in the region, changes in the degree of urbanisation and changes in housing tenure. For example, the estimated effect of stable single (≤ 35 years) relocations of women in England indicates that young women below the age of 36 reduce their room stress by about 0.3 rooms compared to single stayers that do not change their partnership statuses. Instead of the more complex relocation types, I also tested the effect of a general relocation variable in the model.⁹ The estimated coefficients indicate that individuals that relocate reduce their room stress by -0.10 rooms (t = -4.54), English men by -0.09 rooms (t = -3.90), German women by -0.17 rooms (t = -15.40) and German men by -0.19 rooms (t = -15.49). For theoretical reasons and due to the better model fit, I use the differentiated relocation types in the subsequent analysis.

Young stable single women reduce their room stress by about 0.3 rooms in England and by about 0.1 rooms in Germany on average (cf. Table 6.3). Young, stable single men do not change their room stress significantly if they relocate in England and Germany. Older singles significantly increase their room stress, if they relocate. This effect is not significant for male movers in Germany. That improvements are

⁶ All LR tests are conducted for models without clustered standard errors.

⁷ LR tests for English women: $\chi^2(6) = 165.38$, p = 0.000; English men: $\chi^2(6) = 161.15$, p = 0.000; German women: $\chi^2(6) = 715.69$, p = 0.000; German men: $\chi^2(6) = 569.47$, p = 0.000.

⁸ I tested quadratic up to quintic polynomials, as well as natural cubic splines with 3 to 7 knots to model the effect of age. I tested these specifications in models only including age and period. I tested different positions for the function with 7 knots. I compared the Akaike information criterion (AIC) values for all models and selected the age specification with a relative good model fit in both countries and for both genders (AIC values not shown here). I decided for a natural cubic spline specification with 7 knots at age 23, 32, 38, 45, 52, 61, and 75. This specification is used in all models throughout the chapter.

⁹ Including the relocation types instead of a general relocation variable significantly increases model fit for all groups (LR-tests for English women: $\chi^2(5) = 125.07$, p = 0.000; English men: $\chi^2(5) = 131.68$, p = 0.000; German women: $\chi^2(5) = 386.44$, p = 0.000; German men: $\chi^2(5) = 203.76$, p = 0.000.).

	Model 6.1 England		Mode Germ	l 6.2 Jany
	Women	Men	Women	Men
Southeast ENG/East GER	0.641***	0.693***	0.124	0.037
	(4.82)	(5.09)	(1.42)	(0.43)
Urbanisation (ref: Rural)				
Metropolitan	0.070	0.053	-0.096^{*}	-0.068
	(0.70)	(0.48)	(-2.01)	(-1.36)
Urban	-0.171	-0.185	0.006	0.003
	(-1.77)	(-1.79)	(0.17)	(0.09)
Partnership status (ref: Single)				
Cohabiting	-0.316^{***}	-0.142^{*}	-0.355^{***}	-0.241^{***}
	(-6.21)	(-2.04)	(-10.92)	(-6.50)
Married	-0.470^{***}	-0.236***	-0.484^{***}	-0.304***
	(-10.20)	(-3.60)	(-16.57)	(-8.24)
Child aged 1 to 9	-0.346***	-0.270***	-0.236***	-0.195***
c	(-10.89)	(-7.66)	(-11.26)	(-8.56)
Family events (ref: No event)	· /	. ,	· /	· · · ·
Child born	0.105***	0.092**	0.206***	0.219***
	(3.82)	(3.13)	(10.92)	(11.09)
Child left	-0.513***	-0.534***	-0.432***	-0.456***
	(-22.75)	(-20.43)	(-22.40)	(-21.86)
Job mobility (ref: No mobility)	× /	· /	× /	· /
Promotion	0.016	0.028	0.000	0.026**
	(1.13)	(1.83)	(0.02)	(2.79)
Demotion	0.012	0.009	0.008	0.005
	(0.70)	(0.46)	(0.63)	(0.42)
Job status (ref: Inactive)	(0110)	(0)	(0.00)	(***=)
Employed	0.027	-0.089*	-0.027	-0.003
Linpioyed	$(1 \ 13)$	(-2.42)	(-1.78)	(-0.13)
Unemployed	-0.025	-0.163***	-0.034	0.004
Chemployed	(-0.61)	(-3.35)	(-1.87)	(0.19)
Household income (log)	-0.069***	-0.107***	-0.031***	-0.053***
Household meome (log)	(-3.30)	(-5.05)	(-5.29)	(-7.12)
Housing tenure (ref: Owned with	mortgage)	(5.05)	(3.2))	(,.12)
Social housing	0.600***	0 590***	0.805***	0 763***
Social nousing	(8.53)	(6.89)	(20.79)	(18.91)
Rented privately	0.363***	0.368***	0 740***	0 743***
remed privatery	(5.79)	(5.11)	(22.85)	(21.62)
Owned outright	0.012	0.017	0 104***	0 103***
Gwiled buttight	(0.32)	(0.41)	(4.02)	(3.76)
	(0.52)	(0.41)	(+.02)	(3.70)
	continued of	overleaf		

Table 6.3 Fixed-effects models of room stress

continued				
	Mode Engl	l 6.1 and	Mode Gern	el 6.2 nany
	Women	Men	Women	Men
Relocation type (ref: Stayer)				
Stable single (\leq 35 years)	-0.266^{**}	-0.064	-0.102^{**}	-0.031
Stable single (\geq 36 years)	0.228***	0.274**	0.083**	0.041
Stable couple	(3.62) -0.188***	(2.77) -0.197^{***}	(2.77) -0.288^{***}	(0.86) -0.279^{***}
Single to couple	(-6.92) 0.056	(-7.02) 0.048	(-19.74) -0.291^{***}	(-18.92) -0.232^{***}
	(0.64)	(0.60)	(-6.72)	(-5.04)
Dissolved couple	(2.73)	(4.24)	0.159*** (4.75)	$(2.12)^{*}$
Other	-0.033	-0.040	0.027	-0.041
Constant	-0.846	(-0.74) -1.059	-0.374	(-1.33) -0.190
	(-0.87)	(-1.42)	(-1.06)	(-0.43)
Observations	44,027	37,947	119,790	104,523
Individuals	4,827	4,302	14,730	13,295
With change in outcome	3,925	3,488	10,442	9,352
LL Null-model	-53,996.37	-47,998.65	-145,213.12	-125,836.83
LL	-51,422.23	-45,953.30	-137,270.51	-119,558.40
within R ²	0.11	0.10	0.12	0.11

Data: BHPS waves 1-18, SOEP v26 waves 8-25 (individual level, unweighted)

Note: Linear fixed-effects regression model, dependent variable: room stress, unstandardised coefficients, t statistics in parentheses; *** significant at 0.1% two-tailed, ** significant at 1%, * significant at 5%; models also include covariates period and age not shown here.

more likely for young singles than for older singles is in line with past literature (e.g. Clark and Huang 2003; Morrow-Jones and Wenning 2005). Partners in stable couples significantly reduce their room stress in both countries. English women and men in stable couples reduce their room stress by about 0.2 rooms more than single stayers. Partners in stable couples in Germany reduce their room stress by about 0.3 rooms more than single stayers. Men in both countries reduce their room stress the strongest, if they relocate in a stable couple. However, for English men the difference to relocating as a young single is not significant¹⁰ and for German men the difference to relocating and forming a new couple is not significant.¹¹

¹⁰ F(1, 4301) = 1.45, p = 0.228.

¹¹ F(1, 13294) = 0.98, p = 0.322.

English women reduce their room stress the most if they relocate as young singles, but the difference to stable couples is not significant.¹² German women reduce their room stress the most if they relocate and form a new couple. Again, the difference to relocations in stable couples are not significant.¹³ Greater reductions in room stress for families and couples than for singles have also been found in the past (e.g. Feijten and Mulder 2002). Relocations of couples and relocations of singles to couples reduce room stress significantly more in Germany than in England.¹⁴ Older singles increase their room stress more in England than in Germany, if they relocate.¹⁵

If singles relocate and form a couple in England, they do not significantly change their room stress compared to single stayers. However, the effects of these relocations should be assessed together with a change in partnership status. Being in a partnership reduces room stress clearly. Those individuals that cohabit live in lower room stress compared with years in which they do not cohabit. Married individuals live in even lower room stress. The differences between married and cohabiting individuals are significant for women, but not for men in both countries.¹⁶ These findings confirm earlier results on the positive association of cohabitation and marriage with the quality of locations (Morrow-Jones and Wenning 2005). For English women and men, single to couple relocations do not significantly affect room stress, but women and men significantly reduce their room stress if they relocate and cohabit or marry.¹⁷ In Germany, single women reduce their room stress by about 0.3 rooms and men reduce their room stress by about 0.2 rooms if they relocate and form a union. Relocations that are associated with the dissolution of a union increase room stress for women and men in both countries. Especially men in England are adversely affected and increase their room stress by about 0.5 rooms more than single stayers, if they dissolve their union and relocate.

These findings provide evidence for and against a number of hypotheses put forward in Section 2.3 with regard to room stress. First, I hypothesise in H 1.1 that single to couple relocations and relocations of couples reduce room stress more than

¹² F(1, 4826) = 0.74, p = 0.391.

¹³ F(1, 14729) = 0.00, p = 0.953.

¹⁴ I test the difference between coefficients by stacking the samples and fully interacting the model with dummies for England and Germany. I then used Wald tests to evaluate the difference between coefficients. For women, stable couple: F(1, 19556) = 10.41, p = 0.001; for women, single to couple: F(1, 19556) = 12.56, p = 0.000; for men, stable couple: F(1, 17596) = 6.81, p = 0.009; for men, single to couple: F(1, 17596) = 9.22, p = 0.002.

¹⁵ For women: F(1, 19556) = 4.37, p = 0.037; for men: F(1, 19556) = 4.49, p = 0.034.

¹⁶ English women: F(1, 4826) = 11.89, p = 0.001; English men: F(1, 4301) = 3.39, p = 0.066; German women: F(1, 14729) = 19.30, p = 0.000; German men: F(1, 13294) = 3.73, p = 0.053.

¹⁷ Tests for combined effect of single to couple relocation and cohabiting: English women: F(1, 19556) = 165.06, p = 0.000; English men: F(1, 17596) = 63.30, p = 0.000. Tests for combined effect of single to couple relocation and married: English women: F(1, 19556) = 244.89, p = 0.000; English men: F(1, 17596) = 83.74, p = 0.000.

stable single relocations. I cannot reject this hypothesis for Germany. In England, I can reject the hypothesis regarding young singles which do not differ significantly in their changes in room stress in case of relocations from couples. Second, H 1.2 about the positive effect of marriage on the quality of location is supported. Third, H 1.4 states that relocations that are a) associated with union dissolution increase room stress and I expect this effect b) to be stronger for women than for men and c) to be persistent over the life course. I cannot reject the first part of the hypothesis, because women and men in England and Germany increase their room stress on average, if they relocate. The effect is stronger for women than men in Germany. but the difference is not statistically significant.¹⁸ In England, I find a significant difference between women and men, but against my expectation men increase their room stress more than women.¹⁹ Thus, I can reject the second part of H 1.4. To test the last part of the hypothesis, I include a variable measuring if an individual ever divorced before in the model. The results show no significant difference between English men and German women for the years before and after the first divorce in average room stress.²⁰ English women live in 0.20 (t = 2.61) rooms higher room stress after the first divorce on average and German men live in 0.12 (t = -1.97) rooms lower room stress after the first divorce. Thus, while I find an immediate, negative effect of relocations associated with union dissolution, most individuals seem to adapt in the long run after a divorce (cf. also Figure 6.2). The last part of H 1.4 must be rejected. This is in contrast to research regarding other quality indicators, e.g. Feijten and Mulder (2005) show that individuals are permanently less likely to live in home ownership after union dissolution.

	Model 6.3 England		Model Germ	l 6.4 any
	Women	Men	Women	Men
Child aged 1 to 9	-0.338***	-0.211^{***}	-0.134**	-0.010
	(-6.47)	(-3.89)	(-2.75)	(-0.18)
Household income (log)	-0.069^{**}	-0.106^{***}	-0.027^{***}	-0.047^{***}
_	(-3.28)	(-4.99)	(-4.31)	(-6.27)
Child*Income	-0.001	-0.009	-0.015^{*}	-0.026***
	(-0.17)	(-1.14)	(-2.11)	(-3.33)

Table 6.4 Fixed-effects models of room stress with interaction of income and presence of children

Data: BHPS 1-18, SOEP v26 waves 8-25 (individual level, unweighed)

Note: Linear fixed-effects regression model, dependent variable: room stress, unstandardised coefficients, t statistics in parentheses; *** significant at 0.1% two-tailed, ** significant at 1%, * significant at 5%; models also include all covariates presented in Table 6.3.

¹⁹ F(1, 9128) = 3.85, p = 0.050.

²⁰ English men: b = -0.148 (t = -1.35); German women: b = 0.052 (t = 1.08)

¹⁸ To test the differences in coefficients by gender I proceed as described in Footnote 14. F(1, 28024) = 1.56, p = 0.211.

Room stress of individuals that live with children aged 1 to 9 is about 0.3 rooms less in England and 0.2 rooms less in Germany on average compared to the same individual not living with children (cf. Table 6.3). Again, this reflects earlier findings on the positive association of being in a family and the quality of locations (Dieleman and Everaers 1994; Feijten and Mulder 2002). In Table 6.4, interaction effects of the presence of children and income are presented. For England, I do not find a significant interaction of these variables. Independently from income, individuals are in reduced room stress if children are present. In Germany, I find a significant interaction between the presence of children and income. The main effect of income for women and men in Germany shows that individuals with no children aged 1 to 9 significantly reduce their room stress if they have a higher income. The main effect of children aged 1 to 9 shows that in the hypothetical case of an individual with no household income, the presence of at least one child would reduce room stress for women by about 0.1 rooms but would not significantly reduce room stress for German men. The interaction effects of children aged 1 to 9 and income shows that women and men in Germany reduce their room stress significantly more in the presence of children if they have a higher income. The findings partly support H 1.3 about the a) positive effect of children on the quality of a location and b) a positive interaction effect between children and income. The first part of the hypothesis is supported in both countries. Individuals live in lower room stress in years in which a child aged 1 to 9 lives in the same household. The second part of the hypothesis is only supported for Germany and has to be rejected for England. The effect of children aged 1 to 9 is getting larger with increasing income only in Germany. Coming back to the models in Table 6.3, I find childbirth and a child leaving the parental home to have a strong effect on room stress. Individuals that experience a childbirth in their households increase their room stress by about 0.1 rooms in England. In Germany, room stress increases by about 0.2 rooms. If a child leaves the household, room stress of individuals is reduced by about 0.5 rooms in England and by about 0.4 rooms for women and 0.5 rooms for men in Germany.

In H 1.5, I stated that becoming unemployed or being demoted should increase room stress, while becoming employed and being promoted should reduce room stress on average. I do not find support for this hypothesis. German men increase their room stress slightly if they experience a promotion, but no other group is significantly affected by a promotion or demotion. Thus, individual labour income changes are not related to changes in room stress. English men live in dwellings with lower room stress if they are employed or unemployed compared to being inactive. The difference between employed and unemployed is statistically not significant.²¹ English women as well as German women and men do not change their room stress on average if they change their job statuses. Overall, job mobility has only small

²¹ F(1, 4301) = 2.88, p = 0.090.

effects on room stress, but I find employment and unemployment to have stronger effects on men's room stress in England than in Germany.²²

So far, I implicitly made rather restrictive assumptions about the effect of relocations on room stress. 1) I assumed that the effect of relocation is immediate and permanent. 2) I assumed that each subsequent relocation has the same additive effect on room stress. 3) I assumed that the effect of residential relocation on room stress is constant within subgroups of the sample. To test the first assumption, I allow the effect of relocations to vary over time. To this end, I include lagged variables in my model measuring the occurrence of a residential relocation not only at t, but also at t - 1, t - 2, and at t - 3 or earlier. My data does not allow me to test the second assumption. The assumption could be tested by separately identifying the effect of first and later relocations in the residential trajectory, but neither the BHPS nor the SOEP provides retrospective information about residential relocations. The third assumption is tested by interacting relocations with income to allow the effects of relocations to vary by income level.

I present estimation results of models including lagged variables of different relocation types in Figure 6.2. The figure only shows the coefficients of the lagged event variables and their 95 per cent confidence interval. For each relocation type, the immediate effects of the event taking place at t, i.e. in the year just before the interview, at the very left and the lagged effects up to t - 3, i.e. the relocation is at least 3 years ago at the time of the interview, at the very right are shown. Thus, by looking at each group of four coefficients from left to right the effect of relocations can be followed over time. Only coefficients for which the presented confidence interval does not include 0 are statistically significant.

For stable couples, the effect of relocations diminishes over time. In England, the effect is no longer significant if the relocation was at least three years ago. In Germany, even those stable couples that relocated at least three years ago are still in lower room stress compared to singles stayers. Single to couple relocations have a persisting effect on room stress in Germany. Even if the relocation is three or more years ago, individuals that relocate to form a couple are still in lower room stress than single stayers that do not change their partnership statuses. This is also true for English women, but not for English men whose room stress is not affected by this type of relocation at any point in time. Young singles in both countries and of both genders do not show a clear pattern regarding the effect of relocations on room stress over time. It seems that in Germany the effect of relocation. The positive effect of these relocations on room stress in England is essentially constant over time.

The negative effect of relocations that are associated with a union dissolution vanishes over time for all but German women. English women and men face a strong increase in room stress if they just relocated and dissolved their union in the year before the interview. Already in the second year after a relocation, men are

²² Employment: F(1, 17596) = 4.10, p = 0.043; unemployment: F(1, 17596) = 9.69, p = 0.002.



Data: BHPS waves 1-18, SOEP v26 waves 8-25 (individual level, unweighted) Note: Linear fixed-effects regression model, dependent variable: room stress; graph shows unstandardised coefficients and 95 %-confidence interval for event occured at t (very left), t-1, t-2, t-3 or earlier (very right); models also include all covariates presented in Table 6.3.

Fig. 6.2 Effects of relocation types on room stress over time

no longer in significantly higher room stress than single stayers that do not change their partnership statuses. English women's room stress is no longer significantly different if the relocation is at least two years ago. Germans need more time to reduce their room stress after a relocation associated with a union dissolution. German men do not live in significantly higher room stress anymore only after the relocation is at least three years ago. For German women, the effect of relocations with union dissolutions also decreases over time. Even three or more years after the relocation took place, German women are still in significantly higher room stress than single stayers that did not change their partnership statuses.

6.2.2 Economic Resources and Housing Tenure

Housing is a good traded in markets and even though welfare states intervene in these market exchanges, the quality of housing that individuals can purchase is assumed to be strongly depended on individuals' economic resources. In H 2.1 a), I expect that individuals that increase their income are likely to reduce their room stress. In Table 6.3, I show that income changes have this effect on room stress. If individuals have a higher household income they are in lower room stress than the same individuals with lower income on average. English women and men reduce their room stress by about 0.1 rooms, German women by only slightly more than 0.0 rooms and German men by about 0.1 rooms on average if their log household income increases by 1 unit. This is in accordance with H 2.1 a). Higher incomes are associated with lower room stress, albeit only a slight difference. English women and men reduce their room stress about twice as much with an increase in income than individuals in Germany. This difference across countries is only statistically significant for men.²³ This is in accordance with H 4.3 which states that income and quality of housing are more strongly associated in England than in Germany, because housing is more decommodified in Germany than in England. I discuss this issue further in Section 9.1.3. My findings regarding the negative association of income and room stress are in accordance with the literature (Kendig 1984; Clark, Deurloo and Dieleman 1997, 2000; Buck 2000a).

In H 2.1 b), I expect that the effect of relocations on room stress varies by income. I expect individuals with higher incomes to reduce their room stress more through residential relocations than similar individuals with lower incomes. To test this expectation, I interact relocation types with income and I present the results in Table 6.5. The main effect of income indicates that for stayers in both countries, room stress is lower on average in years in which they have higher incomes. For example, English women reduce their room stress by about 0.1 rooms if their log household income increases by 1 unit. The main effects for the different relocation types show the effect of relocations if the individual has no household income, which is not sensible to interpret. The interaction effects of various types of relocations and income indicate how the effect of relocations linearly vary by log household income. For example, the interaction effect for stable single (≤ 35 years) and income indicates how the effect of these relocations on room stress compared to the reference category of single stayers changes by household income.

I find several interaction terms to be significantly different from 0, and thus conclude that the effect of these types of relocations on room stress depends on the level of income. Young single relocations for women in England increase their room stress on average if they have a higher income. This is counter-intuitive, but the main effect of young stable single relocations is about -2.3 for women in England and, thus, for most observed income values these relocations will still reduce room stress

 $^{^{23}}$ F(1, 17596) = 5.84, p = 0.016.

	Model 6.5 England		Model Germ	6.6 any
	Women	Men	Women	Men
Household income (log)	-0.090^{***} (-4.21)	-0.102^{***} (-4.71)	-0.029^{***} (-4.85)	-0.049^{***} (-6.68)
Relocation type (ref: Stayer)	· · · ·	· /	· /	· · · ·
Stable single (\leq 35 years)	-2.285^{***}	-0.508	-0.706	0.431
	(-3.63)	(-0.80)	(-1.58)	(0.85)
Stable single (\geq 36 years)	-0.977	-0.295	0.222	0.946*
	(-1.24)	(-0.44)	(0.84)	(2.24)
Stable couple	0.917*	0.182	0.133	0.326
-	(2.56)	(0.49)	(0.93)	(1.51)
Single to couple	0.988	-0.921	1.311*	2.350***
	(1.09)	(-1.01)	(2.42)	(3.68)
Dissolved couple	-1.044	0.657	0.308	-0.628
	(-1.82)	(0.98)	(1.16)	(-1.28)
Other	-0.113	0.508	-0.231	-0.210
	(-0.25)	(1.45)	(-1.44)	(-0.80)
Interaction with income				
Stable single (\leq 35 years)	0.300***	0.069	0.087	-0.064
	(3.37)	(0.77)	(1.38)	(-0.91)
Stable single (\geq 36 years)	0.171	0.077	-0.020	-0.123^{*}
	(1.51)	(0.82)	(-0.52)	(-2.11)
Stable couple	-0.149^{**}	-0.051	-0.057^{**}	-0.082^{**}
	(-3.08)	(-1.01)	(-2.93)	(-2.80)
Single to couple	-0.126	0.130	-0.215^{**}	-0.346^{***}
	(-1.01)	(1.08)	(-2.93)	(-4.00)
Dissolved couple	0.182^{*}	-0.027	-0.021	0.097
	(2.16)	(-0.30)	(-0.56)	(1.46)
Other	0.009	-0.078	0.038	0.025
	(0.14)	(-1.60)	(1.61)	(0.66)

Table 6.5 Fixed-effects models of room stress with interaction of relocation type and income

Data: BHPS waves 1-18, SOEP v26 waves 8-25 (individual level, unweighted)

Note: Linear fixed-effects regression model, dependent variable: room stress, unstandardised coefficients, t statistics in parentheses; *** significant at 0.1% two-tailed, ** significant at 1%, * significant at 5%; models also includes all covariates presented in Table 6.3.

on average. The main effect may also indicate that low-income women reduce their room stress more than high-income women, because they experience higher room stress before the relocation. Relocations of older, male singles in Germany also interact significantly with income. With each unit increase in log household income, these relocations reduce room stress by an additional 0.1 rooms.

With increasing income, stable couples reduce their room stress to a greater degree if they relocate on average. This interaction is significant for women in England and both genders in Germany, but not for men in England. In Figure 6.3, I plot the marginal effects of stable couple relocations on room stress by income based on the samples of women.²⁴ The marginal effects and the 95%-confidence intervals are shown. Marginal effects are significant if the confidence interval does not include 0. For English women, stable couple relocations only have a significant effect on room stress above PPP-\$ 800 household income. Below this income, relocations of stable couples do not significantly affect room stress. The marginal effects of stable couple relocations on room stress become more negative with increasing income. An English woman living in a household with PPP-\$ 1,000 income reduces room stress by about 0.1 rooms with a relocation on average. A woman living in a household with PPP-\$ 3,000 income reduces room stress by about 0.3 rooms. The predicted marginal effects for English men do not change as much by income as for women, but they show a similar picture: relocations of stable couples have a greater impact on improvement of room stress with higher incomes. I also find this to be true in Germany, where women in stable couples reduce their room stress more in case of relocation with higher household income. For example, women with a household income of PPP-\$ 1,000 reduce their room stress by slightly more than 0.3 rooms on average. Women with a household income of PPP-\$ 3,000 reduce their room stress by slightly less than 0.3 rooms.

The findings show that all stable couples except the lowest income groups reduce their room stress on average if they relocate. However, the reduction in room stress is clearly stratified by income. The returns of relocations increase with growing household incomes. Couples with high household incomes reduce their room stress substantially more than couples with low incomes. The stratification of returns by income seems to be stronger in England than in Germany. This finding again supports H 4.3 about the stronger association of income and quality of locations in England by comparison.

For Germany, I find significant interactions of single to couple relocations and income. Single women and men in Germany that relocate and form a couple reduce their room stress much more with growing income. Women and men with low income are likely to increase their room stress if they relocate and form a couple in Germany, which is apparent from the significant and positive main effect for single to couple relocations. Finally, relocations that are associated with a union dissolution interact positively with income to impact room stress for women in England. Thus, women with higher incomes reduce their room stress less than women with lower incomes. This finding is unexpected and one explanation may be that women with high household incomes have lived in relatively low room stress before. After union dissolution and relocation, these women are initially in small dwellings and this may drive the positive interaction effect. Over time, these women will reduce their room stress again and adapt to their new situations on average (cf. Figure 6.2).

Housing markets are structured by housing tenures and access to each tenure is strongly dependent on current and potential economic resources of individuals. At

²⁴ The findings for men are similar to the findings for women. The graphs are generated with code suggested in Brambor, Clark and Golder (2006).



Data: BHPS waves 1-18, SOEP v26 waves 8-25 (unweighted)

Note: Based on linear fixed-effects regression models shown in Table 6.5, dependent variable: room stress; Thin lines indicate 95 %-confidence interval of marginal effects.

Fig. 6.3 Marginal effect of stable couple relocations on room stress conditional on household income, women

the same time, housing tenures are strongly correlated with average housing quality. In general, home ownership offers the best housing quality, but to purchase a home individuals need economic resources. In H 2.2, I expect changes into home ownership to reduce room stress the most, changes to private rented dwellings to reduce room stress moderately, and changes to social housing to reduce room stress the least. My analysis shows support of this hypothesis. For example, a stable couple that relocates from social housing to a dwelling owned with mortgage reduces room stress by about $0.8 \ (\approx -0.188 + (-1) * 0.600)$ rooms on average in England (cf. Table 6.3, based on women sample). In Germany, the same couple would reduce room stress even by about $1.0 \ (\approx -2.88 + (-1) * 0.805)$ room (based on women sample). In Germany, changes in housing tenure also considerably contribute to the explained variance of the model as mentioned above. The models show that both rental tenures are associated with higher room stress in England and Germany com-

pared to owning. The differences are especially strong in Germany and coefficients are at least marginally significantly different across countries.²⁵

In England, the effects of living in social housing or privately rented dwellings differ significantly.²⁶ Individuals in social housing are in higher room stress than if they are in privately rented dwellings. In Germany, social housing and privately rented dwellings do not differ much in average room stress and only for women the coefficients for the two tenures are significantly different.²⁷ In England, living in dwellings owned outright does not significantly affect room stress compared to living in dwellings owned with mortgages. In Germany outright owners experience slightly higher room stress compared to owning with mortgages.

Figure 6.4 shows predicted changes in room stress for women in case of relocation by tenure of origin and tenure of destination.²⁸ The predictions are based on OLS regression models with change in room stress between t - 1, i.e. the year before the interview, and t as the dependent variable.²⁹ Three findings are especially relevant. First, social housing in both countries is associated with high room stress, but social tenants can improve their situations on average if they relocate independently of their tenures of destination. Second, private tenants change their room stress differently in England and Germany. In England, private tenants reduce their room stress significantly less than social tenants in most cases, while there is no difference in Germany on average. Private tenants in England only change their room stress significantly, if they buy a home. In Germany, private tenants also reduce their room stress on average if they continue to rent. Third, owners in both countries change their room stress similarly. Differences between outright owners and owners with mortgages are not statistically significant in England or Germany. Expectations about cross-national differences formulated in H 2.3 are further discussed in Section 9.1.2.

6.2.3 Situational Conditions of Decision Making

H 3.1 states that a) individuals that relocate long distances will have less improvement in room stress than individuals relocating short distances, but I expect that b)

²⁵ For women, social housing: F(1, 19556) = 6.49, p = 0.011; for women, rented privately: F(1, 19556) = 28.76, p = 0.000; for women, owned outright: F(1, 19556) = 3.86, p = 0.049; for men, social housing: F(1, 17596) = 3.37, p = 0.066; for men, rented privately: F(1, 17596) = 22.15, p = 0.000; for men, owned outright: F(1, 17596) = 3.06, p = 0.080.

²⁶ English women: F(1, 4826) = 11.97, p = 0.001; English men: F(1, 4301) = 7.01, p = 0.008.

²⁷ German women: F(1, 14729) = 8.84, p = 0.003; German men: F(1, 13294) = 0.79, p = 0.373.

²⁸ Due to space restrictions I only report findings for women. The predicted room stress values for men are very similar.

²⁹ All independent variables except relocation variables shown in Table 6.3 are included. In addition the lagged housing tenure from t - 1 is included.



Data: BHPS waves 1-18, SOEP v26 waves 8-25 (unweighted)

Note: Based on linear OLS regression models with dependent variable change in room stress and all covariates shown in Table 6.3; whiskers indicate 95 %-confidence interval; only movers.



the effect of distance is stronger regarding neighbourhood quality than regarding room stress. H 3.2 stated that individuals improve the quality of their locations less on average, if they relocate involuntarily compared to voluntarily. This is due to the constrained search process for involuntary relocations. A new location has to be found quickly and individuals may be more likely to relocate to the next best location instead of considering several locations and choosing the best one. In Table 6.6, I present models testing these hypotheses with regard to room stress.³⁰ To capture involuntary relocations, I include subjective reasons for relocating into the models. I consider the reasons 'notice given by landlord', 'housing costs too high' and 'partnership dissolution'.³¹ It can be assumed that individuals' choices to relocate or to

 $^{^{30}}$ In this and subsequent sections of the chapter, a general relocation variable is used instead of the differentiated relocation types.

³¹ The reason 'housing costs too high' cannot clearly be identified in the BHPS data is because the question asked about size and cost: 'Wanted smaller/cheaper accommodation'.

stay are limited in these cases. I contrast involuntary relocations with voluntary relocations of various distances, i.e. relocations that are not motivated by one of the three reasons for involuntary relocation. Distances of relocations are only available after 2000 in the SOEP and I drop earlier years from the analysis in the present subsection.

Voluntary relocations reduce room stress for almost all groups and distances of relocations on average (cf. Table 6.6). Only relocations that are longer than 30 km do not significantly affect room stress for women and men in England. Relocations have similar effects on room stress irrespective of their distances. I observe only two significant differences between coefficients: First, women in England that relocate between 1 and 5 km reduce their room stress more than women that relocate more than 30 km.³² Second, women in Germany that relocate between 1 and 5 km reduce their room stress significantly more than women that relocate less than 1 km.³³ Men also increase their room stress the most if they relocate between 1 and 5 km, but there are no significant differences compared to other coefficients. Relocations between 1 and 5 km may provide access to a broader variety of housing than shorter relocations, but at the same time these relocations are too short to substantially disrupt individual social networks and, thus, may be especially attractive for movers that only want to improve their housing situations. All in all, the distance of relocations has little impact on the change in room stress associated with relocation in most cases. Thus, H 3.1 can partly be rejected.

In contrast to voluntary relocations, involuntary relocations are mainly associated with increases in room stress (cf. Table 6.6). Individuals that state that they relocated because of a union dissolution increase their room stress in both countries. For example, German women increase their room stress by 0.4 rooms more than stayers on average if they relocate due to a union dissolution. This findings is in accordance with earlier evidence of the negative effect of relocations that are associated with union dissolution presented in Section 6.2.1. Individuals that relocate after they receive a notice from their landlords significantly increase their room stress in England compared to stayers. English women and men increase their room stress by 0.4 rooms on average. In Germany, involuntary relocations due to termination of the rent contract do not affect average room stress of movers. Neither German women nor men change their room stress significantly in these cases. However, relocations after termination of the rent contract reduce room stress significantly less than voluntary relocations for women in Germany, but not for men.³⁴ These differences across countries may be due to stronger legal protection of tenants in Germany. The gender difference may be driven by groups of single women such as single mothers

 $^{^{32}}$ F(1, 3450) = 5.26, p = 0.022.

 $^{^{33}}$ F(1, 12081) = 7.59, p = 0.006.

 $^{^{34}}$ I test jointly whether involuntary relocations are different from voluntary relocations of all distances for German women: F(4, 12081) = 3.45, *p* = 0.008; German men: F(4, 10816) = 1.16, *p* = 0.327.

that are more vulnerable to terminations of the rent contract than single men. Relocations that are motivated by housing costs that were too high at the former location also increase room stress on average in both countries. In England, women and men increase their room stress by 0.4 (women) and 0.5 (men) rooms respectively if they have to relocate from their former location as they can no longer pay the housing costs. In Germany, the effect of these relocations is smaller. Women and men increase their housing stress by 0.1 to 0.2 rooms on average. These findings support H 3.2 and the findings are also in accordance with past research showing the negative effects of involuntary relocations (e.g. Kleinhans 2003).

	Model Engla	Model 6.7 England		l 6.8 any
	Women	Men	Women	Men
Voluntary relocation				
< 1 km	-0.192^{**}	-0.203^{**}	-0.160^{***}	-0.192^{***}
	(-2.79)	(-2.80)	(-6.46)	(-7.05)
1 < 5 km	-0.241***	-0.213***	-0.254***	-0.259***
	(-4.71)	(-3.89)	(-9.14)	(-8.52)
5 < 30 km	-0.142^{*}	-0.199**	-0.226***	-0.240***
	(-2.37)	(-3.16)	(-6.51)	(-6.52)
30+ km	0.015	-0.064	-0.236***	-0.235***
	(0.15)	(-0.60)	(-4.90)	(-4.85)
Involuntary relocation		× /	× /	· /
Separation	0.481***	0.803***	0.368***	0.375***
-	(3.33)	(4.50)	(7.50)	(6.55)
Notice given	0.391**	0.348**	-0.014	-0.104
C C	(2.98)	(2.62)	(-0.20)	(-1.19)
Housing costs	0.397***	0.497***	0.146***	0.175***
-	(3.45)	(3.87)	(3.54)	(3.67)

Table 6.6	Fixed-effects	models of room	stress for volu	untary and	involuntary	relocations

Data: BHPS 11-18, SOEP v26 waves 18-25 (individual level, unweighed)

Note: Linear fixed-effects regression model, dependent variable: room stress, unstandardised coefficients, t statistics in parentheses; *** significant at 0.1% two-tailed, ** significant at 1%, * significant at 5%; models also include all covariates presented in Table 6.3 except relocation variables.

6.2.4 Immigrant Households and Room Stress

Ascribed statuses such as ethnicity can cause discrimination in the housing market and, thus, immigrants may be constrained in their relocation behaviours. Not only immigrants themselves, but also individuals living with immigrants in a common household may suffer from this discrimination. Therefore, I measure the immigrant status at the household level (cf. Section 4.5.2 for a detailed description of the variable). Under the assumption that immigrants have similar preferences to natives, I hypothesise that immigrant households reduce their room stress less than natives if they relocate due to discrimination. I also expect first generation immigrant households to differ to a greater degree from natives than second generation immigrant households (H 4.4). To support this hypothesis, I should find a negative interaction effect of residential relocations and first and second generation immigrant households. I also should find first generation immigrant households to reduce their room stress less than second generation immigrant households are constrained by discrimination, I should also find a negative interaction between income and immigrant status with regard to room stress. Immigrants who possess the same economic resources as natives who do not make it to high quality locations are presumably facing some discrimination from gatekeepers at those locations.

I test these expectations in Model 6.9 and Model 6.10 in Table 6.7. The main coefficients for household income show the effect of changes in income on room stress of individuals in native households. For women and men in native households in both countries, increases in income significantly reduce room stress on average. In the fixed-effects regression models it is not possible to include the timeconstant main effect for first and second generation immigrant households (cf. Section 4.6.1).³⁵ The interaction terms of income and immigrant status for women in England indicate no significant differences between natives, first and second generation immigrant households. For English men, first generation immigrant households reduce their room stress significantly less than natives with increasing incomes. Male second generation immigrant households do not differ with respect to the effect of income on room stress from individuals in native households. For Germany, I find female first and second generation immigrant households to reduce their room stress significantly less than natives with higher income. In fact, testing the combined main and interaction effect shows that women's room stress in immigrant households is not significantly affected by changes in income.³⁶ This is also true for men in first generation immigrant households in Germany.³⁷ Men in second generation immigrant households reduce their room stress significantly on average if their incomes increase.³⁸ Thus, I find evidence for Germany that individuals in immigrant households are constrained by discrimination in their housing situations. Under the assumption of similar preferences, individuals in immigrant households

- 37 F(1, 13294) = 2.40, p = 0.122.
- 38 F(1, 13294) = 4.09, p = 0.043.

³⁵ Native individuals may leave immigrant households, but it can be expected that this is a very small group of individuals and I do not further examine this group.

³⁶ First generation: F(1, 14727) = 0.02, p = 0.902; Second generation: F(1, 14727) = 0.31, p = 0.577.

that increase their incomes cannot use these resources to reduce their room stress as much as natives. I do not find evidence for discrimination in England.

	Model 6.9 England		Model Germ	6.10 any
	Women	Men	Women	Men
Household income (log)	-0.080^{***}	-0.125***	-0.037***	-0.057***
Income*1 st gen. immigrant	(-3.30) 0.008	(-4.89) 0.049^*	(-6.20) 0.039**	0.030
Income*2 nd gen, immigrant	(0.44) -0.035	(2.21) 0.028	(2.98) 0.046**	(1.78) 0.021
	(-1.34)	(1.36)	(3.14)	(1.20)
Relocation	(-4.12)	(-2.66)	(-10.70)	-0.164
Relocation*1 st gen. immig.	-0.067	-0.147^{*}	-0.100^{***}	-0.072^{**} (-2.72)
Relocation*2 nd gen. immig.	(-0.92) -0.047 (-0.53)	(-2.02) -0.126 (-1.38)	(-0.165^{***}) (-3.99)	(-2.72) -0.156^{***} (-3.63)

Table 6.7	Fixed-effects	models of ro	oom stress	bv ir	nmigrant	status
					(m) · · · ·	

Data: BHPS 1-18, SOEP v26 waves 8-25 (individual level, unweighted)

Note: Linear fixed-effects regression model, dependent variable: room stress; unstandardised coefficients, t statistics in parentheses; *** significant at 0.1% two-tailed, ** significant at 1%, * significant at 5%; models also include covariates presented in Table 6.3.

The main effect for relocations shows that women and men in native households in both countries significantly reduce their room stress if they relocate. For English women, I do not find a significant interaction of relocation and immigrant status. Individuals in first and second generation immigrant households are as likely as English women in native households to reduce their room stress with residential relocations. Men in first generation immigrant households in England reduce their room stress by about 0.1 rooms more on average than English men in native households. Men in second generation immigrant households in England do not change their room stress significantly from men in native households in case of relocation. Individuals in immigrant households in Germany reduce their room stress significantly more than Germans in native households, if they relocate. This is true for women and men in first as well as second generation immigrant households. For example, women in first generation immigrant households reduce their roomstress by 0.1 rooms more on average if they relocate compared to women in native households. These findings reject H 4.4 with regard to changes in room stress. Contrary to my expectations, individuals in immigrant households reduce their room stress not less than individuals in native households if they relocate — in Germany they reduce their room stress even more than individuals in native households.

6.2.5 Structural Conditions in Housing Markets

Up to now, only individual and household level variables have been considered. I now extend the analysis by considering regional characteristics and structural conditions in the housing markets.³⁹ Most indicators on housing markets in England are only available after 1999. I drop earlier years from the analysis in the present section. Individuals are clustered in housing markets. In the panel model, this clustering is difficult to take into account in a straightforward manner and, thus, the estimated standard errors for variables at the housing market level may be underestimated. Instead, I estimate an alternative multi-level model with random intercepts at the individual and housing market level to yield unbiased standard errors for the housing market variables. The membership in certain housing markets is not constant over time, however, because individuals may relocate between different housing markets. Hence, individuals can be clustered in more than one housing market in the observation period. This kind of data structure can be accommodated by multiple membership multi-level models (Browne, Goldstein and Rasbash 2001; Windzio 2008). In multiple membership models the membership in different units is weighted by the time which respondents are exposed to the respective unit (cf. Section 4.6.2 for more details on these models). The multiple membership multi-level model is reported in Table A.6 in the appendix. I only highlight general characteristics of the model and differences to the panel regression model here.

According to the fixed-effects panel regression reported in Table 6.8, women that have relocated at some point in the past between the Southeast of England and the rest of England live in 0.6 rooms greater room stress on average than the same individual before the relocation. English men's room stress does not change significantly if they relocate between the Southeast and the rest of England. Neither German women nor men live in significantly different room stress, if they have moved between East and West Germany. The degree of urbanisation does not seem to have a substantial effect on average room stress. Individuals that relocate between rural, urban and metropolitan areas do not live in more or less crowded dwellings comparing years before and after the relocation. The findings are broadly similar in the multi-level model (cf. Table A.6).

I consider three indicators that further describe the structural conditions in the regional housing market: share of owned dwellings, number of new constructions and the population balance over the last five years (cf. Section 4.5.2). In the fixed-effects panel regression model presented in Table 6.8, I find only a few structural conditions to be relevant for explaining changes in room stress. In Germany, individuals are more likely to reduce their room stress if the share of owned dwellings in their regional housing markets increases — either for the reason that individuals relocate

³⁹ Housing markets refer to the geographical unit of counties (*Kreise und kreisfreie Städte*) in Germany. Germany is divided into 413 counties. In England, housing markets also refer to the geographical unit of 88 counties (cf. Section 4.3.3).

Model 6.11 England		Model Germ	6.12 any
Women	Men	Women	Men
0.594^{*}	0.430	-0.072	-0.127
(2.55)	(1.60)	(-0.70)	(-1.19)
-0.003 (-0.02)	-0.008 (-0.04)	-0.161 (-1.81)	-0.035 (-0.39)
(-0.077)	(-0.221)	(-0.022)	0.035
-0.101^{***}	(-0.084^{*})	-0.157^{***}	-0.176^{***}
(-3.30) -0.017	(-2.49) -0.009	(-10.90) -0.008^{***}	(-11.33) -0.009^{***}
(-1.60) -0.005	(-0.91) -0.008	(-4.97) 0.000	(-5.09) 0.002
(-1.07) 0.006 (1.78)	$(-1.45) \\ 0.002 \\ (0.50)$	(0.13) 0.004^{*} (2.56)	(0.54) 0.003 (1.38)
22,041 3,614 2,670 -23,706.53 -22,908.90 0.07	19,177 3,197 2,347 -21,094.39 -20,488.56 0.06	73,054 12,415 8,259 -78,672.00 -75,438.67	64,507 11,089 7,362 -69,764.12 -67,160.68
	$\begin{array}{r} \textbf{Model}\\ \textbf{Engl}\\ \textbf{Women}\\ \hline 0.594^{*}\\ (2.33)\\ -0.003\\ (-0.02)\\ -0.077\\ (-0.49)\\ -0.101^{***}\\ (-3.30)\\ -0.017\\ (-1.60)\\ -0.005\\ (-1.07)\\ 0.006\\ (1.78)\\ \hline 22,041\\ 3,614\\ 2,670\\ -23,706.53\\ -22,908.90\\ 0.07\\ \end{array}$	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$\begin{array}{c c c c c c c c } \mbox{Model 6.11} & \mbox{Model 6.11} & \mbox{Model 6.11} & \mbox{Germ} \\ \hline \begin{tabular}{ c c c c c } \hline \begin{tabular}{ll c c c c } \hline \begin{tabular}{ll c c c c } \hline \begin{tabular}{ll c c c } \hline \begin{tabular}{ll c c c c c } \hline \begin{tabular}{ll c c c c } \hline \begin{tabular}{ll c c c c c } \hline \begin{tabular}{ll c c c c c c } \hline \begin{tabular}{ll c c c c c c } \hline \begin{tabular}{ll c c c c c c c c c c c c c c c c c c $

Table 6.8 Fixed-effects models of room stress with housing market variables

Data: BHPS 10-18, ONS various years, SOEP v26 waves 17-25 (individual level, unweighted), INKAR 2010

Note: Linear fixed-effects regression model, dependent variable: room stress, unstandardised coefficients, t statistics in parentheses; *** significant at 0.1% two-tailed, ** significant at 1%, * significant at 5%; models also include all covariates shown in Table 6.3.

to a different housing market or because the conditions in the housing market itself change over time. If the share of owned dwellings increases by 1 percentage point, the average room stress for individuals falls very slightly but significantly. Individuals relocating from the housing market with the lowest share of owned dwellings (8.3 per cent) to the market with the highest share (88.6 per cent) would reduce their room stress by about 0.8 rooms on average in Germany. In England, the share of owned dwellings does not affect room stress according to the fixed-effects model. The findings from the multi-level model diverge from the fixed-effects model. According to the former, the share of owned dwellings affects individuals' levels of room stress in England. Women and men that live in housing markets with a higher share of owned dwellings have a lower room stress after controlling for average individual room stress over time. Differences between the multi-level model and the fixed-effects model seem to be due to smaller standard errors in the multi-level

specification. Larger standard errors in the fixed-effects regression models result mainly from the smaller effective sample size of these models, as only individuals with change in the outcome and independent variables are used to draw inference. These findings give support to H 4.2 about the positive effect of average regional housing quality on individuals' room stress. In housing markets with a higher share of high quality housing, individuals have more opportunities to relocate to better locations. This confirms earlier findings for the Netherlands (Feijten and Mulder 2005).

The number of new constructions has no effect on changes in room stress. Thus, increases in the supply of housing does not seem to cause changes in the average room stress in a housing market. This finding is also supported in the multi-level model. The population balance over the last five years is an indicator for the demand side in the regional housing market. In the fixed-effects specification, only German women are significantly affected by the population balance. Their room stress increases very slightly, if over the last five years more people moved into the region or were born relative to the number of people that died or left the region. Changing the population balance from the minimum in Germany (-26.82 persons per 1,000 inhabitants) to the maximum (31.02 persons per 1,000 inhabitants) would increase the room stress by about 0.2 rooms. In the multi-level model, the coefficients of the population balance for all groups are very similar to the fixed-effects model, but again the standard errors are smaller. Therefore, according to the multi-level model, English women, German women and German men are all significantly affected by the population balance in their room stress. The more the population grows in a region, the more likely individuals experience room stress. These findings do not clearly support or reject H 4.1 about the effects of changes in the regional supply and demand of housing on room stress. There seem to be some support for the part about demand: A positive change in population growth is associated with positive changes in room stress. The part about supply can be rejected as a positive change in number of constructions is not associated with negative changes in average room stress.

The estimation of the random term of the multi-level model shows substantial variation of intercepts between individuals (cf. Table A.6). Thus, after conditioning for all covariates, individuals still differ significantly in levels of time-constant average room stress. I also find significant variation of intercepts between housing markets in Germany. There seems to be unobserved heterogeneity in housing markets that causes divergence in average regional room stress. For example, regional variation in the average household structure or typical dwelling size may be captured here. I do not find significant variation of intercepts between housing markets in England. Thus, conditioned on the independent variables, the average regional room stress does not vary significantly between markets in England.

6.3 Summary

I found individual life course conditions to be important in explaining room stress changes. First, relocations in general reduce room stress, but the effect of relocations varies by life course stage and associated life course events. Relocations of couples and relocations to form a couple household reduce room stress more than other types of relocations. Married couples experience especially low room stress. Union dissolutions increase room stress. The presence of children reduces room stress in the long run, while childbearing initially increases room stress. Thus, in the growingfamily phase individuals may experience higher room stress, but subsequent relocations reduce room stress. The positive effects of stable couples' relocations and the negative effects of relocations associated with union dissolutions fade out over time as individuals adapt their housing situations. Transitions in the job trajectory do not cause changes in room stress. These findings are evidence that with regard to room stress, outcomes of residential relocations are contingent upon life course events and stages. Their outcomes are shaped by individuals' life courses. Disparities result between life course stages such as the pre-family phase and the family phase and from adverse life events such as union dissolution.

At the same time, outcomes are co-determined by individuals' resources. Household income has a strong effect on room stress. First, changes in income are negatively associated with changes in room stress. Thus, individuals that increase their household incomes are also likely to reduce their room stress on average. Second, the contingent event of residential relocations is shaped by individual resources. At least for stable couples, relocations reduce room stress more with greater household income. In Germany, economic resources moderate the way that families with children can adapt their housing situations. Individuals that have the necessary resources to purchase property and relocate into ownership reduce their room stress more than individuals that stay within the rental sector. Social housing is associated with the highest room stress in both countries. Stratification of room stress by tenure is stronger in Germany than in England. My findings clearly show disparities in the effect of relocations on room stress dependent on individual resources. Individuals with high economic resources and individuals that can purchase their own homes reduce their room stress to a much greater degree than individuals without these resources. For individuals that lack these resources, relocations negatively affect the housing situation and increase room stress.

Structural conditions in the housing market have little effect on average room stress of individuals. The regional degree of urbanisation is not associated with room stress. The demand in the regional housing market is weakly associated with room stress. Individuals are in higher room stress on average, if they are in housing markets with above average population growth compared to housing markets with lower population growth. Changes in the supply of housing in the market do not affect average room stress. I found some evidence that the share of owned dwellings in a market is negatively associated with average room stress, but the findings were not clear.

Individuals in households with immigrants in England are not significantly different from individuals in native households in their room stress. In Germany, individuals in immigrant households are in higher room stress on average than individuals in native households in a cross-sectional perspective. For Germany, I found some evidence that individuals in immigrant households are not able to transform income increases into lower room stress and this may indicate discrimination in the housing market. However, if individuals in immigrant households relocate in Germany, they reduce their room stress even more than natives. I conclude that structural conditions have only a weak impact on room stress of individuals. Disparities in changes in room stress are relatively independent from conditions in the housing market. Only in housing markets with above-average population growth, do individuals have a harder time to reduce their room stress levels. Disparities in room stress between migrants and natives in Germany are reduced through residential relocations.

I found some evidence for the importance of situational conditions of decision making. First, involuntary relocations increase room stress on average. Thus, individuals that have to relocate and are restricted in their search process relocate to more crowded dwellings than individuals that are less restricted in their search on average. I find almost no support for a stratification of relocation outcomes by the distance of relocation. Individuals that relocate only 1 to 5 km reduce their room stress the most, but differences compared to shorter and longer relocations are only small. I did not test for intra-household differences and gender roles with regard to room stress.

Chapter 7 Neighbourhood Quality and Residential Relocations

Individuals' dwellings are located in certain neighbourhoods, which constitute an important characteristic of the residential location (cf. Section 3.3.1). Local neighbourhoods are complex social entities that can provide important resources and opportunities for their residents. Ideally, neighbourhoods are places of social interaction with neighbours, provide essential goods and services, as well as safe and healthy environments. Neighbourhoods are stratified by the amount of opportunities and resources that they offer and this may affect their residents' life chances. Living in less resourceful neighbourhoods with fewer opportunities is associated with negative outcomes for the residents in various areas such as health, education and employment. Individuals may relocate to better neighbourhoods to increase life chances, and, in general, voluntary relocations can be expected to improve neighbourhood quality. However, past research shows that this path is not open to all and following from the theoretical framework described in Section 2.3, systematic differences in the outcomes of residential relocations across the population can be expected.

In order to investigate different outcomes I first examine descriptive evidence in Section 7.1. Next, I use multivariate methods to formally test the hypotheses (Section 7.2). First, I examine the effect of individual life course conditions (Section 7.2.1). Second, I scrutinise inequalities that result from unequal economic resources and disparities between housing tenures (Section 7.2.2). Third, I test for the effects of distances of relocations and involuntary relocations on changes in neighbourhood quality (Section 7.2.3). Fourth, potential inequalities in changes in neighbourhood quality for individuals in immigrant households and native households are analysed in Section 7.2.4. Fifth, the effect of structural conditions in regional housing markets is considered (Section 7.2.5). In Section 7.3, I summarise my findings.

My measures of neighbourhood quality are based on additional data sources that are linked to the BHPS and SOEP data sets. I use the Indices of Multiple Deprivation (IMDs) for England and the Microm data set for Germany to construct the neighbourhood quality measures. I describe the construction of these measures in detail in Section 4.5.1.3. A higher value on these measures indicates better neighbourhoods. Data from respondents in the SOEP cannot be linked to information at the neighbourhood level prior to 2000, and, thus, the sample in the present chapter is more restricted than in the previous analyses, as the observation period is limited to the years 2000 to 2008.¹

7.1 Descriptive Results

I now turn to the descriptive analysis of neighbourhood quality in England and Germany. I present descriptive results that show how my objective neighbourhood quality measure is related to subjective perceptions of neighbourhood quality in Germany to validate the quality measure that has not been used in prior research (Section 7.1.1).² Next, I show summary statistics about changes in neighbourhood quality over individuals' life courses (Section 7.1.2). In Section 7.1.3, I describe average changes in neighbourhood quality after residential relocations for particular groups of movers.

7.1.1 Objective and Subjective Neighbourhood Quality in Germany

Table 7.1 summarises the general neighbourhood quality measure, which is an 'objective' measure of quality, by subjective satisfaction measures for Germany in 2004. The subjective measures are based on questions about the perceived neighbourhood quality asked in the SOEP in 2004. Clearly, those respondents that report certain problems with their neighbourhood are living in objectively worse neighbourhoods compared to those not reporting the respective problems. This difference is especially strong regarding safety in the neighbourhood. Those feeling unsafe live in neighbourhoods with a 0.36 standard deviation lower level of quality on average than those that feel safe. In Table 7.2, the subjective evaluation of the last relocation of the household is compared with the change in objective neighbourhood quality. The subjective measures are based on respondents' self-reported satisfaction with their present dwelling compared to the last one and these measures are surveyed every wave. In general, I find subjective changes in neighbourhood quality to correspond to objective changes in quality with regard to the general neigh-

¹ I report descriptive statistics for all variables in the appendix in Table A.4 for England and Table A.4 for Germany.

 $^{^2}$ The English neighbourhood data has been used repeatedly in the past (e.g. Rabe and Taylor 2010).

bourhood and environmental conditions. For example, those that report a subjective improvement of neighbourhood quality also improve their objective neighbourhood quality by 0.25 standard deviations on average. On the contrary, those reporting a subjective deterioration of neighbourhood quality also live in objectively worse neighbourhoods that have -0.60 standard deviations lower quality compared to the last location.

Not all subjective measures correspond with the objective measure, though. Those that report a deterioration of their contacts to neighbours do not change their objective neighbourhood quality significantly different from those that report no change. This may indicate that contact with neighbours is not closely associated with the general quality of the neighbourhood. In low- as well as high-quality neighbourhoods, residents may interact socially with their neighbours. In addition, those respondents that report a deterioration in their access to public transportation compared to their last location increase their objective neighbourhood quality by 0.45 standard deviations on average. The last finding indicates an important caveat in interpreting subsequent findings for Germany. The neighbourhood quality index that I constructed is not necessarily measuring the availability of infrastructure facilities that may be important for some residents. For example, less wealthy households without cars may need public transportation in their neighbourhoods to organise their everyday life. Wealthier households with cars may not use public transportation and, thus, are not affected by a lack of transportation in the neighbourhood.

Subjective evaluation	Average neighbourhood quality			
-	Yes	-	No	
Feel unsafe	-0.36	***	0.03	
Green spaces lacking	-0.23	***	0.08	
Noise is a problem	-0.11	***	0.10	
Air pollution is a problem	-0.11	***	0.10	
No visits with neighbours	-0.10	***	0.06	

 Table 7.1
 Average neighbourhood quality by subjective neighbourhood evaluation in 2004, Germany

Data: SOEP v26 wave 21 (household level, cross-sectionally weighted), Microm 2010 Note: Difference: *** significant at 0.1% two-tailed, ** significant at 1%,* significant at 5%.

Therefore, in interpreting subsequent results for Germany, it must be kept in mind that the neighbourhood index only measures certain characteristics of the neighbourhood. On average, the neighbourhood quality measure will have a higher value in purely residential, suburban areas with high shares of long-time residents. Mixed urban areas with a high turnover of residents will be mostly measured as 'bad' neighbourhoods. These objective neighbourhood quality may not always be congruent with subjective neighbourhood quality, e.g. younger individuals may prefer to live in objectively 'worse' neighbourhoods in urban areas. Future research may validate my findings by using additional subjective quality indicators.

 Table 7.2
 Average change in neighbourhood quality by change in subjective neighbourhood evaluation, Germany

Subjective evaluation	Average	Average change in neighbourhood quality					
	worsened		no change		improved		
Neighbourhood	-0.60	***	-0.02	***	0.25		
Environmental conditions	-0.43	***	-0.04	***	0.34		
Contact with neighbours	0.02		0.04		0.14		
Access to public transportation	0.45	***	0.12	***	-0.19		

Data: SOEP v26 waves 17-25 (household level, cross-sectionally weighted), Microm 2010 Note: Differences to no change: *** significant at 0.1% two-tailed, ** significant at 1%,* significant at 5%.

7.1.2 Patterns in Changes in Neighbourhood Quality over the Life Course

Similar to Section 6.1.2 regarding room stress, I now present statistics to describe variation in neighbourhood quality and changes in neighbourhood quality over parts of the life course. The within-standard deviation of neighbourhood quality in England is 0.226 (CI: [0.225, 0.228]). Slightly more than two thirds of individuals in England differ not more than about 0.2 standard deviations from their individual mean over their observed life course. Variation over the life course is slightly higher in Germany, where the within-standard deviation is 0.271 (CI: [0.270, 0.272]). However, due to the different construction of the neighbourhood quality variable and the much smaller geographical and population size of neighbourhoods in Germany, these differences may mainly be a statistical artefact. The between-deviation is considerably higher than the within-deviation and close to 1 for both countries as a result of the standardisation of the neighbourhood quality measure.³

Figure 7.1 shows the risk of relocating into worse neighbourhoods. The proportion of downward relocations is relatively stable across age groups in Germany at 46 per cent. In England, I observe the youngest age group to have a relatively high risk of relocating into worse neighbourhoods with about 44 per cent. Those aged 26 to 35 are considerably less likely to relocate to worse neighbourhoods and the risk

 $^{^{3}}$ England: between-SD = 0.983, CI: [0.967, 1.000]; Germany between-SD = 0.970, CI: [0.961, 0.979]. The between-deviation is not exactly 1, since the within-deviation has been extracted from the data.



Data: BHPS waves 10-18, SOEP v26 waves 17-25 (household level, cross-sectional weights) Note: Whiskers indicate 95 %-confidence interval of proportion.

Fig. 7.1 Proportion of downward relocations (neighbourhood quality) by age group

increases for the next two age groups before it decreases at late life. Thus, I observe more of a uniform risk of relocating to worse neighbourhoods in Germany than in England. In all age groups except those aged 46 to 55, German movers are more likely to relocate into worse neighbourhoods than English movers.

7.1.3 Average Changes in Neighbourhood Quality After Relocations by Groups

Table 7.3 shows the average neighbourhood quality as well as the average change in neighbourhood quality for those who relocate broken down by several covariates that are measured at the first observation after the relocation. For Germany and England, the average neighbourhood quality for the whole population is close to 0, due to the standardisation of the variable.⁴ The average change in neighbourhood quality associated with a relocation is 0.07 standard deviations in Germany. This change is quite small, but significantly different from 0 at the 95 per cent confidence

⁴ The mean of the standardised variable is not exactly 0, as the results are weighted.

level. For England, movers do not change their neighbourhood quality on average. Thus, relocations in England tend to be horizontal rather than vertical with regard to neighbourhood quality. About 8 per cent of movers in England and 13 per cent of movers in Germany reduce their neighbourhood quality by at least 1 standard deviation. About 13 per cent of movers in England and 17 per cent of movers in Germany increase their neighbourhood quality by at least 1 standard deviation.

In both countries, I find strong differences in average neighbourhood quality by region. Individuals in Southeast England live in 0.48 standard deviations higher neighbourhood quality than those in the rest of England on average. This difference may be driven by the higher average economic resources in the Southeast (Anyadike-Danes 2004). East Germans live in -1.03 standard deviations lower neighbourhood quality on average than individuals in West Germany. This difference is striking, because the purchasing power variable, which is used to construct the neighbourhood quality measure for Germany, has been standardised separately for East and West Germany controlling for average differences in economic resources. Residential stability is slightly higher in West than in East Germany. The difference between East and West Germany is mainly driven by much higher prevalence of donations, which is used as a proxy for social and economic capital, and national newspapers, which is used as a proxy for cultural capital, in West Germany (cf. Section 4.3.2.2 for details on the neighbourhood data). In both countries, there are no significant differences between the regions in the change of neighbourhood quality after relocations. Thus, while I find the average neighbourhood quality to differ by region, relocations' effects on neighbourhood quality do not seem to be conditioned by regional characteristics.

In England, individuals in metropolitan areas live in -1.03 standard deviations lower neighbourhood quality compared to individuals in rural areas on average. Individuals in urban areas live in -0.61 standard deviations lower neighbourhood quality than individuals in rural areas. Relocations that end in rural areas increase neighbourhood quality by 0.09 standard deviations on average, which is significantly more than relocations that end in urban or metropolitan areas. The strong differences in average neighbourhood quality by degree of urbanisation are mainly caused by the higher population density in urban and metropolitan areas compared to rural areas (Adams 1992). The higher population density leads to, for example, more traffic, less green space, and more crime. Rural areas are also characterised by greater residential stability, as individuals are less mobile on average. Residential stability is assumed to improve social networks in neighbourhoods (Sampson and Groves 1989).

In Germany, individuals in metropolitan and rural areas live in significantly lower quality neighbourhoods compared to urban areas. The differences between metropolitan and rural areas are not statistically significant. Relocations that end in all three types of areas improve neighbourhood quality to the same degree on average. The finding that neighbourhood quality in rural areas is lower than in urban areas in Germany is unexpected. Due to the lower population density in rural areas, one

Group	Average neighbourhood quality and change England Germany					
	Quality	Change	Quality	Change		
Total	-0.01	0.00	0.01	0.07		
Region						
Southeast England/East Germany *	0.30	-0.02	-0.82	0.08		
Rest	-0.18	0.00^{\dagger}	0.21	0.07^{\dagger}		
Urbanisation						
Metropolitan	-0.42	-0.04^{\dagger}	-0.24	0.06^{\dagger}		
Urban *	0.00	-0.03	0.19	0.06		
Rural	0.61	0.09	-0.22	0.12^{\dagger}		
Immigrant in HH						
Yes	-0.08	-0.06^{\dagger}	-0.07	0.06^{\dagger}		
No *	0.01	0.01	0.03	0.07		
Tenure						
Social housing	-0.78	-0.05	-0.55	-0.06		
Rented privately	-0.12	-0.05	-0.25	0.04		
Owned with mortgage *	0.10	0.05	0.26	0.26		
Owned outright	0.19	-0.03^{\dagger}	0.43	0.27^{\dagger}		
Life stage						
Single (\leq 35 years old)	-0.43	-0.22	-0.33	-0.05		
Single (\geq 36 years old)	-0.24	0.12^{\dagger}	-0.10	-0.03		
Single parent	-0.32	-0.09	-0.26	-0.04		
Couple parent *	0.04	0.08	0.07	0.19		
Childless couple	0.10^{\dagger}	-0.02	0.10^{\dagger}	0.09		
Other	-0.31	$0.25^{\dagger \S}$	-0.10^{\dagger}	0.67^{+8}		
Employment status						
Employed *	0.04	0.01	0.04	0.12		
Unemployed	-0.58	-0.16	-0.50	-0.04		
Inactive	-0.07	-0.02^{\dagger}	0.04^{\dagger}	-0.02		
Household income						
1 st quintile (lowest)	-0.37	-0.12	-0.33	-0.02		
2 nd	-0.27	0.04^{\dagger}	-0.11	0.07^{\dagger}		
3 rd *	-0.09	-0.00	0.06	0.17		
4 th	0.10	0.01^{+}	0.21	0.08^{\dagger}		
5 th quintile (highest)	0.37	0.05^{+}	0.46	0.18^{+}		

 Table 7.3 Average neighbourhood quality and change in quality by group and country

Data: BHPS waves 10-18 (individual level, cross-sectional weights), IMD 2004 & 2007, SOEP v26 waves 17-25 (individual level, cross-sectional weights), Microm 2010

Note: Change only for movers, characteristics measured after relocation, * indicates reference category, † indicates mean not statistically different from reference category; § : N <100.

would expect to find rural areas to have better neighbourhood quality than urban and metropolitan areas just as in England. Taking a closer look at the indicators that are used in the neighbourhood quality measure for Germany reveals that rural areas have a much lower prevalence of donations and national newspapers than urban and metropolitan areas. The differences in these two indicators seem to drive most of the average differences in neighbourhood quality between areas with different degrees of urbanisation in Germany. The degree of urbanisation variable should be treated with caution, though, as it is measured at the level of counties. Within counties, smaller areas that can be characterised as metropolitan, urban or rural may co-exist.

In both countries, individuals in immigrant households live in significantly lower quality neighbourhoods on average than individuals in native households. In both countries, the difference in neighbourhood quality between the two types of households is roughly 0.1 standard deviations. This is in accordance with earlier findings that showed lower average neighbourhood quality for immigrants in Germany (Drever 2004) and England (Clark and Drinkwater 2002). However, changes in neighbourhood quality after relocations are not significantly different between individuals in immigrant and native households. Once individuals in immigrant households relocate, they seem to improve their neighbourhood as much as individuals from native households in both countries.

Tenants in social housing live in the worst neighbourhoods on average in both countries compared to others. This is in accordance with earlier literature showing that social housing tends to be concentrated in low-quality neighbourhoods in both countries (e.g. for England Fitzpatrick and Pawson 2007; for Germany Häußermann and Siebel 2000). The average neighbourhood quality is higher for private tenants than for social tenants, but is still well below the average neighbourhood quality of owners. The difference between social tenants and outright owners is 0.97 standard deviations in England and 0.98 standard deviations in Germany. In Germany, individuals that relocate into home ownership improve their neighbourhood quality more on average than both rental tenures. Individuals that relocate into ownership with a mortgage in England also improve their neighbourhood quality, but individuals that relocate into outright ownership do not improve their neighbourhood quality.

Singles and single parents live in neighbourhoods with below average quality in both countries. Old singles are better off than young singles and single parents. With the exception of old singles in England, individuals ending their relocation in any of these household types reduce their neighbourhood quality. In contrast, couple parents and childless couples live in above average neighbourhoods in both countries. Couple parents also improve their neighbourhood quality by 0.08 standard deviations in England and by 0.19 standard deviations in Germany after relocations on average. Childless couples improve their quality significantly in Germany, but not so in England. The number of other households is too small to draw reliable conclusions from the data. These findings show that especially couples live in neighbourhoods whose qualities are above average. Couple parents increase their
neighbourhood quality in both countries after relocation and the improvement is markedly higher in Germany than in England.

My findings corroborate what is found in the literature that an increasing concentration of low-income and unemployed individuals live in lower quality neighbourhoods (e.g. for England Meen et al. 2005: 25ff; for Germany Friedrichs and Triemer 2008: 111ff). The unemployed live in neighbourhoods more than half a standard deviation lower in quality than the employed on average in both countries. The inactive live in significantly lower quality neighbourhoods than the employed in England, but not in Germany. The employed improve their neighbourhood quality on average in both countries, while the unemployed reduce their neighbourhood quality through relocations. The inactive in Germany change their neighbourhood quality significantly less than the employed, but in England the difference is not significant. Part of the effect of unemployment on neighbourhood quality is likely to be caused by income losses.

Income has a strong effect on neighbourhood quality according to my descriptive results. The higher the income of individuals, the better their neighbourhood qualities in both countries. Individuals in the lowest income quintile live in neighbourhoods that are 0.37 standard deviations below the general average in England and 0.33 standard deviations below the general average in Germany. The difference between the highest and lowest income quintile in neighbourhood quality is more than 0.70 standard deviations in both countries. While levels of neighbourhood quality are clearly associated with income, changes in neighbourhood quality are not obviously related to levels of income. Individuals in the lowest income quintile in both countries reduce their neighbourhood quality on average if they relocate, while individuals in all other income quintiles at least relocate horizontally, i.e. they do not change their neighbourhood quality. Individuals do not increase their neighbourhood quality more with higher incomes. Improvements in neighbourhood quality seem to be equally likely for most income groups with the exception of the lowest income quintiles.

7.2 Multivariate Results

I now turn to the multivariate analysis to provide further evidence regarding the hypotheses.⁵ In the multivariate analysis, I again mostly apply fixed-effects regression models with clustered standard errors (cf. Section 4.6.1).⁶ Due to the caveats

⁵ In contrast to Chapter 6, I was not able to test for cross-country differences in the present chapter, since the data for England and Germany had to be analysed separately due to data protection.

⁶ Due to the multi-level structure in my data, I am confronted with a complexly nested data structure observations. Observations are clustered within individuals, neighbourhoods and housing markets. This clustering may bias the estimated standard errors and thereby yield misleading significance tests. I account for the first clustering by estimating robust standard errors that allow

of fixed-effects regression I again refrain from making causal claims in the present analysis and talk about associations between variables instead (cf. Section 4.6.1). Also, the estimated coefficients are only based on those respondents for which a change in the neighbourhood variable is measured, i.e. individuals that change their neighbourhood in the observation period by relocating.

7.2.1 Individual Life Course Conditions

I present estimation results of fixed-effects regressions on neighbourhood quality in Table 7.4.⁷ Model 7.1 for England and Model 7.2 for Germany have an explained within-variance of 9 per cent for women and men.⁸ In both countries, the degree of

⁷ Similar to my approach in Chapter 6, I ran several diagnostic tests for the full models reported in Table 7.4. 1) I tested whether the more efficient random-effect model would be preferable, but the Hausman test rejected the assumption that the random-effect model is consistent with the fixed-effects estimations for all models. 2) I tested for autocorrelation and heteroscedasticity. Both tests indicated that clustered standard errors should be applied. 3) I analysed the variance inflation factor (VIF) for a pooled OLS regression model to test for multicollinearity. Only the age variables showed a VIF exceeding the common cut-off value of 10. Thus, multicollinearity did not seem to affect the estimation. 4) I tested for a non-response bias by including a lead indicator and a lagged indicator of non-response in my model. The lead indicators of non-response were not significant for women and men in both countries (English women: b = -0.018, t = -1.35; English men: b = -0.009, t = -0.66; German women: b = 0.001, t = 0.12; German men: b = -0.003, t = -0.46). The lagged indicator of non-response was only significant for English women (English women: b = 0.026, t = 2.22; English men: b = 0.009, t = 0.76; German women: b = 0.018, t = 1.68; German men: b = -0.009, t = -0.73). English women were in significantly higher quality neighbourhoods, if they did not take part in the last interview. Thus, due to non-response I may slightly underestimate the effect of relocations on neighbourhood quality for English women. The models controlled for non-response showed very similar results to the full models. 5) Finally, I examined influential individual year-observations with large residuals in a pooled OLS regression model. After checking for data inconsistencies, I tested whether excluding the cases changes estimation results. All changes in coefficients were minuscule and therefore I decided to keep the cases in the estimation sample.

⁸ I only report the full models here. I tested if each covariate has a significant effect on the model fit in a stepwise procedure using LR tests. A number of variables do not increase model fit significantly. For English women and men, the variables child aged 10 to 17, child moved out, as well as the variables measuring job mobility and job status did not increase model fit (not shown here). For English women, the variables child born and household income also did not increase model fit. For German women and men, the variables child aged 10 to 17, as well as the variables measuring job

for clustering at the individual level. I do not consider the second clustering a problem for the estimation of standard errors, because on average a building sector in the Microm dataset contains only 1.02 interviewed households and a street sector contains on average 1.4 households of the SOEP sample (Dittmann and Goebel 2010). In England, a neighbourhood (LSOA) contains 2.0 households of the BHPS sample in 2000 and 1.7 households in 2008 on average. In Section 7.2.5, I test for the robustness of my results by estimating alternative multi-level models to account for potential bias due to the third clustering.

urbanisation has the strongest explanatory power and contributes about 7 percentage points to the explained within-variance for English women and men, about 3 percentage points for German women and about 5 percentage points for German men in the full model. The inclusion of the different relocation types increases model fit significantly compared to a model without relocation types, but does not increase the explained variance considerably.⁹ All models also include the controls period and region as well as a natural cubic spline specification of age and these are not reported in the tables.¹⁰ The effect of period is minuscule. The effects of region in both countries is similar to earlier descriptive findings. I do not find a significant effect of age in the fully specified model in England.¹¹ For Germany, on the other hand, age still has a significant effect on neighbourhood quality even in the full model.¹² For German women and men, neighbourhood quality increases strongly up to their 40s and stagnates thereafter.

I include the differentiated types of relocations in the models (cf. Table 7.4).¹³ These differentiated types of residential relocations are interactions of partnership status and residential relocation (cf. Section 6.2.1 for more details regarding the interpretation). Most differentiated types of relocations have no significant effect on neighbourhood quality compared to single stayers. Relocations are mostly hori-

⁹ Women in England: $\chi^2(6) = 23.99$, p = 0.001; men in England: $\chi^2(6) = 48.45$, p = 0.000. Women in Germany: $\chi^2(6) = 153.88$, p = 0.000; men in Germany: $\chi^2(6) = 164.06$, p = 0.000.

¹⁰ I tested quadratic up to quintic polynomials, as well as natural cubic splines with 3 to 7 knots to model the effect of age. I tested these specifications in models only including age and period. I tested different positions for the function with 7 knots. I compared the AIC values for all specifications and selected the model with the best fit (not shown). For England, a natural cubic spline function with 7 knots at age 22, 32, 39, 46, 54, 63, and 77 offered the best fit for both genders. In Germany, a natural cubic spline function with 7 knots at age 26, 36, 42, 49, 57, 65, and 76 was selected for women. For men, a natural cubic spline function with 7 knots at age 20, 30, 40, 50, 60, 70, and 80 was selected. These specifications are used in all models throughout the chapter.

¹¹ I test whether all age coefficients are commonly equal to 0. For both genders, H₀ cannot be rejected (women: F(6, 3470)= 1.43, p = 0.200; men: F(6, 3051)= 1.89, p = 0.079). The model with age provides a better model fit than without age, however (LR test for women: $\chi^2(6) = 37.67$, p = 0.000; men: $\chi^2(6) = 29.63$, p = 0.000).

¹² Again, I test that all age coefficients are commonly equal to 0. For both genders, H₀ can be rejected (women: F(6, 11853)=7.64, p = 0.000; men: F(6, 10617)=7.57, p = 0.000). In addition, the inclusion of age improves the model fit significantly (LR test for women: $\chi^2(6) = 271.22$, p = 0.000; men: $\chi^2(6) = 230.72$, p = 0.000).

¹³ I also estimate the general effect of all types of relocations on changes in neighbourhood quality. For English women b = 0.006 (t = 0.47), English men b = 0.027 (t = 2.19). German women b = 0.045 (t = 5.35), German men b = 0.047 (t = 5.09). Including the relocation types instead of a general relocation variable significantly increases model fit for all groups (LR-tests for English women: $\chi^2(5) = 23.44$, p = 0.000; English men: $\chi^2(5) = 32.69$, p = 0.000; German women: $\chi^2(5) = 41.56$, p = 0.000; German men: $\chi^2(5) = 51.86$, p = 0.000).

mobility and job status did not increase model fit. For German men, the variable child moved out also did not increase model fit. To provide evidence against some of my hypotheses and to be able to compare effects of variables on neighbourhood quality with effects on room stress, I included all variables in the full models.

zontal instead of vertical with regard to neighbourhood quality. For English women, no type of relocation significantly affects neighbourhood quality. Even relocations in stable couples, which increase neighbourhood quality for all other subsamples, are not associated with changes in neighbourhood quality. English men increase their neighbourhood quality by about 0.04 standard deviations, if they relocate in a stable couple. No other type of relocation has a significant effect on English men's neighbourhood quality. German women and men relocate to neighbourhoods with about 0.06 standard deviations higher quality if they relocate in a stable couple. For German women and men, relocations in the other category increase neighbourhood quality on average, but the rest of the relocation types do not affect neighbourhood quality.

I do not find evidence for H 1.1 that states that relocations of couples and singles that become couples improve neighbourhood quality more than relocations of singles. None of the coefficients for relocations are significantly different from each other except relocations of older singles and couple relocations in Germany.¹⁴ It is important to keep in mind that neighbourhood quality can only change in the event of residential relocations due to the construction of the neighbourhood quality measure (cf. Section 4.5.1.3). At the same time, certain covariates can only meaningfully change in value if a relocation takes place, e.g. changes from metropolitan to rural area necessarily induce residential relocations.

H 1.4 a) states that individuals reduce the quality of their locations on average if their relocation is associated with a union dissolution. According to H 1.4 b) this effect should be stronger for women than for men and c) should persist over the life course. I do not find a significant effect of relocations associated with union dissolution in the fixed-effects specification. In contrast to the situation regarding room stress, ex-partners that relocate are not more likely to reduce their neighbourhood quality than other individuals. However, the effect of these relocations must be seen in combination with the partnership status variables to grasp the whole effect of relocations into or out of partnerships. Married women in England and Germany live in higher quality neighbourhoods in years in which they are married on average compared to being single, but the same is not true for men in both countries. German women also live in better quality neighbourhoods, if they cohabit compared to living without a partner. The coefficients for cohabiting and being married are not significantly different for women and men in England and Germany.¹⁵ I have to reject H 1.2 stating that married couples live in better neighbourhoods than cohabiting couples for England and Germany. Similar results have been found for the US where transitions to cohabitation and marriage are also not associated with better neighbourhood quality (Sampson and Sharkey 2008). To assess the combined effect of relocations and transitions in the partnership trajectory on neighbourhood quality,

¹⁴ German women: F(1, 11853) = 4.23, p = 0.040; German men: F(1, 10617) = 6.38, p = 0.012.

¹⁵ English women: F(1, 3470) = 3.23, p = 0.072; English men: F(1, 3051) = 3.69, p = 0.055; German women: F(1, 11853) = 0.03, p = 0.857; German men: F(1, 10617) = 0.05, p = 0.829.

7.2 Multivariate Results

	Model Engla	7.1 nd	Model Germ	7.2 any
	Women	Men	Women	Men
Urbanisation (ref: Rural)				
Metropolitan	-0.697^{***}	-0.680^{***}	-0.452^{***}	-0.512^{***}
-	(-7.28)	(-7.37)	(-5.35)	(-5.58)
Urban	-0.408***	-0.341***	0.044	0.073
	(-4.79)	(-4.25)	(0.51)	(0.79)
Partnership status (ref: Single)	× /	· /	× /	· · /
Cohabiting	0.021	-0.001	0.091***	0.039
C	(0.67)	(-0.03)	(3.88)	(1.63)
Married	0.063*	0.052	0.087***	0.043
	(2.45)	(1.46)	(4.50)	(1.91)
Child aged (ref: No child)	(-)			
1 to 9	0.002	0.040**	0.035**	0.034**
	(0.18)	(2.86)	(2.83)	(2.72)
10 to 17	0.001	0.002	-0.017	-0.004
	(0.10)	(0.12)	(-1.88)	(-0.47)
Family events (ref: No event)	(****)	(***=)	(()
Child born	0.014	0.005	-0.001	0.003
	(1.26)	(0.52)	(-0.07)	(0.23)
Child left	0.013	0.018*	-0.005	0.003
	(1.33)	(2.07)	(-0.64)	(0.36)
Iob events (ref: No event)	(1.55)	(2.07)	(0.01)	(0.50)
Promotion	-0.001	-0.002	0.003	0.000
Tomotion	(-0.12)	(-0.30)	(0.69)	(0.05)
Demotion	0.007	-0.001	0.005	0.006
Demotion	(0.96)	(-0.19)	(0.88)	(1.06)
Job status (ref: Inactive)	(0.90)	(0.1))	(0.00)	(1.00)
Employed	-0.004	0.019	-0.002	0.005
Employed	(-0.49)	(1.18)	(-0.21)	(0.66)
Unemployed	0.003	-0.020	(-0.002)	-0.001
Unemployed	(0.18)	(-0.96)	(-0.21)	(-0.07)
Household income (log)	0.006	-0.009	0.011	0.015
Household medile (log)	(0.55)	(-1.03)	(1.44)	(1.84)
Housing tenure (ref: Owned with	(0.55)	(1.05)	(1.44)	(1.04)
Social housing		-0.007	_0 108***	-0.174***
Social nousing	(-0.62)	(-0.14)	(-6.61)	(-5.48)
Rented privately	0.129**	0.086*	_0.137***	_0 123***
Reflice privatery	(2.87)	(1.97)	(-5.82)	(-4.91)
Owned outright	_0.023	-0.054**	(-5.62)	_0.010
Owned Outright	(-1.20)	(-2.73)	(-1.18)	(-1.27)
	(-1.20)	(-2.13)	(-1.10)	(-1.27)

Table 7.4	Fixed-effects	model	of neighb	ourhood o	quality

continued					
	Mode Eng	el 7.1 land	Mode Germ	l 7.2 any	
	Women	Men	Women	Men	
Relocation type (ref: Stayer)					
Stable single (\leq 35 years)	0.096	0.003	0.052	-0.001	
	(1.49)	(0.04)	(1.74)	(-0.03)	
Stable single (\geq 36 years)	0.016	0.016	0.005	-0.028	
	(0.32)	(0.25)	(0.20)	(-0.83)	
Stable couple	-0.001	0.042**	0.060***	0.062***	
	(-0.07)	(2.87)	(5.48)	(5.43)	
Single to couple	-0.015	0.050	0.056	0.035	
	(-0.31)	(0.72)	(1.73)	(1.04)	
Dissolved couple	-0.009	-0.103	-0.016	0.019	
	(-0.17)	(-1.84)	(-0.52)	(0.55)	
Other	-0.009	0.008	0.052^{*}	0.077^{**}	
	(-0.23)	(0.27)	(2.26)	(3.22)	
Constant	-0.234	-0.123	-0.505^{**}	-0.280	
	(-0.93)	(-0.42)	(-2.95)	(-0.94)	
Observations	21,898	19,033	70,824	62,646	
Individuals	3,471	3,052	11,854	10,618	
With change in outcome	1,266	1,132	3,789	3,374	
LL Null-model	1,813.42	1,658.02	-6,383.79	-3,878.76	
LL	2,892.40	2,533.15	-2,996.64	-811.10	
within R ²	0.09	0.09	0.09	0.09	

Data: BHPS 10-18 (individual level, unweighed), IMD 2004 & 2007, SOEP v26 waves 17-25 (individual level, unweighed), Microm 2010

Note: Linear fixed-effects regression model, standardised dependent variable: neighbourhood quality, unstandardised coefficients, t statistics in parentheses; *** significant at 0.1% two-tailed, ** significant at 1%, * significant at 5%; models also include covariates period, region, age not shown here.

I test coefficients for these variables together — namely single to couple relocations and cohabiting; single to couple relocations and married; dissolved couple relocations and cohabiting*-1; dissolved couple relocations and married*-1. For English women, none of the combined effects is significantly different from 0.¹⁶ English women do not change neighbourhood quality in case of partnership changes that are linked to residential relocations. For English men, a separation of a married couple associated with a relocation significantly reduces neighbourhood quality for

¹⁶ Single to couple and cohabiting: F(1, 3470) = 0.01, p = 0.916; single to couple and married: F(1, 3470) = 0.75, p = 0.386; dissolved couple and cohabiting*-1: F(1, 3470) = 0.25, p = 0.616; dissolved couple and married*-1: F(1, 3470) = 1.63, p = 0.202.

the ex-partner. Other combined effects of relocations and partnership changes are not significantly associated with changes in neighbourhood quality.¹⁷

For German women, all of these combinations are significant.¹⁸ Relocations associated with forming a new couple increase neighbourhood quality, while relocations associated with dissolution of a couple reduce neighbourhood quality on average both for cohabiting and married couples. For German men, none of these combinations is significant.¹⁹ Thus, German men do not experience a significant change in neighbourhood quality, if they relocate and change their partnership status. Considering all this evidence, I argue that the first part of H 1.4 cannot be rejected with regard to neighbourhood quality. Similar results have been found in the US (Crowder and South 2005). Also, German women seem to be affected by union dissolution and associated relocations more negatively than men. This supports the second part of H 1.4 about gender differences for Germany, but not for England where men seem to be affected more strongly than women. Finally, I do not find evidence that the effect of union dissolutions is persisting over the life course for most groups. In a fixed-effects regression in which I compare years in which individuals are not divorced to years in which individuals are divorced, I only find English men to be negatively affected by divorces.²⁰

I hypothesised that individuals with children will live in better neighbourhoods than childless individuals on average and that the positive effect of children on neighbourhood quality will increase with income (H 1.3). I find the presence of children aged 1 to 9 to positively affect neighbourhood quality except for English women. On average, English men as well as German women and men live in better quality neighbourhoods in years in which they live together with young children. Older children aged 10 to 17 are not associated with changes in neighbourhood quality for any subsample. The family events of childbirth and a child leaving the household have mostly no effect on neighbourhood quality. Only English men significantly increase their neighbourhood quality in years in which a child left the household. Taken together, the evidence for the first part of H 1.3 is ambiguous. The presence of young children seems to be related to increases in neighbourhood quality, but childbirths are not associated with immediate improvements in neighbourhood quality. Older children do not affect changes in neighbourhood quality.

¹⁷ Single to couple and cohabiting: F(1, 3051) = 0.42, p = 0.519; single to couple and married: F(1, 3051) = 1.96, p = 0.162; dissolved couple and cohabiting*-1: F(1, 3051) = 2.69, p = 0.101; dissolved couple and married*-1: F(1, 3051) = 6.74, p = 0.010.

¹⁸ Single to couple and cohabiting: F(1, 10617) = 14.69, p = 0.000; single to couple and married: F(1, 10617) = 13.77, p = 0.000; dissolved couple and cohabiting*-1: F(1, 10617) = 8.39, p = 0.004; dissolved couple and married*-1: F(1, 10617) = 7.72, p = 0.006.

¹⁹ Single to couple and cohabiting: F(1, 10617) = 3.11, p = 0.078; single to couple and married: F(1, 10617) = 3.50, p = 0.062; dissolved couple and cohabiting*-1: F(1, 10617) = 0.23, p = 0.633; dissolved couple and married*-1: F(1, 10617) = 0.33, p = 0.568.

²⁰ English women: b = -0.030 (t = -0.55); English men: b = -0.142 (t = -2.83); German women: b = -0.042 (t = -1.06); German men: b = -0.018 (t = -0.43).

Sampson and Sharkey (2008) also find a positive association between household size and neighbourhood quality. Other research has shown that childbirth improves neighbourhood quality on average (Rabe and Taylor 2010). Part of the effect of childbirth may be captured by other covariates in my model, e.g. relocations of stable couples or tenure changes.

The second part of H 1.3 stated that the effect of children on neighbourhood quality will increase with income. While all parents may have preferences for highquality neighbourhoods to raise their children, only those with sufficient income may be able to meet these preferences. I test this part of the hypothesis by interacting income and the number of children in models reported in Table 7.5. The main effect of at least one child aged 1 to 9 in the household indicates the average change in neighbourhood quality if an individual starts to live with a young child and has no household income. This main effect is only significantly different from 0 for English women, which indicates that English women reduce their neighbourhood quality on average if they live together with a young child in the hypothetical case of having no household income. Accordingly, the main effect of at least one child aged 10 to 17 shows the average change in neighbourhood quality, if an older child starts to live in the household. This main effect shows that English women improve their neighbourhood quality on average if they have no income and start to live together with an older child. German women reduce their neighbourhood quality on average, if they have no income and start living with an older child.

	Mode Engl	Model 7.3 England		7.4 any
	Women	Men	Women	Men
Child aged				
1 to 9	-0.044^{*}	0.040	0.047	0.050
	(-2.16)	(1.39)	(1.38)	(1.33)
10 to 17	0.263*	0.081	-0.390***	-0.174
	(2.00)	(0.54)	(-3.52)	(-1.45)
Household income (log)	0.009	-0.008	0.003	0.012
	(0.82)	(-0.86)	(0.35)	(1.44)
1 to 9*income	0.007**	-0.000	-0.002	-0.002
	(2.58)	(-0.01)	(-0.38)	(-0.44)
10 to 17*income	-0.035^{*}	-0.011	0.050***	0.023
	(-2.00)	(-0.53)	(3.41)	(1.43)

 Table 7.5
 Fixed-effects model of neighbourhood quality with interactions of income and number of children

Data: BHPS 10-18 (individual level, unweighed), IMD 2004 & 2007, SOEP v26 waves 17-25 (individual level, unweighed), Microm 2010

Note: Linear fixed-effects regression model, standardised dependent variable: neighbourhood quality, unstandardised coefficients, t statistics in parentheses; *** significant at 0.1% two-tailed, ** significant at 1%, * significant at 5%; models also include all covariates shown in Table 7.4.

7.2 Multivariate Results

The main effect for household income shows no significant effect of changes in income on changes in neighbourhood quality for individuals without children in the household. For English women, I find both interactions' effects to be statistically significant. With increasing income, the presence of young children buffers neighbourhood quality losses for English women. On the contrary, the presence of older children impedes neighbourhood quality improvements with increasing income for English women. In Germany, the presence of older children reduces neighbourhood quality less with increasing income. Income and children in the household are not significantly interacted for men in England and Germany. Thus, I do not find clear evidence for the second part of H 1.3. For women, income moderates the effect of children in the household, but not always in the expected way. As predicted, women in England improve their neighbourhood quality more if they have young children and higher incomes. The same is true for German women with older children. Unexpectedly, the positive effect of older children on neighbourhood quality in England decreases with higher incomes. This may be due to the fact that women with higher incomes are more likely to relocate from less densely populated areas to more densely populated areas once their children are older.

I expect that transitions into unemployment and demotions reduce neighbourhood quality on average, while individuals that become employed or experience a promotion are more likely to improve their neighbourhood quality than individuals that do not experience the respective job mobility (H 1.5). Contrary to my expectations, I find that promotions, demotions and employment status do not have significant effects on changes in neighbourhood quality. Transitions in the job trajectory do not affect changes in neighbourhood quality. Part of the unemployment effect on neighbourhood quality may be captured by the household income, but excluding income from the models does not change the effect size for the unemployment variable (not shown here).

My findings are in contrast to earlier results by Rabe and Taylor (ibid.) which show that the event of unemployment for male partners reduces objective and subjective neighbourhood quality significantly for couples that relocate in England, and by Bolt and Kempen (2010) who find the unemployed to be more likely to relocate to segregated areas in the Netherlands. My different findings may be due to the fixed-effects model specification which only estimates effects for deviations from the individual-specific mean. For example, Sampson and Sharkey (2008) find no effect of employment changes or changes in job prestige on neighbourhood quality in the US in accordance with my findings.

7.2.2 Economic Resources and Housing Tenure

H 2.1 a) states that individuals with positive changes in income are likely to improve the quality of their neighbourhood by relocating to better areas. In contrast to H 2.1

a), I do not find a significant association between changes in income and changes in neighbourhood quality for any subsample. In years in which individuals have a higher income than in other years, their neighbourhood quality is not higher on average. It is puzzling that the effect of income on neighbourhood quality is so small and insignificant, as neighbourhood quality is expected to be positively associated with land rent, which affects rents and property prices. Higher rents and property prices may only be affordable with higher income.

Even though my findings seem to be counter-intuitive, they are consistent with past research. While the level of income has been found to be positively associated with improvements in neigbourhood quality (South and Crowder 1997a; Crowder and South 2005; Clark, Deurloo and Dieleman 2006), changes in income are only weakly associated with changes in neighbourhood quality (South and Crowder 1997a; Sampson and Sharkey 2008). Potential reasons for the insignificant income effect may be that, first, individuals want to be sure that the change in income is permanent, before they relocate and burden themselves with the associated costs. Second, individuals may not depend on increases in household income, since they can access mortgages to relocate to better neighbourhoods.²¹ Third, income may only be important for certain quintiles of the income distribution, e.g. only high-income individuals may live in significantly better quality neighbourhoods. Fourth, income may not be necessary to relocate to better neighbourhoods for the reason that neighbourhood quality is actually not positively associated with the price of housing in the area.

Table 7.6 presents alternative model specifications which test for these potential explanations. In Model 7.5 and Model 7.6, I present results from a pooled cross-sectional OLS regression with clustered standard errors. Clearly, individuals with higher income live in better quality neighbourhoods. The earlier fixed-effects regression results showed that the *same* individual is not likely to be in a higher quality neighbourhood, if she or he has a higher income compared to years with lower income. The pooled cross-sectional model shows that an individual is in a higher quality neighbourhood compared to *other* (and the same) individuals with lower income. This is evidence that neighbourhood quality is positively associated with the price of housing in the area.

The association of neighbourhood quality and the price of housing in an area is directly supported by results from Model 7.11 and Model 7.12 based again on fixed-effects regressions, which show that the same individual lives in higher quality neighbourhoods in years in which she or he has higher housing costs. The variable housing costs measures mortgage and rent payments. The positive effect of housing costs on neighbourhood quality shows that individuals that pay mortgage may not be as dependent on their current incomes to increase neighbourhood quality as nonmortgage holders. Individuals that anticipate income changes at some point in the

²¹ Individuals may also be wealthy and use their own capital to purchase homes, but wealth is not regularly measured in the SOEP and BHPS and I do not test this explanation.

future may already increase the resources they spend on housing with a mortgage. Depending on access to mortgages in a housing market, individuals may not even expect an increase in their income to take up a mortgage. Housing costs and neighbourhood quality may be endogenous, though, and both explained by unobserved variables such as the presence of high quality dwellings.

Model 7.7 and Model 7.8 are also estimated using fixed-effects regressions. In these models, the income variable is averaged over a three year time window. Hence, the variable measures more lasting income changes than the original income variable. The models show that more lasting income changes are significantly associated with changes in neighbourhood quality for all subsamples except English men. If the average household income over a three year time window increases, neighbourhood quality improves on average. For example, English women live in about 0.03 standard deviations better neighbourhoods, if their log household income averaged over three years increases by 1 unit. This is evidence that not immediate income changes, but more lasting income changes of individuals trigger relocations to better neighbourhoods on average.

Finally, I test for changes in neighbourhood quality by income quintiles in Model 7.9 and Model 7.10. In England, I do not find individuals that change between income quintiles to change their neighbourhood quality significantly different from those staying in the third income quintile. In Germany, I find individuals that enter the lowest income quintile to reduce their neighbourhood quality significantly compared to those staying in the third quintile. Other income groups are not associated with changes in neighbourhood quality. Thus, in Germany immediate income changes seem only to affect neighbourhood quality for the lowest income group. Individuals that change between more affluent income groups do not change their neighbourhood quality on average.

I expect that individuals improve their neighbourhood quality more through relocations if they have higher incomes compared to the same individuals with lower incomes (H 2.1 b)). To test this expectation, I interact income with the various residential relocation types to see whether the returns to relocation vary by income (cf. Table A.9 in the appendix). I only find a significant interaction effect of income and stable couple relocations in Germany. German couples with higher incomes improve their neighbourhood quality more than low-income couples. The impact of other relocation types does not vary by income in Germany. In England, the effects of all types of relocations are independent from changes in income. The relationship between income and neighbourhood quality for stable couple relocations is plotted in Figure 7.2 for women in both countries.²² For example, at an equalised monthly household income of about PPP-\$ 2,500 the marginal effect of a stable couple relocation on neighbourhood quality is about 0.09 in Germany. This marginal effect is

²² For English men, the marginal effects are slightly higher than for English women (not shown here). The marginal effects for German men are almost identical to the effects for German women.

	England		Germa	any
	Women	Men	Women	Men
	Model	7.5 ^a	Model	7.6 ^a
Household income (log)	0.203***	0.190***	0.296***	0.314***
	(9.27)	(8.38)	(17.61)	(18.05)
	Mode	17.7	Model	7.8
Average household income	0.029*	-0.001	0.041**	0.038*
(3 years, log)	(1.97)	(-0.07)	(2.69)	(2.49)
	Mode	1 7.9	Model	7.10
Household income (ref: 3rd quin	tile)			
1 st quintile (lowest)	0.001	0.004	-0.016^{*}	-0.020^{*}
	(0.08)	(0.33)	(-2.05)	(-2.51)
2 nd quintile	-0.003	0.005	-0.004	-0.009
	(-0.38)	(0.61)	(-0.78)	(-1.69)
4 th quintile	0.006	0.005	-0.004	0.001
	(0.70)	(0.66)	(-0.79)	(0.11)
5 th quintile (highest)	0.019	-0.000	0.003	-0.004
	(1.64)	(-0.03)	(0.32)	(-0.44)
	Model	7.11	Model	7.12
Housing costs	0.003*	0.006**	0.006***	0.005***
	(2.09)	(3.17)	(4.05)	(3.75)

Table 7.6 Fixed-effects models of neighbourhood quality with alternative income specifications

Data: BHPS 10-18 (individual level, unweighed), IMD 2004 & 2007, SOEP v26 waves 17-25 (individual level, unweighed), Microm 2010

Note: Linear fixed-effects regression model expect ^a cross-sectional OLS, standardised dependent variable: neighbourhood quality, unstandardised coefficients, t statistics in parentheses; *** significant at 0.1% two-tailed, ** significant at 1%, * significant at 5%; models also include all covariates shown in Table 7.4 except household income.

statistically significant, which is indicated by the confidence interval that does not include 0.

From Figure 7.2 it is clear that the effect of stable couple relocations on neighbourhood quality increases with income only in Germany. In England no marginal effect is significantly different from zero.²³ With each additional PPP-\$ of income, couples in Germany improve their neighbourhood quality more than couples with less income in case of relocation. Only those with an equalised monthly household income above about PPP-\$ 1,100 improve their neighbourhood quality significantly. Couples with a household income below about PPP-\$ 1,100 do not change

 $^{^{23}}$ For English men, marginal effects above an income of about PPP-\$ 1,250 are significantly different from 0, but the overall interaction term is not significant.



Data: BHPS waves 10-18 (individual level, unweighted), IMD 2004 & 2007, SOEP v26 waves 17-25 (individual level, unweighted), Microm 2010

Note: Based on linear fixed-effects regression models shown in Table A.9, standardised dependent variable: neighbourhood quality; Thin lines indicate 95 %-confidence interval of marginal effects.



their neighbourhood quality significantly. Low-income couples do not change their neighbourhood quality with residential relocations in Germany. Only for those couples with medium and high household incomes relocations improve neighbourhood quality. This is evidence for the stratification of neighbourhood quality after relocation through the market mechanism in support of H 2.1 b) for Germany. Thus, relocations may widen the disparities in neighbourhood quality for different income groups. At least for stable couples in Germany, the inequality of neighbourhood quality across income groups apparent in the descriptive analysis is widened through relocations (cf. Table 7.3).

In H 2.2, I expect that individuals that relocate to social housing improve their neighbourhood quality the least, relocations to private tenancy moderately improve neighbourhoods and relocations into ownership improve neighbourhood quality the strongest on average. The expectation is confirmed by my model for Germany (cf. Table 7.4). Individuals that own their homes live in significantly better neighbour-



Data: BHPS waves 10-18 (individual level, unweighted), IMD 2004 & 2007, SOEP v26 waves 17-25 (individual level, unweighted), Microm 2010

Note: Based on linear OLS regression models with dependent variable change in neighbourhood quality and all covariates shown in Table 7.4; whiskers indicate 95 %-confidence interval; only movers.

Fig. 7.3 Predicted changes in neighbourhood quality by tenure of origin and destination after relocation, women

hoods compared to the same individuals living in other housing tenures. Private tenants live in significantly better neighbourhoods than social tenants.²⁴ The difference between owning with or without a mortgage is not significant. The housing tenure has the second largest effect on neighbourhood quality after the degree of urbanisation in Model 7.2. For England, H 2.2 has to be rejected based on the fixed-effects regression results. I do not find owners with mortgages to live in significantly different quality neighbourhoods than outright owners or social tenants, and private tenants live in better neighbourhoods than owners on average. Female private tenants live in neighbourhoods with about 0.13 standard deviations (0.09 standard deviations for men) higher quality compared to owners with mortgages in England. Female social tenants live in significantly worse neighbourhoods than private tenants in England,

²⁴ German women: F(1, 11853) = 11.72, p = 0.001; German men: F(1, 10617) = 7.10, p = 0.008.

but men have a similar neighbourhood quality whether they live in social housing or privately rented accommodations.²⁵ The findings regarding private tenants in England contradict my expectations and my descriptive findings. It has to be noted that the estimated effect is only based on those relocating into or out of dwellings rented privately. As only 7 per cent of the English sample live in the private rent sector, this group may be very selective (cf. Table A.4).

In Figure 7.3, I show predicted changes in neighbourhood quality for women in case of relocation by tenure of origin and tenure of destination to further test my hypothesis.²⁶ The predicted values are based on OLS regression models in which change in neighbourhood quality between t-1 and t is the dependent variable.²⁷ The findings presented in Figure 7.3 partly support H 2.2. Individuals that relocate to social housing improve their neighbourhood quality the least on average, individuals that relocate into privatley rented dwellings improve their neighbourhood quality more and individuals that relocate into dwellings owned with mortgages improve their quality the most in both countries. Relocations into dwellings owned outright improve neighbourhood quality less than relocations into dwellings owned with mortgages in England, but improve neighbourhood quality more in Germany. The figure also partly provides evidence for H 2.3, as relocations between privately rented dwellings only improve neighbourhood quality in Germany but not in England. Further, relocations from privately rented accommodation to owner dwellings improve neighbourhood quality in Germany, but not in England. These cross-country differences are further discussed in Section 9.1.2.

7.2.3 Situational Conditions of Decision Making

Based on my theoretical considerations, I assume that relocations' outcomes may be conditioned by the information available to movers.²⁸ First, information about the new neighbourhood may be too costly to obtain. The farther the new neighbourhood is away, the more expensive information gathering is likely to be. Therefore, H 3.1 states that individuals that relocate long distances will improve their neighbourhood quality less than individuals relocating short distances. On the grounds that neighbourhoods are complex entities, it may be harder to obtain accurate information about a neighbourhood than about the available rooms in a dwelling. Thus, I additionally assume in H 3.1 that the effect of relocation distances is stronger regarding

²⁵ English women: F(1, 3470) = 8.04, p = 0.005; English men: F(1, 3051) = 3.16, p = 0.075.

²⁶ The predicted neighbourhood quality for men is very similar.

²⁷ All independent variables except relocation types shown in Table 7.4 are included. In addition the lagged housing tenure from t - 1 is added to the model.

 $^{^{28}}$ In this and subsequent sections of the chapter, a general relocation variable is used instead of the differentiated types of relocations.

neighbourhood quality than regarding room stress. Second, if individuals have to relocate involuntary, they may not have the necessary time to consider several alternative locations and relocate to any available dwelling. In these cases, I expect individuals to be more likely to reduce their neighbourhood quality (cf. H 3.2).

I test for these hypotheses in models that include the distance of voluntary relocations and indicators for involuntary relocations (cf. Table 7.7).^{29,30} Voluntary relocations in England have no significant effect on women's changes in neighbourhood quality irrespective of relocation distance compared to stayers. For English men, only relocations that are between 1 and 5 km significantly increase neighbourhood quality by about 0.05 standard deviations. Other voluntary relocations do not affect changes in neighbourhood quality. The effects of different relocation distances are also not significantly different from each other for English women and men. In Germany, voluntary relocations of women improve neighbourhood quality if they are between 1 and 29 km. Relocations shorter than 1 km or farther than 29 km do not change neighbourhood quality compared to staying put. For German men, all relocations that are shorter than 30 km improve neighbourhood quality.

These findings refute H 3.1 for England. The effects of relocations on neighbourhood quality for men are only weakly dependent on the distance of relocations. In general, short-distance relocations as well as long-distance relocations do not improve neighbourhood quality. For Germany, I cannot reject H 3.1 so clearly. The distance of relocations matters for changes in neighbourhood quality. Medium-distance relocations do not improve quality of neighbourhoods on average, while long-distance relocations do not improve neighbourhood quality in Germany, which is in accordance with my expectations. Very short relocations hardly change neighbourhood quality, which may be explained by the similarity of neighbourhoods that are close to each other. H 3.1 also states that the effect of distance on changes in the quality of locations should be stronger with regard to neighbourhood quality than room stress. My findings support this expectation for Germany. Longer relocation distances seem to increase the costs of information regarding neighbourhoods, but not so much regarding the number of rooms in a dwelling (cf. Section 6.2.3).

I also test for the influence of involuntary relocations on neighbourhood quality. To capture involuntary relocations, I include subjective reasons for relocating into the regression models (cf. Table 7.7). I consider the subjective reasons 'separation', 'notice given' and 'housing costs too high'. Involuntary relocations do not change neighbourhood quality significantly compared to staying put or compared to relocating voluntarily for women and men in England. Individuals who responded that their relocations were due to separation, termination of their rent contracts or

²⁹ The relocation distance is not available for Germany for the year 2000.

³⁰ The inclusion of the information covariates significantly increase the model fit compared to a model only considering a simple relocation measure (LR test for English women: $\chi^2(7) = 22.55$, p = 0.001; English men: $\chi^2(7) = 19.79$, p = 0.003; German women: $\chi^2(7) = 133.47$, p = 0.000; German men: $\chi^2(7) = 46.31$, p = 0.000).

7.2 Multivariate Results

	Mode Eng	Model 7.13 England		7.14 any
	Women	Men	Women	Men
Voluntary relocation				
< 1 km	-0.012 (-0.58)	0.030 (1.30)	0.018 (1.34)	0.032^{*} (2.30)
1 < 5 km	0.015 (0.67)	0.047* (2.08)	0.069*** (3.81)	0.076*** (3.75)
5 < 30 km	0.040	0.038 (1.43)	0.105***	0.066**
30+ km	-0.002 (-0.04)	-0.004 (-0.10)	0.051 (1.74)	(0.009) (0.29)
Involuntary relocation	(0.0.1)	(0.10)	(11)	(0.22)
Separation	-0.049 (-0.70)	-0.108 (-1.54)	-0.086^{**} (-2.66)	-0.028 (-0.75)
Notice given	(-0.022) (-0.35)	-0.007 (-0.11)	-0.084 (-1.48)	-0.061 (-1.01)
Housing costs	-0.067 (-1.10)	(-0.009) (-0.18)	(-0.018) (-0.63)	(-0.011) (-0.33)
Observations	21,879	18,996	65,280	57,725
Individuals	3,471	3,050	11,823	10,590
With change in outcome	1,259	1,119	3,393	3,009
LL INUII-MODEI	1,845.05	1,/48.01	-2,141.93	- /9/.88
within R ²	2,916.71 0.09	2,391.45 0.08	404.83 0.09	0.10

 Table 7.7 Fixed-effects models of neighbourhood quality for voluntary relocations by distance and for involuntary relocations

Data: BHPS waves 10-18 (individual level, unweighted), IMD 2004 & 2007, SOEP v26 waves 17-25 (individual level, unweighted), Microm 2010

Note: Linear fixed-effects regression model, standardised dependent variable: neighbourhood quality; unstandardised coefficients, t statistics in parentheses; *** significant at 0.1% two-tailed, ** significant at 1%, * significant at 5%; models also include all covariates presented in Table 7.4 except relocation variables.

housing costs which are too high are not more likely to reduce their neighbourhood quality than other individuals. This rejects the expectation that individuals may relocate to worse neighbourhoods if they do not have a long search period before their relocation compared to voluntary relocations or stayers.

In Germany, only women who responded that their relocations were motivated by a separation reduce their neighbourhood quality compared to women staying put. Compared with voluntary relocations, all types of involuntary relocations reduce neighbourhood quality more for German women.³¹ German men do not change their neighbourhood quality significantly compared to stayers or voluntary movers. Thus, even if individuals relocate involuntary, they are not more likely to reduce their neighbourhood quality on average. H 3.2 can be rejected for both countries. Only German women seem to be affected by involuntary relocations — at least to the degree that they improve their neighbourhood quality less than if they would have relocated voluntarily. This particularity indicates that women in Germany are more vulnerable to forced relocations than men or women in England. However, significant effects for German women which are not significant for the other subsamples may also be due to the considerably larger sample of German women compared with German men and English women and men. All in all, there is only moderate evidence that information about potential neighbourhoods of destination is relevant to predict variation in neighbourhood quality changes.

7.2.4 Immigrant Households and Neighbourhood Quality

I hypothesise that individuals in immigrant households improve the quality of their neighbourhoods through relocations less than individuals in native households (H 4.4). I test for the effect of immigrant status on neighbourhood quality by interacting the relocation event with the immigrant status, thereby allowing the event to have heterogeneous effects on the respective groups. Due to the smaller sample compared to the analysis of changes in room stress, I do not differentiate first and second generation immigrant households in the present analysis. H 4.4 is supported if I find a negative interaction effect of immigrant status and relocation. This would indicate that individuals in immigrant households improve their neighbourhood quality less through relocation than individuals in native households.

Table 7.8 shows estimation results from a fixed-effects regression testing for the effect of immigrant status on neighbourhood quality outcomes in England and Germany. The main effect of household income shows the effect of income changes on changes in neighbourhood quality for individuals in native households. Income changes are not associated with neighbourhood quality changes in any subsample. This finding is similar to the results from the general model presented earlier in Section 7.2.2 due to the high share of natives in the samples. The interaction term between household income and immigrant status indicates no significant interactions. The effect of income for individuals in native households is not statistically different from the effect for individuals in native households. This is in contrast to findings regarding room stress (cf. Section 6.2.4). There it was shown that some

 $^{^{31}}$ I tested for each type of voluntary relocation, if the coefficient is significantly different from all distance categories of voluntary relocations. Test for separation: F(4, 11822) = 6.73, p = 0.000; for notice given: F(4, 11822) = 4.96, p = 0.001; for housing costs: F(4, 11822) = 4.96, p = 0.001.

individuals in immigrant households reduce their room stress less than individuals in native households with increasing income. However, if the models are run separately by immigrant status, I find that changes in income significantly and positively affect changes in neighbourhood quality for individuals in immigrant households in Germany (not shown here).³² I do not find differences in England when estimating models separately by immigrant status.

The main effect for residential relocations indicates the average change in neighbourhood quality for individuals in native households in case of relocations. Again, the estimated coefficients are quite similar to the ones presented before. Men in native households in England, and women and men in native households in Germany all increase their neighbourhood quality on average, if they relocate. Women in native households in England do not change their neighbourhood quality significantly if they relocate. With the exception of English women, all individuals in immigrant households improve their neighbourhood quality less after relocations than individuals in native households, but the differences by immigrant status are not significant. Relocations affect individuals in immigrant and native households similarly in both countries. These findings reject H 4.4. With regard to neighbourhood quality, immigrant households do not seem to be discriminated against in the housing market if they relocate to a new neighbourhood.³³ My findings are in contrast to results from the US that consistently show a race gap in changes in neighbourhood quality where especially African Americans are improving their neighbourhood quality less than other ethnic groups (Crowder and South 2005; South, Crowder and Chavez 2005; Sampson and Sharkey 2008). This may indicate that discrimination and structural inequalities between ethnic groups are stronger in the US than in England and Germany.

The findings should be treated with caution, though. Individuals in immigrant households live in worse neighbourhoods on average in both countries (cf. Table 7.3). This may mainly be due to systematic differences between immigrant and native households. For example, about 37 per cent of immigrant households in Germany reside in metropolitan areas, compared to only 30 per cent of native households.³⁴ Part of the differences between immigrant and native households.³⁴ Part of the differences between immigrant and native households may be driven by the former more often living in metropolitan areas with lower average neighbourhood quality. However, the apparent disparities may also be caused by past discrimination and as long as immigrant households do not improve their neighbourhood quality *more* than natives, these disparities will not diminish nor will the effects of previous discrimination become evident in longitudinal regression models.

³² Estimated coefficients of household income (log) for women in immigrant households: 0.042 (t = 2.44); women in native households: -0.001 (t=-0.09); men in immigrant households: 0.051 (t=-0.09); men in immigrant households: 0.051

⁽t = 2.89); men in native households: 0.007 (t = 0.73).

³³ The estimated effects of relocations are similar, if the models are estimated separately by immigrant status (not shown here).

 $^{^{34}}$ Test of interdependence of immigrant status and degree of urbanisation: $\chi^2(2)=401.51,\,p=0.000$

In addition, the number of relocations among individuals in immigrant households is relatively small.³⁵ At the same time the group of immigrants is very heterogeneous. Thus, a range of effects for subgroups of immigrant households may make the estimation of an average effect across all subgroups difficult, i.e. the average effect may hide strong effects for specific immigrant groups.³⁶ For example, Lersch (2013) finds Turkish immigrants in Germany to improve their neighbourhood quality less than German natives and other immigrants when relocating. Finally, the hypothesis relies on the assumption that immigrants and natives have similar preferences. Preferences may be heterogeneous and immigrants may prefer to live close to co-ethnics. Future research should investigate potential differences in preferences. Thus, the results that I present are restricted in their degree of generalisability due to the limitations of the data with regard to immigrant status. I reject H 4.4 based on the presented results. Further research is necessary to validate this.

	Mode Eng	el 7.15 land	Model Germ	7.16 any
	Women	Men	Women	Men
Household income (log)	0.007	-0.007	0.010	0.015
	(0.65)	(-0.76)	(1.40)	(1.80)
Income*immigrant	-0.006	-0.009	0.005	-0.002
-	(-1.25)	(-1.27)	(0.68)	(-0.22)
Relocation	0.004	0.029*	0.047***	0.050***
	(0.30)	(2.19)	(5.09)	(4.95)
Relocation*immigrant	0.012	-0.012	-0.008	-0.014
-	(0.36)	(-0.35)	(-0.44)	(-0.72)

Data: BHPS 10-18 (individual level, unweighted), IMD 2004 & 2007, SOEP v26 waves 17-25 (individual level, unweighted), Microm 2010

Note: Linear fixed-effects regression model, standardised dependent variable: neighbourhood quality; unstandardised coefficients, t statistics in parentheses; *** significant at 0.1% two-tailed, ** significant at 1%, * significant at 5%; models also include all covariates presented in Table 7.4.

³⁵ In Germany, 1,118 women and 1,017 men in immigrant households relocated at least once in my observation window. In England, only 217 women and 209 men in immigrant households relocated at least once.

³⁶ I do not further differentiate the very heterogeneous group of immigrants, due to the small sample size of immigrants in the data. A differentiation by first and second generation immigrants, by visible or invisible ethnic differences, or by ethnic origin may provide further insights in differences within the group of immigrants.

7.2.5 Structural Conditions in Housing Markets

H 4.1 postulates that individuals are less likely to improve their neighbourhood quality if they are in housing markets with high demand for dwellings and individuals are more likely to improve their neighbourhood quality if they are in housing markets with high supplies of dwellings. In H 4.2, I expect that individuals are more likely to improve their neighbourhood quality if they are in housing markets with many high-quality locations. I use panel regression models to test these hypotheses, then I turn to multiple membership multilevel regression models to control for the clustering in various housing markets at the end of the section (cf. Section 4.6.2 for more details on these models).

Before testing the hypotheses, I discuss the findings regarding the degree of urbanisation which has the strongest effect on neighbourhood quality in England and after East Germany the second strongest effect in Germany in the full model presented earlier (cf. Table 7.4). The strong difference in average neighbourhood quality between urban and rural areas is supported elsewhere (Adams 1992; Ham and Feijten 2008; McCulloch 2012). After including the housing market variables, the effect of degree of urbanisation is clearly reduced (cf. Table 7.9). Thus, important differences between metropolitan, urban and rural areas are captured in the additional housing market variables. The inclusion of the share of owned dwellings changes the effect size of metropolitan areas the strongest. The effect of urban areas on neighbourhood quality changes only slightly after inclusion of additional housing market variables.

Controlling for the housing market covariates, relocating from rural to metropolitan areas in Germany on average reduces neighbourhood quality by about 0.19 standard deviations for women, which is not statistically different from 0, and 0.28 standard deviations for men. Germans living in urban areas have a slightly but not significantly higher neighbourhood quality than if they live in rural areas. In England, individuals in metropolitan and urban areas live in lower average neighbourhood qualities than the same individuals in rural areas. For example, English females live in 0.45 standard deviations lower quality if they are in metropolitan areas compared with rural areas. This is initial evidence for the importance of the regional context in neighbourhood quality outcomes.

The share of owned dwellings in the housing market as a proxy for high-quality locations has a strong effect on neighbourhood quality outcomes. With each percentage point increase in owned dwellings in the housing market, the average neighbourhood quality increases by about 0.01 standard deviations for women and men in Germany. In England, the neighbourhood quality improves by about 0.03 standard deviations for women and 0.02 for men.³⁷ Thus, by relocating from a housing market with 40 per cent owned dwellings (minimum in England) to a housing market

³⁷ Differences between England and Germany should not be overstated, as the variable share of owned dwellings is constructed differently in both countries (cf. Section 4.5.2).

with 84 per cent (maximum in England), the average neighbourhood quality of a English women increases by 1.32 standard deviations. The effect is larger in England than in Germany and may indicate that local opportunities in England are more closely associated with the prevalence of ownership in the area, while in Germany areas without a high share of ownership do not necessarily have a low neighbourhood quality. The latter may be due to the overall lower share of owned homes in Germany. H 4.2 is clearly supported for both countries. Living in a housing market with more home ownership is associated with higher average neighbourhood quality. In a similar way, Kearns and Mason (2007) show that the share of owners in a neighbourhood is positively associated with the quality of the neighbourhood.³⁸

Bailey and Livingston (2008) find individuals in tight housing markets to be less likely to leave deprived neighbourhoods. In contrast, I find that a growing population in the housing market over the last five years is positively associated with neighbourhood quality for all subsamples except English men. Thus, a higher demand on the housing market improves neighbourhood quality. This refutes the first part of H 4.1 stating that neighbourhood outcomes are better in housing markets with low demand. One possible explanation for this finding may be reverse causality. Housing markets with high average neighbourhood quality are more attractive for individuals outside the housing market. The population in the housing market grows through inward-relocations and, thus, I find a positive association between the two variables.

The number of newly constructed dwellings in the housing market has no significant impact on neighbourhood quality in all samples except for English women. The second part of H 4.1 must be rejected. This may be due to several factors. First, newly constructed dwellings may not be associated with good neighbourhood quality, since social networks between the new occupants must grow over time. Second, new dwellings within urban and metropolitan areas may not necessarily be constructed in good neighbourhoods, but rather on vacant plots that can be scattered throughout the city. Third, the number of new dwellings may be significant only in interaction with the demand side of the housing market. For example, the number of new dwellings may be important only in a housing market with high demand due to a positive population balance. Therefore, I tested for an interaction effect of the number of new constructions and the population balance in Germany and England. The interaction effects were not significant for women and men in both countries (not shown).

The findings from the fixed-effects panel models and the multiple membership multi-level models (reported in Table A.10 in the appendix) are similar regarding the housing markets variables. The housing market variables' coefficients are almost identical. The standard errors in the multi-level model are estimated to be much smaller. This is due to the larger effective analytic sample in the multi-level

³⁸ The share of owners in a market has also been found to affect transition into ownership positively (Flippen 2010).

	Model 7.17 England		Model Germ	7.18 any
	Women	Men	Women	Men
Urbanisation (ref: rural)				
Metropolitan	-0.446^{***}	-0.471^{***}	-0.193	-0.282^{*}
	(-4.20)	(-4.22)	(-1.64)	(-2.21)
Urban	-0.372***	-0.321***	0.101	0.110
	(-4.37)	(-4.03)	(1.19)	(1.20)
Relocation	0.000	0.023	0.043***	0.046***
	(0.01)	(1.91)	(5.16)	(5.02)
Owned dwellings	0.028***	0.023***	0.006**	0.005^{*}
	(4.39)	(3.98)	(2.71)	(2.26)
New constructions	0.004*	0.001	0.000	0.001
	(2.02)	(0.36)	(0.38)	(0.80)
Population balance	0.006**	0.002	0.007***	0.006***
	(2.85)	(1.03)	(7.38)	(6.84)

Table 7.9 Fixed-effects models of neighbourhood quality with housing market variables

Data: BHPS 10-18 (individual level, unweighted), IMD 2004 & 2007, ONS various years, SOEP v26 waves 17-25 (individual level, unweighted), Microm 2010, INKAR 2010 Note: Linear fixed-effects regression model, standardised dependent variable: neighbourhood quality, unstandardised coefficients, t statistics in parentheses; *** significant at 0.1% two-tailed, ** significant at 1%, * significant at 5%; models also include all covariates presented in Table 7.4.

regression model, because individuals without change are also included in the analysis. This causes the effect of the population balance for men in England to become significant. The interpretation of the effect of the share of owned dwellings and the population balance does not change from the panel regression model. For Germany, the effects of metropolitan and urban areas are significantly different from rural areas in the multi-level model. Individuals in metropolitan areas live in lower average neighbourhood quality and individuals in urban areas live in higher average neighbourhood quality compared to rural areas. In general, taking into account the clustering of individuals in housing markets does not change the interpretation of my data substantially.

The variance estimation for the random intercepts at the level of housing markets indicates that there is considerable variance between markets after controlling for the degree of urbanisation and housing market variables (cf. Table A.10 in the appendix). For example, about 20 per cent ($\approx 0.145/[0.145+0.523+0.052]$) of the unexplained variance of neighbourhood quality for English women in the model is between housing markets. Potential covariates that may reduce the unexplained variance at the level of housing markets could be the vacancy rate, average rent level or share of social housing in the market. Further research should investigate the effects of these variables. The considerable share of unexplained variance between housing markets indicates that it is necessary to control for the clustering of indi-

viduals within housing markets, but even more important is the clustering at the individual level. Around 70 per cent of the unexplained variance in all models are at the individual level and 10 per cent of the variance is specific to individual-year observations.

7.3 Summary

With regard to neighbourhood quality, relocations are mostly horizontal and relocations themselves have only a small effect on changes in neighbourhood quality. Conditions of relocations cause important disparities in neighbourhood quality outcomes. I find that individual life course conditions are relevant to explain changes in neighbourhood quality. In both countries, stable couples are more likely to improve their neighbourhood quality through relocations than any other group of movers with the exception of English women who do not change neighbourhood quality at all if they relocate. There is evidence that relocations that are associated with union dissolution reduce neighbourhood quality. The presence of young children seems to be associated with higher neighbourhood quality, but childbirths are not associated with immediate improvements. The unemployed are in lower average neighbourhood qualities on average, but do not change their neighbourhood quality significantly less than the employed if they relocate. I do not find other transitions in the job trajectory to be important in explaining neighbourhood quality changes. Disparities in neighbourhood quality outcomes after relocations may result from these processes. First, individuals live in worse neighbourhoods in early life and improve their neighbourhoods substantially only when forming partnerships and families in Germany. In England, singles also improve their neighbourhood quality. Individuals at least temporarily reduce their neighbourhood quality after union dissolution. Second, disparities exist between individuals that are employed and unemployed, but these average differences are not traced in particular changes in neighbourhood quality.

All in all, I find strong support for the importance of economic resources, but the link between economic resources and neighbourhood quality is not straightforward. Cross-sectional evidence shows that neighbourhood quality is strongly associated with household income. At the same time, I find immediate changes in income to explain little variation in neighbourhood quality changes. More persisting changes in household income as well as housing costs are associated with changes in neighbourhood quality. I find that relocating stable couple families improve their neighbourhood quality more in Germany if they have more income. I do not find a significant interaction of income and relocations in England. In Germany, a clear stratification of neighbourhood quality by housing tenure is observed with owned dwellings being located in better neighbourhoods on average than rented dwellings. In England, living in owned dwellings is not associated with better neighbourhood quality on average. Hence, disparities in neighbourhood outcomes result from different investment in locations that depend partly on individual economic resources, but also on access to external resources in the form of mortgages. Relocations are increasingly beneficial with growing income for couples in Germany. In Germany, relocations into owned dwellings strongly improve neighbourhood quality. Thus, relocations widen disparities in neighbourhood quality between groups with different access to economic resources.

I find clear evidence for the importance of structural conditions for neighbourhood quality changes at the level of housing markets. The regional degree of urbanisation has a strong effect on average neighbourhood qualities. More densely populated areas are associated with lower average neighbourhood quality. I find the average quality of locations in the housing market to affect average neighbourhood outcomes. Supply and demand of dwellings do not have expected effects on neighbourhood quality. I find only weak evidence that individuals in immigrant households are structurally constrained in their relocations to better neighbourhoods in Germany, but no evidence for constrained behaviour of individuals in immigrant households in England. I conclude that structural conditions cause disparities in relocation outcomes, because they offer different opportunities for movers. Individuals that relocate into rural areas in England or rural and urban areas in Germany are much more likely to improve their neighbourhood quality than individuals relocating into metropolitan areas. Relocations in housing markets with high shares of owned dwellings improve neighbourhood quality more than relocations in markets with low shares of owners.

I find only weak or no evidence for the importance of situational conditions of decision making to explain neighbourhood quality changes. In England, short-distance relocations as well as long-distance relocations do not improve neighbourhood quality on average. For Germany, the distance of relocations matters for changes in neighbourhood quality and short- as well as long-distance relocations do not improve neighbourhood quality while medium-distance relocations improve neighbourhood quality. Involuntary relocations are not found to affect neighbourhood quality significantly with the exception of relocations after union dissolutions for German women.

Chapter 8 Long-Distance Relocations and Subsequent Employment of Dual-Earner Couples

Long-distance relocations are often occupationally motivated and individuals are assumed to relocate to places that offer job opportunities to increase their life-time earnings (e.g. Becker 1995: 53). Individuals in dual-earner couples are constrained in their mobility, because both partners' careers have to be considered in the decision of whether and where to relocate. It is unlikely that both partners will receive equally good job offers at a new location at the same time, because job opportunities are dispersed in geographical space and emerge at relatively random times (Mincer 1978). Therefore, long-distance relocations have divergent effects on subsequent employment of both partners. In this chapter, I examine the potential difference in occupational returns between partners in dual-earner couples by analysing whether both partners maintain employment after long-distance relocations (cf. Section 4.5.1.4 for a detailed description of the dependent variables).

Long-distance relocations are assumed to have a stronger effect on subsequent employment than short-distance relocations, since the distance to the old place of work will increase substantially after long-distance relocations in most cases and individuals usually prefer to limit their commuting (e.g. Boyle, Feng and Gayle 2009). I treat relocations across borders of Local Authority Districts (LADs) in England and relocations across borders of counties in Germany as long-distance relocations (cf. Section 4.5.1.1 for more details). In contrast to earlier chapters, I differentiate West and East Germany in the present analysis to test H 5.3 which states that the negative effect of long-distance relocations on women's careers is largest in West Germany, moderate in East Germany and smallest in England. This hypothesis is motivated by the cross-country differences in gender relations described in Section 4.1.4. I use the term 'contexts' to refer to these three institutional and cultural settings.

The sample for the main part of the present analysis is more restricted than in previous chapters. First, I only consider stable couples in which one partner is the head of the household. In stable couples both partners live together in the same household at t and t + 1. The potential bias resulting from focussing on stable couples is discussed in Section 8.2.3, where I do not find substantially different results controlling for this selectivity. Second, since I focus on dual-earner couples, I only include couples in which each partner works more than 10 and less than 81 hours per week at t.¹ I exclude individuals that work less than 11 hours on average, because I consider their employment as minor for the economic situation of the couple.² I exclude individuals with more than 80 work hours per week, as in the SOEP these cases are treated as implausible and are given missing values. Third, I analyse couples in which both partners are at least 20 years old and not older than 55 years to only include individuals in their prime working age. I exclude self-employed individuals and individuals with a second job, due to problems measuring their work hours and labour income. After all these operations, my sample consists of 2,093 unique English couples which contribute 20,507 individual-year observations, 3,661 West German couples which contribute 27,859 individual-year observations and 1,512 East German couples which contribute 13,374 individual-year observations to the sample.³

In contrast to earlier chapters, overtime changes in individual outcomes of relocations are of minor interest in the present chapter. Instead, my aim is to show which average inter-individual differences in the effects of relocations on subsequent employment exist between women and men. In the following, I first describe the average proportion of partners from dual-earner couples that leave employment after long-distance relocations for different groups of couples in univariate analyses (Section 8.1.1). In the multivariate section, I analyse determinants of leaving employment after relocations in couples using actor-partner interdependence models (APIMs; Section 8.2). I summarise my findings in Section 8.3.

8.1 Descriptive Results

In general, couples are found to be less mobile than singles in the literature and couples are especially unlikely to relocate over long distances (e.g. Clark, Deurloo and Dieleman 2000).⁴ Table 8.1 shows that only about 2 per cent of couples relocate over long distances in England, West and East Germany in a given year on average. Annual mobility rates are significantly lower for dual-earner couples compared to male-breadwinner couples in Germany, but not in England. The term dual-earner

¹ For the descriptive analysis I also include single-earner couples as a reference category.

 $^{^2}$ I test whether further restricting the sample to respondents working at least 30 hours a week affects my results and it does not (Section 8.2.3).

³ I report descriptive statistics for all variables for England in Table A.11, for West Germany in Table A.12, and for East Germany in Table A.13 in the appendix.

⁴ See also Chapter 5.

couples refers to couples in which both partners work between 11 and 80 hours a week. One of the causes for less frequent long-distance relocations among dualearner couples may be the problem of finding adequate new jobs for both partners after relocations. Since dual-earner couples are less mobile, it can be expected that partners in these couples are more likely to be tied stayers than partners in other couples, i.e. they stay even though they would receive positive returns from relocating (Nisic 2010). Couples in England are more likely to relocate long distances on average than couples in Germany. This is especially apparent in the share of couples that experience at least one long-distance relocation in the observation period. Due to the limited observation window, this information gives only a rough indication of expected lifetime mobility. About 13 per cent of all couples in England relocate at least once, compared with only 8 per cent of couples in West Germany and 9 per cent of couples in East Germany.

Group by context	Share of all couples	Relocations each year	Relocate at least once	Number of observed relocations
	in p	per cent of couj	ples	relocations
		E	ngland	
All couples	100.0	2.3	13.2	552
Labour participation				
Male-breadwinner	17.5	2.8	11.8	105
Dual-earner	51.7	2.3	15.2	292
	West Germany			
All couples	100.0	1.6	8.0	596
Labour participation				
Male-breadwinner	30.5	1.7	7.5	188
Dual-earner	38.1	1.3	8.0	231
		East	Germany	
All couples	100.0	1.6	9.2	187
Labour participation				
Male-breadwinner	22.1	2.2	14.4	46
Dual-earner	48.2	1.2	7.2	83

Table 8.1	Mobility	rates and	number	of observe	d long-distanc	e relocations	by type of	f couple
-----------	----------	-----------	--------	------------	----------------	---------------	------------	----------

Data: BHPS waves 1-18, SOEP v26 waves 8-25 (household level, cross-sectional weights, number of observed relocations unweighed)

Note: Only long-distance relocations. Female-breadwinner couples omitted due to small number of cases.

The present analysis deals with dual-earner couples only. Both partners work at least 11 and not more than 80 hours in about 52 per cent of couples in England, 38

per cent of couples in West Germany and 48 per cent of couples in East Germany. The last column in Table 8.1 indicates the small number of long-distance relocations that I observe for dual-earner couples. This is a clear limitation of my analysis. Due to the small case numbers, differences between stayers and movers and differences within the group of movers are difficult to identify. In general, panel data sets such as the BHPS and SOEP offer only limited opportunities to analyse rare events such as long-distance relocations (cf. Section 4.3.1). On the other hand, panel data sets offer rich information on the conditions of these events and allow over-time analyses. To increase the number of observed long-distance relocations and to profit from the benefits of panel data, most of the multivariate analysis is run on pooled samples of English, West and East German couples.

About 40 per cent of long-distance relocations of dual-earner couples in England are job-motivated for at least one partner (Table 8.2). The share of job-motivated long-distance relocations is even higher in West (49 per cent) and East Germany (62 per cent). These shares are much higher than the share of all all occupationally motivated relocations irrespective of distance. For example, only 16 per cent of all relocations of (single- and dual-earner) stable couples in England and 11 per cent in Germany are occupationally motivated. For England, the data allows differentiation between couples which relocate only for the career of one partner or which relocate for the careers of both partners. Only 5 per cent of mover couples relocate for the career of the female partner, while 21 per cent relocate for the male partner's career. In these cases, the other partner can be considered a tied mover, if she or he relocates for the sake of her or his partner's career and negative outcomes for their own careers may be expected (Taylor 2007). Another 14 per cent of dual-earner couples relocate over long distances for both partners' careers in England. Thus, for a high share of long-distance movers, labour market outcomes may be the motivation to relocate in the first place and long-distance movers may be selective in the way that only those individuals relocate that expect positive returns from relocations. This potential selectivity in long-distance movers is further addressed in Section 8.2.3.

8.1.1 Leaving Employment for Movers

Table 8.3 shows the average proportion of individuals leaving employment for women and men for all dual-earner couples by mover status in England, West and East Germany. The term movers only refers to individuals that experience a *longdistance* relocation. The term stayers refers to individuals that do not relocate, or that relocate only within their county or LAD of origin. I test for statistical differences in each category by gender and by mover status. Table 8.3 shows the average rate of individuals leaving employment until t + 1, conditional on being employed at t. Among stayers, women in dual-earner couples are significantly more likely to leave employment until t + 1 and become inactive, unemployed, or enter education

Group by context	Share of job-motivated relocations	Number of couples	
	in per cent		
England			
Total	39.7	105	
Only for woman job motivated	5.0	23	
Only for man job motivated	20.6	59	
For both partners job motivated	14.1	23	
West Germany ^a			
Total	49.4	116	
East Germany ^a			
Total	62.2	22	

Table 8.2 Job-motivated long-distance relocations in dual-earner couples

Data: BHPS waves 1-18, SOEP v26 waves 8-25 (household level, cross-sectional weights, number of observed relocations unweighed)

Note: Only long-distance relocations.^a: In the SOEP, individual motivations for relocations are not observed.

than men. About 4 per cent of English, female stayers leave employment while only 2 per cent of English men do so. In West Germany, 4 per cent of women and 3 per cent of men in dual-earner couples leave employment until t + 1 on average. Leaving employment for women and men is twice as likely in East Germany compared with West Germany. In England and West Germany, long-distance movers are more likely to leave employment, but the differences by mover status are only statistically significant for women. For English women, the proportion of those leaving employment more than triples for movers compared to stayers. About 13 per cent of female long-distance movers are no longer employed after the relocation. This share is significantly higher compared to the share of English, male movers that leave employment. In West Germany, about 10 per cent of female long-distance movers are no longer employed after a relocation. I do not find a significant difference in the proportion leaving employment after relocations between West German women and men.

In East Germany, women are slightly but not significantly more likely to leave employment after long-distance relocations compared to stayers. For men in East Germany, the opposite is true and only 2 per cent of long-distance movers leave employment, while 6 per cent of stayers are not longer employed at t + 1. This may indicate that East Germans are more likely to relocate to maintain employment than West German and English movers which is also apparent in the higher share of East Germans that relocate long distances due to occupational reasons (cf. Table 8.2). Indeed, before a relocation the regional unemployment is twice as high in East Germany (M = 18.3, SD = 0.69, N = 37) compared to West Germany (M = 9.31, SD = 0.29, N = 150). At the same time, East Germans (M = -1.52, SD = 1.67, N = 21) relocate to regions with relatively lower unemployment rates compared to their previous location than West Germans on average (M = -1.29, SD = 0.43, N = 84).⁵ This may indicate that East Germans relocate to better opportunities in the labour market. However, the case numbers for these descriptive statistics are small, because regional unemployment rates are only available after 1998 in Germany.

Gender by context	Proportion leaving employment		
	Stayer	Mover	
England			
Women	0.04*** ###	0.13*	
Men	0.02	0.05	
West Germany			
Women	0.04** #	0.10	
Men	0.03	0.05	
East Germany			
Women	0.08^{*}	0.11 [§]	
Men	0.06	0.02^{\S}	

Table 8.3 Average proportion leaving employment by mover status

Data: BHPS waves 1-18, SOEP v26 waves 8-25 (cross-sectional weights)

Note: Only dual-earner couples. Difference between women and men: *** significant at 0.1% two-tailed, ** significant at 1%,* significant at 5%. Difference between stayer and mover: ### significant at 0.1% two-tailed, ## significant at 1%,# significant at 5%. \S : N <100.

These findings show that long-distance relocations of dual-earner couples seem to be more disruptive for women's careers than for men's careers and the findings are in accordance with past literature (e.g. Taylor 2007). While men continue to work in the same job or change to a new job quickly, e.g. as they relocate after having received a job offer in the new location, a considerable share of women seem to leave employment after long-distance relocations. For women in dual-earner couples, relocations seem to be speculative, i.e. they often do not have a job offer before the relocation. This supports the argument that men are mostly leading spouses, i.e. couples relocate for men's careers, and women are often trailing spouses, i.e. they follow their partners, in dual-earner couples. Women may also initially not work after a long-distance relocation, as according to traditional gender roles women are more likely to be responsible for setting up the new household, e.g. by furnishing the new dwelling and buying home appliances (Belch and Willis 2002).

While women have a higher immediate probability of leaving employment in England and West Germany if they experience a long-distance relocation, they are

⁵ It may be expected that East-to-West relocations contribute to this difference, but I only observe 9 dual-earner couples that relocate from East to West Germany (and 7 couples that relocate from West to East Germany).

not more likely to remain out of employment at the second interview after a relocation took place (Table 8.4). This is in accordance with past literature that shows that the negative effect of long-distance relocations on women's employment is only temporarily (e.g. Clark and Withers 2002). In East Germany, women that experience a long-distance relocation are slightly more likely to have left employment until t + 2than stayers, but the difference is not statistically significant. These findings show that at least for England and West Germany the disruptive effect of long-distance relocations on employment of women is only temporary. While initially being not employed at the new location, women take up employment again soon after. Nevertheless, this short-term interruption may contribute to, inter alia, discontinuous female work histories that are associated with worse long-term outcomes regarding labour market attachment, career advancement and pension rights compared to continuously employed women (Taylor 2007).

Gender by context	Proportion leaving employment until t+2		
·	Stayer	Mover	
England			
Women	0.07***	0.06	
Men	0.03	0.03	
West Germany			
Women	0.06***	0.06	
Men	0.04	0.05	
East Germany			
Women	0.09	0.12^{\S}	
Men	0.08	0.08^{\S}	

Table 8.4	Average proportio	n leaving employment	until t+2 by mover status
-----------	-------------------	----------------------	---------------------------

Data: BHPS waves 1-18, SOEP v26 waves 8-25 (cross-sectional weights)

Note: Only dual-earner couples. Difference between women and men: *** significant at 0.1% two-tailed, ** significant at 1%,* significant at 5%. Difference between stayer and mover: ### significant at 0.1% two-tailed, ## significant at 1%,# significant at 5%. \S : N <100.

8.2 Multivariate Results

I now further analyse leaving employment after relocations using multivariate regression methods to control for third variables that may confound the descriptive results. I present a general APIM of leaving employment in Section 8.2.1 (cf. Section 4.6.3 for a detailed description of APIMs), before testing for differences between England, West and East Germany (Section 8.2.2). Finally, I check the robustness of my findings by addressing potential weaknesses of my modelling strategy (Section 8.2.3). All continuous independent variables in the multivariate analysis are mean-centred.

8.2.1 Gross Effect of Long-Distance Relocations

Table 8.5 shows estimation results for the logistic regression APIM with the dependent variable leaving employment.⁶ Model 8.1 is a general model, in which I estimate a gross effect of long-distance relocations that is constrained to be equal for England, and West and East Germany. For a parsimonious presentation of results, the separate coefficients for women and men are presented in columns 2 and 4, but they are estimated in one pooled model. In column 3 and 5, I additionally report average marginal effects (AMEs; cf. Section 4.6.3). I test for each pair of AMEs, to see if significant gender differences exist. Gender differences are indicated with the number symbol '#'. The substantially interesting independent variable measures the occurrence of long-distance relocations. Including the long-distance relocation variable significantly increases model fit.⁷

The model indicates that women's odds of leaving employment are about 3.3 $(\approx e^{1.200})$ times higher in England, and West and East Germany on average, if they relocate long distances compared to staying. This is significantly different from having no effect at the 99.9 per cent confidence level. The AMEs show that for women with average population characteristics the probability to leave employment is about 3 percentage points higher if they relocate a long distance compared to staying. For women, long-distance relocations have the second largest effect on women's chances to leave employment in the model after having a continuous work history. For men the odds of leaving employment after long-distance relocations are about 1.9 ($\approx e^{0.649}$) times higher compared to stayers and this is significantly different from having no effect. Men's probability to leave employment is about 2 percentage points higher if they relocate a long-distance compared to staying. In comparison to other variables in the model, relocations have only the fifth largest effect on the chances to leave employment for men. The difference between women and men in the effects of long-distance relocations is only marginally significant.⁸ Thus, I find both women and men to have a higher probability to leave employment after long-distance relocations than stayers. The negative effect seems to be stronger for

⁶ I estimate all multilevel logistic regression models in this chapter with 5 integration points, where the number of integration points increases precision but also computation time, which is extensive for the APIMs. I re-estimated the main Model 8.1 using 20 integration points. Results did not differ substantially between the two estimations and AMEs changed at most in the second decimal place.

⁷ LR test: $\chi^2(2) = 44.07$, p = 0.000.

⁸ $\chi^2(1) = 2.89, p = 0.089.$

		Mo Poole	del 8. ed sam	1 ple	
	Wom	nen		Mei	n
	b	AME		b	AME
Age	-0.110^{*}	-0.003^{*}		-0.253***	-0.007^{***}
	(-2.06)	(-2.06)		(-3.58)	(-3.57)
Age ² /100	0.167*	0.005*		0.367***	0.010***
0	(2.45)	(2.44)		(4.32)	(4.30)
Married (ref: Cohabiting)	0.249*	0.007*	##	-0.170	-0.005
	(2.19)	(2.19)		(-1.26)	(-1.26)
Child aged (ref: No child)	~ /			· /	· /
0 to 4	0.338**	0.009**	#	0.044	0.001
	(3.26)	(3.24)		(0.29)	(0.29)
5 to 10	0.051	0.001		0.323**	0.009**
	(0.59)	(0.59)		(2.84)	(2.83)
Owner	-0.185^{*}	-0.005^{*}		-0.175	-0.005
	(-2.24)	(-2.23)		(-1.70)	(-1.70)
Education (ref: Intermediate)	· /	· · · ·		· /	· /
Basic	0.091	0.002		0.067	0.002
	(0.96)	(0.96)		(0.57)	(0.57)
Higher	-0.020	-0.001		-0.133	-0.003
-	(-0.20)	(-0.20)		(-0.94)	(-0.94)
Continuous work history	-1.418^{***}	-0.038^{***}		-1.934^{***}	-0.052^{***}
-	(-8.33)	(-8.03)		(-6.33)	(-6.20)
Occupational position (ref: Pro	ofessional/mana	ger/technician)			
Skilled non-manual	0.331***	0.009**		0.232	0.006
	(3.68)	(3.64)		(1.39)	(1.39)
Skilled manual	0.828^{***}	0.022***		0.661***	0.018***
	(6.43)	(6.27)		(5.20)	(5.10)
Partly & unskilled pos.	0.828***	0.022***		0.731***	0.020***
	(6.28)	(6.12)		(3.91)	(3.87)
Time with employer (years)	-0.034^{***}	-0.001^{***}		-0.047^{***}	-0.001^{***}
	(-5.17)	(-4.95)		(-7.76)	(-7.32)
Permanent position	-0.860^{***}	-0.023^{***}		-0.971^{***}	-0.026^{***}
	(-9.35)	(-8.71)		(-7.78)	(-7.41)
Work hours	-0.011^{**}	-0.000^{**}	#	-0.029^{***}	-0.001^{***}
	(-2.87)	(-2.85)		(-4.13)	(-4.09)
Partner: Age	-0.005	-0.000		0.086	0.002
	(-0.09)	(-0.09)		(1.24)	(1.24)
Part.: Age ² /100	-0.010	-0.000		-0.097	-0.003
-	(-0.14)	(-0.14)		(-1.11)	(-1.11)
	continu	ued overleaf			

Table 8.5 Actor-partner interdependence model of leaving employment until t + 1

continued					
	Model 8.1				
	***	Poc	oled sampl	e	
	Wom	en		Me	en ANTE
	b	AME		b	AME
Part.: Education (ref: Intermed	iate)				
Basic	0.038	0.001		0.244	0.007
	(0.40)	(0.40)		(1.96)	(1.96)
Higher	-0.043	-0.001		0.145	0.004
C	(-0.40)	(-0.40)		(1.13)	(1.13)
Part.: Continuous work hist.	-0.342	-0.009		-0.511*	-0.014^{*}
	(-1.25)	(-1.25)		(-2.17)	(-2.16)
Part.: Occupational position (r	ef: Professional	/manager/tec	hnician)		
Skilled non-manual	-0.036	-0.001		0.053	0.001
	(-0.28)	(-0.28)		(0.48)	(0.48)
Skilled manual	0.074	0.002		0.031	0.001
	(0.77)	(0.77)		(0.19)	(0.19)
Partly & unskilled pos.	0.033	0.001		0.359*	0.010*
	(0.20)	(0.20)		(2.08)	(2.07)
Part.: Time with employer	-0.000	-0.000		0.003	0.000
	(-0.08)	(-0.08)		(0.38)	(0.38)
Part.: Permanent position	0.081	0.002		-0.007	-0.000
-	(0.65)	(0.65)		(-0.06)	(-0.06)
Part.: Work hours	0.008	0.000		0.014**	0.000**
	(1.48)	(1.48)		(2.76)	(2.74)
Long-distance relocation	1.200***	0.032**	*	0.649*	0.017*
0	(6.85)	(6.68)		(2.53)	(2.52)
Constant	-3.815***	· · · ·		-4.258***	· · · ·
	(-12.35)			(-9.88)	
Random effects					
Variance Intercents	0 776***			1 264***	
variance intercepts	(5.28)			(5.70)	
Covariance Intercepts	(3.20)	() 462***	(3.70)	
Covariance intercepts		,	(3.89)		
Observations			50.540		
Individuals			11.663		
Couples			5.320		
LL			7,343.64		

Data: BHPS waves 1-18, SOEP v26 waves 8-25 (individual level, unweighted)

Note: Multi-level logistic regression model with random intercepts, binary dependent variable: left employment between *t* and *t* + 1 (coded 1), otherwise (coded 0), unstandardised coefficients, z statistics in parentheses; *** significant at 0.1% two-tailed, ** significant at 1%, * significant at 5%; difference between women's and men's AMEs: ### significant at 0.1% two-tailed, ## significant at 1%, # significant at 5%; model also includes the follwing covariates not shown here: country-specific year dummies and region.

women than for men, but my model does not provide evidence for a significant difference by gender.

These effects are controlled for individuals' human capital, occupational positions, family statuses, regions, country-specific period effects, and characteristics of their partners. In addition to these factors, the decision to relocate may be associated with individuals' attachment to the labour market (Taylor 2007). That is to say that respondents with low labour market attachment may be more likely to relocate and, at the same time, may be more likely to leave employment. To control for this, I consider whether respondents have a continuous work history which I interpret as an indication for high labour market attachment (cf. Section 4.5.2). For women and men, this variable has the largest effect on leaving employment until t + 1. Respondents with a more continuous work history, i.e. respondents that have spent more time in employment since they left education, have a much lower probability to leave employment.

The findings also take into account unobserved, time-constant individual differences in the likelihood to leave employment and the correlation of these characteristics between partners by including random intercepts and allowing correlation of these intercepts in couples. The model presented with these random effects offers a better fit than a model without random effects.⁹ I find statistically significant variance of intercepts between individuals. This shows that time-constant unobserved individual heterogeneity affects the likelihood to leave employment. For example, certain individuals may have better unobserved skills. I also find a significant correlation of random intercepts within couples. The significant correlation may be an indication of homogamy and common fate. By not taking unobserved characteristics of individuals and couples into account, the effects of long-distance relocations on women are slightly overestimated. A binary logistic regression without random effects estimates women to have a 4 percentage point higher probability to leave employment if they relocate compared to staying, while for men the AMEs are similar in both model specifications The gender difference in the logistic regression model without random effects is also not statistically significant.¹⁰

The other variables in the model behave as expected and I do not further discuss them here. My model shows that characteristics of respondents' partners have little impact on the probability to leave employment. For theoretical reasons, I do not exclude these variables from the model.¹¹ I only find few significant gender differences in the AMEs. Married women are more likely to leave employment than cohabiting women, while married men are not significantly more likely to leave employment than cohabiting men. Young children increase the risk of leaving em-

⁹ LR test: $\chi^2(3) = 122.34$, p = 0.000.

¹⁰ $\chi^2(1) = 2.91, p = 0.088.$

¹¹ Including the partner variables does not increase model fit significantly (LR test: $\chi^2(22) = 20.49$, p = 0.553).
ployment for women, but not for men. Men that work longer hours reduce their risk to leave employment to a greater degree than women that work longer hours.

The findings from Model 8.1 support H 5.1 which states that partnered women that relocate are negatively affected in their career compared to women staying put. The findings show that women are indeed more likely to leave employment if they relocate compared to staying. This is also true for men and is in accordance with past literature about job-motivated relocations (Taylor 2007). The findings provide only weak support for H 5.2 about gendered outcomes of long-distance relocations. While women's chances to leave employment increase more after long-distance relocations compared to men, the null hypothesis that this difference is equal to 0 cannot be rejected. As I control for human capital model. These findings do not yet allow direct support of one or another of the theoretical explanations for gender differences and do not allow differentiation of the effects of long-distance relocations in England, and West and East Germany. These extensions are addressed in the next sections.

	Model 8.2 Pooled sample				
	Won b	nen AME	Me b	en AME	
Long-distance relocation	0.298 (1.35)	0.006 (1.35)	$0.026 \\ (0.08)$	$0.001 \\ (0.08)$	

Table 8.6 Actor-partner interdependence model of leaving employment until t + 2

Data: BHPS waves 1-18, SOEP v26 waves 8-25 (individual level, unweighed) Note: Multi-level logistic regression model with random intercepts, binary dependent variable: left employment between *t* and *t* + 1 (coded 1), otherwise (coded 0), unstandardised coefficients, z statistics in parentheses; *** significant at 0.1% two-tailed, ** significant at 1%, * significant at 5%; difference between women's and men's AMEs: ### significant at 0.1% two-tailed, ## significant at 1%, # significant at 5%; model also includes all covariates shown in Table 8.5.

In Section 8.1.1 in the descriptive results, I show that while some groups of movers are more likely to leave employment immediately after a relocation, at the second interview after a relocation movers are mostly not more likely to leave employment than stayers. These descriptive results are corroborated by the multivariate results for Model 8.2 presented in Table 8.6. Model 8.2 includes the same independent variables as Model 8.1, but the dependent variable is leaving employment until t + 2 instead of leaving employment until t + 1. The model shows no statistically significant effects of long-distance relocations between t and t + 1 on being out of employment at t + 2. The effects for women and men are also not significantly

different from each other.¹² Thus, the finding in past literature that employment disruptions after long-distance relocations are mostly temporary is supported by my analysis.

In Section 3.4.2 women's relative resources in the couple was described as a potential explanation for gendered outcomes of relocations. I test this explanation by allowing the effects of long-distance relocations to vary between couples in which female partners earn at least 50 per cent of the couples' annual labour incomes and couples in which female partners earn not more than 25 per cent of the couples' labour incomes. Couples in which female partners earn more than 25 per cent but less than 50 of the couples' labour incomes are the reference category. None of the interaction effects are significantly different from 0 (not shown here). Thus, neither women nor men in couples in which she earns at least 50 per cent or not more than 25 per cent of the couples' annual labour incomes have a significantly higher chance to leave employment after long-distance relocations than women and men in couples in which she earns more than 25 per cent of the labour incomes.

In previous research it has also been suggested that gender role attitudes may be causing unequal employment outcomes after relocations. I test whether women in egalitarian couples are less adversely affected by relocations than women in nonegalitarian couples. The models are estimated separately by country, because the gender role attitudes variables are constructed differently from the BHPS and SOEP data (cf. Section 4.5.2). In England, the variable egalitarian couple is coded 1 if both partners agree that both partners should contribute to the household income and coded 0 otherwise. In Germany, the variable egalitarian couple is coded 1, if the male partner is responsible for at least one third of the time that the couple spends on housework and childcare on a regular Sunday and coded 0 otherwise.¹³ West and East Germany are not further differentiated, since the case number of long-distance relocations differentiated by West and East Germany and by egalitarian and nonegalitarian couples would be too small to gain reliable estimates. The interaction effects of long-distance relocations and egalitarian couples do not differ significantly from 0, but the effects' directions are as expected. English and German women in egalitarian couples are less likely to leave employment after long-distance relocations than women in non-egalitarian couples. English and German men in egalitarian couples are more likely to leave employment after long-distance relocations than men in non-egalitarian couples. The measures of gender role attitudes need to be further improved to corroborate these findings.

¹² Test of AMEs: $\chi^2(1) = 0.23$, p = 0.630.

¹³ The variable for Germany is based on previous work by Jürges (2006). Time use on a Sunday may be a weak proxy for gender role attitudes, however. For example, men may be likely to spend 'quality time' with their children on a Sunday, while not contributing substantially to childcare on weekdays. Unfortunately, the SOEP does not offer any more direct attitudinal measures. The measures may be also difficult to compare across countries, as it can be expected that gender-equal attitudes may be more likely than gender-equal behaviour.

8.2.2 Institutional Context

In H 5.3, I expect that gender differences will be strongest in West Germany, moderate in East Germany and weakest in England.¹⁴ In Model 8.3 in Table 8.7, I test this hypothesis by interacting long-distance relocations with dummies for West and East Germany. The main effect for long-distance relocations shows the difference in log odds of leaving employment between movers and stayers in England. For women and men in England the chances to leave employment are higher, if they relocate over a long distance. The difference between women and men in England is not statistically significant.¹⁵ The main effect for West Germany shows that West German stayers are less likely to leave employment than English stayers, but only for women the coefficient shows a significant difference. In contrast, East German stayers are more likely to leave employment than English stayers, but the coefficients are not significantly different from 0.

None of the interaction terms is statistically significant at conventional levels. However, the interaction for relocations and East Germany is marginally significant and negative for women and men, i.e. movers in East Germany are less likely to leave employment than movers in England. The non-significance of these coefficients may be mainly due to the low case number of movers. For West Germany the coefficients are so small that even with moderately larger case numbers they may not turn significant. The marginal effects in the lower panel in Table 8.7 show that relocations in East Germany do not increase the probability for women to leave employment. In West Germany, the probability of leaving employment for women that relocate a long distance is about 4 percentage points higher than for those not relocating. Men in West and East Germany do not have a higher probability of leaving employment, if they relocate. The AMEs for English women indicate that their probability of leaving employment is 8 percentage points (z = 3.39) higher after long-distance relocations compared to staying (AMEs not shown in table). For English men, the probability is increased by 4 percentage points (z = 1.69). Thus, I cannot support H 5.3 about cross-country differences in the effects of long-distance relocations. While women in East Germany seem to be less likely to leave employment in accordance with my expectations, English women are not less likely to leave employment than West German women on average.

I use Model 8.3 to predict the probability of leaving employment for women and men in England, East and West Germany grouped by their mover status holding all other variables at their means. The predictions are only based on the fixed part of the model and are presented in Figure 8.1. In all three contexts, female stayers have a higher predicted probability to leave employment than male stayers, but the difference is only significant in England. Women that relocate have a much higher probability to be out of work than female stayers in England and West Germany.

¹⁴ This section partly builds on (Lersch 2013, forthcoming).

¹⁵ Test of AMEs: $\chi^2(1) = 2.16$, p = 0.142.

8.2 Multivariate Results

	Model 8.3 Pooled sample				
	Women Men				
	b				
Long-distance relocation	1.380*** 0.934*				
	(5.50) (2.40)				
West Germany	-0.660^{*} -0.277				
	(-2.37) (-0.69)				
West*Relocation	-0.024 -0.065				
	(-0.06) (-0.12)				
East Germany	0.300 0.709				
	(1.06) (1.75)				
East*Relocation	-1.027 -1.530				
	(-1.92) (-1.76)				
	AME of relocation at				
West					
No	0.067*** 0.032				
	(3.49) (1.58)				
Yes	0.038** 0.022				
	(2.63) (1.36)				
East					
No	0.065*** 0.035				
	(4.22) (1.92)				
Yes	0.013 -0.022				
	(0.62) (-0.96)				

Table 8.7 Actor-partner interdependence models of leaving employment until $t + 1$ by containing the second sec
--

Data: BHPS waves 1-18, SOEP v26 waves 8-25 (individual level, unweighted) Note: Multi-level logistic regression model with random intercepts, binary dependent variable: left employment between *t* and *t* + 1 (coded 1), otherwise (coded 0), unstandardised coefficients, z statistics in parentheses; *** significant at 0.1% two-tailed, ** significant at 1%, * significant at 5%; model also includes all covariates shown in Table 8.5.

About 1 in 11 English women in dual-earner couples that relocate a long-distance are predicted to leave employment, while only about 1 in 50 female stayers are predicted to leave employment in England. In West Germany, about 1 in 100 women in dual-earner couples that do not relocate leave employment, but about 1 in 20 women that relocate are predicted to leave employment. In East Germany, I do not observe significant differences in the predicted probabilities to leave employment between female movers and stayers. In both groups, about 1 in 25 women are predicted to leave employment. For men, the probability to leave employment does not differ significantly between stayers and movers in England, West or East Germany.

Figure 8.1 also shows differences in the predicted probability of leaving employment across countries. Among female stayers, women in East Germany are most



Data: BHPS waves 1-18, SOEP v26 waves 8-25 (individual level, unweighted) Note: Prediction based on Model 8.3 in Table 8.7; whiskers indicate 95 %-confidence interval. Negative values of confidence intervals are constrained to 0.

Fig. 8.1 Predicted probabilities for leaving employment by context, gender and relocation status

likely to leave employment, followed by women in West Germany and England. These differences reflect general variation in the economic situation, labour demand, and job mobility across the three contexts. Against my expectations, female movers in England have the highest predicted probability to leave employment, while female movers in West and East Germany have a similar and lower predicted probability to leave employment. However, due to the wide confidence intervals of all three predictions, one cannot be sure that the differences are not just pure chance. It may be expected that at least the difference between England on the one hand and West and East Germany on the other hand may turn significant, if the case number of long-distance relocations would be higher. Nevertheless, the findings again do not support H 5.3. Cross-country differences are further discussed in Section 9.1.4.

8.2.3 Robustness Checks

I perform a number of additional analyses to test the robustness of my findings. I only describe the main findings here without showing full estimation results. First, in the descriptive results I show that a high share of long-distance relocations are motivated by occupational reasons and, thus, only those respondents may select into relocations that expect to receive positive returns from relocating. For example, individuals may have unobserved skills that make them more likely to receive a job offer at a new location and because individuals know about these skills, they are more likely to relocate. I test for potential selection into the group of movers by estimating two-stage regression models separately for women and men in both countries, since the selection into relocations may be different for these subgroups.¹⁶ In the first stage, I estimate pooled probit models with the dependent variable longdistance relocation to retrieve Inverse Mills' Ratios (IMRs) which can be used in a second step to control for selectivity (Brüderl 2000: 621; cf. for an application to relocations: Taylor 2007).¹⁷ To gain reliable estimates, I include an identifying variable in this equation that is the number of household members, which affects the likelihood to relocate over a long distance but not the probability to leave employment at t + 1. In a second step, I run random-effects panel regression models with the same covariates as reported for Model 8.1 and include the IMRs as additional regressors to control for positive or negative selection into relocations. The signs of the coefficients of IMRs in the second stage equation can be interpreted as the direction of correlation between unobserved heterogeneity in the first and second equation.

For men, the coefficients of the IMRs are statistically insignificant. However, for English men, the coefficient for relocations is no longer significant after controlling for selection into relocations, while the effect size does not change. For German men, the coefficient for long-distance relocations does not change. For women in both countries, I find well-estimated, negative coefficients for the IMRs. That is to say, that women that are more likely to relocate over long-distances due to unobserved characteristics are less likely to leave employment. For example, these unobserved characteristics may be related to women's motivation to pursue a career. Correcting for this selection reduces the effect sizes of long-distance relocations on leaving employment for women, but the key findings do not change: women in dualearner couples are more likely to leave employment after long-distance relocations compared to stayers.

¹⁶ I cannot differentiate West and East Germany due to the small number of observed relocations in East Germany.

¹⁷ The IMR λ is the ratio of the probability density function over the cumulative distribution function of a distribution: $\lambda(.) = \phi(.)/\Phi(.)$ (Cameron and Trivedi 2010: 560f).

Second, I only consider stable couples in the present analysis, i.e. couples that live in the same household at t and t + 1.¹⁸ It may be expected that partners that would be especially negatively affected by long-distance relocations rather break up with their partner or form a second household, i.e. they become a living-aparttogether couple (LAT). My estimates for the relocation variable may be biased if unobserved characteristics are correlated with long-distance relocations and the decision to continue living in the same household. I run multi-level bivariate probit models separately by gender and context to test whether this selection process may affect my coefficients for long-distance relocations The bivariate probit model is one way to handle associations across equations by allowing the error terms in both equations to be correlated (Brüderl 2000: 635). Here, the first equation models whether individuals leave employment (outcome equation) and the second equation models whether individuals still live together in a common household at t + 1(selection equation). In the selection equation, I additionally include the variable ever divorced to facilitate identification of the equation system. The effects of longdistance relocations on leaving employment are robust in the model controlling for selectivity and the main results hold: Women in England and West Germany are more likely to leave employment after relocations, women in East Germany and men in West and East Germany are not significantly more likely to leave employment after relocations. Due to the small number of dual-earner couples that separate, the estimation of the selection equation is hampered and especially the estimation of the random effects proves problematic. Due to the small case number, the model does not converge for English men.

Third, both partners may be affected by exogenous shocks in the regional labour market that make them lose their job and relocate to a different region. These unobserved shocks may bias my estimates for the long-distance relocation variable. I attempt to control for such shocks by estimating an alternative model in which I include the regional unemployment rate in the year before individuals may leave employment I measure the regional unemployment rate at the level of counties in Germany and LADs in England. I do not include this variable in the general model, since regional unemployment rates are only available after 1998 for England. In addition, regions add another level of data and as individuals may change regions over time, modelling these data becomes very complicated. For the present analysis, I simply add the regional unemployment rate as an additional characteristic at the level of couples to the model, but the standard errors for the regional unemployment rate are likely to be underestimated as I do not account for the clustering of observations in counties and LADs.

The results show that the regional unemployment rate has a positive effect on the probability to leave employment at t + 1, but only men are statistically significantly

¹⁸ I exclude 1,002 individual-year observations from the analytic sample on the grounds that these individuals do not live together with the same partner at *t* and t + 1. In 261 of these cases, at least one partner relocates a long-distance between *t* and t + 1.

affected by the regional unemployment rate. Men's probability to leave employment increases by 0.1 percentage points with a 1 percentage point increase in the regional unemployment rate. Adding the control regional unemployment rate does not substantially alter the estimated effect of long-distance relocations on the probability to leave employment. Women and men both remain more likely to leave employment after long-distance relocations than stayers after adding the regional unemployment rate to the model. The sizes of the coefficients change, however; and the AMEs for women are smaller, while the AMEs for men are larger than reported in Table 8.5. The difference between women and men is reduced and remains statistically non-significant.

Fourth, my analytic sample includes respondents with a wide range of working hours. I re-estimate Model 8.1 and restrict the sample to all those dual-earner couples in which both partners work between 30 and 80 hours a week. Thus I exclude respondents with more than 10 and less than 30 working hours — mainly women — and this makes the analytic sample more homogeneous. The estimation results show that the effect of long-distance relocations for this more homogeneous sample is very similar to the results presented before. Women's probability to leave employment is 3 percentage points higher if they relocate a long-distance compared to stayers. Men's probability of leaving employment is 1 percentage point higher. Both AMEs are only marginally different from the AMEs presented in Table 8.5. Thus, respondents that work longer hours are slightly less likely to leave employment after long-distance relocations, but the difference is very small and does not affect my interpretation of the data.

For a final robustness check, I use the euclidean metric between individuals' locations to identify long-distance relocations.¹⁹ As my variable of long-distance relocations only measures whether respondents relocate across administrative borders of counties and LADs, this proxy may not adequately capture long-distance relocations and may also be not directly comparable across countries due to the different size of administrative units. Instead, I now consider the actual distance of individuals' relocations and a binary relocation variable that takes the value 1, if respondents relocate at least 50 km (cf. Section 4.5.1.1). The average distance of relocations across administrative borders is 45.78 km (median: 14.22 km) in England and 45.04 km (median: 13.47 km) in Germany. Thus, relocations across administrative borders seem to be comparable in distance across countries. Only 23 per cent of the relocations across administrative borders are relocations farther than 50 km and I observe only 112 relocations of dual-earner couples that are farther than 50 km. This may indicate that relocations across administrative borders may not increase commuting distance to a degree that leads individuals to change their places of work after a relocation.

¹⁹ I only use data from 2000 onwards, as the geo-coded addresses are only available for the SOEP starting in 2000. For this reason, I use the proxy variable long-distance relocation for the main analysis.

I re-estimate Model 8.1 using the relocation farther than 50 km variable instead of the relocation across administrative borders variable as the independent variable. The alternative model shows that women and men are significantly more likely to leave employment if they relocate farther than 50 km. Women's probability to leave employment is 5 percentage points higher if they relocate farther than 50 km. This change in probability is slightly larger compared with relocations across administrative borders. This may be due to the fact that relocations across administrative borders are mostly shorter than 50 km and may not make a change in the work place necessary. Similarly, I also find a higher AME for men when comparing the effect of relocations farther than 50 km with the effect of relocations across administrative borders. The former type of relocations increases the probability to leave employment for men by 6 percentage points, while the latter type of relocations longer than 50 km for men is also slightly higher than for women, but the coefficients are not significantly different.

These additional findings regarding relocations that are farther than 50 km have two implications for the general interpretation of my findings. First, the results that I present regarding relocations across administrative borders are likely to underestimate the effect of long-distance relocations on the probability of leaving employment. Due to the fact that most of the relocations across administrative borders are relatively short, the presented effects may be smaller than expected when only analysing 'true' long-distance relocations. Second, the alternative model indicates that the effects for relocations across administrative borders may over-estimate gender differences. Thus, long-distance relocations may have less clearly gendered effects on the chances to leave employment, if only 'true' long-distance relocations would be considered. However, relying on earlier literature that repeatedly showed gendered effects of long-distance relocations, the smaller gender difference in relocations that are farther than 50 km may be also due to the very small number of relocations. Results should be interpreted cautiously due to this small number of relocations.

8.2.4 Changes in Wage Rates

To examine employment outcomes of relocations for those that stay in employment, I analyse immediate effects of long-distance relocations on wage rates at t + 1 and medium-term effects until t + 2. I use simultaneous equation models to control for the selection into employment.²⁰ The models show that long-distance relocations

²⁰ The substantially interesting equation regresses log wage rates at the next interview on longdistance relocations controlled for demographic characteristics, human capital, occupational position, employment status and the work history. The equation allows the intercept to vary across individuals. The second equation models the probability of respondents to provide valid information regarding wage rates at t + 1. This selection equation additionally includes variables measuring the

do not have an impact on wage rates of partners in dual-earner couples on average at t + 1 and t + 2 (not shown here). After a relocation, the wage rates of movers do not differ significantly from the wage rates of stayers. If women manage to maintain employment or find a new position right after a relocation, their wage rates are not adversely affected by the relocation compared to wage rates of women that do not relocate.

The findings regarding women are similar to previous research that shows that the negative effect of long-distance relocations on female careers is mainly due to a reduction in labour participation but not due to changes in wage rates for those that maintain employment (e.g. Jacobsen and Levin 1997; LeClere and McLaughlin 1997; Boyle, Feng and Gayle 2009). The findings for men contrast with earlier research including all men that finds significant positive returns of long-distance relocations regarding wage rates (e.g. Böheim and Taylor 2007). This indicates that men in dual-earner couples may be more constrained in their relocation behaviour than other men and, thereby, are less able to relocate to better job offers.

8.3 Summary

In the present chapter, I test the theoretical expectations regarding determinants of disparities in relocation outcomes focusing on gender differences in occupational returns to long-distance relocations. I find that long-distance relocations have disruptive effects for women's and men's careers in England and West Germany. Longdistance movers are more likely to leave employment than stayers on average. Thus, I can support H 5.1 about the negative effect of long-distance relocations on women's employment at least for these two contexts. However, neither female nor male long-distance movers have lower average wage rates after a relocation compared with stayers. If women maintain employment, they are not adversely affected in their career measured through the hourly wage rate. These findings show that longdistance relocations of dual-earner couples of which a large share are occupationally motivated do not necessarily yield immediate positive returns regarding employment. This is in contrast to past findings regarding positive average returns for all couples and singles. This may indicate that dual-earner couples are constrained in their mobility and disparities between individuals in dual-earner couples and other couples result.

The negative effect of long-distance relocations on employment is only temporary and at the second interview after a relocation, female movers are as likely to be

presence of children aged 0 to 4 and 5 to 10 in the household and variables measuring the partner's occupational position. These variables are assumed to affect the respondents' supply of labour, but not the wage offers received by the respondents. I run the models separately by gender and country, as it can be assumed that the selection into employment after long-distance relocations is different for women and men in both countries based on the findings that I presented before.

employed as female stayers. Thus, I find some evidence that disparities in relocation outcomes occur between women and men in dual-earner couples in accordance with H 5.2, but disparities are smaller than expected. If women maintain employment, their returns of relocations in wage rates are not different from men. Female movers are also able to enter employment again soon if they left immediately after long-distance relocations. Nevertheless, even short disruptions of the career may have negative long-term effects, e.g. with regard to pension rights.

For the effects of long-distance relocations in the context of England, West and East Germany, I find gender differences to be strongest in England against my expectations in H 5.3. In England, the difference in the effects of long-distance relocations on leaving employment is marginally significant between women and men, where women are more likely to leave employment. In West and East Germany, differences between women and men are weaker. In West Germany, only women are more likely to leave employment after long-distance relocations. In East Germany, neither women nor men are significantly more likely to leave employment after long-distance relocations for these unexpected findings are further discussed in the next chapter in Section 9.1.4.

Due to the small number of observed long-distance relocations, the statistical power in my analysis is limited and the standard errors were relatively large in the analysis. Effects that may have a substantial impact on individuals may have not turned out to be statistically significant due to these large standard errors. Therefore, the findings in the present chapter should be treated with caution and further validations with hopefully larger samples are necessary. A further limitation of the present chapter is due to the proxy for long-distance relocations. In Section 8.2.3, I showed that many relocations across administrative borders cover only short distances and, therefore, my analysis may underestimate the effect of long-distance relocations on labour market outcomes. At the same time, gender differences may be overestimated.

Chapter 9 Outcomes of Residential Relocations and Structural Conditions: England and Germany Compared

Individuals' life courses are embedded in geographical places, historical times and institutional settings, and individual agency is always conditioned by opportunities and constraints resulting from this embeddedness. In life course research, mostly the institutions of the education system, the labour market and the pension system have been analysed in the way they "channel individuals to varying extents from participation in one to participation in the next institution according to a country's specific institutional regime" (Krüger 2003: 36). The term life course regime is used to describe these "temporal structures of relatively high stability for structuring life courses" (Wingens and Reiter 2011: 190). Life courses without being deterministic (ibid.: 191). These life course regimes have been mostly studied in their effects on individuals' job and family trajectories in the past (e.g. Leisering and Schumann 2003; Schaeper and Falk 2003).

The basic idea of the present chapter is that what is true for the job and family trajectories of individuals is also true for their residential trajectories. The "steering mechanisms" (Diewald and Mayer 2009: 6) of life course regimes also generate regularities and uniformities in individuals' residential trajectories and channel individuals through particular residential locations at particular times in their lives. Figure 9.1 provides the general intuition about the links between life course regimes and regularities in individuals' residential trajectories: Institutional features of life course regimes, e.g. in the labour market, affect the occurrences, timings, durations and outcomes of life events and stages. Institutional features do not affect individuals homogeneously, but are moderated by individual heterogeneities, e.g. gender (Diewald and Faist 2012). Different life course trajectories are interrelated. Thereby, institutional features that affect one trajectory may also indirectly affect other trajectories.

The aim of the present chapter is to identify institutional features in life course regimes and characteristics of typical family and job trajectories that generate regu-



Fig. 9.1 Life course regimes and residential trajectories

larities in residential trajectories in England and Germany. These regularities of residential trajectories may generate particular disparities in outcomes of relocations between individuals and also over time for the same individuals. Regularities are revealed through the typical characteristics of residential trajectories in both countries. In the following, these characteristics are compared across countries, and linked to particular institutional features and patterns in family and job trajectories in England and Germany. I draw from the empirical findings presented in the previous chapters to identify such regularities.

To this end, I proceed as follows. In Section 2.3, hypotheses are derived about the effect of contextual constraints and opportunities on individuals' relocations and disparities in their outcomes. These hypotheses are informed by the case description in Section 4.1 that highlights the institutional features of England and Germany with regard to their housing markets, labour markets, welfare systems and gender relations. In Section 9.1, I weigh the evidence for each of my cross-national hypotheses drawing from findings presented in Chapter 6 and Chapter 7. These findings mainly relate to differences in the English and German housing markets, labour markets and welfare systems. Then, I refer to findings presented in Chapter 8 to examine the effect of divergent societal gender relations in England, and West and East Germany

on disparities between women and men in the effects of relocations on employment and wages. Based on the assessment of my hypotheses I sketch the most important features of the life course regimes that affect relocations and residential trajectories in Section 9.2. I conclude in Section 9.3.

Due to the complexity of structural conditions that affect relocations, the present comparison can only touch upon particular differences in the institutional features of both countries that are highlighted in my hypotheses. Using the available information I make an argument about why these differences may be associated with particular observed differences in English and German residential trajectories. Competing explanations based on different sets of institutional particularities may be possible for each hypothesis, but cannot be further tested in the present analysis. To strengthen my argumentation, I refer to the wider literature on differences between life courses in England and Germany and show how my results relate to previous findings (cf. also Section 4.1.3). I also point at potential directions for future research to further analyse cross-national differences. In addition, it has to be stressed that my conclusions in this chapter are based on information from rather small sections (maximum of 18 years) of individual life courses only, as I am using prospective panel data instead of retrospective life history data. I generalise from these observed sections of various individuals to general patterns over the entire life course. This generalisation may be biased by cohort effects.

9.1 Institutional Features and Their Effects on Residential Relocations

In the previous chapters, I examined the effect of residential relocations on several outcomes: room stress, i.e. the number of available rooms divided by the room need; neighbourhood quality, i.e. the quality of individuals' direct local environment at their places of residence; and chances to leave employment (cf. Section 4.5.1 for more details on the dependent variables). In this section, I provide evidence for and against my cross-national hypotheses with regard to these outcomes.

I briefly recapitulate these hypotheses. Due to relatively easy access to home ownership in England, the dominant norm of being a owner, and a higher risk of downward social mobility, I expect that the changes in quality of locations over the life course follow distinct patterns in England and Germany. More concretely, I expect that a) on average improvements in the quality of locations through relocations for young singles and childless individuals are more likely in England than in Germany; b) improvements in the quality of locations through relocations are more likely in the growing-family phase in Germany than in England; and c) downward relocations, i.e. relocations that reduce the quality of locations, are more likely in England than in Germany (H 1.7). In past literature on life course regimes, the

average English life course has been described as less stable than the average German life course (Mayer 2005: 30). Stable means that individuals experience fewer changes over their life courses and changes are homogeneous across individuals. Following this literature, I expect that the quality of location is more stable over the life course in Germany than in England on average, especially in the mid and late life course (H 1.6). Due to the different tenure structure in both countries, I expect housing tenure to affect changes in the quality of locations distinctively in England and Germany. I hypothesise that a) relocations from tenancy to ownership and b) relocations between privately rented dwellings improve quality of locations stronger in Germany than in England and that c) relocations between owned homes improve quality more in England than in Germany (H 2.3).

Since the welfare system is less protective in England and housing is more decommodified in Germany than in England, I expect changes in the quality of locations as a result of relocations to depend more on income in England than in Germany (H 4.3). Last, I hypothesise that gender inequality in effects of long-distance relocations on careers will be weakest in England, modest in East Germany, and strongest in West Germany on average (H 5.3). This is due to divergence in policies in England and Germany that affect the division of labour between women and men and differences in gender norms between England, and West and East Germany.

9.1.1 Quality of Locations over the Life Course

The patterns of changes in the quality of locations over the life course and the associations between family and residential trajectories diverge between England and Germany and this supports the assumption that the residential trajectory is affected distinctively by the institutional context in both countries. Firstly, I find support for H 1.7 a) with regard to room stress but not with regard to neighbourhood quality. Concerning room stress, the evidence shows that young singles already relocate to more spacious dwellings in England, but not in Germany, and that the differences between singles and couples are stronger in England than in Germany. These differences may be due to particular patterns in job trajectories. In England, average entry age in the labour market is lower (Schmitt 2012). Early labour market entry may give young singles in England the economic stability and planning security to improve the quality of their locations early in life. Young singles in England may also be able to invest more into housing due to easier access to mortgages (Diamond and Lea 1992: 118ff). H 1.7 a) is not supported with regard to neighbourhood quality however. Young singles in England do not improve their neighbourhood quality if they relocate. This may be due to the fact that young singles may have preferences for metropolitan and urban locations that are associated with a lower neighbourhood quality. This is supported by my descriptive evidence about a lower average neighbourhood quality for young singles.

In contrast to England where singles also improve their locations, improvements in the quality of locations are closely linked to forming or being in a couple in Germany. I find single to couple and stable couple relocations to reduce room stress more strongly in Germany than in England. Relocations of stable couples reduce room stress more persistently in Germany than in England. This indicates that the investment into housing of couples is more long-term oriented in Germany than in England. German couples are more likely to relocate to dwellings that offer more rooms even after future household size increases compared to the pre-relocation situation, while English couples seem to be more likely to adapt to momentary room needs, e.g. anticipating childbirth, and do not relocate to more spacious dwellings. All this is evidence in accordance with H 1.7 b).

My findings are in accordance with past literature that shows that couples in England are more likely to live in home ownership, which offers better quality of locations on average, from the start of their partnerships. This is also due to the social norm of becoming a home owner before marriage. Couples in Germany are more likely to make the transition into ownership during their partnerships and the transition to ownership takes place later in their partnerships than in other countries on average (Clark, Deurloo and Dieleman 1997; Mulder 2006: 294). The idea of building a joint nest for couples may also be stronger in Germany, where couples stay together for longer on average and traditional families based on marriage play a more important role in welfare and family policies, than in England, where unions are less stable and individualistic values are supported by policies (Ostner and Lewis 1995: 184; Bussemaker and Kersbergen 1999: 18; Blossfeld and Drobnič 2001: 43).

My findings show that being in a couple is more important than having children for improvements in the quality of locations in Germany. The presence of children aged 0 to 9 reduces room stress more in England than in Germany, while childbirth is associated with a greater increase in room stress in Germany than in England. This evidence contradicts H 1.7 b). Thus, the event of union formation and the stage of being in a union are more important conditions for improvements in the quality of location in Germany than in England, instead of the growing family phase. This may also be due to the fact that average age at first birth is higher in England than in Germany, while average age at first consensual union is lower in England than in Germany (Ermisch and Francesconi 2000; Konietzka and Huinink 2003; Schmitt 2012). This may indicate that union formation and childbearing are distinct events in England, while they are closer associated in time in Germany and, therefore, I do not find a unique effect of childbirth in Germany separately of being in a couple.

With regard to neighbourhood quality, stable couple relocations improve neighbourhood quality for English men, but single to couple relocations do not increase neighbourhood quality. Similarly, mainly stable couples improve their neighbourhood quality through relocations in Germany. The difference between single to couple and stable couple relocations may again be due to divergent preferences. Couples that move together are mostly in early stages of their life courses and they may have stronger preferences for urban and metropolitan areas with lower neighbourhood quality than stable couples that relocate and which are likely to be in a later life course stage. Young children improve neighbourhood quality on average in Germany, but in England only men improve their neighbourhood quality if at least one young child lives in the household. These findings provide some evidence for H 1.7 b) regarding neighbourhood quality.

Regarding H 1.7 c), I only find evidence for a higher risk of downward relocations in England compared to Germany with regard to room stress. English individuals in all age groups are significantly more likely to increase their room stress if they relocate compared with German movers. This is in accordance with findings from other life course research that shows, e.g., higher risks of downward job mobility in England compared to Germany (Allmendinger and Hinz 1998). The risk of downward relocations is especially high in late life in England compared to Germany. This corresponds with findings regarding stronger variation in room stress changes in late life in England compared to Germany as described below. The higher risk of downward relocations regarding room stress in England may be due to, first, higher chances for downward social mobility (Allmendinger and Hinz 1998; Jarvis and Jenkins 1998: 151). Second, more frequent increases in household size due to higher average fertility in England may increase room stress more often than in Germany.

Concerning changes in neighbourhood quality, English movers are found to be less at risk to reduce their neighbourhood quality than German movers. Thus, H 1.7 c) can only be supported with regard to room stress and has to be rejected with regard to neighbourhood quality. As stated above, the cross-national differences regarding neighbourhood quality may partly be statistical artefacts, but may also be due to genuine differences between neighbourhood quality and other characteristics of residential locations: First, preferences regarding neighbourhoods may be more heterogeneous across the population than preferences regarding rooms in a dwelling and these preferences may override other structural conditions. That is to say that as individuals differ strongly in their preferences for certain neighbourhoods, these divergent preferences already explain a large part of the variation in outcomes of relocations and institutional differences may not play a big role in explaining divergence. Second, neighbourhood quality depends more on the local context than room stress on average, e.g. in metropolitan areas neighbourhood quality is lower on average than in rural areas while the average level of room stress does not differ. Thus, the local environment of relocations will play a larger role in neighbourhood quality changes and may also override structural conditions that affect neighbourhood quality in England and Germany. Third, a potential explanation may be that neighbourhoods in Germany are more heterogeneous than in England. However, this explanation is contradicted by findings about less spatial segregation in Germany compared to England (Musterd 2005).

I find partial evidence for H 1.6 about the stability of life courses only for room stress, but not for neighbourhood quality. The estimated within-individual deviation for room stress is significantly higher in England than in Germany. Thus, individuals

in England are more likely to change their room stress over the observed section of their life course than individuals in Germany. This may be partly due to the higher frequency of transitions in family trajectories in England. These findings fit well with the broader life course literature on the differences between the English liberal market economy regime and the German conservative continental regime. Again, I do not find the same picture regarding cross-national differences in changes in neighbourhood quality. Against my expectation in H 1.6, individuals in Germany show less uniformity of the residential trajectory regarding neighbourhood quality than in England.

9.1.2 Role of Housing Tenure

The opportunities and constraints that result from institutional particularities in national housing markets affect disparities in outcomes and the timing of relocations in the life course in England and Germany. My findings especially support the assumption that the housing tenure structure in each national housing market influences individuals' outcomes after relocations depending on their tenures of origin and destination. In England privately rented accommodations are only a transitive tenure that most individuals exit to buy property as soon as possible or that is used to bridge the waiting time for social housing (Kemp and Keoghan 2001). In Germany, on the other hand, the privately rented tenure offers attractive dwellings (Mulder and Wagner 1998). Relocations between rented dwellings are an integral part of residential trajectories in Germany and individuals can relocate between rented accommodations to improve their qualities of location. In Germany, relocations into ownership are singular events in most life courses, since repeated purchases of property are constrained by conditions in the housing market, e.g. penalties for early repayment of mortgages and once-only subsidies for purchasing homes (Voigtländer 2009). Therefore, individuals in Germany are more likely to use these singular events to increase the qualities of their locations substantially. In England, repeated relocations between owned homes are more common and, thus, transitions between rented and owned accommodations are less significant events, because owners can still trade up their dwellings later in life.

More specifically, I find a similar hierarchy in the quality of locations by housing tenure in both countries in a cross-sectional perspective. Outright owned dwellings are associated with the lowest room stress and the highest neighbourhood quality on average. Tenants in social housing live in the worst neighbourhoods and highest room stress in both countries. Dwellings owned with a mortgage and privately rented dwellings are between the other two tenures regarding room stress and neighbourhood quality, and dwellings owned with a mortgage offer better quality than privately rented dwellings on average. Turning to the longitudinal findings, the housing tenure variables are important in explaining within-individual change in room stress in Germany, while in England tenure explains only little within-change. The fixed-effects models show that relocations into social housing and privately rented dwellings increase room stress in both countries. Relocating into social housing is associated with stronger increases in room stress than relocations into the private rental sector in England, but there are no significant differences between changes into privately rented dwellings and social housing in Germany. Based on models that take into account the tenure of origin and destination, I test H 2.3. Private tenants in England are found to only change their room stress significantly, if they buy a home. Germans that relocate from rented accommodation into ownership reduce their room stress more strongly than English movers. Germans that relocate within the privately rented sector also reduce their room stress to a greater degree than English movers. English home owners that relocate to another own home reduce their room stress more than German home owners. This is evidence in favour of H 2.3 regarding room stress.

Regarding changes in neighbourhood quality, I find changes to social housing and privately rented dwellings to reduce neighbourhood quality in Germany compared to dwellings owned with mortgages. In contrast, changes to social housing have no effect on changes in neighbourhood quality and changes to privately rented dwellings increase neighbourhood quality on average in England. Findings from models that include the tenures of origin and destination show that relocations between privately rented dwellings improve neighbourhood quality in Germany but not in England. Relocations between rented accommodations and owned dwellings improve neighbourhood quality on average in Germany but not in England. I find relocations between owned dwellings in England to be horizontal, while in Germany these relocations reduce neighbourhood quality on average. Thus, the expectations formulated in H 2.3 are mostly supported with the exception that neighbourhood quality does not change if movers own before and after the relocation in England.

I only test one hypothesis regarding the housing market in the present study. The importance of the housing market for relocation outcomes clearly goes beyond this one aspect. For example, the housing market has a strong effect on the entry age into home ownership and, thus, affects the timing of important changes in the quality of locations for individuals. In England, the entry age is considerably lower than in Germany and this may explain some of the patterns in individuals' residential trajectories (Freeman 1997: 162). Deviation in room stress changes between individuals in England may be relatively high in their 20s, as some individuals already buy their own properties, while in Germany most individuals in this age group still live in a rented accommodation. Future research is needed to take a closer look at various design features of the English and German housing markets to assess their impact on relocations and relocation outcomes.

9.1.3 Decommodification of Housing

I find average room stress to be more stratified by income groups in Germany than in England. In England, the highest income quintile has about twice as many excess rooms, i.e. rooms beyond the objective room needs, than the lowest income quintile, while in Germany the highest income quintile has about three times as many excess rooms as the lowest income quintile. Thus, cross-sectional evidence does not show an equalisation of room stress in both countries. Even though I find a strong stratification of room stress by income, most individuals in England and Germany live in very low room stress in a historical perspective, and compared to their objective room needs, individuals in both countries live in objectively sufficiently spacious dwellings. In 2008, only 3 per cent of individuals in England and slightly under 8 per cent of individuals in Germany had a positive room stress, i.e. less rooms available than necessary according to their objective room needs.

In a longitudinal perspective, positive changes in household income reduce room stress more in England than in Germany. English individuals reduce their room stress about twice as much with an increase in income as individuals in Germany. The interaction effect of income and residential relocation of stable couples is also larger for women in England than in Germany. These findings indicate that changes in room stress are more strongly affected by income changes in England than in Germany. The findings support H 4.3 regarding room stress. However, the overall level of income seems to be more important in explaining changes in room stress in Germany than in England, as the level of income is not associated with changes in room stress in England, but the level is positively associated with changes in room stress in Germany. The last findings are in accordance with the general picture of social mobility in Germany drawn by Brückner (2004). The overall level of income, which can be interpreted as an indicator for a particular track in life, is important in explaining changes in room stress, while individual deviations from this track, i.e. current income changes, do not affect room stress strongly. In contrast, the findings for England show that a particular track in life indicated by the level of income is not as relevant as in Germany for explaining changes in room stress. Rather, momentary deviations from this track have stronger effects on room stress changes.

According to my descriptive findings, average neighbourhood quality differs strongly by income quintiles in both countries. The higher the income of individuals, the better their neighbourhood quality and the differences by income quintile are similar in England and Germany. These findings are akin to the results for room stress and again indicate that both welfare systems do not provide equal access to good neighbourhoods. However, I do not find a significant association between changes in income and changes in neighbourhood quality for both countries. While individuals with higher income live in better neighbourhood quality on average, individuals that increase their income do not improve their neighbourhood quality on average. Analyses with alternative measures for economic resources show that there is a more complex association between resources and neighbourhood quality in both countries. Individuals live in higher quality neighbourhoods in years in which they have higher housing costs. Changes in more permanent income (averaged over three years) are also positively associated with changes in neighbourhood quality. I only find a significant interaction effect of income and stable couple relocations in Germany. German couples with higher incomes improve their neighbourhood quality more than low-income couples. The impact of other relocation types does not vary by income in Germany and I do not find interaction effects of relocations and income to be significant in England. My findings do not support H 4.3 regarding neighbourhood quality.

Overall the findings tentatively support the idea that the stronger de-commodification in Germany weakens the association between changes in income and changes in the quality of locations to some degree. This support is stronger for changes in room stress than for changes in neighbourhood quality. This may be due to the fact that floor space is directly affecting housing costs either because rent increases or due to increases in construction costs and property prices. In most cases, floor space is highly correlated with the number of available rooms in a dwelling. Neighbourhood quality, on the other hand, may not be as closely correlated with land rent. For example, suburban or rural areas have a high neighbourhood quality on average, but at the same time housing costs are lower as these areas are less densely populated. In contrast, urban and metropolitan areas have a low neighbourhood quality on average, but housing costs are high due to the higher population density. Future research may account for the average housing prices in an area to better model the effect of income changes on neighbourhood quality.

My findings regarding changes in income and changes in room stress and neighbourhood quality resonate with other research showing the relative importance of individual resources in English compared to German life courses (Allmendinger and Hinz 1998; Mayer 2005). The "primacy of the market" (Arts and Gelissen 2002: 141) in the English welfare system can also be found in the housing market and links the residential trajectories of individuals more closely with their individual economic resources than in Germany. This also implies that individual misfortune, e.g. union dissolutions, may have stronger immediate negative effects on individuals' residential trajectories in England than in Germany, and for recovery individuals need to participate in the labour market (Vandecasteele 2011). Indeed, my findings show a more negative initial effect of dissolutions on quality of locations in England than in Germany. However, the negative life event of unemployment does not have significant effects on room stress or neighbourhood quality in either country according to my results.

9.1.4 Gender Relations in England, and West and East Germany

In H 5.3, I expect gender inequality in effects of long-distance relocations on careers to be smallest in England, modest in East Germany, and greatest in West Germany on average. In Chapter 8, I test H 5.3 and find that England shows the greatest gender inequality in the effects of long-distance relocations on leaving employment, but in general I find no indication of significant differences in the effects of long-distance relocations across countries. Some evidence for gender inequality in West Germany and no evidence for East Germany is found. The difference between the two parts of Germany is in accordance with the theoretical expectations, but the strong gender inequality in England is unexpected. Regarding wage rates of movers and stayers, no gender differences in any of these contexts is apparent.

First, my hypothesis may have underestimated the importance of gender role attitudes in the three examined contexts. More egalitarian attitudes in East Germany compared to the other two contexts may increase the probability for couples to consider both careers in their relocation decisions in this context. Second, my hypothesis may have overstated the differences in gender relations between England and West Germany and may have paid too little attention to the particularities of long-distance relocations of dual-earner couples in both countries. While welfare policies in England do not hinder partnered women to work, they also provide little direct incentive for an equal division of labour. Especially in situations such as before long-distance relocations in which the labour division may be re-arranged and one partner's career has to be prioritised (Pixley and Moen 2003), this institutional context may not be supportive enough for women to maintain their employment and traditional divisions of labour prevail, i.e. women leave employment to accommodate the careers of their partners.

Gender differences may also be intensified through less extensive child care provision in England compared to West and especially East Germany. Mothers, which traditionally are responsible for child care in couples, may therefore spend more time to arrange new external child care after a long-distance relocation in England than in Germany. Only after child care has been arranged, women take up employment at the new location. This argument would be supported by the finding that women in the second year after a relocation are no longer more likely to be out of work compared to stayers. Future research should consider differences between childless couples and couples with young children to further examine this potential cause for cross-national differences.

Dual-earner couples are less prevalent in West and East Germany than in England. Therefore, dual-earner couples in West and East Germany may be a more selective group of couples than in England. Dual-earner couples in Germany may be more egalitarian on average and they may be more likely to consider both partners careers in their relocation decision. The descriptive statistics also show that couples in which she earns at least 50 per cent of the couple's labour income are twice as frequent in East Germany than in England and West Germany (cf. Table A.13 in the appendix). For these couples, it may be an economic necessity to consider both partners' careers in their relocation decisions. In addition, East German women have more continuous work histories and work longer hours than English and West German women according to the descriptive statistics. These average differences hint at the stronger average labour market attachment of East German women. The frequencies of egalitarian gender role attitudes are not comparable across countries due to the divergent measurement of attitudes and the frequencies of egalitarian couples are very similar in West and East Germany.¹

Regarding the general differences in the chances to leave employment after relocations irrespective of gender, the institutional features of labour markets may play an important role. Due to the higher job mobility and easier access to new jobs, individuals in England may be more willing to relocate and to give up employment initially, since they are more likely to find new employment soon. This may also explain why men in England are slightly more likely to leave employment after relocations compared to men in West and East Germany. In the latter two contexts, access to new jobs is more difficult and this fact may deter potential movers from relocating if they have no new job offers. This interpretation of the findings would be in accordance with other life course research showing the higher instability of English job trajectories (Mayer 2005). Movers in East Germany may also be more likely to relocate to maintain employment due to the unfavourable labour market conditions in East Germany and, thus, the positive effect of relocations on chances to leave employment is weaker in this context than in England and West Germany.

My findings show that the overall effect of relocations on chances to leave employment depends on the institutional context of relocations. For example, while movers are much more likely to leave employment in England than stayers, movers are just as likely as stayers to leave employment in East Germany. The institutional contexts affect women's and men's life courses differently in England and West Germany, but disparities in outcomes of relocations between women and men are relatively similar in these contexts. Stronger differences are apparent between England and West Germany on the one hand, compared to East Germany on the other hand. In the latter context, institutional and cultural differences translate into more equal outcomes for women and men after relocations. These more equal outcomes may be at least partly driven by economic necessity to maintain two incomes, but are also likely to be a consequence of the more egalitarian gender role attitudes in this context and the stronger labour market attachment of East German women.

¹ Past research has shown that East Germans have more egalitarian attitudes than West Germans on average (Adler and Brayfield 1996). Again, this may hint at problems with the variable time use as a proxy for gender role attitudes (cf. Chapter 8, Footnote 13).

9.2 Life Course Regimes and Residential Relocations

The above tests of the cross-national hypotheses provide insights that help to sketch features of life course regimes and average patterns in family and job trajectories that affect residential relocations and shape regularities in residential trajectories. In the following I elaborate features and patterns in four areas: 1) family trajectories; 2) job trajectories; 3) welfare systems; and 4) housing markets. These four aspects of life course regimes seem to be most influential for residential trajectories and relocations. These aspects also depend on each other (cf. Figure 9.1).

First, life course regimes differ in the way they 'schedule' events in the family trajectory and, thereby, the frequency and timing of events differ across regimes. For example, the average age at leaving the parental nests varies considerably between countries and can be related to differences in welfare systems' degree of individualisation among other factors (Aassve et al. 2002). This variance in the age at the beginning of the independent residential trajectory will shape the following mobility over the life course (Vidal 2011). Fertility events are shaped, inter alia, by the compatibility of careers and family responsibilities which relies on the institutions of the labour market and welfare systems (Krüger 2003). The age at first birth as well as the overall number of children will affect the relocation behaviour of individuals and their choices of locations, because families in general prefer better locations for their children (Rowlands and Gurney 2000). Finally, the timing and frequency of union formation and dissolution affects the frequency of relocations and their outcomes. For example, as English individuals form their first consensual unions later than German individuals on average, they already improved their locations as singles, while in Germany improvements come with being in a partnership. A higher frequency of partner events in England increases the frequency of relocation events, the average risk of downward relocations (at least with regard to room stress) and the overall variation in the quality of locations over the life course.

More abstractly, the welfare domain of life course regimes can be distinguished by the degree of being individual-centred or oriented toward family units. For example, the German labour market and welfare system provide family units rather than individuals with income. The normative and monetary importance of families in welfare systems affects the role that the family status plays in relocation decisions and outcomes. The present study shows that being in a partnership and having children has a stronger positive association with improvements after relocations in Germany than in England. Normative concepts about residential locations at particular ages and for particular types of households also vary across life course regimes. This is well researched for the normative age for leaving the parental nest (e.g. Billari and Liefbroer 2007). Research also shows that other relocations are influenced by age norms, e.g. relocations to suburbia for families (e.g. Brun and Fagnani 1994). These norms will shape individual preferences and behaviours regarding relocations. For example, the age norm for relocating to ownership differs strongly between England and Germany and makes improvements in housing at an early age more likely in England (Freeman 1997: 162).

Second, institutions of the educational system and the labour market shape the entry age into the labour market, the frequency of job mobility, and the prospects for income mobility. These institutions affect women and men differently and may (re-) produce gendered divisions of labour (Krüger 2003; Leisering and Schumann 2003; Mayer 2005). It is expected that the first job is not only important due to the labour income, but also since an important new life stage begins. Income security may be higher than while being in education and depending on the tenure of the first job an immediate relocation may be less likely than before taking up the first job. Individuals in England may improve their quality of locations earlier in life, because they enter the labour market at younger ages on average. While I do not find direct effects of job mobility on room stress or neighbourhood quality, income mobility clearly impacts relocation outcomes with regard to room stress. Income mobility is more important in England than in Germany for improving room stress through relocations and part of the higher average income mobility in England.

Women's and men's life courses in couples are linked and life course regimes affect the degree to which each partner's life course is in accordance with traditional gender roles, e.g. by promoting female labour market participation. Life course regimes shape the way that job trajectories and residential trajectories are intertwined for both genders and, thereby, generate gender disparities in relocation outcomes. My findings clearly support this assumption and show that the institutional and cultural context affects labour market outcomes of women and men in dual-earner couples after long-distance relocations. For example, if dual-earner couples relocate in England, the women are likely to experience a disruption of their careers at least temporarily. This interruption may affect the future job trajectory negatively. Thereby, gender inequalities may be re-produced and in future relocation decisions the female partner may have even less influence. If dual-earner couples relocate in East Germany, such adverse effects are less likely.

Third, the institutions of the welfare system in the life course regime moderate the effect of negative life course events such as union dissolution and unemployment on individuals lives. DiPrete (2002) mentions several dimensions that should be differentiated in this regard: 1) the likelihood of negative life course events in a certain context; 2) the immediate consequences of these events; and 3) the degree to which recovery is facilitated after negative events. Depending on how these features are reflected in life course regimes, negative life course events will have particular impacts on residential trajectories. For example, in England dissolutions are relatively frequent and have immediate strong negative consequences for the quality of locations, broadly speaking. However, individuals recover quickly and individuals reach their pre-dissolution quality of locations again quickly. In the long run, negative effects of dissolutions on the quality of locations will be relatively small in England. Welfare systems also differ in the degree to which individuals' living standards depend on their market incomes. The more welfare systems decommodify housing, the less the quality of locations will depend on individuals' market incomes. The findings show that the stronger decommodification of housing in Germany reduces the association between income changes and changes in room stress, but not so regarding neighbourhood quality.

Fourth and especially relevant for relocations but hardly examined in life course research, the institutions of the housing market generate uniformity and regularities in residential trajectories. Housing markets provide vacant dwellings or lots with specific layouts and sizes at particular places for given prices. Individuals are constrained in their choices by these vacancies and outcomes of relocations will depend on the quality of these vacancies At the same time, individuals influence supply with their current demands, but the housing market reacts relatively slowly to changes in the demand side compared to other markets (Mayer 1998: 37). Evidence for the importance of these factors was strong in England and Germany at the regional level, but it can also be expected that the supply of dwellings at the national level affects relocation outcomes, as dwellings have certain regular characteristics within housing markets. For example, dwellings may be more often owned than rented in particular housing markets. In these markets, individuals are more likely to relocate to owned homes than in other markets with lower shares of owned homes, as is the case with England compared to Germany.

Relevant for relocations are also the rules of transactions in the housing market, e.g. rules for repayment of mortgages, and the legal position of market participants. For example, it was shown that the stronger legal protection of tenants in Germany compared to England may affect relocation outcomes after involuntary relocations positively. Markets also differ in the degree to which they are open and non-discriminatory. Regarding the housing market, immigrants are especially vulnerable to discrimination. My findings show that the German housing market is less integrative and immigrants experience divergent changes in the quality of their locations after relocations compared to natives. This is not so much the case in England.

Cohort effects are neglected in the present analysis. With regard to housing the size of cohorts may be especially relevant, as "it is not single individuals but populations which are allocated to and are streamlined through the institutional fabric of society across the lifetime" (Mayer 2003: 466). If housing markets provide a given number of vacancies that fit preferences of certain age groups, the size of particular cohorts will determine the demand for these dwellings while supply changes only slowly due to the particularities of the housing market. The effect of cohort sizes may be less relevant for housing compared with education or mating, because the housing market is less age-specifically structured than for example the educational system. Still cohort size may matter. In addition, particular shared biographical experiences of cohorts, e.g. wars, may have distinct effects on their future relocation behaviour. One more recent experience is the economic crisis beginning in 2008 that will shape the residential biographies of those entering the housing market in Eng-

land around that time, who have already been labelled 'generation rent' (National Centre for Social Research 2011).

9.3 Summary

To generalise from these two concrete examples, it seems that societies with more general social mobility, i.e. higher job and income mobility as well as more frequent family changes, also exhibit higher rates of residential mobility and more variation in outcomes. These societies also show higher risks of downward relocations. Residential trajectories in England are more volatile than in Germany at least with respect to room stress and labour market outcomes, and this is in accordance with other research on life courses in both countries (Allmendinger and Hinz 1998; Mayer 2005). Thus, the typical residential trajectories in both countries seem to comply with the logic of the general life course regimes of a liberal market regime in England and a conservative continental regime in Germany. This association is intuitive, as relocations are triggered by family changes and transitions in the job trajectory.

So far comparative research providing empirical evidence for this association is rare. The findings presented in the present chapter contribute some additional evidence, but the conclusions about life course regimes and regularities in residential trajectories are preliminary and build on only two country cases and face some data limitations. Due to the complexity of the institutional settings in England and Germany, causal links between institutional design features and observed outcomes cannot be established clearly. The present study shares this limitation with most other cross-national comparative research. To identify causal effects, it may be more promising to use difference-in-difference research designs that allow analysis of the effects of changes in particular institutional characteristics over time within one context (Angrist and Pischke 2009: 227ff). However, as some institutional characteristics do not change or change only incrementally, cross-national research provides the second best alternative to analyse institutional context effects and to test the degree of generality of findings. With regard to the residential dimensions of life course regimes, much more remains to be done in the future to better understand institutional context effects that impact residential trajectories.

Also, it is apparent that only some dimensions of the residential trajectory fit well with the general ideas about uniformity and regularities in life courses in England and Germany. For example, the findings differ strongly from prior findings with regard to neighbourhood quality where I find that German life courses are more volatile than English ones. However, this may also be due to the divergent construction of the neighbourhood quality measures. The similarities between typical residential trajectories in both countries should also not be neglected. Many of my findings show striking similarities between residential relocations and their outcomes in England and Germany. A final question remains of how these findings may be applied to other countries, e.g. the US. The US is more similar to England than to Germany concerning institutional characteristics that may affect residential trajectories, but also shows important differences compared to England. For example, home owner rates are similar in the US compared to England (United States Census Bureau 2011), but cushioning of negative life course events is weaker in the US than in England and recovery after negative life course events may be quicker in the US (DiPrete 2002). Residential trajectories in the US show divergent regularities compared to England: Residential mobility is much higher. About 15 per cent of the population moved between 2008 and 2009 in the US (Ihrke, Faber and Koerber 2011). Women have been found to be less negatively affected by long-distance relocations in the US than in the whole UK (e.g. Cooke et al. 2009), while immigrants seem to be less integrated in the US housing market than in the UK (Musterd 2005). More cross-national research including England, Germany and other countries should clarify this divergence.

Chapter 10 Conclusion

This study aims to answer the research question: Which individual, household and contextual conditions shape the outcomes of relocations over the life course? The main hypothesis is that the outcomes of relocations are systematically stratified. Relocations are not positive life course events per se, but are contingent on individual and structural conditions that are further described below. Thereby, relocations differ in their effects on the goal attainment of individuals with regard to their "physical wellbeing and social approval" (Lindenberg 1996: 169). Relocations where individuals move to higher quality places improve wellbeing and increase social approval more than relocations to relatively worse locations. In the present study, the quality of different locations is measured through room stress, neighbourhood quality and risk of leaving employment. These qualities refer to the dwelling, the neighbourhood and the regional labour market respectively. By answering the above research question, the present work contributes innovatively to the literature on residential relocations, social inequality and the life course. More specifically, my analysis extends previous research 1) by treating relocations as contingent life course events that affect social inequality; 2) by providing a more holistic analysis of relocation outcomes; 3) by using panel data and appropriate longitudinal methods; and 4) by adding a cross-nationally comparative perspective.

In this concluding chapter, I provide an answer to the research question by discussing my hypotheses and the empirical evidence (Section 10.1). I acknowledge limitations of my empirical analysis (Section 10.2) and sketch remaining questions for future research (Section 10.3). I then describe selected policy implications of my findings (Section 10.4). I close with some final reflections on the association between residential relocations and social inequality (Section 10.5).

10.1 Discussion of Hypotheses and Empirical Evidence

My theoretical framework is rooted in the life course perspective, but also incorporates elements from the institutional approach, the rational choice approach and the behavioural approach. I assume that individuals relocate voluntarily, because transitions in their life courses reduce the fit between their preferences and their respective current locations. They intentionally decide about whether and where to relocate within the limits of their intellectual capacities and available information. Even though the relocation decision is intentional, I assume that the outcomes of relocations are shaped by various conditions which cause some relocations to be more beneficial, i.e. contributing more to individuals' goal attainment than other relocations. I assume that outcomes of relocations are contingent on resources and restrictions at the individual level as well as constraints and opportunities at the structural level. I identify five sets of conditions that I expect to be especially important to explain variation in outcomes of relocations and that I focus on in my empirical analysis: At the individual level, I examine 1) individual life course conditions; 2) economic resources; and 3) situational conditions of decision making. At the structural level, I consider 4) conditions in the housing market; and 5) gendered roles in couples (which also relate to the individual level). I expect some of these conditions to be mediated by the national institutional and cultural contexts. In Table 10.1, I summarise the empirical evidence for the hypotheses that are derived from these sets of conditions.¹ For each hypothesis, I provide keywords which summarise the content of the hypothesis and I indicate whether the hypothesis is supported in my empirical analysis regarding the different outcomes.

First of all, my empirical analysis supports the main hypothesis that residential relocations are contingent life course events. In contrast to previous research that mainly treated relocations as genuinely positive events, my results show strong variation in outcomes of relocations. While many individuals improve the quality of their locations by relocating, individuals differ strongly in their degree of improvement. In addition, a substantial share of movers relocate horizontally, i.e. they do not change the quality of their locations significantly. This is especially the case regarding neighbourhood quality. Residential relocations also have negative outcomes for a considerable share of movers. Hence, it is relevant to analyse the determinants of this variation to explain disparities in the outcomes of relocations.

My findings show the following general tendencies. Events and statuses in the family trajectory and economic resources shape the outcomes of relocations over the life course most substantially. Relocations especially improve the quality of locations, if they are associated with forming a couple or family and if couples relocate. Singles and ex-partners after union dissolution are less likely to improve the quality of their locations. However, relocations of couples are not necessarily equally beneficial for both partners. Women are more likely to be adversely affected in their ca-

¹ For a more detailed description of these hypotheses cf. Section 2.3.

H Keyword	Room stress	Neighbourhood quality	Leave employment ^a
Life course conditions			
1.1 Couple	+/-	-	(not tested)
1.2 Married	+	-	
1.3 Children	+	+/-	
1.4 Dissolution	+/-	+/-	
1.5 Job mobility	-	-	
1.6 Stability*Context	+	-	
1.7 Pattern*Context	+	-	
Economic resources			
2.1 Income*relocation	+/-	+/-	(not tested)
2.2 Tenure	+	+	
2.3 Tenure*Context	+	+	
Situational conditions			
3.1 Distance	-	+/-	
3.2 Involuntary relocation	+	-	
Conditions in housing market			
4.1 Demand/supply	+/-	-	(not tested)
4.2 Average quality	+	+	
4.3 Income*Context	+	-	
4.4 Immigrant	-	+/-	
Gendered roles			
5.1 Female tied movers	(not tested)		+
5.2 Gendered outcomes			+/-
5.3 Gend. out.*Context			+/-

Table 10.1 Evaluation of hypotheses

Note: + indicates clear support for H, +/- indicates ambiguous findings regarding H, - indicates that H rejected. ^a: Only for dual-earner couples.

reers by long-distance relocations in cultural contexts with more traditional gender ideologies. Furthermore, outcomes are clearly stratified by economic resources and by housing tenure. Individuals improve the quality of their locations more, if they have more resources and if they are able to buy homes. The latter events are turning points in the residential trajectories for most individuals and bring about a strong improvement in the qualities of their locations. With regard to contextual conditions, the impact of the family trajectory on the residential trajectory is mediated by the life course regime and the link between economic resources and quality of locations is weakened by higher decommodification of housing. The tenure structure in the national housing market as well as regional conditions in housing markets are important in explaining outcomes of relocations. I now provide more details regarding these general tendencies with regard to the five sets of conditions and cross-national dissimilarities.

The most important determinants of variation in relocation outcomes are individual life course conditions. At which times in their life courses individuals relocate and with whom they relocate with substantially affects outcomes of relocations. In general, couples are more likely to improve the quality of their locations than singles and, on average, the improvement is more lasting for couples than for singles. Individuals with children improve the quality of their locations more than childless individuals. Relocations that are associated with union dissolutions have a negative effect on the quality of locations. Events in the job trajectory are not important for explaining changes in the quality of locations, but unemployed individuals live in lower quality on average. These findings support the idea of interdependent life course trajectories and show the especially close link between the family trajectory and the residential trajectory. The latter trajectory is instrumental and individuals often relocate to accommodate changes in their family trajectories. Such relocations improve the quality of neighbourhoods especially if individuals form a family and build a nest together. On the one hand, this may be due to social norms about appropriate locations for couples and children, their preferences, and risk-adverse behaviour of families, i.e. families only relocate if they are confident in improving their locations. On the other hand, this may be due to pooled resources of partners and long-time investment in dwellings, since couples are less likely to relocate again.

Improvements in the quality of locations are more strongly associated with family events with regard to room stress than with regard to the quality of neighbourhoods. This is also true for other explanatory variables, and, in general, my models explain changes in room stress better than changes in neighbourhood quality. This may be the case, because neighbourhood quality is a rather subjective quality that is not adequately captured with 'objective' neighbourhood quality measures as discussed in Section 10.2. Subjective evaluation of the neighbourhood may vary considerably across the life course.

Economic resources are important determinants of variation in relocation outcomes. Good locations are scarce and as they are traded in markets, individuals with more economic resources are more likely to purchase better locations and relocate there than individuals with less resources. Thus, economic inequalities are translated into locational inequalities. The more household income couples have, the more they improve the quality of their locations after relocations. Thus, relocations amplify inequalities in the quality of locations for couples. The housing tenure of individuals is closely related to their economic resources and is clearly associated with the quality of locations. Individuals that relocate into ownership improve their locations much more than individuals that relocate into rented accommodations. Social tenants live in the worst quality of locations. Thus, relocations that change the housing tenure of individuals are important turning points in individuals' residential trajectories.

Situational conditions of decision making affect variation in relocation outcomes less than other factors mentioned before. Involuntary relocations are associated with less beneficial outcomes, but the negative effects are only moderate. However, my

analysis focussed mainly on voluntary relocations and a more thorough analysis of involuntary relocations may show a higher impact of these events on relocation outcomes. In addition, even those relocations that I treat as being voluntary may differ in their degree of voluntariness, i.e. some relocations may only be motivated by individual preferences while other relocations are motivated by preferences and external conditions. The degree of voluntariness may affect outcomes, but this cannot be tested with the available data. Longer distances of relocations are modestly associated with less positive changes in neighbourhood quality, but distances do not affect changes in room stress. I do not find evidence for local adjustment relocations after long-distance relocations. Thus, most individuals can be expected to evaluate changes in their qualities of location even in situations where search costs are high or the time for a search is limited.

The opportunities and constraints for movers in regional *housing markets* affect disparities in outcomes. The more high-quality locations that exist in a region, the higher are the chances of individuals to improve their locations through relocating. Regional demand and supply of dwellings are less relevant for relocation outcomes. Due to the higher population density in urban and metropolitan areas, relocations to these areas reduce neighbourhood quality compared to relocations to rural areas, on average. This is not the case for room stress, where average room stress does not differ between urban and rural areas. Movers do not face trade-offs for different dimensions of their qualities of locations in housing markets. That is to say that they do not face deterioration in one quality dimension if they improve another quality dimension. On the contrary, relocation outcomes are mostly complementary and movers improving one dimension of their locations mostly also improve other dimensions. The positive association between improvements in room stress and neighbourhood quality is especially strong.

Immigrants in Germany differ in their relocation outcomes from natives. They are less likely to improve their neighbourhoods through relocations. This may partly be due to restrictions that immigrant movers face in the housing market, e.g. due to discrimination by gate keepers such as real estate agents. Divergence between immigrants and natives may also be caused by differences in their preferences for particular neighbourhoods. However, immigrants do not always have worse relocation outcomes than natives, e.g. relocations reduce inequalities in room stress for immigrants stronger than for natives in Germany.

Disparities in relocation outcomes in dual-earner couples arise due to *gendered roles* of both partners. Women that relocate over long distances are more likely than stayers to leave employment controlling for a wide range of covariates including human capital and employment history in England and West Germany. The negative effect is stronger for women than for men that relocate. The latter are more likely to leave employment only in England. Long-distance relocations do not affect average wage rates in dual-earner couples and there are also no significant gender differences in the effects on wage rates. These findings support the assumption that relocation decisions of households are not solely based on rational utility maximising, but

are likely to be influenced by gendered roles of household members and the subtle power that results from these roles for men. This underlines that intra-household disparities in relocation outcomes should be examined when analysing relocation outcomes. However, direct tests of the impact of gender role attitudes on relocation outcomes do not show significant effects.

While the above findings are similar in England and Germany and show common determinants of relocation outcomes in both contexts, the findings also hint at dissimilarities across *national and cultural contexts*. First, life course conditions have divergent effects on changes in the quality of locations in England and Germany. In accordance with previous, general literature on life course regimes, average residential trajectories in England show more changes within individuals and more heterogeneous changes between individuals regarding room stress. Individuals in England are also more likely than in Germany to experience downward relocations. These findings do not hold regarding neighbourhood quality. English singles are more likely to improve the quality of their locations early in their lives than in Germany, while in the latter context individuals improve the quality of their locations through relocations when they are in couples later in their life courses. Thus, national institutional contexts mediate the way that family trajectories and residential trajectories are linked. While job mobility in general is of little importance for explaining changes in the quality of locations, the effect of job mobility is larger in England than in Germany. This shows the effect of the institutional context of the labour market on relocation outcomes.

Second, the association of the family and the residential trajectories is fundamentally shaped by opportunities in the housing market. Housing markets differ strongly between England and Germany at the national level. Ownership is much more prevalent in England than in Germany and, consequently, turning points from rented to owned dwellings occur earlier in life in England - also for singles. In Germany, these turning points are more closely associated with being in a couple and having children. In Germany, relocations while being in ownership are rare and many individuals improve the quality of their locations while relocating between rented dwellings. In England, repeated relocations between owned homes are not as rare and individuals are likely to trade up their properties over their life courses. The German housing market is more decommodified than the English one and, hence, income is less important in explaining changes in the quality of locations in Germany. This again is in accordance with the wider literature on the liberal market life course regime in England and the conservative life course regime in Germany, where the former regime has a stronger emphasis on free market exchange than the latter regime. Finally, differences between immigrants and natives are less pronounced in the English housing market which may indicate less discrimination against immigrants compared to the German market.

Third, regarding the cultural context of relocations, I find disparities in the effects of relocations on employment for women and men in England and West Germany, but not in East Germany. I argue that this may be due to, first, the more equal gender role attitudes in East Germany, since individuals are embedded in gendered social structures. Second, East Germans are exposed to more adverse labour market conditions which make relocations to maintain employment more likely. Both arguments show that the cultural context and conditions in the labour market affect the intrahousehold disparities in relocation outcomes.

The theoretical framework is mostly supported by the findings and the evidence underlines the central role of the life course perspective, since life course conditions are key to understand the outcomes of relocations. Also very helpful in understanding the variation in relocation outcomes are individual resources and restrictions. In addition, the consideration of structural determinants has proven fruitful at the regional level as well as at the national level. These conditions of relocations help to understand "regularities 'produced' by institutions and structural opportunities" (Diewald and Mayer 2009: 6), which is the main focus of sociological life course research. However, the theoretical framework leaves space for further improvement especially with regard to the actual decision making of individuals which has been mainly treated as a black-box in the present framework. While certain conditions of the decision making situation, e.g. the search costs, are considered, the motivations, intentions and preferences of individuals are not taken into account. The lack of preferences and intentions in the theoretical framework may partly explain why the predictive power of the framework is smaller for neighbourhood quality than for room stress.

10.2 Limitations of the Analyses

Measuring neighbourhood quality presents some empirical hurdles and the analysis in Chapter 7 is limited by the difficulties in comparing the neighbourhood quality measures across countries due to their divergent constructions, and, thus, some of the differences across countries may be statistical artefacts. The data for England and Germany are available for divergently sized geographical units. The neighbourhood units in England are much larger than in Germany. In addition, I exclude changes in neighbourhoods from my analysis by averaging neighbourhood quality over time. Thereby, important processes such as invasion of new residents and succession of old residents are not captured. This is a clear limitation, especially as processes of gentrification, i.e. invasion through high-income residents and succession of low-income households in neighbourhoods, have gained more and more attention in public debate (e.g. Hesselmann 2011). This limitation can only partly be overcome with the available data, because changes in neighbourhood quality cannot be captured adequately with the IMD for England and only after 2004 with the Microm data for Germany.

More generally, my analysis calls into question the relevance of 'objective' neighbourhood quality measures based on data such as the IMD or Microm data.

Over the life course, preferences for neighbourhood quality seem to differ strongly. While it was argued in Section 3.3.1 that low 'objective' neighbourhood quality may affect residents negatively and independently from their preferences, their overall wellbeing may be not negatively affected if they indeed prefer to live in these areas. Thus, one-catch-all measures based on the social composition of neighbourhoods may not be adequate to capture relevant neighbourhood qualities for individuals over their life courses. More promising may be to use indicators of environmental hazards in the neighbourhood. Causal associations between these 'objective' hazards and individuals' wellbeing independent of their preferences may be stronger than with other neighbourhood characteristics.

An additional problem of commonly used 'objective' neighbourhood quality measures is that they are highly negatively correlated with population density. Thus, urban areas have lower average neighbourhood quality than rural areas and this raises the question whether the concept of neighbourhood quality tells us more about the local conditions of individuals' lives than the categories of urban and rural areas would tell us. In future research, quality measures that take the relative position of neighbourhoods within their local environments into account may be an alternative way to tackle this shortcoming.

Although the BHPS and SOEP offer very rich data, both data sets have severe limitations with regard to the analysis of residential relocations: 1) The number of observed relocations is rather small, especially with regard to long-distance relocations. Therefore the analysis of subgroups of movers is hampered. 2) The small case number of immigrants also makes analysis of this subpopulation problematic especially for England and regarding neighbourhood quality changes. Future research for England could make use of the ethnic boost sample for the Understanding Society Panel, which substitutes the BHPS. 3) The BHPS and SOEP do not provide retrospective life history data with regard to residential locations. The lack of retrospective data does not only limit the analysis of determinants of residential relocations, but may also prevent the identification of important causes for variation in relocations' outcomes. For example, repeated movers with more experience may be more likely to improve their quality of location if they relocate. The lack of retrospective data also precludes the analysis of complete individual residential trajectories. Instead only generic residential trajectories that are based on observed sections of various individual life courses are analysed. 4) The BHPS and SOEP offer hardly any information about residential preferences and intentions. Thus, these data do offer only limited opportunity for opening the black-box of decision making regarding relocations.

The present analysis treats relocations as events rather than processes as described by Kley (2011). Thus, only individuals that relocate are analysed, but not individuals that consider or even plan a relocation, but do not realise these aspirations. The latter group of respondents is relevant to analyse, since they may receive positive returns from relocating somewhere else, but are not able to realise these returns. This relates to the discussion about the concept of motility "as the capacity
of entities [...] to be mobile in social and geographic space" (Kaufmann, Bergman and Joye 2004: 750). For example, some immigrants may be restricted in their mobility — at least in Germany — and may not be able to improve the quality of their location by relocating to better locations in the same way as natives. Involuntary stayers are also relevant regarding dual-earner couples, as a high share of partners in dual-earner couples can be expected to be tied stayers, i.e. individuals that would receive positive returns from relocation but stay for the sake of their partners' careers. The SOEP does not allow identification of involuntary stayers, while in the BHPS respondents state their relocation desires and preferences and this additional information may be included in future analyses. However, both data sets offer only limited possibilities to model processes of relocations.

A further limitation of the analysis concerns the question of unequal consequences of certain relocation outcomes for individuals. For example, an increase in room stress of 1 room may have substantially different effects for individuals that spend a lot of time in their dwelling than for individuals that spend only limited time in their dwellings. Similarly, neighbourhood quality changes may be more relevant for individuals that spend more time in their neighbourhoods. Time use data from the BHPS and SOEP may be used to infer some information about the time spent in a dwelling and neighbourhood, but this information is relatively imprecise.

Finally, I refrain from making causal claims based on my empirical evidence. My research design utilises longitudinal data and panel regression methods, which are one step closer to controlling for non-random treatment than cross-sectional methods. Still, unobserved time-varying heterogeneity, e.g. health problems, and self-selection, e.g. due to time trends in particular variables, may bias the coefficients of interest and obscure potential causal relationships. Therefore, I only describe associations between variables and treat these associations as evidence for or against my theoretical model, which is based on assumptions about causal mechanisms. To directly test causal mechanisms to answer specific research questions, more elaborated research designs and econometric methods have to be applied in future research.

10.3 Open Questions for Future Research

Some open questions for future research emerge from the above mentioned limitations of the present analysis and from additional considerations. The present work shows that outcomes of relocations vary over the life course and initial analyses of life course patterns indicate distinct changes in the quality of locations in England and Germany. However, the analyses are mainly focused on relocations as isolated events in the residential trajectory of individuals and are limited due to missing life history data on residential trajectories. One avenue for future research is to extend these analyses in time by examining complete life histories of individuals. To do so research may utilise life history data such as the German Life History Study, the 1970 British Cohort Study and the SHARELIFE data to investigate life histories and to examine cohort effects. Especially the SHARELIFE data set promises to be an inspiring data source for cross-national research in residential trajectories. However, the analysis of longer sections of individuals' life courses will come at the costs of having less detailed information about relocations within the trajectory than when using rich panel data, e.g. with regard to individuals' incomes. In this regard, the present analysis complements such life history studies by analysing single events in depth.

A second avenue for future research is to extend the analysis in geographical scope by considering more country cases. My conclusions about structural conditions of relocation outcomes at the national level are only exploratory and based on two countries so far. Further validation of cross-country differences is necessary. At the end of Chapter 9, I briefly consider potential implications of my research for the US. Another interesting country case is Sweden. Both countries have a highly mobile population, but differ in important dimensions of their institutional design from England and Germany, e.g. with regard to their welfare systems. At the same time, both countries offer good data sources, e.g. the Panel Study of Income Dynamics in the US and register data in Sweden, for conducting longitudinal analysis of residential outcomes. It would be interesting to analyse these countries to establish more reliable conclusions about the effects of national institutional contexts on residential relocations and their outcomes, and about divergent residential trajectories over the life course in different national settings.

Third, I do not consider the decision process of movers, intentions to relocate that are not realised and subjective evaluation of outcomes of residential relocations in the present analysis. It would be worthwhile to consider these issues in future research to better understand the individual decision processes and evaluations of locations that lead to particular outcomes within the limits of structural constraints. To this end, survey instruments must be used that explicitly measure relocations as processes (e.g. Kley 2009: 77ff). These instruments are not yet available in largescale panel data sets, but a part of the analysis of subjective evaluations may be undertaken with the BHPS and SOEP data, e.g. see Section 7.1.1 with regard to subjective neighbourhood evaluation. This extension of the present research would be especially important to better understand disparities in relocation outcomes between natives and immigrants, since the present analysis does not allow differentiation between disparities in outcomes of immigrants that are due to divergent preferences and disparities that are due to different restrictions of relocation behaviours. Additional qualitative data collection may be necessary to fully understand individuals' decision processes and their subjective evaluations. Measuring relocations as processes would also enable researchers to identify individuals that cannot realise their relocation intentions.

Finally, the present analysis may be extended by considering more recent periods. The present analysis ends in 2008 for practical reasons. However, the situation in England may change considerably after the credit crunch following the economic crisis in 2008. In Germany, the economic crisis did not have the same impact in the housing market, mainly for the reason that there is no German subprime mortgage market. However, numerous German metropolitan areas have seen steady increases in average rents in recent years (Grossarth 2012). These dynamics in local housing markets may have far-reaching consequences for disparities in relocation outcomes due to unequal economic resources of individuals. The effects of these changes in the English and German housing markets may be unveiled by extending the observation window up to the most recent waves of the SOEP and by considering further waves from the BHPS subsample of the Understanding Society Panel.

10.4 Policy Implications of Findings

From my findings a number of policy implications can be drawn. I restrict myself to highlight four implications that seem to be especially important. First, my empirical analysis repeatedly underlines that social housing in England and Germany is associated with low quality neighbourhoods and high room stress. Social tenants are excluded from general improvements in room stress in both countries in the last years. These findings are in line with earlier research that shows the increasing marginalisation of social housing in both countries (e.g. Lee and Roseman 1999). Therefore, policy makers should strive to enable social tenants to relocate to privately rented dwellings by allocating sufficient housing benefits to those individuals that are willing to relocate. However, not all social tenants may be willing or able to leave social housing, e.g. because they are discriminated against in the private market or housing benefits are not sufficient to cover rents in the private rental sector in tense housing markets such as in London (Vickery 2012). For these tenants, sufficient numbers of dwellings that meet minimum quality requirements, e.g. at least one room per resident, must be sustained in the social housing sector as a functioning safety net. Thus, policy makers should not further privatise social housing. It also must be a policy goal to invest in the improvement of the existing social housing stock, e.g. to tackle deficits in the built environment. To sustain social housing, to increase the quality of the existing stock and to allocate more housing benefits comes at the costs of increased public spending in times of austerity measures, but investing in the future of social tenants now may save society from long-term costs of adverse health outcomes, lower educational attainments, or exposure to crime of social tenants in increasingly unfit dwellings and bad neighbourhoods.

Second, housing markets are notoriously slow in reacting to changes in the demand structure and adequate dwellings for singles lack in both countries. Singles are in higher room stress than couples and are less likely to change their room stress in cases of relocations. This may mainly be due to inadequate layouts of most dwellings for the needs of singles (e.g. Watson 1986). Neighbourhood quality is lower on average for singles as well, but this may mainly be due to their preferences for more densely populated areas. Considering the growing share of single households in England and Germany — especially in urban areas — (Statistisches Bundesamt 2012a; Office for National Statistics 2012b), policy makers should promote the construction of dwellings with suitable layouts for singles in good quality neighbourhoods to reduce inequalities between family and non-family households. As being a single is a life course stage and not a permanent status for most individuals, these dwellings should be designed in such a way that they can be flexibly merged to form larger dwelling units that also meet the room needs of larger households. Planing neighbourhoods that mix buildings with multiple rented dwellings, detached owned houses and houses with shared ownership will increase the opportunities of non-traditional families and singles to relocate to better neighbourhoods. Introducing this mix of dwelling types in metropolitan and urban areas may also increase the attractiveness of these areas for families and initiate positive neighbourhood change.

Third, my research shows that immigrants in both countries, but especially in Germany, live in lower quality of locations. In Germany, immigrants reduce their room stress through relocations, but evidence suggests that they cannot improve their neighbourhood quality. First, policy makers should combat any kind of discrimination in the housing market. Second, immigrant households may face similar problems as singles with regard to layouts of dwellings that do not meet their room needs as immigrant households are larger on average and more often multi-generational than native households. Therefore, the construction of flexible rental accommodation with layouts that are suitable for larger families and multigenerational families should be promoted. Again, dwelling units should be designed in ways that they can easily be merged to accommodate larger households. In addition, policy makers should intensify their efforts to improve neighbourhood quality in urban areas with high concentration of immigrants, e.g. by reducing throughtraffic, by stimulating local civic initiatives and by allocating loans and funds for the renovation of the built environment. These measures have to be accompanied by strict rent regulation and legal protection of sitting renters to avoid succession processes.

To reduce gender inequalities in labour market outcomes after long-distance relocations for dual-earner couples, policy makers must unequivocally support the equal participation of women and men in the labour market. To this end, goals in the private sphere such as family formation must become reconcilable with goals in one's career for both partners (Diewald 2010: 38). To reconcile partnership and family on the one hand, and occupational mobility requirements on the other hand, several concrete policy recommendations can be made: 1) Some relocations may be avoided by commuting. To facilitate commuting, subsidized public transportation should be expanded. Support for individual transport should be reduced to increase the sustainability of commuting. 2) To mitigate the negative effects of long-distance relocations, only necessary relocations should be promoted and the awareness of potentially negative consequences of relocations for partners should be increased (cf. also Hofmeister and Schneider 2009). 3) Large companies and public employers should be motivated to run dual-career programs to support employment of trailing male and female spouses. 4) The establishment of second homes as an alternative to relocations of whole households should be supported. To this end, the financial burden of commuting and establishing a second home should be further reduced, e.g. through tax breaks. For families, sufficient public child care has to be made available to help temporarily single parents to reconcile family and paid work. 5) To facilitate long-distance commuting and shuttling over the weekend, employers may introduce more flexible time policies. Allowing workers in the knowledge and service sectors, where attendance at the work place may not be permanently necessary, to work part of the week at home and avoiding meetings on Monday mornings and Friday afternoons for all workers may enable more employees to travel to their partners and families over the weekend.

10.5 Final Reflections on Relocations and Social Inequality

Residential locations are an important dimension of social inequality and individuals' places of residence affect their life chances. Places affect individuals beyond other dimensions of social inequality. Individuals living in crowded housing, bad neighbourhoods and stagnating regions can be expected to have worse outcomes regarding, inter alia, their health, subjective wellbeing and economic status than the same individuals living in spacious housing, good neighbourhoods and prosperous regions. The residential dimension of inequality is closely associated with other dimensions, foremost economic inequality. Individuals with less economic resources are also likely to live in low-quality places. However, this association is mediated by various factors such as life course conditions, opportunities and constraints in the housing market, and welfare state interventions. Therefore, it is relevant to analyse determinants of inequality in the residential dimension as a distinct aspect of social inequality. To this end, the analysis of residential relocations is paramount as these events lead individuals to live in particular locations.

While previous research has mainly focused on positive effects of relocations, i.e. improvements in the quality of locations, the present study shows that the outcomes of relocations are contingent on individual resources and restrictions and structural opportunities and constraints. Improvements in the quality of their locations occur mainly if individuals relocate together with a partner, have children, have high economic resources, buy property or relocate out of urban or metropolitan areas. Deteriorations in the quality of locations are most likely if individuals dissolve their union, have to leave their own home, have few economic resources, or relocate into urban or metropolitan areas. Thus, relocations do not automatically improve individual life chances and a significant share of movers are worse off after their relocations compared to their pre-relocation situations. Looking at between-individual differ-

ences, the present study finds relocations to amplify particular inequalities within households between women and men, between low- and high-income households and between immigrant households and native households. Thus, the effects of relocations on social inequalities can only be grasped if these conditions are taken into account.

However, even in post-industrial societies residential relocations are relatively rare events. Few individuals relocate frequently and a high share of the population relocates not more than once or twice in their lives. On the one hand, this rareness of residential relocations limits the importance of the event at the societal scale. On the other hand, the rareness of the event also increases the importance of each single event for individuals' life courses and may have far-reaching repercussions for their lives. Therefore, it is time for life course scholars interested in social inequalities to take the situatedness of life courses in particular places and changes between these places seriously to better understand residential relocations' effects on individuals' lives. The present analysis makes a further step towards a better understanding of these processes. More steps need to follow.

Appendix A Additional Tables

A.1 Research Design and Methods

Recoded variable name Original variable name							
	BHPS	SOEP					
Residential relocations	Xhhmove, Xmovest	qpader, qhader					
Relocations types	Xhhmove, Xmovest, Xage,	qpader, qhader, gebjahr, partz00,					
	Xmastat, Xsppid	partnr00					
Long-distance relocations	Xlad	qkkz_rek					
Room stress	Xhsroom, Xage	room00, gebjahr					
Wage rate	Xfimnl, Xjbhrs, Xjbotpd	labgro00, qp4901, qp5501, qp5401, rp5602					
Women	sex	sex					
Age	Xage	gebjahr					
Age of HH members	Xage	gebjahr					
Education	Xcasmin	casmin00					
Household income (log)	Xhhyneti, Xage	i1110200, gebjahr					
Years in residence	Xplnowy4	moveyr00					
Life course stage	Xhhtype, Xage	typ2hh00, gebjahr					
Child born	Xlwstat, Xrel	qp14221					
Child moved out	Xnwstat, Xrel	qp14217					
Partners moved together	Xlwstat, Xrel	partnr00, qnetto					
Partners moved apart	Xnwstat, Xrel	partnr00, qnetto					
Partner died	Xnwstat, Xrel	qnetto, partnr00, partz00					
1st/2nd generation immig-	maborn, mabrn, paborn, pabrn,	migback					
rant HH	plbornc						
Housing costs	Xxphsn	rent00, qh32					
Employment status	Xjbstat	lfs00, stib00					
Housing tenure	Xtenure	owner00, rsubs00					
Partnership status	Xmastat	partz00					
Promotion, demotion	Xfimnl, Xjbhrs, Xjbotpd	labgro00, qp4901, qp5501,					
		qp5401, rp5602					
Became unemployed	Xjbstat lfs00						
Region	Xregion	qsampreg					
Relocation distance	Xdistmov	distance					
	continued overleaf						

Table A.1 Recoded and original variable names

	continued										
Recoded variable name	Original va	riable name									
	BHPS	SOEP									
Involuntary relocation	Xmovy1, Xmovy2	qh1001, qh1006, qh1011									
Her relative income	Xfiyrl, Xsppid	i1111000, partnr00									
Egalitarian couple	Xopfamd	rp0206, rp0209, rp0212									
Occupational status	Xjbisco	is8800									
Time with employer	Xcjsbgy4	qerwzeit									
Permanent position	Xjbterm, Xjbterm1	qp3501									
Continuous work history	Xleshsm, Xleshem, Xleslen,	from pbiospe: zensor, endy,									
	Xleshst, Xleshsy4, Xleshey4,	beginy, spelltyp, spellnr; from									
	Xcjsbly, Xjbstat, Xjspno, Xjhstat,	artkalen: spellnr, spelltyp, begin,									
	Xjhbgm, Xjhendm, Xjhbgy4,	end, zensor									
	Xjhendy4										
Work hours	Xjbhrs	qp4901, qp5501									

A.2 Setting the Scene: Occurrence and Characteristics of Relocations

Variables		Wo	men		Men			
	М	SD	MIN	MAX	М	SD	MIN	MAX
Relocation	0.07	0.25	0.00	1.00	0.07	0.26	0.00	1.00
Southeast ENG/East GER	0.32	0.47	0.00	1.00	0.32	0.46	0.00	1.00
Time in residence/100	0.12	0.12	0.00	0.79	0.11	0.11	0.00	0.77
Urbanisation (ref: Rural)								
Metropolitan	0.40	0.49	0.00	1.00	0.38	0.49	0.00	1.00
Urban	0.35	0.48	0.00	1.00	0.35	0.48	0.00	1.00
Age/100	0.47	0.15	0.18	0.79	0.48	0.15	0.18	0.79
Partnership status (ref: Single)								
Cohabiting	0.12	0.33	0.00	1.00	0.13	0.34	0.00	1.00
Married	0.66	0.48	0.00	1.00	0.73	0.44	0.00	1.00
Child aged (ref: No child)								
1 to 9	0.24	0.43	0.00	1.00	0.23	0.42	0.00	1.00
10 to 17	0.22	0.41	0.00	1.00	0.20	0.40	0.00	1.00
Family events (ref: No event)								
Child born	0.03	0.18	0.00	1.00	0.04	0.19	0.00	1.00
Child left	0.04	0.20	0.00	1.00	0.04	0.19	0.00	1.00
Job mobility (ref: No mobility)								
Promotion	0.15	0.36	0.00	1.00	0.17	0.38	0.00	1.00
Demotion	0.10	0.29	0.00	1.00	0.11	0.31	0.00	1.00
Became unemployed	0.01	0.12	0.00	1.00	0.02	0.16	0.00	1.00
Job status (ref: Inactive)								
Employed	0.57	0.49	0.00	1.00	0.72	0.45	0.00	1.00
Unemployed	0.02	0.14	0.00	1.00	0.04	0.19	0.00	1.00
Dual-earner couple	0.38	0.49	0.00	1.00	0.46	0.50	0.00	1.00
Income (log)	7.30	0.65	0.00	10.96	7.38	0.65	0.00	10.96
Housing tenure (ref: Owned with	mortgag	ge)						
Social housing	0.16	0.37	0.00	1.00	0.13	0.33	0.00	1.00
Rented privately	0.07	0.25	0.00	1.00	0.07	0.26	0.00	1.00
Owned outright	0.27	0.44	0.00	1.00	0.25	0.43	0.00	1.00
Room stress	-2.00	1.60	-22.00	4.50	-2.10	1.63	-18.00	4.50

 Table A.2 Descriptive statistics, England (Chapter 5)

Data: BHPS waves 1-18 (unweighted)

Variables		Wo	men		Men			
	М	SD	MIN	MAX	М	SD	MIN	MAX
Relocation	0.06	0.24	0.00	1.00	0.06	0.24	0.00	1.00
Southeast ENG/East GER	0.26	0.44	0.00	1.00	0.27	0.44	0.00	1.00
Time in residence/100	0.14	0.14	0.00	0.79	0.14	0.14	0.00	0.79
Urbanisation (ref: Rural)								
Metropolitan	0.21	0.41	0.00	1.00	0.21	0.41	0.00	1.00
Urban	0.47	0.50	0.00	1.00	0.46	0.50	0.00	1.00
Age/100	0.48	0.15	0.18	0.79	0.49	0.14	0.18	0.79
Partnership status (ref: Single)								
Cohabiting	0.09	0.29	0.00	1.00	0.10	0.30	0.00	1.00
Married	0.70	0.46	0.00	1.00	0.77	0.42	0.00	1.00
Child aged (ref: No child)								
1 to 9	0.21	0.41	0.00	1.00	0.22	0.41	0.00	1.00
10 to 17	0.22	0.42	0.00	1.00	0.22	0.41	0.00	1.00
Family events (ref: No event)								
Child born	0.02	0.15	0.00	1.00	0.03	0.16	0.00	1.00
Child left	0.02	0.15	0.00	1.00	0.02	0.15	0.00	1.00
Job mobility (ref: No mobility)								
Promotion	0.11	0.31	0.00	1.00	0.14	0.34	0.00	1.00
Demotion	0.07	0.25	0.00	1.00	0.08	0.28	0.00	1.00
Became unemployed	0.03	0.17	0.00	1.00	0.04	0.20	0.00	1.00
Job status (ref: Inactive)								
Employed	0.52	0.50	0.00	1.00	0.69	0.46	0.00	1.00
Unemployed	0.06	0.23	0.00	1.00	0.06	0.24	0.00	1.00
Dual-earner couple	0.34	0.47	0.00	1.00	0.38	0.49	0.00	1.00
Income (log)	7.25	0.94	0.00	10.85	7.31	0.93	0.00	11.17
Housing tenure (ref: Owned with	mortgag	ge)						
Social housing	0.05	0.22	0.00	1.00	0.05	0.21	0.00	1.00
Rented privately	0.50	0.50	0.00	1.00	0.48	0.50	0.00	1.00
Owned outright	0.20	0.40	0.00	1.00	0.20	0.40	0.00	1.00
Room stress	-1.19	1.53	-14.50	5.00	-1.21	1.53	-14.50	5.00

 Table A.3 Descriptive statistics, Germany (Chapter 5)

Data: SOEP v26 waves 17-26 (unweighted)

A.3 Roomstress and Residential Relocations

Variable		We	omen			N			
	Μ	SD	MIN	MAX	Μ	SD	MIN	MAX	
Room stress	-1.97	1.62	-22.00	4.50	-2.07	1.65	-18.00	4.50	
Southeast England	0.32	0.47	0.00	1.00	0.32	0.46	0.00	1.00	
Urbanisation									
Metropolitan	0.40	0.49	0.00	1.00	0.38	0.49	0.00	1.00	
Rural	0.25	0.43	0.00	1.00	0.26	0.44	0.00	1.00	
Age	46.66	15.50	18.00	79.00	47.40	14.93	18.00	79.00	
Immigrant in HH	0.18	0.38	0.00	1.00	0.17	0.38	0.00	1.00	
Partnership status									
Cohabiting	0.13	0.33	0.00	1.00	0.14	0.35	0.00	1.00	
Married	0.64	0.48	0.00	1.00	0.71	0.45	0.00	1.00	
Family events (ref: No event)									
Child born	0.03	0.18	0.00	1.00	0.04	0.19	0.00	1.00	
Child left	0.04	0.21	0.00	1.00	0.04	0.21	0.00	1.00	
Child aged 0 to 9	0.24	0.43	0.00	1.00	0.23	0.42	0.00	1.00	
Employment status									
Employed	0.57	0.50	0.00	1.00	0.72	0.45	0.00	1.00	
Unemployed	0.02	0.14	0.00	1.00	0.04	0.20	0.00	1.00	
Job mobility									
Promotion	0.15	0.36	0.00	1.00	0.17	0.38	0.00	1.00	
Demotion	0.10	0.29	0.00	1.00	0.11	0.31	0.00	1.00	
Highest education									
Basic	0.39	0.49	0.00	1.00	0.34	0.47	0.00	1.00	
High	0.32	0.47	0.00	1.00	0.39	0.49	0.00	1.00	
Household income (log)	7.30	0.66	0.00	10.96	7.38	0.67	0.00	10.96	
Tenure									
Social housing	0.17	0.37	0.00	1.00	0.13	0.34	0.00	1.00	
Rented privately	0.07	0.26	0.00	1.00	0.08	0.27	0.00	1.00	
Owned outright	0.26	0.44	0.00	1.00	0.25	0.43	0.00	1.00	
Relocation type									
Stable single (≤ 35 years)	0.01	0.09	0.00	1.00	0.01	0.08	0.00	1.00	
Stable single (> 36 years)	0.01	0.08	0.00	1.00	0.01	0.07	0.00	1.00	
Stable couple	0.05	0.21	0.00	1.00	0.05	0.22	0.00	1.00	
Single to couple	0.01	0.08	0.00	1.00	0.01	0.08	0.00	1.00	
Dissolved couple	0.01	0.08	0.00	1.00	0.01	0.08	0.00	1.00	
Other	0.02	0.12	0.00	1.00	0.02	0.13	0.00	1.00	
	со	ntinued	overleaf						

 Table A.4 Descriptive statistics, England (Chapter 6)

		contir	nued					
Variable	Women					Ν	Ien	
	Μ	SD	MIN	MAX	Μ	SD	MIN	MAX
Distance of relocation								
< 1 km	0.02	0.15	0.00	1.00	0.02	0.15	0.00	1.00
1 < 5 km	0.03	0.17	0.00	1.00	0.03	0.17	0.00	1.00
5 < 30 km	0.02	0.15	0.00	1.00	0.02	0.15	0.00	1.00
30+ km	0.01	0.12	0.00	1.00	0.02	0.13	0.00	1.00
Involuntary relocation								
Separation	0.00	0.07	0.00	1.00	0.00	0.06	0.00	1.00
Notice given	0.01	0.07	0.00	1.00	0.01	0.07	0.00	1.00
Housing costs	0.00	0.07	0.00	1.00	0.00	0.06	0.00	1.00
Population balance	3.98	5.55	-11.58	34.46	4.03	5.56	-11.58	34.46
New constructions	7.28	3.23	0.00	30.97	7.32	3.21	0.00	30.97
Buildings with 1-2 dwellings	69.97	7.49	39.69	83.80	70.24	7.31	39.69	83.80

Data: BHPS waves 1-18 (individual level, unweighted)

Variable	Women Men							
	Μ	SD	MIN	MAX	Μ	SD	MIN	MAX
Room stress	-1.18	1.55	-14.50	6.00	-1.20	1.54	-14.50	6.00
East Germany	0.26	0.44	0.00	1.00	0.26	0.44	0.00	1.00
Urbanisation								
Metropolitan	0.21	0.41	0.00	1.00	0.21	0.41	0.00	1.00
Rural	0.32	0.47	0.00	1.00	0.32	0.47	0.00	1.00
Age	47.51	15.03	18.00	79.00	48.43	14.21	18.00	79.00
immigrant in HH	0.21	0.41	0.00	1.00	0.22	0.41	0.00	1.00
Partnership status								
Cohabiting	0.10	0.30	0.00	1.00	0.10	0.30	0.00	1.00
Married	0.69	0.46	0.00	1.00	0.76	0.43	0.00	1.00
Family events (ref: No event)								
Child born	0.03	0.16	0.00	1.00	0.03	0.17	0.00	1.00
Child left	0.03	0.16	0.00	1.00	0.02	0.15	0.00	1.00
Child aged 0 to 9	0.22	0.41	0.00	1.00	0.22	0.41	0.00	1.00
Employment status								
Employed	0.52	0.50	0.00	1.00	0.69	0.46	0.00	1.00
Unemployed	0.06	0.24	0.00	1.00	0.06	0.24	0.00	1.00
Job mobility								
Promotion	0.11	0.31	0.00	1.00	0.14	0.35	0.00	1.00
Demotion	0.06	0.24	0.00	1.00	0.08	0.28	0.00	1.00
Highest education								
Basic	0.45	0.50	0.00	1.00	0.45	0.50	0.00	1.00
High	0.16	0.37	0.00	1.00	0.22	0.41	0.00	1.00
Household income (log)	7.25	0.92	0.00	10.85	7.31	0.91	0.00	11.17
	со	ntinued	overleaf	•				

 Table A.5 Descriptive statistics, Germany (Chapter 6)

		contir	nued						
Variable	Women					Men			
	Μ	SD	MIN	MAX	Μ	SD	MIN	MAX	
Tenure									
Social housing	0.05	0.22	0.00	1.00	0.05	0.22	0.00	1.00	
Rented privately	0.50	0.50	0.00	1.00	0.49	0.50	0.00	1.00	
Owned outright	0.20	0.40	0.00	1.00	0.20	0.40	0.00	1.00	
Relocation type									
Stable single (\leq 35 years)	0.01	0.08	0.00	1.00	0.01	0.08	0.00	1.00	
Stable single (\geq 36 years)	0.01	0.09	0.00	1.00	0.00	0.07	0.00	1.00	
Stable couple	0.04	0.21	0.00	1.00	0.05	0.21	0.00	1.00	
Single to couple	0.01	0.08	0.00	1.00	0.01	0.07	0.00	1.00	
Dissolved couple	0.01	0.09	0.00	1.00	0.01	0.08	0.00	1.00	
Other	0.01	0.11	0.00	1.00	0.01	0.11	0.00	1.00	
Distance of relocation									
< 1 km	0.01	0.12	0.00	1.00	0.01	0.12	0.00	1.00	
1 < 5 km	0.01	0.11	0.00	1.00	0.01	0.11	0.00	1.00	
5 < 30 km	0.01	0.10	0.00	1.00	0.01	0.10	0.00	1.00	
30+ km	0.01	0.07	0.00	1.00	0.01	0.07	0.00	1.00	
Involuntary relocation									
Separation	0.01	0.08	0.00	1.00	0.01	0.08	0.00	1.00	
Notice given	0.00	0.06	0.00	1.00	0.00	0.06	0.00	1.00	
Housing costs	0.01	0.09	0.00	1.00	0.01	0.08	0.00	1.00	
Population balance	9.17	5.06	1.14	52.46	-0.33	6.85	-26.82	31.02	
New constructions	-0.34	6.85	-26.82	31.02	9.20	5.05	1.14	52.46	
Buildings with 1-2 dwellings	47.59	21.76	8.30	88.60	47.78	21.71	8.30	88.60	

Data: SOEP v26 waves 8-25 (individual level, unweighted)

	Model Engl	A.3.1 and	Model Germ	A.3.2 hany
	Women	Men	Women	Men
Urbanisation (ref: rural)				
Metropolitan	0.094	0.097	-0.020	-0.039
L	(1.78)	(1.69)	(-0.67)	(-1.27)
Urban	0.096*	0.029	0.015	-0.000
	(2.35)	(0.67)	(0.71)	(-0.02)
Partnership status (ref: Single)				< / /
Cohabiting	-0.392^{***}	-0.198^{***}	-0.396***	-0.375^{***}
8	(-11.32)	(-4.84)	(-21.22)	(-18.42)
Married	-0.556***	-0.362***	-0.471***	-0.368***
	(-18.77)	(-9.54)	(-32.92)	(-20.84)
Child aged 1 to 9	-0.255***	-0.151***	-0.186***	-0.167***
	(-10.65)	(-5.74)	(-12.69)	(-10.68)
Family events (ref: No event)	(10.00)	(0.7.1)	(12:0))	(10100)
Child born	0.138***	0.188***	0.175***	0.185***
	(4.11)	(5.57)	(7.38)	(7.48)
Child left	-0.464***	-0.472^{***}	-0.397***	-0.429***
	(-15.77)	(-15.15)	(-18.13)	(-17.17)
Job mobility (ref: No mobility)	(10177)	(10110)	(10110)	(1/11/)
Promotion	0.028	0.048**	0.003	0.052***
110110101	(1.56)	(2.66)	(0.27)	(4.48)
Demotion	0.017	0.013	-0.010	0.013
	(0.79)	(0.59)	(-0.65)	(0.90)
Iob status (ref: Inactive)	(0.75)	(0.57)	(0.05)	(0.90)
Employed	-0.009	-0.066*	-0.040**	0.013
Employed	(-0.42)	(-2.00)	(-3.24)	(0.76)
Unemployed	-0.069	-0.016	0.020	0.043*
Chemployed	(-1.42)	(-0.32)	(1.04)	(1.98)
Household income (log)	_0.109***	-0.116***	_0.363***	-0.432^{***}
Household meonie (log)	(-7.91)	(-7.77)	(-32.86)	(-35.69)
Housing tenure (ref: Owned with	(-7.91)	(-7.77)	(-52.80)	(-33.09)
Social housing	0.058***	0 00/***	1 052***	1 082***
Social nousing	(23.00)	(21.55)	(38.02)	(36.70)
Ponted privately	(23.90)	0.535***	0.854***	0.012***
Refited privately	(11.55)	(12.60)	(61.24)	(60.76)
Owned outright	(11.33)	(12.09) -0.066*	(01.24) -0.080***	-0.040^{*}
Gwiled outright	(-0.077)	(-2.20)	(-6.02)	(2.55)
Palacetion	(-2.70)	(-2.20)	(-0.02)	(-2.33)
NEIOCAUOII	-0.106	-0.095	-0.147	-0.146
	(-4./1)	(-3.88)	(-11.11)	(-10.33)
	continued of	overleaf		

Table A.6 Multiple membership multi-level models of room stress with housing market variables

	continu	ied		
	Model Engl	A.3.1 and	Model Gern	A.3.2 nany
	Women	Men	Women	Men
Owned dwellings	-0.019***	-0.016***	-0.010***	-0.010***
	(-6.14)	(-5.00)	(-16.12)	(-16.15)
New constructions	-0.002	-0.006	0.001	0.002
	(-0.70)	(-1.55)	(0.68)	(0.88)
Population balance	0.006**	0.003	0.004**	0.003*
1	(3.06)	(1.38)	(3.03)	(2.47)
Constant	0.421	1.219***	1.757***	2.446***
	(1.36)	(3.30)	(12.49)	(13.68)
Random part variance estimation				
Housing market	0.023	0.011	0.043***	0.034***
-	(1.75)	(1.28)	(5.64)	(5.36)
Individual	1.560***	1.716***	0.961***	0.967***
	(37.71)	(35.56)	(60.34)	(56.82)
Observation	0.599***	0.615***	0.725***	0.727***
	(96.15)	(89.49)	(175.52)	(167.25)
Observations	22,041	19,177	73,054	64,507
Individuals	3,445	3,117	9,609	8,846
Housing markets	114	114	441	440
DIC	54,503.662	48,005.484	192,801.725	170,550.153

Data: BHPS waves 10-18, ONS several years, SOEP v26 waves 17-25 (individual level, un-weighted), INKAR 2010

Note: Random intercept multiple-membership multi-level model (level 1: individual-years, level 2: individuals, level 3: housing markets), dependent variable: room stress, unstandardised coefficients, t statistics in parentheses; *** significant at 0.1% two-tailed, ** significant at 1%, * significant at 5%; models also include covariates period, age and region not shown here.

A.4 Neighbourhood Quality and Residential Relocations

Variable		Wor	nen			М		
	Μ	SD	MIN	MAX	М	SD	MIN	MAX
Neighbourhood quality	-0.01	1.01	-3.27	5.41	0.02	0.99	-3.16	5.41
Southeast England	0.32	0.47	0.00	1.00	0.31	0.46	0.00	1.00
Urbanisation								
Metropolitan	0.38	0.49	0.00	1.00	0.37	0.48	0.00	1.00
Rural	0.26	0.44	0.00	1.00	0.27	0.44	0.00	1.00
Age	47.24	15.23	18.00	79.00	48.26	14.75	18.00	79.00
Immigrant	0.16	0.37	0.00	1.00	0.16	0.37	0.00	1.00
Partnership status								
Cohabiting	0.15	0.36	0.00	1.00	0.16	0.37	0.00	1.00
Married	0.62	0.49	0.00	1.00	0.69	0.46	0.00	1.00
Family events (ref: No event)								
Child born	0.03	0.18	0.00	1.00	0.04	0.19	0.00	1.00
Child left	0.04	0.20	0.00	1.00	0.05	0.21	0.00	1.00
Child in HH (ref: No child)								
age 1 to 9	0.24	0.43	0.00	1.00	0.23	0.42	0.00	1.00
age 10 to 17	0.22	0.42	0.00	1.00	0.20	0.40	0.00	1.00
Employment status								
Employed	0.59	0.49	0.00	1.00	0.73	0.44	0.00	1.00
Unemployed	0.02	0.13	0.00	1.00	0.03	0.16	0.00	1.00
Job mobility								
Promotion	0.17	0.37	0.00	1.00	0.19	0.39	0.00	1.00
Demotion	0.10	0.30	0.00	1.00	0.11	0.31	0.00	1.00
Highest education								
Basic	0.34	0.47	0.00	1.00	0.30	0.46	0.00	1.00
High	0.35	0.48	0.00	1.00	0.40	0.49	0.00	1.00
Household income (log)	7.47	0.65	0.00	10.96	7.54	0.66	0.00	10.96
Tenure								
Social housing	0.14	0.35	0.00	1.00	0.12	0.32	0.00	1.00
Rented privately	0.07	0.26	0.00	1.00	0.07	0.26	0.00	1.00
Owned outright	0.29	0.45	0.00	1.00	0.28	0.45	0.00	1.00
Relocation type								
Stable single (< 35 years)	0.01	0.09	0.00	1.00	0.01	0.08	0.00	1.00
Stable single (\geq 36 years)	0.01	0.08	0.00	1.00	0.01	0.07	0.00	1.00
Stable couple	0.04	0.21	0.00	1.00	0.05	0.22	0.00	1.00
Single to couple	0.01	0.07	0.00	1.00	0.01	0.08	0.00	1.00
Dissolved couple	0.01	0.08	0.00	1.00	0.01	0.08	0.00	1.00
Other	0.01	0.12	0.00	1.00	0.02	0.13	0.00	1.00
	conti	inued o	verleaf					

 Table A.7 Descriptive statistics, England (Chapter 7)

	continued									
Variable	Women					Μ	len			
	М	SD	MIN	MAX	Μ	SD	MIN	MAX		
Distance of relocation										
< 1 km	0.02	0.15	0.00	1.00	0.02	0.15	0.00	1.00		
1 < 5 km	0.03	0.16	0.00	1.00	0.03	0.17	0.00	1.00		
5 < 30 km	0.02	0.15	0.00	1.00	0.02	0.15	0.00	1.00		
30+ km	0.01	0.11	0.00	1.00	0.01	0.12	0.00	1.00		
Involuntary relocation										
Separation	0.00	0.07	0.00	1.00	0.00	0.06	0.00	1.00		
Notice given	0.00	0.07	0.00	1.00	0.00	0.07	0.00	1.00		
Housing costs	0.01	0.07	0.00	1.00	0.00	0.06	0.00	1.00		
Population balance	4.52	4.18	-11.58	17.16	4.56	4.13	-11.58	17.16		
New constructions	7.29	3.23	0.00	30.97	7.32	3.21	0.00	30.97		
Buildings with 1-2 dwellings	70.18	7.16	39.69	83.80	70.45	6.97	39.69	83.80		

Data: BHPS waves 10-18 (individual level, unweighted)

Variable		Women				Men				
	Μ	SD	MIN	MAX	Μ	SD	MIN	MAX		
Neighbourhood quality	-0.00	1.00	-2.53	3.72	0.00	1.00	-2.62	3.72		
East Germany	0.25	0.43	0.00	1.00	0.25	0.43	0.00	1.00		
Urbanisation										
Metropolitan	0.29	0.45	0.00	1.00	0.29	0.45	0.00	1.00		
Rural	0.22	0.42	0.00	1.00	0.23	0.42	0.00	1.00		
Age	48.80	14.95	18.00	79.00	49.89	14.26	18.00	79.00		
Immigrant	0.19	0.40	0.00	1.00	0.20	0.40	0.00	1.00		
Partnership status										
Cohabiting	0.10	0.31	0.00	1.00	0.11	0.31	0.00	1.00		
Married	0.67	0.47	0.00	1.00	0.73	0.44	0.00	1.00		
Family events (ref: No event)										
Child born	0.03	0.16	0.00	1.00	0.03	0.16	0.00	1.00		
Child left	0.02	0.15	0.00	1.00	0.02	0.15	0.00	1.00		
Child in HH (ref: No child)										
age 1 to 9	0.20	0.40	0.00	1.00	0.19	0.39	0.00	1.00		
age 10 to 17	0.21	0.41	0.00	1.00	0.20	0.40	0.00	1.00		
Employment status										
Employed	0.53	0.50	0.00	1.00	0.67	0.47	0.00	1.00		
Unemployed	0.05	0.22	0.00	1.00	0.06	0.24	0.00	1.00		
Job mobility										
Promotion	0.11	0.31	0.00	1.00	0.14	0.34	0.00	1.00		
Demotion	0.07	0.25	0.00	1.00	0.08	0.27	0.00	1.00		
	cont	inued o	verleaf							

 Table A.8 Descriptive statistics, Germany (Chapter 7)

continued										
Variable		We	omen			N	Men			
	Μ	SD	MIN	MAX	Μ	SD	MIN	MAX		
Highest education										
Basic	0.43	0.50	0.00	1.00	0.45	0.50	0.00	1.00		
High	0.17	0.37	0.00	1.00	0.23	0.42	0.00	1.00		
Household income (log)	7.41	0.51	0.00	10.85	7.47	0.51	0.00	11.17		
Tenure										
Social housing	0.04	0.19	0.00	1.00	0.03	0.18	0.00	1.00		
Rented privately	0.48	0.50	0.00	1.00	0.46	0.50	0.00	1.00		
Owned outright	0.23	0.42	0.00	1.00	0.23	0.42	0.00	1.00		
Relocation type										
Stable single (\leq 35 years)	0.01	0.09	0.00	1.00	0.01	0.08	0.00	1.00		
Stable single (\geq 36 years)	0.01	0.09	0.00	1.00	0.01	0.08	0.00	1.00		
Stable couple	0.04	0.20	0.00	1.00	0.04	0.21	0.00	1.00		
Single to couple	0.01	0.08	0.00	1.00	0.01	0.08	0.00	1.00		
Dissolved couple	0.01	0.09	0.00	1.00	0.01	0.08	0.00	1.00		
Other	0.01	0.11	0.00	1.00	0.01	0.10	0.00	1.00		
Distance of relocation										
< 1 km	0.02	0.15	0.00	1.00	0.02	0.15	0.00	1.00		
1 < 5 km	0.02	0.14	0.00	1.00	0.02	0.14	0.00	1.00		
5 < 30 km	0.02	0.13	0.00	1.00	0.02	0.12	0.00	1.00		
30+ km	0.01	0.09	0.00	1.00	0.01	0.09	0.00	1.00		
Involuntary relocation										
Separation	0.01	0.09	0.00	1.00	0.01	0.08	0.00	1.00		
Notice given	0.00	0.05	0.00	1.00	0.00	0.05	0.00	1.00		
Housing costs	0.01	0.10	0.00	1.00	0.01	0.09	0.00	1.00		
Population balance	-0.34	6.84	-26.82	31.02	-0.33	6.85	-26.82	31.02		
New constructions	7.12	4.23	-0.10	36.50	7.14	4.23	-0.10	36.50		
Buildings with 1-2 dwellings	47.60	21.77	8.30	88.60	47.78	21.71	8.30	88.60		

Data: SOEP v26 waves 17-25 (individual level, unweighted)

	Mode Eng	l A.4.1 land	Model Germ	A.4.2 any
	Women	Men	Women	Men
Household income (log)	0.003	-0.008	0.010	0.013
Relocation type (ref: Stayer)	(0.57)	(-1.21)	(1.51)	(1.65)
Stable single (< 35 years)	0.273	0.746	0.618	0.155
g (<u> </u>	(1.50)	(1.10)	(1.30)	(0.42)
Stable single (> 36 years)	-0.853	-0.006	0.284	-0.433
	(-1.15)	(-0.01)	(0.97)	(-1.15)
Stable couple	-0.251	-0.071	-0.567**	-0.753***
1	(-1.26)	(-0.33)	(-2.93)	(-3.80)
Single to couple	-1.198	0.947	0.711	-0.020
	(-1.82)	(1.26)	(1.32)	(-0.04)
Dissolved couple	0.014	-0.675^{*}	-0.318	-0.140
	(0.08)	(-2.27)	(-1.06)	(-0.31)
Other	-0.155	-0.120	0.052	0.573*
	(-0.47)	(-0.66)	(0.31)	(2.18)
Interaction with income				
Stable single (\leq 35 years)	-0.026	-0.106	-0.081	-0.022
	(-1.00)	(-1.16)	(-1.22)	(-0.42)
Stable single (\geq 36 years)	0.120	0.003	-0.039	0.055
	(1.18)	(0.05)	(-0.96)	(1.07)
Stable couple	0.033	0.015	0.084**	0.110***
	(1.25)	(0.52)	(3.22)	(4.10)
Single to couple	0.155	-0.119	-0.087	0.007
	(1.76)	(-1.20)	(-1.22)	(0.10)
Dissolved couple	-0.004	0.075	0.043	0.021
	(-0.14)	(1.89)	(0.99)	(0.35)
Other	0.021	0.018	0.000	-0.070
	(0.46)	(0.73)	(0.00)	(-1.91)

Table A.	Fixed-effects	models of	neighbourhood	quality with	income	interactions
14010 110.	i inca cirecto	mouch of	neignooumoou	quality mith	meonie	meenomo

Data: BHPS 10-18 (individual level, unweighted), IMD 2004 & 2007, SOEP v26 waves 17-25 (individual level, unweighted), Microm 2010

Note: Linear fixed-effects regression model,standardised dependent variable: neighbourhood quality, unstandardised coefficients, t statistics in parentheses; *** significant at 0.1% two-tailed, ** significant at 1%, * significant at 5%; models also include all covariates shown in Table 7.4.

	Model Engl	A.4.3 and	Model	A.4.4
	Women	Men	Women	Men
Southeast ENG/East GER	0.491*** (22.68)	0.313*** (12.95)	-0.901^{***} (-45.15)	-0.837^{***} (-40.39)
Urbanisation (ref: Rural)	()			
Metropolitan	-0.463***	-0.476^{***}	-0.179^{***}	-0.272^{***}
Ĩ	(-23.78)	(-23.89)	(-7.04)	(-10.38)
Urban	-0.419***	-0.370***	0.120***	0.119***
	(-26.02)	(-23.33)	(7.55)	(6.79)
Partnership status (ref: Single)	· · /	· · · ·		
Cohabiting	0.013	0.021	0.091***	0.046***
8	(1.10)	(1.56)	(11.55)	(5.61)
Married	0.070***	0.085***	0.09***	0.057***
	(6.82)	(6.59)	(12.53)	(7.05)
Child aged (ref: No child)	()	()	()	
1 to 9	0.004	0.038***	0.028***	0.03***
	(0.49)	(4.61)	(4.67)	(4.98)
10 to 17	-0.001	-0.002	-0.018***	-0.005
	(-0.14)	(-0.31)	(-3.37)	(-0.98)
Family events (ref: No event)	(••••••)	((•••••)	(0.50)
Child born	0.015	0.004	0.001	0.002
	(1.47)	(0.38)	(0.15)	(0.20)
Child left	0.010	0.009	-0.005	0.003
	(1.14)	(1.00)	(-0.71)	(0.32)
Job events (ref: No event)	(111.)	(1100)	(0001)	(0102)
Promotion	0.000	-0.003	0.002	-0.002
1101101101	(0.01)	(-0.60)	(0.49)	(-0.47)
Demotion	0.006	-0.004	0.005	0.004
Demotion	(1.00)	(-0.56)	(0.93)	(0.82)
Job status (ref: Inactive)	(1.00)	(0.50)	(0.95)	(0.02)
Employed	-0.003	0.027**	0.002	0.01
Employed	(-0.49)	(2.74)	(0.33)	(1.64)
Unemployed	-0.005	-0.018	-0.005	-0.005
enemployed	(-0.33)	(-1.17)	(-0.76)	(-0.68)
Household income (log)	0.013**	0.006	0.033***	0.039***
Household meonie (10g)	(3.16)	(1.30)	(7.44)	(8.19)
Housing tenure (ref: Owned with	mortgage)	(1.50)	(7.1-1)	(0.17)
Social housing	-0.120^{***}	-0.085***	-0.261***	-0.245***
Social nousing	(-8.28)	(-5.43)	(-23.75)	(-21.01)
Rented privately	0.095***	0.044**	-0.187^{***}	-0.177***
Rented privately	(7.62)	(3, 23)	(-28.75)	(-26.29)
Owned outright	-0.011	-0.047***	0.004	0.005
C mice outright	(-1.20)	(-5.14)	(0.69)	(0.81)
	(1.20)	(3.17)	(0.07)	(0.01)
	continued of	overleaf		

 Table A.10
 Multiple membership multi-level models of neighbourhood quality with housing market variables

	continu	ied			
	Model Engl	A.4.3 and	Model Germ	A.4.4 any	
	Women	Men	Women	Men	
Relocation	0.002	0.024**	0.04***	0.042***	
	(0.34)	(3.34)	(8.58)	(8.73)	
Owned dwellings	0.029***	0.025***	0.005***	0.005***	
	(25.09)	(21.58)	(10.73)	(9.04)	
New constructions	0.004***	0.001	0.001	0.001	
	(4.31)	(0.90)	(1.39)	(1.08)	
Population balance	0.006***	0.002**	0.007***	0.007***	
-	(7.01)	(2.47)	(16.49)	(15.24)	
Constant	-2.571***	-2.088***	-1.071***	-0.612***	
	(-22.08)	(-15.03)	(-14.95)	(-6.16)	
Random part variance estimation					
Housing markets	0.145***	0.165***	0.167***	0.161***	
	(5.16)	(5.16)	(11.84)	(11.74)	
Individuals	0.523***	0.531***	0.518***	0.522***	
	(39.93)	(36.86)	(71.81)	(68.26)	
Observations	0.052***	0.053***	0.076***	0.072***	
	(96.56)	(89.06)	(171.07)	(159.52)	
Observations	21,898	19,033	70,598	62,408	
Individuals	3,471	3,052	11,818	10,579	
Housing markets	87	87	413	412	
DIC	844.66	988.22	29,963.88	23,246.26	

Data: BHPS waves 10-18 (individual level, unweighted), IMD 2004 & 2007, ONS several years, SOEP v26 waves 17-25 (individual level, unweighted), Microm 2010, INKAR 2010 Note: Random intercept multiple-membership multi-level model (level 1: individual-years, level 2: individuals, level 3: housing markets), standardised dependent variable: neighbourhood quality, unstandardised coefficients, t statistics in parentheses; *** significant at 0.1% two-tailed, ** significant at 1%, * significant at 5%; models also include covariates period, age and region not shown here.

A.5 Long-Distance Relocations and Subsequent Employment of Dual-Earner Couples

Variable	Women			Men				
	Μ	SD	MIN	MAX	М	SD	MIN	MAX
Leaves employment until $t + 1$	0.04	0.20	0.00	1.00	0.02	0.14	0.00	1.00
Leaves employment until $t + 2$	0.07	0.25	0.00	1.00	0.03	0.17	0.00	1.00
Long-distance relocation	0.03	0.16	0.00	1.00	0.03	0.16	0.00	1.00
Southeast England	0.29	0.46	0.00	1.00	0.30	0.46	0.00	1.00
Age	37.32	8.68	20.00	55.00	39.15	8.65	20.00	55.00
Married	0.79	0.40	0.00	1.00	0.79	0.41	0.00	1.00
Child in HH								
0 to 4	0.17	0.38	0.00	1.00	0.17	0.37	0.00	1.00
5 to 10	0.22	0.42	0.00	1.00	0.23	0.42	0.00	1.00
Owner	0.89	0.31	0.00	1.00	0.89	0.31	0.00	1.00
Education								
Basic	0.22	0.42	0.00	1.00	0.24	0.43	0.00	1.00
Higher	0.38	0.49	0.00	1.00	0.42	0.49	0.00	1.00
Continuous work history	0.82	0.20	0.00	1.00	0.94	0.14	0.00	1.00
Occupational position								
Skilled non-manual	0.49	0.50	0.00	1.00	0.13	0.33	0.00	1.00
Skilled manual	0.05	0.23	0.00	1.00	0.33	0.47	0.00	1.00
Partly & unskilled occupations	0.05	0.23	0.00	1.00	0.05	0.21	0.00	1.00
Time with employer	3.89	4.64	0.00	36.00	4.91	6.03	0.00	40.00
Permanent position	0.96	0.20	0.00	1.00	0.98	0.14	0.00	1.00
Work hours	33.33	11.21	11.00	80.00	45.24	8.60	11.00	80.00
Partner: Age	39.06	8.68	20.00	55.00	37.38	8.68	20.00	55.00
Part.: Education								
Basic	0.24	0.43	0.00	1.00	0.23	0.42	0.00	1.00
Higher	0.42	0.49	0.00	1.00	0.38	0.49	0.00	1.00
Part.: Continuous work history	0.94	0.13	0.00	1.00	0.82	0.20	0.00	1.00
Part.: Occupational position								
Skilled non-manual	0.13	0.34	0.00	1.00	0.49	0.50	0.00	1.00
Skilled manual	0.33	0.47	0.00	1.00	0.05	0.22	0.00	1.00
Partly & unskilled occupations	0.05	0.22	0.00	1.00	0.06	0.24	0.00	1.00
Part.: Time with employer	4.90	6.01	0.00	40.00	3.89	4.65	0.00	36.00
Part.: Permanent position	0.98	0.15	0.00	1.00	0.96	0.21	0.00	1.00
Part.: Working hours	45.08	8.59	11.00	80.00	32.88	11.33	11.00	80.00
Egalitarian couple	0.25	0.43	0.00	1.00	0.25	0.43	0.00	1.00
Her share of couples labour income								
She earns AL 50 %	0.19	0.39	0.00	1.00	0.18	0.38	0.00	1.00
She earns not than 25 %	0.28	0.45	0.00	1.00	0.30	0.46	0.00	1.00
Regional unemployment rate	5.49	2.22	0.70	18.90	5.49	2.22	0.70	18.90

 Table A.11 Descriptive statistics, England (Chapter 8)

Data: BHPS waves 1-18 (individual level, unweighted)

Table A.12	Descriptive	statistics,	West	Germany	(Chapter	8)
------------	-------------	-------------	------	---------	----------	---	---

Variable	Women				Men				
	Μ	SD	MIN	MAX	Μ	SD	MIN	MAX	
Leaves employment until $t + 1$	0.04	0.20	0.00	1.00	0.03	0.16	0.00	1.00	
Leaves employment until $t + 2$	0.06	0.24	0.00	1.00	0.04	0.19	0.00	1.00	
Long-distance relocation	0.02	0.12	0.00	1.00	0.02	0.12	0.00	1.00	
Age	38.34	8.16	20.00	55.00	40.86	8.29	20.00	55.00	
Married	0.83	0.38	0.00	1.00	0.83	0.37	0.00	1.00	
Child in HH (ref: No child)									
0 to 4	0.09	0.29	0.00	1.00	0.09	0.29	0.00	1.00	
5 to 10	0.22	0.42	0.00	1.00	0.22	0.42	0.00	1.00	
Owner	0.48	0.50	0.00	1.00	0.47	0.50	0.00	1.00	
Education									
Basic	0.40	0.49	0.00	1.00	0.50	0.50	0.00	1.00	
Higher	0.12	0.33	0.00	1.00	0.17	0.38	0.00	1.00	
Continuous work history	0.75	0.22	0.00	1.00	0.90	0.14	0.00	1.00	
Occupational position									
Skilled non-manual	0.40	0.49	0.00	1.00	0.13	0.34	0.00	1.00	
Skilled manual	0.11	0.31	0.00	1.00	0.45	0.50	0.00	1.00	
Partly & unskilled occupations	0.10	0.30	0.00	1.00	0.06	0.23	0.00	1.00	
Time with employer	8.39	7.60	0.00	40.80	12.24	9.63	0.00	42.30	
Permanent position	0.74	0.44	0.00	1.00	0.76	0.43	0.00	1.00	
Work hours	30.75	10.16	10.06	70.00	40.87	5.35	11.82	80.00	
Partner: Age	40.83	8.28	20.00	55.00	38.41	8.20	20.00	55.00	
Part.: Education									
Basic	0.50	0.50	0.00	1.00	0.40	0.49	0.00	1.00	
Higher	0.17	0.38	0.00	1.00	0.12	0.33	0.00	1.00	
Part .: Continuous work history	0.90	0.14	0.00	1.00	0.75	0.22	0.00	1.00	
Part.: Occupational position									
Skilled non-manual	0.13	0.34	0.00	1.00	0.40	0.49	0.00	1.00	
Skilled manual	0.45	0.50	0.00	1.00	0.11	0.32	0.00	1.00	
Partly & unskilled occupations	0.06	0.23	0.00	1.00	0.10	0.30	0.00	1.00	
Part.: Time with employer	12.10	9.56	0.00	42.30	8.32	7.62	0.00	40.80	
Part.: Permanent position	0.75	0.43	0.00	1.00	0.73	0.44	0.00	1.00	
Part.: Working hours	40.88	5.56	10.92	80.00	30.59	10.18	10.06	70.00	
Egalitarian couple	0.63	0.48	0.00	1.00	0.63	0.48	0.00	1.00	
Her share of couples labour income									
She earns AL 50 %	0.16	0.36	0.00	1.00	0.15	0.36	0.00	1.00	
She earns not than 25 %	0.27	0.44	0.00	1.00	0.27	0.45	0.00	1.00	
Regional unemployment rate	9.51	3.38	2.40	25.60	9.50	3.39	2.40	25.60	

Data: SOEP v26 waves 8-25 (individual level, unweighted)

Variable	Women				Men				
	Μ	SD	MIN	MAX	Μ	SD	MIN	MAX	
Leaves employment until $t + 1$	0.07	0.25	0.00	1.00	0.06	0.23	0.00	1.00	
Leaves employment until $t + 2$	0.08	0.27	0.00	1.00	0.07	0.26	0.00	1.00	
Long-distance relocation	0.01	0.11	0.00	1.00	0.01	0.11	0.00	1.00	
Age	39.27	7.93	20.00	55.00	41.56	8.02	20.00	55.00	
Married	0.86	0.34	0.00	1.00	0.86	0.34	0.00	1.00	
Child in HH									
0 to 4	0.09	0.28	0.00	1.00	0.09	0.28	0.00	1.00	
5 to 10	0.24	0.43	0.00	1.00	0.24	0.42	0.00	1.00	
Owner	0.45	0.50	0.00	1.00	0.45	0.50	0.00	1.00	
Education									
Basic	0.08	0.27	0.00	1.00	0.16	0.37	0.00	1.00	
Higher	0.37	0.48	0.00	1.00	0.24	0.43	0.00	1.00	
Continuous work history	0.84	0.16	0.00	1.00	0.91	0.11	0.00	1.00	
Occupational position									
Skilled non-manual	0.35	0.48	0.00	1.00	0.09	0.29	0.00	1.00	
Skilled manual	0.08	0.28	0.00	1.00	0.54	0.50	0.00	1.00	
Partly & unskilled occupations	0.05	0.21	0.00	1.00	0.07	0.26	0.00	1.00	
Time with employer	9.32	8.31	0.00	40.50	9.86	9.24	0.00	40.40	
Permanent position	0.70	0.46	0.00	1.00	0.73	0.44	0.00	1.00	
Work hours	37.63	7.31	10.46	70.85	42.56	5.08	10.31	77.00	
Partner: Age	41.59	8.08	20.00	55.00	39.23	7.89	20.00	55.00	
Part.: Education									
Basic	0.16	0.36	0.00	1.00	0.08	0.27	0.00	1.00	
Higher	0.25	0.43	0.00	1.00	0.37	0.48	0.00	1.00	
Part.: Continuous work history	0.91	0.11	0.00	1.00	0.84	0.16	0.00	1.00	
Part.: Occupational position									
Skilled non-manual	0.09	0.28	0.00	1.00	0.34	0.48	0.00	1.00	
Skilled manual	0.54	0.50	0.00	1.00	0.08	0.28	0.00	1.00	
Partly & unskilled occupations	0.07	0.25	0.00	1.00	0.05	0.22	0.00	1.00	
Part.: Time with employer	9.74	9.17	0.00	40.60	9.24	8.31	0.00	40.50	
Part.: Permanent position	0.72	0.45	0.00	1.00	0.69	0.46	0.00	1.00	
Part.: Working hours	42.61	5.28	10.31	77.00	37.61	7.37	10.38	70.85	
Egalitarian couple	0.61	0.49	0.00	1.00	0.61	0.49	0.00	1.00	
Her share of couples labour income									
She earns AL 50 %	0.34	0.47	0.00	1.00	0.34	0.47	0.00	1.00	
She earns not than 25 %	0.12	0.33	0.00	1.00	0.13	0.33	0.00	1.00	
Regional unemployment rate	18.77	3.02	10.10	31.40	18.81	3.04	10.10	31.40	

Table A.13 Descriptive statistics, East Germany (Chapter 8)

Data: SOEP v26 waves 8-25 (individual level, unweighted)

References

- Aarland, K. and Nordvik, V. (2009). On the Path to Homeownership: Money, Family Composition and Low-income Households. *Housing Studies* 24 (1), 81–101.
- Aassve, A., Billari, F. C., Mazzuco, S. and Ongaro, F. (2002). Leaving home: A comparative analysis of ECHP data. *Journal of European Social Policy* 12 (4), 259–275.
- Abbott, A. D. (1997). On the concept of turning point. *Comparative Social Research* 16, 85–105.
- Abraham, M., Auspurg, K. and Hinz, T. (2010). Migration Decisions Within Dual-Earner Partnerships: A Test of Bargaining Theory. *Journal of Marriage and Family* 72 (4), 876–892.
- Adams, R. E. (1992). Is happiness a home in the suburbs? The influence of urban versus suburban neighborhoods on psychological health. *Journal of Community Psychology* 20 (4), 353–372.
- Adler, M. A. and Brayfield, A. (1996). East-West differences in attitudes about employment and family in Germany. *Sociological Quarterly* 37 (2), 245–260.
- Alba, R. D. and Logan, J. R. (1993). Minority Proximity to Whites in Suburbs: An Individual-Level Analysis of Segregation. *American Journal of Sociology* 98 (6), 1388–1427.
- Allison, P. D. (1982). Discrete-Time Methods for the Analysis of Event Histories. Sociological Methodology 13, 61–98.
- (1999). Comparing Logit and Probit Coefficients Across Groups. Sociological Methods and Research 28 (2), 186–208.
- (2009). Fixed effects regression models. Los Angeles: Sage Publications.
- Allmendinger, J. and Hinz, T. (1998). Occupational careers under different welfare regimes: West Germany, Great Britain and Sweden. In: Leisering, L. and Walker, R. (eds.). *The dynamics of modern society*. Bristol: Policy Press, 63–84.
- Andersson, G. (2003). Dissolution of unions in Europe: A comparative overview. MPIDR working paper WP-2003-004. Rostock: Max Planck Institute for Demographic Research. URL: http://www.demogr.mpg.de/papers/working/wp-2003-004.pdf (visited on 11/09/2012).
- Andersson, R., Musterd, S., Galster, G. and Kauppinen, T. M. (2007). What Mix Matters? Exploring the Relationships between Individuals' Incomes and Different Measures of their Neighbourhood Context. *Housing Studies* 22 (5), 637–660.
- Andreß, H.-J., Borgloh, B., Bröckel, M., Giesselmann, M. and Hummelsheim, D. (2006). The Economic Consequences of Partnership Dissolution: A Comparative Analysis of

Panel Studies from Belgium, Germany, Great Britain, Italy, and Sweden. *European Sociological Review* 22 (5), 533–560.

- Angrist, J. D. and Pischke, J.-S. (2009). Mostly harmless econometrics: An empiricist's companion. Princeton: Princeton University Press.
- Anyadike-Danes, M. (2004). The Real North–South Divide? Regional Gradients in UK Male Non-employment. *Regional Studies* 38 (1), 85–95.
- Apps, P. and Rees, R. (2005). Gender, Time Use, and Public Policy over the Life Cycle. Oxford Review of Economic Policy 21 (3), 439–461.
- Arts, W. and Gelissen, J. (2002). Three worlds of welfare capitalism or more? A state-of-theart report. *Journal of European Social Policy* 12 (2), 137–158.
- Atkinson, R. and Kintrea, K. (2001). Disentangling Area Effects: Evidence from Deprived and Non-deprived Neighbourhoods. *Urban Studies* 38 (12), 2277–2298.
- Bailey, N. and Livingston, M. (2008). Selective Migration and Neighbourhood Deprivation: Evidence from 2001 Census Migration Data for England and Scotland. Urban Studies 45 (4), 943–961.
- Balchin, P. (1996a). Introduction. In: Balchin, P. (ed.). Housing policy in Europe. London: Routledge, 1–22.
- (1996b). The United Kingdom. In: Balchin, P. (ed.). *Housing policy in Europe*. London: Routledge, 210–228.
- Barrett, A. L., Oropesa, R. S. and Kanan, J. W. (1994). Neighborhood Context and Residential Mobility. *Demography* 31 (2), 249–270.
- Battu, H., Ma, A. and Phimister, E. (2008). Housing Tenure, Job Mobility and Unemployment in the UK. *Economic Journal* 118, 311–328.
- Baum, S., Arthurson, K. and Rickson, K. (2010). Happy People in Mixed-up Places: The Association between the Degree and Type of Local Socioeconomic Mix and Expressions of Neighbourhood Satisfaction. *Urban Studies* 47 (3), 467–485.
- Bauman, Z. (2000). Globalization: The human consequences. New York: Columbia University Press.
- Beblo, M. (2001). Bargaining over time allocation: Economic modeling and econometric investigation of time use within families. Heidelberg: Physica-Verlag.
- Becker, G. S. (1995). *Human capital: A theoretical and empirical analysis, with special reference to education.* 3rd ed. Chicago: University of Chicago Press.
- Belch, M. A. and Willis, L. A. (2002). Family decision at the turn of the century: Has the changing structure of households impacted the family decision-making process? *Journal of Consumer Behaviour* 2 (2), 111–124.
- Bielby, W. T. and Bielby, D. D. (1992). I Will Follow Him: Family Ties, Gender-Role Beliefs, and Reluctance to Relocate for a Better Job. *American Journal of Sociology* 97 (5), 1241–1267.
- Billari, F. C. and Liefbroer, A. C. (2007). Should I Stay or Should I Go? The Impact of Age Norms on Leaving Home. *Demography* 44 (1), 181–198.
- Bird, G. A. and Bird, G. W. (1985). Determinants of Mobility in Two-Earner Families: Does the Wife's Income Count? *Journal of Marriage and Family* 47 (3), 753–758.
- Bjerk, D. (2009). How Much Can We Trust Causal Interpretations of Fixed-Effects Estimators in the Context of Criminality? *Journal of Quantitative Criminology* 25 (4), 391– 417.
- Blackburn, M. L. (2006). The impact of internal migration on married couples' earnings in Britain, with a comparison to the United States. ISER Working Paper Series 2006-24. Colchester: Institute for Social and Economic Research, University of Essex. URL: https://www.iser.essex.ac.uk/publications/working-papers/iser/2006-24.pdf (visited on 11/09/2012).

- (2010). Internal migration and the earnings of married couples in the United States. Journal of Economic Geography 10 (1), 87–111.
- Blasius, J. and Friedrichs, J. (2007). Internal Heterogeneity of a Deprived Urban Area and its Impact on Residents' Perception of Deviance. *Housing Studies* 22 (5), 753–780.
- Blau, P. M., Duncan, O. D. and Tyree, A. (1967). *The American occupational structure*. New York: Free Press.
- Blood, R. O. and Wolfe, D. M. (1960). Husbands and wives: The dynamics of married living. New York: Free Press.
- Blossfeld, H.-P. (2009). Comparative life course research: Cross-national and longitudinal perspective. In: Elder, G. H. and Giele, J. Z. (eds.). *The craft of life course research*. New York: Guilford Press, 280–306.
- Blossfeld, H.-P. and Drobnič, S. (2001). Theoretical perspectives on couples' careers. In: Blossfeld, H.-P. and Drobnič, S. (eds.). *Careers of couples in contemporary societies*. Oxford: Oxford University Press, 16–50.
- Blossfeld, H.-P. and Rohwer, G. (2002). *Techniques of event history modeling: New approaches to causal analysis.* 2nd ed. Mahwah: Erlbaum.
- Böheim, R. and Taylor, M. P. (2002). Tied down or room to move? Investigating the relationships between housing tenure, employment status and residential mobility in Britain. *Scottish Journal of Political Economy* 49 (4), 369–392.
- (2007). From the dark end of the street to the bright side of the road? The wage returns to migration in Britain. *Labour Economics* 14 (1), 99–117.
- Bolster, A., Burgess, S., Johnston, R., Jones, K., Propper, C. and Sarker, R. (2007). Neighbourhoods, households and income dynamics: A semi-parametric investigation of neighbourhood effects. *Journal of Economic Geography* 7 (1), 1–38.
- Bolt, G. and Kempen, R. v. (2002). Moving Up or Moving Down? Housing Careers of Turks and Moroccans in Utrecht, the Netherlands. *Housing Studies* 17 (3), 401–422.
- (2010). Ethnic Segregation and Residential Mobility: Relocations of Minority Ethnic Groups in the Netherlands. *Journal of Ethnic and Migration Studies* 36 (2), 333–354.
- Bolt, G., Kempen, R. v. and Ham, M. v. (2008). Minority Ethnic Groups in the Dutch Housing Market: Spatial Segregation, Relocation Dynamics and Housing Policy. *Urban Studies* 45 (7), 1359–1384.
- Bonin, H., Eichhorst, W., Florman, C., Hansen, M. O., Skiöld, L., Stuhler, J., Tatsiramos, K., Thomasen, H. and Zimmermann, K. F. (2008). *Geographic Mobility in the European Union: Optimising its Economic and Social Benefits*. IZA Research Report Series 19. Bonn: Institute for the Study of Labour. URL: http://www.iza.org/en/webcontent/ publications/reports/report_pdfs/report_pdfs/iza_report_19.pdf (visited on 11/09/2012).
- Booth, A. L. and Amato, P. R. (1993). Divorce, Residential Change, and Stress. Journal of Divorce & Remarriage 18 (1), 205–214.
- Bourdieu, P. (1984). *Distinction: A social critique of the judgement of taste*. Cambridge: Harvard University Press.
 - (1989). Social Space and Symbolic Power. Sociological Theory 7 (1), 14–25.
- Bourdieu, P. and Wacquant, L. J. D. (1992). *An invitation to reflexive sociology*. Reprint. Chicago: University of Chicago Press.
- Bowes, A. M., Dar, N. S. and Sim, D. F. (2002). Differentiation in Housing Careers: The Case of Pakistanis in the UK. *Housing Studies* 17 (3), 381–399.
- Box-Steffensmeier, J. M. and Jones, B. S. (2007). *Event history modeling: A guide for social scientists*. Reprint. Cambridge: Cambridge University Press.
- Boyle, P. J., Feng, Z. and Gayle, V. (2009). A New Look at Family Migration and Womens Employment Status. *Journal of Marriage and Family* 71, 417–431.
- Boyle, P. J., Halfacree, K. H. and Robinson, V. (1998). Exploring contemporary migration. Harlow: Longman.

- Boyle, P. J., Cooke, T. J., Halfacree, K. H. and Smith, D. P. (1999). Gender inequality in employment status following family migration in GB and the US: The effect of relative occupational status. *International Journal of Sociology and Social Policy* 19 (9/10/11), 115–150.
- Boyle, P. J., Cooke, T. J., Halfacree, K. and Smith, D. P. (2001). A Cross-National Comparison of the Impact of Family Migration on Women's Employment Status. *Demography* 38 (2), 201–213.
- Boyle, P. J., Cooke, T. J., Halfacree, K. H. and Smith, D. P. (2003). The effect of long-distance family migration and motherhood on partnered women's labour-market activity rates in Great Britain and the USA. *Environment and Planning A* 35 (12), 2097–2114.
- Brainard, J., Jones, A., Bateman, I. and Lovett, A. (2004). Exposure to environmental urban noise pollution in Birmingham, UK. Urban Studies 41 (13), 2581–2600.
- Brambor, T., Clark, W. R. and Golder, M. (2006). Understanding Interaction Models: Improving Empirical Analyses. *Political Analysis* 14 (1), 63–82.
- Brännström, L. (2004). Poor Places, Poor Prospects? Counterfactual Models of Neighbourhood Effects on Social Exclusion in Stockholm, Sweden. Urban Studies 41 (13), 2515– 2537.
- Briggs, X. S. d. and Keys, B. J. (2009). Has Exposure to Poor Neighbourhoods Changed in America? Race, Risk and Housing Locations in Two Decades. *Urban Studies* 46 (2), 429–458.
- Brown, L. A. and Moore, E. G. (1970). The intra-urban migration process: A perspective. *Geografiska Annaler* 52B (1), 1–13.
- Browne, W. J. (2012). MCMC estimation in MLwiN: Version 2.25. Bristol: Centre for Multilevel Modelling, University of Bristol. URL: http://www.bristol.ac.uk/cmm/software/ mlwin/download/mcmc-print.pdf (visited on 21/06/2012).
- Browne, W. J., Goldstein, H. and Rasbash, J. (2001). Multiple membership multiple classification (MMMC) models. *Statistical Modelling* 1 (2), 103–124.
- Browning, C. R. and Cagney, K. A. (2003). Moving beyond Poverty: Neighborhood Structure, Social Processes, and Health. *Journal of Health and Social Behavior* 44 (4), 552– 571.
- Brückner, H. (2004). Gender inequality in the life course: Social change and stability in West Germany, 1975 - 1995. Hawthorne: de Gruyter.
- Brückner, H. and Mayer, K. U. (2005). De-Standardization of the Life Course: What it Might Mean? And if it Means Anything, Whether it Actually Took Place? *Advances in Life Course Research* 9, 27–53.
- Brüderl, J. (2000). Regressionsverfahren in der Bevölkerungswissenschaft [Regression methods in demography]. In: Mueller, U. (ed.). Handbuch der Demographie [Handbook of demography]. Vol. 1. Berlin: Springer, 589–642.
- Brun, J. and Fagnani, J. (1994). Lifestyles and Locational Choices—Trade-offs and Compromises: A Case-study of Middle-class Couples Living in the Ile-de-France Region. *Urban Studies* 31 (6), 921–934.
- Buck, N. H. (2000a). Housing, location and residential mobility. In: Berthoud, R. and Gershuny, J. (eds.). Seven years in the lives of British families. Bristol: Policy Press, 133–160.
- (2000b). Using panel surveys to study migration and residential mobility. In: Rose, D.
 (ed.). *Researching social and economic change*. London: Routledge, 250–272.
- (2001). Identifying Neighbourhood Effects on Social Exclusion. Urban Studies 38 (12), 2251–2275.
- Bundesinstitut für Bau- Stadt- und Raumforschung (2009). Laufende Raumbeobachtung -Raumabgrenzungen: Siedlungsstrukturelle Gemeindetypen. URL: http://www.bbsr. bund.de/nn_1067638/BBSR/DE/Raumbeobachtung/Raumabgrenzungen/Siedlungsstr

ukturelleGebietstypen/Gemeindetypen/gemeindetypen__node.html?__nnn=true (visited on 10/04/2012).

- Burgess, E. W. (2007 [1925]). The growth of the city: An introduction to a research project. In: Park, R. E., Burgess, E. W. and McKenzie, R. D. (eds.). *The city*. Chicago: University of Chicago Press, 47–62.
- Burton-Jeangros, C. and Widmer, E. D. (2009). Cumulative and Compensatory Effects over the Life Course. Swiss Journal of Sociology 35 (2), 183–192.
- Busch-Geertsema, V. (2004). The changing role of the state in German housing housing and social policy. *European Journal of Housing Policy* 4 (3), 303–321.
- Bussemaker, J. and Kersbergen, K. v. (1999). Contemporary Social-Capitalist Welfare states and gender inequality. In: Sainsbury, D. (ed.). *Gender and welfare state regimes*. Oxford: Oxford University Press, 15–46.
- Butler, E. W., Chapin, F., Hemmens, G., Kaiser, E., Stegman, M. and Weiss, S. (1969). *Moving Behavior and Residential Choice: A National Survey*. Washington: Highway Research Board, National Research Council.
- Cadwallader, M. T. (1992). *Migration and residential mobility: Macro and micro approaches*. Madison: University of Wisconsin Press.
- Cairney, J. and Boyle, M. H. (2004). Home ownership, mortgages and psychological distress. *Housing Studies* 19 (2), 161–174.
- Cameron, A. C. and Trivedi, P. K. (2010). *Microeconometrics using Stata*. Revised ed. College Station: Stata Press Publications.
- Castles, S. and Miller, M. J. (2003). *The age of migration: International population movements in the modern world*. 3rd ed. New York: Guilford Press.
- Charles, C. Z. (2003). The dynamics of racial residential segregation. Annual Review of Sociology 29 (1), 167–207.
- Christopher, K., England, P., Smeeding, T. M. and Phillips, K. R. (2002). The Gender Gap in Poverty in Modern Nations: Single Motherhood, The Market, and the State. *Sociological Perspectives* 45 (3), 219–242.
- Clampet-Lundquist, S. and Massey, D. S. (2008). Neighborhood Effects on Economic Self-Sufficiency: A Reconsideration of the Moving to Opportunity Experiment. *American Journal of Sociology* 114 (1), 107–143.
- Clark, K. and Drinkwater, S. (2002). Enclaves, Neighbourhood Effects and Employment Outcomes: Ethnic Minorities in England and Wales. *Journal of Population Economics* 15 (1), 5–29.
- Clark, W. A. V. (1981). Residential mobility and behavioral geography: parallelism or interdependence? In: Cox, K. R. and Golledge, R. G. (eds.). *Behavioral problems in geography revisited*. London: Methuen, 182–205.
- (1992). Comparing cross-sectional and longitudinal analyses of residential mobility and migration. *Environment and Planning A* 24, 1291–1302.
- (2007). Race, Class, and Place: Evaluating Mobility Outcomes for African Americans. Urban Affairs Review 42 (3), 295–314.
- Clark, W. A. V., Deurloo, M. C. and Dieleman, F. M. (1997). Entry to Home-ownership in Germany: Some Comparisons with the United States. *Urban Studies* 34 (1), 7–19.
- (2000). Housing consumption and residential crowding in US housing markets. *Journal* of Urban Affairs 22 (1), 49–63.
- (2003). Housing Careers in the United States, 1968-93: Modelling the Sequencing of Housing States. Urban Studies 40 (1), 143–160.
- (2006). Residential Mobility and Neighbourhood Outcomes. *Housing Studies* 21 (3), 323–342.

- Clark, W. A. V. and Drever, A. I. (2000). Residential mobility in a constrained housing market: Implications for ethnic populations in Germany. *Environment and Planning A* 32 (5), 833–846.
- (2001). Do Immigrants Improve Their Housing Quality When They Move? Evidence from the German Socio-Economic Panel. *Vierteljahrshefte zur Wirtschaftsforschung* 70 (1), 87–94.
- Clark, W. A. V. and Huang, Y. (2003). The life course and residential mobility in British housing markets. *Environment and Planning A* 35, 323–339.
- (2004). Linking Migration and Mobility: Individual and Contextual Effects in Housing Markets in the UK. *Regional Studies* 38 (6), 617–628.
- (2006). Balancing move and work: Women's labour market exits and entries after family migration. *Population, Space and Place* 12 (1), 31–44.
- Clark, W. A. V. and Ledwith, V. (2007). How much does income matter in neighborhood choice? *Population Research and Policy Review* 26 (2), 145–161.
- Clark, W. A. V. and Withers, S. D. (2002). Disentangling the interaction of migration, mobility, and labor-force participation. *Environment and Planning A* 34 (5), 923–945.
- Coggon, D., Barker, D. J. P., Inskip, H. and Wield, G. (1993). Housing in Early Life and Later Mortality. *Journal of Epidemiology and Community Health* 47 (5), 345–348.
- Conley, D. (2001). A Room with a View or a Room of One's Own? Housing and Social Stratification. *Sociological Forum* 16 (2), 263–280.
- Cook, C. and Bruin, M. (1994). Determinants of housing quality: A comparison of white, African-American, and Hispanic single-parent women. *Journal of Family and Economic Issues* 15 (4), 329–347.
- Cook, W. L. and Kenny, D. A. (2005). The Actor–Partner Interdependence Model: A model of bidirectional effects in developmental studies. *International Journal of Behavioral Development* 29 (2), 101–109.
- Cooke, L. P. (2011). Gender-class equality in political economies. New York: Routledge.
- Cooke, L. P. and Gash, V. (2010). Wives' Part-time Employment and Marital Stability in Great Britain, West Germany and the United States. *Sociology* 44 (6), 1091–1108.
- Cooke, T. J. (2003). Family Migration and the Relative Earnings of Husbands and Wives. Annals of the Association of American Geographers 93 (2), 338–349.
- (2008a). Gender role beliefs and family migration. *Population, Space and Place* 14 (3), 163–175.
- (2008b). Migration in a family way. *Population, Space and Place* 14 (4), 255–265.
- Cooke, T. J., Boyle, P. J., Couch, K. and Feijten, P. (2009). A Longitudinal Analysis of Family Migration and the Gender Gap in Earnings in the United States and Great Britain. *Demography* 46 (1), 147–167.
- Courgeau, D. (1985). Interaction between spatial mobility, family and carrer life-cycle: A French survey. *European Sociological Review* 1 (2), 139–162.
- (1989). Family Formation and Urbanization. *Population: An English Selection* 44 (1), 123–146.
- (1995). Migration theories and behavioural models. *International Journal of Population Geography* 1, 19–27.
- Crompton, R. and Lyonette, C. (2006). Work-Life 'Balance' in Europe. *Acta Sociologica* 49 (4), 379–393.
- Crowder, K. D. (2000). The Racial Context of White Mobility: An Individual-Level Assessment of the White Flight Hypothesis. *Social Science Research* 29 (2), 223–257.
- Crowder, K. D. and Downey, L. (2010). Interneighborhood Migration, Race, and Environmental Hazards: Modeling Microlevel Processes of Environmental Inequality. *American Journal of Sociology* 115 (4), 1110–1149.

- Crowder, K. D., Hall, M. and Tolnay, S. E. (2011). Neighborhood Immigration and Native Out-Migration. *American Sociological Review* 76 (1), 25–47.
- Crowder, K. D. and South, S. J. (2005). Race, Class, and Changing Patterns of Migration between Poor and Nonpoor Neighborhoods. *American Journal of Sociology* 110 (6), 1715–1763.
- DaVanzo, J. (1981). Microeconomic approaches to studying migration decisions. In: Jong, G. F. d. and Gardner, R. W. (eds.). *Migration decision making*. New York: Pergamon Press, 90–129.
- Daley, P. O. (1998). Black Africans in Great Britain: Spatial Concentration and Segregation. Urban Studies 35 (10), 1703–1724.
- Dannefer, D. (2003). Cumulative Advantage/Disadvantage and the Life Course: Cross-Fertilizing Age and Social Science Theory. *Journals of Gerontology Series B: Psychological Sciences and Social Sciences* 58 (6), S327–S337.
- DeLeire, T. and Kalil, A. (2005). How Do Cohabiting Couples with Children Spend Their Money? *Journal of Marriage and Family* 67 (2), 286–295.
- Deane, G. D. (1990). Mobility and Adjustments: Paths to the Resolution of Residential Stress. *Demography* 27 (1), 65–79.
- Department for Communities and Local Government (2012b). *Live tables on house building*. London. URL: http://www.communities.gov.uk/housing/housingresearch/housingstatistics/housingstatistics/housebuilding/livetables/ (visited on 02/04/2012).
- Deurloo, M. C., Clark, W. A. V. and Dieleman, F. M. (1990). Choice of Residential Environment in the Randstad. *Urban Studies* 27 (3), 335–351.
- (1994). The move to housing ownership in temporal and regional contexts. *Environment* and Planning A 26 (11), 1659–1670.
- Deurloo, M. C., Dieleman, F. M. and Clark, W. A. V. (1997). Tenure choice in the German housing market: A competing risk model. *Tijdschrift voor Economische en Sociale Geografie* 88 (4), 321–331.
- Dewilde, C. (2003). A life-course perspective on social exclusion and poverty. British Journal of Sociology 54 (1), 109–128.
- Di Salvo, P. and Ermisch, J. (1997). Analysis of the Dynamics of Housing Tenure Choice in Britain. *Journal of Urban Economics* 42 (1), 1–17.
- DiPrete, T. A. (2002). Life Course Risks, Mobility Regimes, and Mobility Consequences: A Comparison of Sweden, Germany, and the United States. *American Journal of Sociology* 108 (2), 267–309.
- DiPrete, T. A. and McManus, P. A. (2000). Family Change, Employment Transitions, and the Welfare State: Household Income Dynamics in the United States and Germany. *American Sociological Review* 65 (3), 343–370.
- Diamond, D. B. and Lea, M. J. (1992). Housing Finance in Developed Countries: An International Comparison of Efficiency. *Journal of Housing Research* 3 (1), 1–271.
- Dieleman, F. M. (2001). Modelling residential mobility: A review of recent trends in research. *Journal of Housing and the Built Environment* 16 (3), 249–265.
- Dieleman, F. M., Clark, W. A. V. and Deurloo, M. C. (1995). Falling out of the home owner market. *Housing Studies* 10 (1), 3–15.
- Dieleman, F. M. and Everaers, P. C. J. (1994). From renting to owning: Life course and housing market circumstances. *Housing Studies* 9 (1), 11–25.
- Dietz, R. D. (2002). The estimation of neighborhood effects in the social sciences: An interdisciplinary approach. *Social Science Research* 31, 539–575.
- Diewald, M. (2010). Lebenslaufregime: Begriff, Funktion und Hypothesen zum Wandel [Life course regime. Concept, function and hypotheses of change]. In: Bolder, A., Epping, R., Klein, R., Reutter, G. and Seiverth, A. (eds.). *Neue Lebenslaufregimes - neue Konzepte der Bildung Erwachsener*? Wiesbaden: VS Verlag für Sozialwissenschaften, 25–41.

- Diewald, M. and Faist, T. (2012). From Heterogeneities to Inequalities: Looking at Social Mechanisms as an Explanatory Approach to the Generation of Social Inequalities. SFB 882 Working Paper Series 1. Bielefeld: DFG Research Center (SFB) 882 From Heterogeneities to Inequalities. URL: http://sfb882.uni-bielefeld.de/sites/default/files/ SFB_882_WP_0001_Diewald_Faist.pdf (visited on 14/08/2012).
- Diewald, M. and Mayer, K. U. (2009). The sociology of the life course and life span psychology: Integrated paradigm or complementing pathways? *Advances in Life Course Research* 14 (1–2), 5–14.
- Dingeldey, I. (2007). Wohlfahrtsstaatlicher Wandel zwischen "Arbeitszwang" und "Befähigung": [Changes in welfare states between coercion and activation]. *Berliner Journal für Soziologie* 17 (2), 189–209.
- Dittmann, J. and Goebel, J. (2010). Your House, Your Car, Your Education: The Socioeconomic Situation of the Neighborhood and its Impact on Life Satisfaction in Germany. *Social Indicators Research* 96 (3), 497–513.
- Doling, J. (1999). De-commodification and Welfare: Evaluating Housing Systems. *Housing*, *Theory and Society* 16 (4), 156–164.
- Dorn, V. (1997). Changes in the social rented sector in Germany. *Housing Studies* 12 (4), 463–475.
- Downs, R. M. (1981). Cognitive mapping: A thematic analysis. In: Cox, K. R. and Golledge, R. G. (eds.). *Behavioral problems in geography revisited*. London: Methuen, 95–122.
- Drever, A. I. (2004). Separate Spaces, Separate Outcomes? Neighbourhood Impacts on Minorities in Germany. *Urban Studies* 41 (8), 1423–1439.
- Drever, A. I. and Clark, W. A. V. (2002). Gaining Access to Housing in Germany: The Foreign-minority Experience. Urban Studies 39 (13), 2439–2453.
- Dunn, J. R. (2000). Housing and Health Inequalities: Review and Prospects for Research. *Housing Studies* 15 (3), 341–366.
- Durkheim, E. (1951 [1897]). Suicide: A study in sociology. New York: Free Press.
- Easterlow, D., Smith, S. J. and Mallinson, S. (2000). Housing for Health: The Role of Owner Occupation. *Housing Studies* 15 (3), 367–386.
- Elder, G. H. (1977). Family History and the Life Course. *Journal of Family History* 2 (4), 279–304.
- (1985). Perspectives on the life course. In: Elder, G. H. (ed.). *Life course dynamics*. Ithaca: Cornell University Press, 23–49.
- (1998). The Life Course as Developmental Theory. *Child Development* 69 (1), 1–12.
- (2003). The life course in time and place. In: Heinz, W. R. and Marshall, V. W. (eds.). Social dynamics of the life course. Hawthorne: Aldine de Gruyter, 57–71.
- Elder, G. H. and O'Rand, A. M. (2009). Adult lives in a changing society. In: Heinz, W. R., Huinink, J. and Weymann, A. (eds.). *The life course reader*. Frankfurt/Main: Campus-Verlag.
- Elder, G. H. and Rockwell, R. C. (1979). The Life-Course and Human Development: An Ecological Perspective. *International Journal of Behavioral Development* 2 (1), 1–21.
- Elias, N. (2003 [1969]). Die höfische Gesellschaft: Untersuchungen zur Soziologie des Königtums und der höfischen Aristokratie [The court society]. Reprint. Frankfurt/Main: Suhrkamp.
- Ellen, I. G. and Turner, M. (1997). Does neighbourhood matter? Assessing recent evidence. *Housing Policy Debate* 8 (4), 833–866.
- Emirbayer, M. and Mische, A. (1998). What Is Agency? American Journal of Sociology 103 (4), 962–1023.
- Engels, F. (1970 [1872]). Zur Wohnungsfrage [The housing question]. In: Marx, K. and Engels, F. (eds.). Ausgewählte Schriften in zwei Bänden [Selected works in two volumes]. Vol. 1. Berlin: Dielz-Verlag, 516–529.

- Ermisch, J. and Francesconi, M. (2000). Cohabitation in Great Britain: Not for long, but here to stay. *Journal of the Royal Statistical Society. Series A (Statistics in Society)* 163 (2), 153–171.
- Ermisch, J. and Halpin, B. (2004). Home ownership and social inequality in Britain. In: Kurz, K. and Blossfeld, H.-P. (eds.). *Home ownership and social inequality in comparative perspective*. Stanford: Stanford University Press, 255–280.
- Esping-Andersen, G. (1990). *The three worlds of welfare capitalism*. Cambridge: Polity Press.
- Eurostat (2011). *Glossary:Equivalised disposable income*. Luxembourg. URL: http://epp.eurostat.ec.europa.eu/statistics_explained/index.php/Glossary:Equivalised_disposable_income (visited on 03/04/2012).
- (2012). Statistics: Dynamic database inquiry. European Commission (ed.). Brussels.
 URL: http://epp.eurostat.ec.europa.eu/portal/page/portal/statistics/search_database
 (visited on 19/04/2012).
- Evans, G. W. (2003). The built environment and mental health. *Journal of Urban Health* 80 (4), 536–555.
- Farwick, A. (2009). Segregation und Eingliederung: Zum Einfluss der räumlichen Konzentration von Zuwanderern auf den Eingliederungsprozess [Segregation and integration. On the influence of spatial concentration of migrants on integration]. Wiesbaden: VS Verlag für Sozialwissenschaften.
- Feijten, P. and Ham, M. v. (2007). Residential mobility and migration of the separated. *Demo-graphic Research* 17 (21), 623–654.
- Feijten, P., Hooimeijer, P. and Mulder, C. H. (2008). Residential Experience and Residential Environment Choice over the Life-course. *Urban Studies* 45 (1), 141–162.
- Feijten, P. and Mulder, C. H. (2002). The Timing of Household Events and Housing Events in the Netherlands: A Longitudinal Perspective. *Housing Studies* 17 (5), 773–792.
 - (2005). Life-course Experience and Housing Quality. *Housing Studies* 20 (4), 571–587.
- Ferrarini, T. (2006). Families, states and labour markets: Institutions, causes and consequences of family policy in post-war welfare states. Cheltenham: Edward Elgar.
- Festinger, L. (2001 [1957]). A theory of cognitive dissonance. Reprint. Stanford: Stanford University Press.
- Fielding, A. J. (1992). Migration and social mobility: South-East England as an escalator region. *Regional Studies* 26, 1–15.
- Fischer, P. A. and Malmberg, G. (2001). Settled people don't move: On life-course and (im-)mobility in Sweden. *International Journal of Population Geography* 7, 357–371.
- Fitzpatrick, S. and Pawson, H. (2007). Welfare safety net or tenure of choice? The dilemma facing social housing policy in England. *Housing Studies* 22 (2), 163–182.
- Flippen, C. A. (2010). The Spatial Dynamics of Stratification: Metropolitan Context, Population Redistribution, and Black and Hispanic Homeownership. *Demography* 47 (4), 845–868.
- Flowerdew, R. (1976). Search Strategies and Stopping Rules in Residential Mobility. *Transactions of the Institute of British Geographers* 1 (1), 47–57.
- (1982). Institutional effects on internal migration. In: Flowerdew, R. (ed.). *Institutions and geographical patterns*. London: Croom Helm, 209–227.
- Flowerdew, R. and Al-Hamad, A. (2004). The relationship between marriage, divorce and migration in a British data set. *Journal of Ethnic and Migration Studies* 30 (2), 339– 351.
- Freeman, A. (1997). A cross national study of tenure patterns, housing costs and taxation and subsidy patterns. Scandinavian Housing and Planning Research 14 (4), 159–174.
- Frees, E. W. (2004). Longitudinal and panel data: Analysis and applications in the social sciences. Cambridge: Cambridge University Press.

- Frick, J. (1996). Lebenslagen im Wandel: Determinanten kleinräumlicher Mobilität in Westdeutschland [Changes in living situations. Determinants of residential mobility in West Germany]. Frankfurt/Main: Campus-Verlag.
- Frick, J. and Grimm, S. (2009). Wohnen in Deutschland nach dem Mauerfall: Eine Analyse für die Jahre 1990 bis 2008 auf Basis der Daten des Sozio-oekonomischen Panels (SOEP) [Housing in Germany after the reunification. An analysis based on the SOEP, 1990-2008]. SOEPpapers 236. Berlin: Deutsches Institut für Wirtschaftsforschung. URL: http://www.diw.de/documents/publikationen/73/diw_01.c.343255.de/diw_sp0236.pdf (visited on 14/08/2012).
- Friedrichs, J. and Blasius, J. (2003). Social norms in distressed neighbourhoods: Testing the Wilson hypothesis. *Housing Studies* 18 (6), 807–826.
- Friedrichs, J. and Triemer, S. (2008). *Gespaltene Städte? Soziale und ethnische Segregation in deutschen Groβstädten [Divided cities? Social and ethnic segregation in German cities]*. Wiesbaden: VS Verlag für Sozialwissenschaften.
- Galster, G., Andersson, R., Musterd, S. and Kauppinen, T. M. (2008). Does neighborhood income mix affect earnings of adults? New evidence from Sweden. *Journal of Urban Economics* 63 (3), 858–870.
- Gangl, M. (2010). Causal Inference in Sociological Research. Annual Review of Sociology 36 (1), 21–47.
- Geist, C. and McManus, P. A. (2008). Geographical Mobility over the Life Course: Motivations and Implications. *Population, Space and Place* 14, 283–303.
- George, L. K. (2009). Conceptualizing and Measuring Trajectories. In: Elder, G. H. and Giele, J. Z. (eds.). *The craft of life course research*. New York: Guilford Press, 163– 186.
- Giesselmann, M. and Windzio, M. (2012). Regressionsmodelle zur Analyse von Paneldaten [Regression models for panel data analysis]. Wiesbaden: VS Verlag f
 ür Sozialwissenschaften.
- Gildemeister, R. (2008). Soziale Konstruktion von Geschlecht: "Doing gender" [Social construction of gender. Doing gender]. In: Wilz, S. M. (ed.). Geschlechterdifferenzen
 Geschlechterdifferenzierungen [Gender differences gender differentiation]. Wiesbaden: VS Verlag für Sozialwissenschaften, 167–198.
- Gobillon, L. and Wolff, F.-C. (2011). Housing and Location Choices of Retiring Households: Evidence from France. *Urban Studies* 48 (2), 331–347.
- Goebel, J., Spieß, C. K., Witte, N. R. J. and Gerstenberg, S. (2008). Die Verknüpfung des SOEP mit MICROM-Indikatoren: Der MICROM-SOEP Datensatz [Linking SOEP and MICROM indicators. The MICROM-SOEP data set]. Data Documentation 26. Berlin: Deutsches Institut für Wirtschaftsforschung. URL: http://www.diw.de/documents/publi kationen/73/diw_01.c.78103.de/diw_datadoc_2007-026.pdf (visited on 22/11/2009).
- Golledge, R. G. and Stimson, R. J. (1997). *Spatial behavior: A geographic perspective*. New York: Guilford Press.
- Goodman, J. L. (1981). Information, uncertainty, and the microeconomic model of migration decision making. In: Jong, G. F. d. and Gardner, R. W. (eds.). *Migration decision making*. New York: Pergamon Press, 130–148.
- Gove, W. R. and Hughes, M. (1983). Overcrowding in the household: An analysis of determinants and effects. New York: Academic Press.
- Gramlich, E., Laren, D. and Sealand, N. (1992). Moving into and out of Poor Urban Areas. Journal of Policy Analysis and Management 11 (2), 273–287.
- Grigg, D. B. (1977). E. G. Ravenstein and the "laws of migration". Journal of Historical Geography 3 (1), 41–54.
- Groh-Samberg, O. (2009). Armut, soziale Ausgrenzung und Klassenstruktur: Zur Integration multidimensionaler und längsschnittlicher Perspektiven [Poverty, social exclusion and

class structure. On the integration of multidimensional and longitudinal perspectives]. Wiesbaden: VS Verlag für Sozialwissenschaften.

- Grossarth, J. (2012). *Mieten driften auseinander*. Frankfurt/Main. URL: http://www.faz.net/ aktuell/wirtschaft/preise-in-deutschland-mieten-driften-auseinander-11685858.html (visited on 23/07/2012).
- Guo, G. (1993). Event-History Analysis for Left-Truncated Data. Sociological Methodology 23, 217–243.
- Haisken-DeNew, J. P. and Frick, J. (2005). Desktop Companion to the German Socio-Economic Panel (SOEP): Version 8.0 - Dec 2005. Berlin: Deutsches Institut für Wirtschaftsforschung. URL: http://www.diw.de/documents/dokumentenarchiv/17/ diw_01.c.38951.de/dtc.409713.pdf (visited on 31/03/2011).
- Halaby, C. N. (2004). Panel Models in Sociological Research: Theory into Practice. Annual Review of Sociology 30, 507–544.
- Hall, P. A. and Soskice, D. (2001). An introduction to Varities of Capitalism. In: Hall, P. A. and Soskice, D. (eds.). Varieties of capitalism. Oxford: Oxford University Press, 1–68.
- Ham, M. v. and Feijten, P. (2008). Who wants to leave the Neighbourhood? The effect of being different from the neighbourhood population on wishes to move. *Environment* and Planning A 40 (5), 1151–1170.
- Ham, M. v. and Manley, D. (2009). Social housing allocation, choice and neighbourhood ethnic mix in England. *Journal of Housing and the Built Environment* 24 (4), 407–422.
- Hardill, I. (2002). Gender, migration and the dual career household. London: Routledge.
- Hardill, I. and Wheatley, D. (2010). Dual Career Couples, Gender and Migration. In: Reuschke, D. (ed.). Wohnen und Gender [Housing and gender]. Wiesbaden: VS Verlag für Sozialwissenschaften, 239–259.
- Harris, D. R. (1999). "Property Values Drop When Blacks Move in, Because...": Racial and Socioeconomic Determinants of Neighborhood Desirability. *American Sociological Review* 64 (3), 461–479.
- Hastings, A. (2009). Neighbourhood Environmental Services and Neighbourhood 'Effects': Exploring the Role of Urban Services in Intensifying Neighbourhood Problems. *Housing Studies* 24 (4), 503–524.
- Hausman, J. A. (1978). Specification Tests in Econometrics. *Econometrica* 46 (6), 1251– 1271.
- Häußermann, H. and Siebel, W. (2000). Soziologie des Wohnens: Eine Einführung in Wandel und Ausdifferenzierung des Wohnens [Sociology of housing. An introduction in change and differentiation of housing]. 2nd ed. Weinheim: Juventa-Verlag.
- Heaton, T. B. and Call, V. R. A. (1995). Modeling Family Dynamics with Event History Techniques. *Journal of Marriage and Family* 57 (4), 1078–1090.
- Hedman, L., Ham, M. v. and Manley, D. (2011). Neighbourhood choice and neighbourhood reproduction. *Environment and Planning A* 43 (6), 1381–1399.
- Heinz, W. R. (2009). Status passages as micro-macro linkages in life course research. In: Heinz, W. R., Huinink, J. and Weymann, A. (eds.). *The life course reader*. Frankfurt/Main: Campus-Verlag, 473–486.
- Heinz, W. R., Huinink, J., Swader, C. S. and Weymann, A. (2009). General Introduction. In: Heinz, W. R., Huinink, J. and Weymann, A. (eds.). *The life course reader*. Frankfurt/Main: Campus-Verlag, 15–30.
- Helbig, M. (2010). Neighborhood does matter! Soziostrukturelle Nachbarschaftscharakteristika und Bildungserfolg [Neighbourhood does matter! Socio-structural neighbourhood characteristics and educational success]. Kölner Zeitschrift für Soziologie und Sozialpsychologie 62, 655–679.

- Helderman, A. C. and Mulder, C. H. (2007). Intergenerational Transmission of Homeownership: The Roles of Gifts and Continuities in Housing Market Characteristics. *Urban Studies* 44 (2), 231–247.
- Herlyn, I. and Herlyn, U. (1983). Wohnverhältnisse in der Bundesrepublik [Housing in Germany]. 2nd ed. Frankfurt/Main: Campus-Verlag.
- Hesselmann, M. (2011). Berlin's gentrification needs radical ideas. London. URL: http:// www.guardian.co.uk/commentisfree/2011/feb/07/berlin-gentrification-prices (visited on 29/08/2012).
- Hipp, J. R. (2007). Block, Tract, and Levels of Aggregation: Neighborhood Structure and Crime and Disorder as a Case in Point. American Sociological Review 72 (5), 659–680.
- Hochschild, A. (1989). *The second shift: Working parents and the revolution at home*. New York: Viking Press.
- Hochstadt, S. (1999). *Mobility and modernity: Migration in Germany, 1820 1989*. Ann Arbor: University of Michigan Press.
- Hofmeister, H. and Schneider, N. F. (2009). Job Mobilities in Europe: Core Findings, Policy Implications and Future Outlook. In: Schneider, N. F. and Collet, B. (eds.). *Causes* and Consequences of Job-Related Spatial Mobility in Cross-National Comparison. Leverkusen: Budrich, 337–356.
- Holland, J. (2012). Home and Where the Heart Is: Marriage Timing and Joint Home Purchase. *European Journal of Population* 28 (1), 65–89.
- Horr, A. (2008). Ethnische und soziale Unterschiede der Wohnungssuche und Wohnortwahl [Ethnic and social differences in residential search and choice]. In: Hillmann, F. and Windzio, M. (eds.). *Migration und städtischer Raum [Migration and urban areas]*. Opladen: Budrich, 175–192.
- Hougaard, P. (2000). Analysis of multivariate survival data. New York: Springer.
- Howden-Chapman, P. (2004). Housing Standards: A Glossary of Housing and Health. *Journal of Epidemiology and Community Health* 58 (3), 162–168.
- Huang, Y. (2003). A room of one's own: Housing consumption and residential crowding in transitional urban China. *Environment and Planning A* 35 (4), 591–614.
- Ihrke, D. K., Faber, C. S. and Koerber, W. K. (2011). Geographical Mobility: 2008 to 2009. Current Population Reports P20-565. Washington and D.C.: United States Census Bureau. URL: http://www.census.gov/prod/2011pubs/p20-565.pdf (visited on 11/09/2012).
- Ineichen, B. (1993). *Homes and health: How housing and health interact*. London: E & FN Spon.
- Jacobsen, J. P. and Levin, L. M. (1997). Marriage and Migration: Comparing Gains and Losses from Migration for Couples and Singles. *Social Science Quarterly* 78 (3), 688– 709.
- (2000). The Effects of Internal Migration on the Relative Economic Status of Women and Men. *Journal of Socio-Economics* 29, 291–304.
- Jansen, S. (2012). What is the worth of values in guiding residential preferences and choices? *Journal of Housing and the Built Environment* 27 (3), 273–300.
- Jarvis, S. and Jenkins, S. P. (1998). Income and poverty dynamics in Great Britain. In: Leisering, L. and Walker, R. (eds.). *The dynamics of modern society*. Bristol: Policy Press, 145–160.
- Jaschinski, I. (2011). Der Übergang in eine nacheheliche Partnerschaft: Eine vergleichende Analyse zwischen Männern und Frauen auf Basis des deutschen Generations and Gender Survey [Re-partnering after divorce in Germany. A comparison between men and women based on an analysis with the Generations and Gender Survey]. Zeitschrift für Familienforschung 23 (2), 219–240.
- Jong, G. F. d. and Fawcett, J. T. (1981). Motivations for Migration: An Assessment and a Value-Expectancy Research Model. In: Jong, G. F. d. and Gardner, R. W. (eds.). *Migration decision making*. New York: Pergamon Press, 13–58.
- Jong, G. F. d. and Gardner, R. W. (1981). Introduction and overview. In: Jong, G. F. d. and Gardner, R. W. (eds.). *Migration decision making*. New York: Pergamon Press, 1–12.
- Jürges, H. (1998). Beruflich bedingte Umzüge von Doppelverdienern: Eine empirische Analyse mit Daten des sozio-ökonomischen Panels [Occupational motivated moves of dualearner couples. An empirical analysis with the SOEP]. Zeitschrift für Soziologie 27, 358–377.
- (2006). Gender ideology, division of housework, and the geographic mobility of families. *Review of Economics of the Household* 4 (4), 299–323.
- Kalter, F. (2000). Theorien der Migration: [Theories of migration]. In: Mueller, U. (ed.). Handbuch der Demographie [Handbook of demography]. Vol. 1. Berlin: Springer, 438– 475.
- Kaufmann, V., Bergman, M. M. and Joye, D. (2004). Motility: Mobility as capital. International Journal of Urban and Regional Research 28 (4), 745–756.
- Kearns, A. and Mason, P. (2007). Mixed Tenure Communities and Neighbourhood Quality. *Housing Studies* 22 (5), 661–691.
- Kearns, A. and Parkes, A. (2003). Living in and leaving poor neighbourhood conditions in England. *Housing Studies* 18 (6), 827.
- Kearns, A., Hiscock, R., Ellaway, A. and MaCintyre, S. (2000). 'Beyond Four Walls': The Psycho-social Benefits of Home: Evidence from West Central Scotland. *Housing Studies* 15 (3), 387–410.
- Kemeny, J. (1995). From Public Housing to the Social Market: Rental Policy Strategies in Comparative Perspective. Abingdon: Routledge.
- (2006). Corporatism and Housing Regimes. *Housing, Theory and Society* 23 (1), 1–18.
- Kemp, P. A. and Keoghan, M. (2001). Movement Into and Out of the Private Rental Sector in England. *Housing Studies* 16 (1), 21–37.
- Kendig, H. L. (1984). Housing careers, life cycle and residential mobility: Implications for the housing market. Urban Studies 21 (3), 271–283.
- Kenny, D. A., Kashy, D. A. and Cook, W. L. (2006). *Dyadic data analysis*. New York: Guilford Press.
- Kilkey, M. and Bradshaw, J. (1999). Lone mothers, economic well-being, and policies. In: Sainsbury, D. (ed.). *Gender and welfare state regimes*. Oxford: Oxford University Press, 147–184.
- Kilper, H. (2004). Regionale Disparitäten: Kein Thema von gestern [Regional disparities. Still relevant today]. In: Müller, B. and Fürst, D. (eds.). *Steuerung und Planung im Wandel [Changes in governance and planing]*. Wiesbaden: VS Verlag für Sozialwissenschaften, 35–52.
- Kirchner, J. (2007). The Declining Social Rental Sector in Germany. European Journal of Housing Policy 7 (1), 85–101.
- Kleinhans, R. (2003). Displaced but still Moving Upwards in the Housing Career? Implications of Forced Residential Relocation in the Netherlands. *Housing Studies* 18 (4), 473– 499.
- Kley, S. (2009). *Migration im Lebensverlauf [Migration in the life course]*. Wiesbaden: VS Verlag für Sozialwissenschaften.
- (2011). Explaining the Stages of Migration within a Life-course Framework. *European Sociological Review* 27 (4), 469–486.
- Kley, S. and Mulder, C. H. (2010). Considering, planning, and realizing migration in early adulthood: The influence of life-course events and perceived opportunities on leaving the city in Germany. *Journal of Housing and the Built Environment* 25 (1), 73–94.

- Kling, J. R., Liebman, J. B. and Katz, L. F. (2007). Experimental Analysis of Neighborhood Effects. *Econometrica* 75 (1), 83–119.
- Kohler, U. and Kreuter, F. (2005). *Data analysis using Stata*. College Station: Stata Press Publications.
- Kohli, M. (2009). The World we forgot: A historical review of the life course. In: Heinz, W. R., Huinink, J. and Weymann, A. (eds.). *The life course reader*. Frankfurt/Main: Campus-Verlag, 64–90.
- Kohn, M. L. (1987). Cross-National Research as an Analytic Strategy: American Sociological Association, 1987 Presidential Address. *American Sociological Review* 52 (6), 713–731.
- Konietzka, D. and Huinink, J. (2003). Die De-Standardisierung einer Statuspassage? Zum Wandel des Auszugs aus dem Elternhaus und des Übergangs in das Erwachsenenalter in Westdeutschland [De-standardisation of a status passage? About the transformation of leaving the parental nest and the transition to adulthood in West Germany]. Soziale Welt 54 (3), 285–312.
- König, W., Lüttinger, P. and Müller, W. (1988). Methodological Foundations and the Construction of a Comparative Educational Scale: A Comparative Analysis of the Development and Structure of Educational Systems. CASMIN Working Paper 12. Mannheim: University of Mannheim.
- Korpi, M., Clark, W. A. V. and Malmberg, B. (2011). The urban hierarchy and domestic migration: The interaction of internal migration, disposable income and the cost of living, Sweden 1993–2002. *Journal of Economic Geography* 11 (6), 1051–1077.
- Kroh, M. (2009). Short-Documentation of the Update of the SOEP-Weights, 1984-2008. Berlin. URL: http://www.diw.de/documents/dokumentenarchiv/17/diw_01.c.342852. de/soep_weights_shortdocu.pdf (visited on 20/08/2012).
- Krüger, H. (2003). The life-course regime: Ambiguities between interrelatedness und individualization. In: Heinz, W. R. and Marshall, V. W. (eds.). Social dynamics of the life course. Hawthorne: Aldine de Gruyter, 33–56.
- Krüger, H. and Baldus, B. (1999). Work, Gender and the Life Course: Social Construction and Individual Experience. *Canadian Journal of Sociology* 24 (3), 355–379.
- Kulu, H. (2002). Socialization and residence: Ethnic return migrants in Estonia. *Environment* and Planning A 34 (2), 289–316.
- (2008). Fertility and spatial mobility in the life course: Evidence from Austria. *Environment and Planning A* 40 (3), 632–652.
- Kulu, H. and Billari, F. C. (2006). Migration to urban and rural destinations in post-Soviet Estonia: A multilevel event-history analysis. *Environment and Planning A* 38 (4), 749– 764.
- Kurz, K. (2004). Home ownership and social inequality in West Germany. In: Kurz, K. and Blossfeld, H.-P. (eds.). *Home ownership and social inequality in comparative perspective*. Stanford: Stanford University Press, 21–60.
- Lampert, H. and Althammer, J. (2004). *Lehrbuch der Sozialpolitik [Introduction to social policy]*. 7th ed. Berlin: Springer.
- LeClere, F. B. and McLaughlin, D. K. (1997). Family migration and changes in women's earnings: A decomposition analysis. *Population Research and Policy Review* 16 (4), 315–335.
- Lee, B. A., Oropesa, R. S. and Kanan, J. W. (1994). Neighbourhood context and residential mobility. *Demography* 31 (2), 249–265.
- Lee, E. S. (1966). A Theory of Migration. Demography 3 (1), 47–57.
- Lee, S. W. and Roseman, C. C. (1999). Migration Determinants and Employment Consequences of White and Black Families, 1985–1990. *Economic Geography* 75 (2), 109– 133.

- Lefebvré, H. (1991 [1974]). The production of space. Oxford: Blackwell Publishing.
- Leisering, L. (2003). Government and the Life Course. In: Mortimer, J. T. and Shanahan, M. J. (eds.). *Handbook of the life course*. New York: Springer, 205–225.
- Leisering, L. and Schumann, K. F. (2003). How institutions shape the German life course. In: Heinz, W. R. and Marshall, V. W. (eds.). *Social dynamics of the life course*. Hawthorne: Aldine de Gruyter, 193–209.
- Lersch, P. M. (2013, forthcoming). Long-Distance Moves and Employment of Women in Dual-Earner Couples in Britain and Germany. *Schmollers Jahrbuch* 133 (2).
- (2013). Place Stratification or Spatial Assimilation? Neighbourhood Quality Changes after Residential Mobility for Migrants in Germany. *Urban Studies* 50 (5), 1011–1029.
- Leventhal, T. and Brooks-Gunn, J. (2003). Moving to Opportunity: An Experimental Study of Neighborhood Effects on Mental Health. *American Journal of Public Health* 93 (9), 1576–1582.
- Levy, H. and Jenkins, S. P. (2012). Documentation for Derived Current and Annual Net Household Income Variables, BHPS waves 1–18: UK Data Archive Study Number 3909. Colchester: UK Data Archive.
- Lewis, J., Knijn, T., Martin, C. and Ostner, I. (2008). Patterns of Development in Work/Family Reconciliation Policies for Parents in France, Germany, the Netherlands, and the UK in the 2000s. *Social Politics: International Studies in Gender, State & Society* 15 (3), 261–286.
- Li, S.-M. (2004). Life course and residential mobility in Beijing, China. *Environment and Planning A* 36 (1), 27–43.
- Lichter, D. T. (1980). Household migration and the labor market position of married women. *Social Science Research* 9 (1), 83–97.
- (1983). Socioeconomic Returns to Migration Among Married Women. Social Forces 62 (2), 487–503.
- Lindberg, E., Hartig, T., Garvill, J. and Garling, T. (1992). Residential-location preferences across the life span. *Journal of Environmental Psychology* 12 (2), 187–198.
- Lindenberg, S. (1983). The New Political Economy: Its potential and limitations for the social sciences in general and for sociology in particular. In: Sodeur, W. (ed.). Ökonomische Erklärung sozialen Verhaltens [Economic explanations for social behaviour]. Duisburg: Verlag der Sozialwissenschaftlichen Kooperative, 1–68.
- (1996). Continuities in the theory of social production functions. In: Ganzeboom, H.
 B. G. and Lindenberg, S. (eds.). *Verklarende sociologie*. Amsterdam: Thesis Publishers, 169–184.
- Lobao, L. M. and Hooks, G. (2007). Advancing the Sociology of Spatial Inequality: Spaces, Places, and the subnational scale. In: Lobao, L. M., Hooks, G. and Tickamyer, A. R. (eds.). *The sociology of spatial inequality*. Albany: State University of New York Press, 29–62.
- Long, L. H. (1974). Women's Labor Force Participation and the Residential Mobility of Families. Social Forces 52 (3), 342–348.
- Lorber, J. (1994). Paradoxes of gender. New Haven: Yale University Press.
- Lowe, R. (2007). *The welfare state in Britain since 1945*. 3rd ed. Houndmills: Palgrave Macmillan.
- Lu, M. (1998). Analyzing migration decisionmaking: Relationships between residential satisfaction, mobility intentions, and moving behavior. *Environment and Planning A* 30 (8), 1473–1495.
- Lück, D. (2008). The impact of gender role attitudes on women's life courses. In: Blossfeld, H.-P. (ed.). *Globalization, uncertainty and women's careers*. Cheltenham: Elgar, 405– 432.

- Ludwig-Mayerhofer, W. and Wroblewski, A. (2004). Eppur si muove? *European Societies* 6 (4), 485–509.
- Lundberg, S. and Pollak, R. A. (2003). Efficiency in Marriage. *Review of Economics of the Household* 1 (3), 153–167.
- MacGuire, C. C. (1981). International housing policies: A comparative analysis. Lexington: Lexington books.
- Magnusson, L. and Özüekren, A. S. (2002). The Housing Careers of Turkish Households in Middle-sized Swedish Municipalities. *Housing Studies* 17 (3), 465–486.
- Mandić, S. (2001). Residential Mobility versus 'In-place' Adjustments in Slovenia: Viewpoint from a Society 'in Transition'. *Housing Studies* 16 (1), 53–73.
- Manion, T. and Flowerdew, R. (1982). Introduction: Institutional approaches in geography. In: Flowerdew, R. (ed.). *Institutions and geographical patterns*. London: Croom Helm, 1–50.
- Markham, W. T., Macken, P. O., Bonjean, C. M. and Corder, J. (1983). A Note on Sex, Geographic Mobility, and Career Advancement. *Social Forces* 61 (4), 1138–1146.
- Marsh, A., Gordon, D., Heslop, P. and Pantazis, C. (2000). Housing Deprivation and Health: A Longitudinal Analysis. *Housing Studies* 15 (3), 411–428.
- Mayer, A. (1998). Theorie und Politik des Wohnungsmarktes: Eine Analyse der Wohnungspolitik in Deutschland unter besonderer Berücksichtigung der ökonomischen Theorie der Politik [Theory and policy of the housing market. An analysis of German housing policy with a focus on political economy]. Berlin: Duncker & Humblot.
- Mayer, K. U. (2003). The Sociology of the Life Course and Life Span Psychology: Diverging or Converging Pathways? In: Staudinger, U. M. and Lindenberger, U. (eds.). Understanding human development. Dordrecht: Kluwer Academic Publishers, 463–481.
- (2004). Whose Lives? How History, Societies, and Institutions Define and Shape Life Courses. *Research in Human Development* 1 (3), 161–187.
- (2005). Life courses and life chances in a comparative perspective. In: Svallfors, S. (ed.). *Analyzing inequality*. Stanford: Stanford University Press, 17–55.
- (2009). New Directions in Life Course Research. Annual Review of Sociology 35 (1), 413–433.
- Mayer, K. U. and Schoepflin, U. (1989). The State and the Life Course. Annual Review of Sociology 15, 187–209.
- McCulloch, A. (2001). Ward-level deprivation and individual social and economic outcomes in the British Household Panel Study. *Environment and Planning A* 33 (4), 667–684.
- (2012). Housing density as a predictor of neighbourhood satisfaction among families with young children in urban England. *Population, Space and Place* 18 (1), 85–99.
- McCulloch, A. and Dex, S. (2001). Married women's employment patterns in Britain. In: Blossfeld, H.-P. and Drobnič, S. (eds.). *Careers of couples in contemporary societies*. Oxford: Oxford University Press, 177–200.
- McGinnity, F. (2004). Welfare for the unemployed in Britain and Germany: Who benefits? Cheltenham: Edward Elgar.
- McGinnity, F. and McManus, P. A. (2007). Paying the price for reconciling work and family life: Comparing the wage penalty for women's part-time work in Britain, Germany and the United States. *Journal of Comparative Policy Analysis: Research and Practice* 9 (2), 115–134.
- McHugh, K. E., Hogan, T. D. and Happel, S. K. (1995). Multiple Residence and Cyclical Migration: A Life Course Perspective. *Professional Geographer* 47 (3), 251–267.
- McKinnish, T. (2008). Spousal Mobility and Earnings. Demography 45 (4), 829-849.
- Meen, G. (2009). Modelling Local Spatial Poverty Traps in England. *Housing Studies* 24 (1), 127–147.

- Meen, G., Gibb, K., Goody, J., McGrath, T. and Mackinnon, J. (2005). *Economic segregation in England: Causes, consequences and policy*. Bristol: Policy Press.
- Megbolugbe, I. F. and Linneman, P. D. (1993). Home Ownership. Urban Studies 30 (4-5), 659–682.
- Memken, J. and Canabal, M. (1994). Housing tenure, structure, and crowding among Latino households. *Journal of Family and Economic Issues* 15 (4), 349–365.
- Menard, S. W. (2008). Applied logistic regression analysis. 2nd ed. Thousand Oaks: Sage Publications.
- Mendenhall, R., Deluca, S. and Duncan, G. J. (2006). Neighborhood resources, racial segregation, and economic mobility: Results from the Gautreaux program. *Social Science Research* 35 (4), 892–923.
- Meyers, M. K., Gornich, J. C. and Ross, K. E. (1999). Public childcare, parental leave, and employment. In: Sainsbury, D. (ed.). *Gender and welfare state regimes*. Oxford: Oxford University Press, 117–146.
- Michelson, W. (1979). Environmental Choice: Human Behavior and Residential Satisfaction. Oxford: Oxford University Press.
- Michielin, F. and Mulder, C. H. (2008). Family events and the residential mobility of couples. *Environment and Planning A* 40 (11), 2770–2790.
- Microm Micromarketing-Systeme und Consult GmbH (2000). Berechnung der Kaufkraft auf Ebene der Marktzelle [Construction of purchasing power at the market level]. Neuss. URL: http://db.dji.de/surveys/docs/2/5/mosaic_kaufkraft_berechnung.pdf (visited on 02/04/2012).
- (2012). Direktmarketing A-Z. Neuss. URL: http://www.microm-online.de/Deutsch/ Microm/Sekundaernavigation/Glossar/M/index.jsp (visited on 02/04/2012).
- Millard-Ball, A. (2002). Gentrification in a Residential Mobility Framework: Social Change, Tenure Change and Chains of Moves in Stockholm. *Housing Studies* 17 (6), 833–856.
- Mills, M. and Blossfeld, H.-P. (2006). Globalization, uncertainty and the early life course: A theoretical framework. In: Blossfeld, H.-P., Klijzing, E., Mills, M. and Kurz, K. (eds.). *Globalization, uncertainty and youth in society*. London: Routledge, 1–24.
- Mincer, J. (1978). Family Migration Decisions. Journal of Political Economy 86 (5), 749– 773.
- Moen, P. and Hernandez, E. (2009). Social convoys: Studying linked lives in time, context, and motion. In: Elder, G. H. and Giele, J. Z. (eds.). *The craft of life course research*. New York: Guilford Press, 258–279.
- Mok, D. (2005). The life stages and housing decisions of young households: An insider perspective. *Environment and Planning A* 37 (12), 2121–2146.
- Mood, C. (2010). Logistic Regression: Why We Cannot Do What We Think We Can Do, and What We Can Do About It. *European Sociological Review* 26 (1), 67–82.
- Morenoff, J. D. (2003). Neighborhood Mechanisms and the Spatial Dynamics of Birth Weight. *American Journal of Sociology* 108 (5), 976–1017.
- Morgan, S. L. and Winship, C. (2010). Counterfactuals and causal inference: Methods and principles for social research. Cambridge: Cambridge University Press.
- Morris, E. W., Crull, S. R. and Winter, M. (1976). Housing norms, housing satisfaction, and the propensity to move. *Journal of Marriage and Family* 38 (2), 309–320.
- Morrow-Jones, H. A. and Wenning, M. V. (2005). The housing ladder, the housing life-cycle and the housing life-course: Upward and downward movement among repeat homebuyers in a US metropolitan housing market. *Urban Studies* 42 (10), 1739–1754.
- Mulder, C. H. (1993). Migration dynamics: A life course approach. Amsterdam: Thesis Publishers.
- (2003). The Housing Consequences of Living Arrangement Choices in Young Adulthood. *Housing Studies* 18 (5), 703–719.

- Mulder, C. H. (2006). Home-ownership and family formation. Journal of Housing and the Built Environment 21 (3), 281–298.
- Mulder, C. H. and Hooimeijer, P. (1999). Residential relocations in the life course. In: Wissen, L. J. G. v. and Dykstra, P. A. (eds.). *Population issues*. New York: Kluwer Academic Publishers, 159–186.
- Mulder, C. H. and Wagner, M. (1998). First-time Home-ownership in the Family Life Course: A West German-Dutch Comparison. Urban Studies 35 (4), 687–713.
- Mullins, D. and Murie, A. (2006). *Housing Policy in the UK*. Houndmills: Palgrave Macmillan.
- Musterd, S. (2005). Social and ethnic segregation in Europe: Levels, causes, and effects. *Journal of Urban Affairs* 27 (3), 331–348.
- Myers, S. M. (1999). Residential Mobility as a Way of Life: Evidence of Intergenerational Similarities. *Journal of Marriage and Family* 61 (4), 871–880.
- Nash, J. (1953). Two-Person Cooperative Games. Econometrica 21 (1), 128–140.
- National Centre for Social Research (2011). Half predict Britain could become nation of renters within a generation: Press release. London. URL: http://www.natcen.ac.uk/me dia-centre/press-releases/2011-press-releases/generation-rent (visited on 04/06/2011).
- Niggebrugge, A., Haynes, R., Jones, A., Lovett, A. and Harvey, I. (2005). The index of multiple deprivation 2000 access domain: A useful indicator for public health? *Social Science & Medicine* 60 (12), 2743–2753.
- Nilsson, K. (2003). Moving into the city and moving out again: Swedish evidence from the cohort born in 1968. *Urban Studies* 40 (7), 1243–1258.
- Nisic, N. (2009). Labour Market Outcomes of Spatially Mobile Coupled Women: Why is the locational context important? *Schmollers Jahrbuch* 129 (2), 203–215.
- (2010). Mitgegangen mitgefangen? Die Folgen von Haushaltsumzügen für die Einkommenssituation von Frauen in Partnerschaften [The tied-mover hypothesis reconsidered—Effects of household migration on the labour market success of partnered women]. Kölner Zeitschrift für Soziologie und Sozialpsychologie 62 (3), 515– 549.
- Nivalainen, S. (2004). Determinants of Family Migration: Short Moves vs. Long Moves. Journal of Population Economics 17 (1), 157–175.
- Noble, M., Wright, G., Dibben, C., Smith, G., McLennan, D., Anttila, C., Barnes, H., Mokhtar, C., Noble, S., Avenell, D., Gardner, J., Covizzi, I. and Lloyd, M. (2003). *The English Indices of Deprivation 2004*. London: Office of the Deputy Prime Minister. URL: http://www.communities.gov.uk/documents/communities/pdf/131209.pdf (visited on 12/06/2010).
- Noble, M., Wright, G., Smith, G. and Dibben, C. (2006). Measuring multiple deprivation at the small–area level. *Environment and Planning A* 38 (1), 169–185.
- Noble, M., McLennan, D., Wilkinson, K., Whitworth, A., Barnes, H. and Dibben, C. (2008). *The English Indices of Deprivation 2007.* London: Communities and Local Government.
- O'Leary, N., Murphy, P. D., Latreille, P., Blackaby, D. H. and Sloane, P. J. (2005). Accounting for Differences in Labour Market Outcomes in Great Britain: Regional Analysis Using the Labour Force Survey. IZA Discussion Papers 1501. Bonn. URL: ftp://repec.iza.org/ RePEc/Discussionpaper/dp1501.pdf (visited on 08/08/2012).
- Office for National Statistics (2007). *Census Geography*. Newport. URL: http://www.statisti cs.gov.uk/geography/census_geog.asp#oa (visited on 06/12/2010).
- (2010). Divorces in England and Wales 2008. Newport. URL: http://www.ons.gov.uk/ ons/rel/vsob1/divorces-in-england-and-wales/2008/divorces-in-england-and-wales-2008.pdf (visited on 26/06/2012).

- (2012a). All releases of Population Estimates for UK, England and Wales, Scotland and Northern Ireland. Newport. URL: http://www.ons.gov.uk/ons/publications/allreleases.html?definition=tcm\%3A77-22371 (visited on 02/04/2012).
- (2012b). Families and households, 2001 to 2011. URL: http://www.ons.gov.uk/ons/ dcp171778_251357.pdf (visited on 23/08/2012).
- (2012c). nomis: Offical Labour Market Statistics. Durham. URL: https://www.nomisw eb.co.uk/ (visited on 13/09/2012).
- (2012b). *Neighbourhood Statistics*. Newport. URL: http://www.neighbourhood.statistic s.gov.uk/dissemination/LeadHome.do (visited on 02/04/2012).
- Organisation for Economic Co-operation and Development (2012). *OECD.Stat Extracts: Employment and Labour Force Statistics*. URL: http://stats.oecd.org/index.aspx (visited on 19/04/2012).
- Oriña, M., Wood, W. and Simpson, J. A. (2002). Strategies of influence in close relationships. Journal of Experimental Social Psychology 38 (5), 459–472.
- Orloff, A. (1996). Gender in the Welfare State. Annual Review of Sociology 22, 51-78.
- Ostner, I. and Lewis, J. (1995). Gender and the evolution of European social policies. In: Leibfried, S. and Pierson, P. (eds.). *European social policy*. Washington: Brookings Institution Press, 159–193.
- Ott, N. (1992). Intrafamily bargaining and household decisions. Berlin: Springer.
- Oxley, M. and Smith, J. (1996). *Housing policy and rented housing in Europe*. London: E & FN Spon.
- Özüekren, A. S. and Kempen, R. v. (2002). Housing Careers of Minority Ethnic Groups: Experiences, Explanations and Prospects. *Housing Studies* 17, 365–379.
- Pailhé, A. and Solaz, A. (2008). Professional outcomes of internal migration by couples: Evidence from France. *Population, Space and Place* 14 (4), 347–363.
- Pallas, A. M. and Jennings, J. L. (2009). Cumulative Knowledge about Cumulative Advantage. Swiss Journal of Sociology 35 (2), 211–229.
- Pampel, F. C. (2008). Logistic regression: A primer. Thousand Oaks: Sage Publications.
- Pan Ké Shon, J.-L. (2010). The Ambivalent Nature of Ethnic Segregation in France's Disadvantaged Neighbourhoods. Urban Studies 47 (8), 1603–1623.
- Parkes, A. and Kearns, A. (2003). Residential Perceptions and Housing Mobility in Scotland: An Analysis of the Longitudinal Scottish House Condition Survey 1991-96. *Housing Studies* 18 (5), 673–701.
- Pevalin, D. J., Taylor, M. P. and Todd, J. (2008). The Dynamics of Unhealthy Housing in the UK: A Panel Data Analysis. *Housing Studies* 23 (5), 679–695.
- Pfau-Effinger, B. (1997). Zum theoretischen Rahmen für die Analyse internationaler Differenzen in der gesellschaftlichen Integration von Frauen [A theoretical framework for the analysis of cross-national differences in the societal integration of women]. In: Hradil, S. (ed.). Differenz und Integration [Difference and integration]: Verhandlungen des 28. Kongresses der Deutschen Gesellschaft für Soziologie in Dresden 1996. Frankfurt/Main: Campus-Verlag, 514–527.
- Pickles, A. R. and Davies, R. B. (1991). The empirical analysis of housing careers: A review and a general statistical modelling framework. *Environment and Planning A* 23 (4), 465–484.
- Pinderhughes, E. E., Nix, R., Foster, E. M. and Damon Jones (2001). Parenting in Context: Impact of Neighborhood Poverty, Residential Stability, Public Services, Social Networks, and Danger on Parental Behaviors. *Journal of Marriage and Family* 63 (4), 941–953.
- Pixley, J. E. and Moen, P. (2003). Prioritizing careers. In: Moen, P. (ed.). It's about time: Couples and careers. Ithaca: ILR Press, 183–200.

- Propper, C., Burgess, S., Bolster, A., Leckie, G., Jones, K. and Johnston, R. (2007). The Impact of Neighbourhood on the Income and Mental Health of British Social Renters. *Urban Studies* 44 (2), 393–415.
- Quigley, J. M. and Weinberg, D. H. (1977). Intra- Urban Residential Mobility: A Review and Synthesis. *International Regional Science Review* 2 (1), 41–66.
- Quillian, L. (2003). How Long are Exposures to Poor Neighborhoods? The Long-Term Dynamics of Entry and Exit from Poor Neighborhoods. *Population Research and Policy Review* 22 (3), 221–249.
- Rabe-Hesketh, S. and Skrondal, A. (2005). *Multilevel and longitudinal modeling using Stata*. College Station: Stata Press Publications.
- Rabe, B. (2011). Dual-earner migration: Earnings gains, employment and self-selection. *Journal of Population Economics* 24 (2), 477–497.
- Rabe, B. and Taylor, M. P. (2010). Residential mobility, quality of neighbourhood and life course events. *Journal of the Royal Statistical Society. Series A (Statistics in Society)* 173 (3), 531–555.
- (2012). Differences in Opportunities? Wage, Employment and House-Price Effects on Migration. Oxford Bulletin of Economics and Statistics 74 (6), 831–855.
- Raddatz, L. and Mennis, J. (2013). Environmental Justice in Hamburg, Germany. Professional Geographer 65 (3), 495–511.
- Rasbash, J., Steele, F., Browne, W. J. and Goldstein, H. (2009a). A user's guide to MLwiN, version 2.1. London: Institute of Education.
- Rasbash, J., Charlton, C., Browne, W. J., Healy, M. and Cameron, B. (2009b). *MLwiN Version* 2.1. Centre for Multilevel Modelling, University of Bristol.
- Raudenbush, S., Brennan, R. T. and Barnett, R. C. (1995). A Multivariate Hierarchical Model for Studying Psychological Change Within Married Couples. *Journal of Family Psychology* 9 (2), 161–174.
- Ravenstein, E. G. (1876). The Birthplace of the People and the Laws of Migration. *The Geographical Magazine* 3, 173–177, 201–206, 229–233.
- (1885). The Laws of Migration. Journal of the Statistical Society of London 48 (2), 167–235.
- (1889). The Laws of Migration. Journal of the Royal Statistical Society 52 (2), 241– 305.
- Rees, P. and Kupiszewski, M. (1999). Internal migration and regional population dynamics in Europe: A synthesis. Strasbourg: Council of Europe.
- Risman, B. J. (2004). Gender As a Social Structure. Gender & Society 18 (4), 429-450.
- Ronald, R. (2008). *The ideology of home ownership: Homeowner societies and the role of housing*. Basingstoke: Palgrave Macmillan.
- Roseman, C. C. (1983). A framework for the study of migration destination selection. *Population and Environment* 6 (3), 151–165.
- Rossi, P. H. (1955). Why families move: A study in the social psychology of urban residential mobility. Glencoe: Free Press.
- Rowlands, R. and Gurney, C. M. (2000). Young Peoples' Perceptions of Housing Tenure: A Case Study in the Socialization of Tenure Prejudice. *Housing, Theory and Society* 17 (3), 121–130.
- Rural Evidence Research Centre (2005). *Defra Classification of Local Authority Districts and Unitary Authorities in England: A Technical Guide*. London. URL: http://archive. defra.gov.uk/evidence/statistics/rural/documents/rural-defn/LAClassifications_technic alguide.pdf (visited on 10/04/2012).
- Ruspini, E. (1998). Women and Poverty Dynamics: the Case of Germany and Britain. *Journal* of European Social Policy 8 (4), 291–316.

- Sainsbury, D. (1999). Gender, policy regimes, and politics. In: Sainsbury, D. (ed.). *Gender* and welfare state regimes. Oxford: Oxford University Press, 245–276.
- Sampson, R. J. and Groves, W. B. (1989). Community Structure and Crime: Testing Social-Disorganization Theory. *American Journal of Sociology* 94 (4), 774–802.
- Sampson, R. J., Morenoff, J. D. and Gannon-Rowley, T. (2002). Assessing "Neighborhood Effects": Social Processes and New Directions in Research. *Annual Review of Sociology* 28, 443–478.
- Sampson, R. J. and Sharkey, P. (2008). Neighborhood Selection and the Social Reproduction of Concentrated Racial Inequality. *Demography* 45 (1), 1–29.
- Sandell, S. H. (1977). Women and the Economics of Family Migration. Review of Economics and Statistics 59 (4), 406–414.
- Schaake, K., Burgers, J. and Mulder, C. H. (2010). Ethnicity at the Individual and Neighborhood Level as an Explanation for Moving Out of the Neighborhood. *Population Research and Policy Review* 29 (4), 593–608.
- Schaeper, H. and Falk, S. (2003). Employment trajectories of East and West German mothers compared: One nation - one pattern? In: Heinz, W. R. and Marshall, V. W. (eds.). Social dynamics of the life course. Hawthorne: Aldine de Gruyter, 143–163.
- Schmitt, C. (2012). Labour market integration, occupational uncertainty, and fertility choices in Germany and the UK. *Demographic Research* 26, 253–292.
- Schonlau, M., Watson, N. and Kroh, M. (2010). Household survey panels: How much do following rules affect sample size? SOEPpapers on Multidisciplinary Panel Data Research 347. Berlin: Deutsches Institut für Wirtschaftsforschung. URL: http://www.diw. de/documents/publikationen/73/diw_01.c.366429.de/diw_sp0347.pdf (visited on 26/07/2011).
- Schönwälder, K. and Söhn, J. (2009). Immigrant Settlement Structures in Germany: General Patterns and Urban Levels of Concentration of Major Groups. *Urban Studies* 46 (7), 1439–1460.
- Settersen, R. A. and Gannon, L. (2009). Structure, Agency, and the space between: On the challenges and contradictions of a blended view of the life course. In: Heinz, W. R., Huinink, J. and Weymann, A. (eds.). *The life course reader*. Frankfurt/Main: Campus-Verlag, 456–472.
- Shaw, C. R. and McKay, H. D. (1969). Juvenile delinquency and urban areas: A study of rates of delinquency in relation to differential characteristics of local communities in American cities. Revised ed. Chicago: University of Chicago Press.
- Shihadeh, E. S. (1991). The Prevalence of Husband-Centered Migration: Employment Consequences for Married Mothers. *Journal of Marriage and Family* 53 (2), 432–444.
- Simmons, J. W. (1968). Changing Residence in the City: A Review of Intraurban Mobility. Geographical Review 58 (4), 622–651.
- Simon, H. A. (1957). Models of man. New York: Wiley.
- Singer, J. D. and Willett, J. B. (2003). *Applied longitudinal data analysis: Modeling change and event occurrence*. Oxford: Oxford University Press.
- Sirgy, M. J., Grzeskowiak, S. and Su, C. (2005). Explaining housing preference and choice: The role of self-congruity and functional congruity. *Journal of Housing and the Built Environment* 20 (4), 329–347.
- Sjaastad, L. A. (1962). The Costs and Returns of Human Migration. Journal of Political Economy 70 (5), 80–93.
- Smith, S. J., Easterlow, D. and Munro, M. (2004). Housing for health: does the market work? *Environment and Planning A* 36 (4), 579–600.
- Smits, J. (1999). Family migration and the labour-force participation of married women in the Netherlands, 1977-1996. *International Journal of Population Geography* 5 (2), 133– 150.

- Sørensen, A. and McLanahan, S. S. (1987). Married Women's Economic Dependency, 1940-1980. American Journal of Sociology 93 (3), 659–687.
- South, S. J. and Crowder, K. D. (1997a). Escaping distressed neighborhoods: Individual, community and metropolitan influences. *American Journal of Sociology* 102 (4), 1040– 1084.
- (1997b). Residential Mobility Between Cities and Suburbs: Race, Suburbanization, and Back-to-the-City Moves. *Demography* 34 (4), 525–538.
- (1998a). Housing Discrimination and Residential Mobility: Impacts for Blacks and Whites. *Population Research and Policy Review* 17 (4), 369–387.
- (1998b). Leaving the 'Hood': Residential Mobility between Black, White, and Integrated Neighborhoods. *American Sociological Review* 63 (1), 17–26.
- South, S. J., Crowder, K. D. and Chavez, E. (2005). Exiting and Entering High-Poverty Neighborhoods: Latinos, Blacks and Anglos Compared. *Social Forces* 84 (2), 873–900.
- Speare, A. (1974). Residential Satisfaction as an Intervening Variable in Residential Mobility. *Demography* 11 (2), 173–188.

StataCorp (2007). Stata Statistical Software: Release 10. College Station.

- Statistisches Bundesamt (2004). Bautätigkeit und Wohnungen: Bestand an Wohnungen [Construction and Housing. Housing Stock]. Fachserie 5 Reihe 3. Wiesbaden.
- (2011). GENESIS-online database. Wiesbaden. URL: https://www-genesis.destatis.de/ genesis/online?operation=sprachwechsel\&option=en (visited on 10/03/2011).
- (2012a). In Deutschland lebte 2011 jede fünfte Person allein: Pressemitteilung vom 11. Juli 2012 – 242/12 [Every fifth German is a single in 2011]. Wiesbaden. URL: https: //www.destatis.de/DE/PresseService/Presse/Pressekonferenzen/2012/Alleinlebende/ pm_allein_PDF.pdf?__blob=publicationFile (visited on 23/08/2012).
- (2012b). Staat & Gesellschaft: Ehescheidungen [State and Society. Divorces]. Wiesbaden. URL: https://www.destatis.de/DE/ZahlenFakten/GesellschaftStaat/Bevoelkerung/Ehescheidungen/Tabellen/DurchschnittlichesAlterGeschiedener.html (visited on 26/06/2012).
- Steele, F. (2011). Multilevel discrete-time event history models with applications to the analysis of recurrent employment transitions. *Australian & New Zealand Journal of Statistics* 53 (1), 1–20.
- Stephens, M., Burns, N. and MacKay, L. (2003). The limits of housing reform: British social rented housing in a European context. *Urban Studies* 40 (4), 767–789.
- Strauß, S. (2008). Volunteering and Social Inclusion: Interrelations between Unemployment and Civic Engagement in Germany and Great Britain. Wiesbaden: VS Verlag f
 ür Sozialwissenschaften.
- TNS Infratest (2008). SOEP 2008 Methodenbericht zum Befragungsjahr 2008 (Welle 25) des Sozio-oekonomischen Panels: [SOEP 2008 methods report for survey year 2008, wave 25]. München: TNS Infratest. URL: http://www.diw.de/documents/dokumentenarchiv/ 17/diw_01.c.340072.de/soepmeth_2008.pdf (visited on 25/08/2010).
- Taylor, M. F., Brice, J., Buck, N. H. and Prentice-Lane, E. (2010). British Household Panel Survey: User Manual Volume A. Introduction, Technical Report and Appendices. Colchester. URL: http://www.iser.essex.ac.uk/bhps/documentation/pdf_versions/volumes/ bhpsvola.pdf (visited on 22/11/2011).
- Taylor, M. P. (2007). Tied Migration and Subsequent Employment: Evidence from Couples in Britain. Oxford Bulletin of Economics and Statistics 69 (6), 795–818.
- Thomas, D. (1990). Intra-Household Resource Allocation: An Inferential Approach. Journal of Human Resources 25 (4), 635–664.
- Tickamyer, A. R. (2000). Space Matters! Spatial Inequality in Future Sociology. Contemporary Sociology 29 (6), 805–813.

- Todaro, M. P. (1969). A Model of Labor Migration and Urban Unemployment in Less Developed Countries. *American Economic Review* 59 (1), 138–148.
- Tomann, H. (1996). Germany. In: Balchin, P. (ed.). *Housing policy in Europe*. London: Routledge, 51–66.
- Townsend, P. (1979). *Poverty in the United Kingdom: A survey of household resources and standards of living*. Reprint. Harmondsworth: Penguin Books.
- Treas, J. and Widmer, E. D. (2000). Married Women's Employment over the Life Course: Attitudes in Cross-National Perspective. Social Forces 78 (4), 1409–1436.
- Turney, K., Clampet-Lundquist, S., Edin, K., Kling, J. R., Duncan, G. J., Ludwig, J. and Scholz, J. K. (2006). Neighborhood Effects on Barriers to Employment: Results from a Randomized Housing Mobility Experiment in Baltimore. *Brookings-Wharton Papers* on Urban Affairs, 137–187.
- Twisk, J. W. R. (2006). Applied multilevel analysis: A practical guide. Cambridge: Cambridge University Press.
- United States Census Bureau (2011). Housing Vacancies and Homeownership: Annual Statistics: 2005. Suitland. URL: http://www.census.gov/hhes/www/housing/hvs/annual05/ ann05ind.html (visited on 05/07/2012).
- Vandecasteele, L. (2011). Life Course Risks or Cumulative Disadvantage? The Structuring Effect of Social Stratification Determinants and Life Course Events on Poverty Transitions in Europe. *European Sociological Review* 27 (2), 246–263.
- Venn, D. (2009). Legislation, collective bargaining and enforcement: Updating the OECD employment protection indicators. OECD Social, Employment and Migration Working 89. Paris: Organisation for Economic Co-operation and Development. URL: http: //www.oecd.org/employment/employmentpoliciesanddata/43116624.pdf (visited on 01/08/2012).
- Vickery, L. (2012). Deepening disadvantages in housing markets for women. *Local Economy* 27 (8), 796–803.
- Vidal, S. T. (2011). Life Course Contingency in the Residential Trajectory: An Analysis of Residential Mobility of Young Adults in Britain. Paper presented at the Population Association of America 2011 Annual Meeting. Washington. URL: http://paa2011.princ eton.edu/download.aspx?submissionId=111345 (visited on 09/07/2012).
- Vlist, A. J. v. d., Gorter, C., Nijkamp, P. and Rietveld, P. (2002). Residential mobility and local housing-market differences. *Environment and Planning A* 34 (7), 1147–1164.
- Voigtländer, M. (2009). Why is the German Homeownership Rate so Low? *Housing Studies* 24 (3), 355–372.
- Voigtländer, S., Berger, U. and Razum, O. (2010). The impact of regional and neighbourhood deprivation on physical health in Germany: A multilevel study. *BMC Public Health* 10 (1), 403.
- Wagner, G. G., Frick, J. and Schupp, J. (2007). The German Socio-Economic Panel Study (SOEP): Scope, Evolution and Enhancements. *Schmollers Jahrbuch* 127 (1), 139–170.
- Wagner, M. (1989). R\u00e4umliche Mobilit\u00e4t im Lebensverlauf: Eine empirische Untersuchung sozialer Bedingungen der Migration [Spatial mobility in the life course. An empirical analysis of the social determinants of mobility]. Stuttgart: Enke.
- Watson, S. (1986). Housing and the family: The marginalization of non-family households in Britain. *International Journal of Urban and Regional Research* 10 (1), 8–28.
- West, C. and Zimmerman, D. H. (1987). Doing Gender. Gender & Society 1 (2), 125–151.
- Wiesemann, L. (2008). Wohnstandortentscheidungen türkischer Migranten im Vergleich [Residential choice of Turkish migrants in comparison]. In: Hillmann, F. and Windzio, M. (eds.). Migration und städtischer Raum [Migration and urban areas]. Opladen: Budrich.

- Williams, R. L. (2000). A Note on Robust Variance Estimation for Cluster-Correlated Data. Biometrics 56 (2), 645–646.
- Wilson, K. L. and Portes, A. (1980). Immigrant Enclaves: An Analysis of the Labor Market Experiences of Cubans in Miami. *American Journal of Sociology* 86 (2), 295–319.
- Wilson, W. J. (1987). The Truly Disadvantaged: The Inner City, the Underclass and Public Policy. Chicago: University of Chicago Press.
- Windzio, M. (2004). Kann der regionale Kontext zur Arbeitslosenfalle werden? Der Einfluss der Arbeitslosigkeit auf die Mobilität zwischen regionalen Arbeitsmärkten in Westdeutschland [Regional context as a trap for the unemployed? The impact of unemployment on inter-regional mobility between labour markets in West Germany]. Kölner Zeitschrift für Soziologie und Sozialpsychologie 56 (2), 257–278.
- (2008). Social Structures and Actors: The Application of Multilevel Analysis in Migration Research. *Romanian Journal of Population Studies* 1 (1), 113–138.
- Wingens, M. and Reiter, H. (2011). The Life Course Approach it's about Time! BIOS 24 (2), 187–203.
- Winstanley, A., Thorns, D. C. and Perkins, H. C. (2002). Moving House, Creating Home: Exploring Residential Mobility. *Housing Studies* 17 (6), 813–832.
- Wolpert, G. (1965). Behavioral aspects of the decision to migrate. Papers and Proceedings of the Regional Science Association 15 (1), 159–169.
- Wooldridge, J. M. (2002). *Econometric analysis of cross section and panel data*. Cambridge: MIT Press.

World Bank (1993). Housing: Enabling markets to work. Washington: World Bank.

- Zelinsky, W. (1971). The Hypothesis of the Mobility Transition. *Geographical Review* 61 (2), 219–249.
- Zhou, M. and Logan, J. R. (1991). In and Out of Chinatown: Residential Mobility and Segregation of New York City's Chinese. *Social Forces* 70 (2), 387–407.
- Zipf, G. K. (1946). The P1 P2/D Hypothesis: On the Intercity Movement of Persons. American Sociological Review 11 (6), 677–686.